SESOIL A SEASONAL SOIL COMPARTMENT MODEL

by M. Bonazountas Precipitation Ram Snow.) J. Wagner 8 6 EVAPOTRANSPIRATION ALVIORI Emporation iste nal Inditionation to Treatment Plant STAL AND AND SOLL FORM A MALINITY HAL LOWER UN OTORIO ED SOL - LOTTE 1981 SUKIAG First KUNDEF Version SAUMATRIS THIR Rechart RUNOFY STREAM, Prederich CHANNEL, ESTUARY. / Arthur D. Little, Inc. Cambridge, Massachusetts

SESQIE MODEL UPDATES

The SESOIL model will be updated periodically (see section 1.1 of this documentation). If you are interested is updates please return this information page (or call) to:

Irene Rickabaugh : (617):864-5770:x3148 Arthur D. Little, Inc. 32/122A Acorn Park Cambridge, Mass. 02140

..Re: SESOIL

Name :

Address:

Tel.# :

Date :



"SESOIL"

A SEASONAL SOIL COMPARTMENT MODEL

Ъy

Marcos Bonazountas Janet M. Wagner

Arthur D. Little, Inc. Cambridge, Massachusetts 02140 617/864-5770

This Model Version Prepared for

United States Environmental Protection Agency Office of Toxic Substances Washington, D.C. 20460 202/755-8068

> Task Manager: Annette Nold EPA Contract No. 68-01-6271

> > December 1981

Arthur D Little, Inc

ABSTRACT

Bonazountas, M. and J. Wagner (1981); "SESOIL: A Seasonal Soil Compartment Model," Arthur D. Little, Inc., Cambridge, Massachusetts 02140 Prepared for U.S. Environmental Protection Agency, Office of Toxic Substances Contract No. 68-01-6271

SESOIL is a "user-friendly" statistical mathematical model designed for longterm environmental pollutant fate simulations that can describe: water transport (quality/quantity); sediment transport (quality/quantity); pollutant transport/transformation; and soil quantity. Simulations are performed for a user specified soil column (designated as compartment), extending between the ground surface and the lower part of the saturated soil zone of a region. The simulation is based upon a three-cycle rationale, each cycle being associated with a number of processes. The three cycles are the: (1) water cycle which takes account of rainfall, infiltration, exfiltration, surface runoff, evapotranspiration, groundwater runoff, snow pack/melt and interception, (2) sediment cycle which takes account of sediment resuspension (because of wind) and sediment washload (because of rain storms), and (3) pollutant cycle which takes account of convection, diffusion, volatilization, adsorption/desorption, chemical degradation/decay, biological transformation/uptake, hydrolysis, photolysis, oxidation, complexation of metals by organics and nutrient cycles. Model development has been sponsored by the U.S. Environmental Protection Agency and has been validated--as an unsaturated soil zone pollutant transport model--at waste land treatment disposal sites. The entire model development has not yet been accomplished; however, certain model features are operational.

Key words: SESOIL, mathematical modeling, pollution, soil quality, groundwater, pathways, land treatment, waste disposal, multi-media modeling.

TABLE OF CONTENTS

Abstract

1.0	INTRODUCTION	1-1
2.0	THE SESOIL MODEL	2-1
3.0	USER'S MANUAL	3-1

APPENDICES

нү ·	Hydrologic Cycle	HY-1
SW	Soil Washload	SW-1
SR	Soil Resuspension	SR-1
vo	Diffusion and Volatilization	VO-1
AD	Adsorption and Desorption	AD-1
DE	Degradation and Decay	DE-1
HD	Hydrolysis of Organic Compounds	HY-1
CE	Cation Exchange	CE-1
CM	Complexation of Metals	CM-1
PH	Photolysis (not in this documentation)	PH-1
FX	Fixation (not in this documentation)	FX-1
BI	Biologic Activities (not in this documentation)	BI-1
NU	Nutrient Cycles	NU-1
PT	Pollutant Transport Cycle	PT-1
ID	Input Data Compilation	ID-1
DF	Data Files and Arrays	DF-1
FC	FORTRAN Code	FC-1
AP	Application Samples	AP-1
RE	References	RE-1
MI	Miscellaneous	MI-1

DISTRIBUTION LIST (of Dec. 31, 1981)

M. Callahan (EPA/OTS)
G. Contos (Versar)
J. Dragun (EPA/OTS)
P. Eagleson (MIT)
G.R. Foster (Purdue Univ.)
G. Harris (ADL)
R. Kickert (EPA/CERL)
A. Nold (EPA/OTS)
M. Slimak/J. Segna (EPA/MDSD)
W. Wood (EPA/OTS)

1.0 INTRODUCTION

TABLE OF CONTENTS

.

1.0	INTRODUCTION	1-2
	1.1 Preamble Notes	1-2
	1.2 Organization of this Documentation	1-3
	1.3 Raison d'Être of SESOIL	1-4
	1.4 Acknowledgements	1-6
	1.5 References	1-7

1.0 INTRODUCTION

1.1 Preamble Notes

- This modeling effort and its documentation have been accomplished at an expense (professional time, all other expenses) of less than \$70,000 and, as such, they should be evaluated or criticized correspondingly.
- This documentation is a preliminary draft and has not been publicly released by the U.S. Environmental Protection Agency; however, it has been circulated to scientists for comments on its technical approach. Since model developers have not published their original work, quotations related to SESOIL or the processes described in this documentation must be referenced (Bonazountas and Wagner 1981) as indicated in the abstract.
- All models are good as long as: (i) users are aware of the assumptions upon which they have been developed; and (ii) they are employed and applied appropriately.
- SESOIL is a "user-friendly" model that can be operated with very few input data, mostly available from government records or other literature (e.g., handbooks). This has been achieved with a sophisticated mathematical description of all SESOIL processes, a task that has exceeded previous similar efforts of the literature. However, "amateur (modelers) can do more harm than city fellers on a farm" (Groundwater 1981); therefore, potential users should be careful when employing this friendly and easy to use package.
- Most environmental models of the literature can be "forced" by their developers to predict almost exactly what their developers desire to predict--via calibration coefficients; this is not a secret among modelers. In that respect, SESOIL is at an advantageous position because no calibration coefficients accompany its theory; however, users should validate model predictions with available data as far as possible. (See also Section 3.4.)
- The authors intend to continue improving SESOIL--both in its newly developed scientific basis and its range of applicability--and to update this documentation as appropriate. In that respect, they: (a) solicit any critical review, and (b) kindly ask users to make sure to have the latest version of the model code.
- SESOIL has been carefully developed and its software has been tested; however, the ultimate responsibility for its use rests with the user, since this is the first version of the model, and developers have not repeatedly applied the model to the real world. However, developers intend to correct any errors which users may report.

- This SESOIL version presents only a subset of all model features. The computer code, therefore, contains control nodes and dummy statements and loops that will facilitate potential future development. It would be inappropriate for a user to modify the code without notifying model developers, because this may result in incorrect calculations.
- Roughly speaking, this version of SESOIL can be employed as: (i) a hydrologic basin model (watershed, unsaturated soil zone, ground-water recharge); and (ii) a pollutant transport model of the unsaturated soil zone, however, interacting (for mass balance purposes) with both the watershed and the groundwater of a soil environment. (See Section 2.0.)
- Strong appreciation is given by model developers to all the people who have supported this effort. (See Section 1.4.)
- Users are advised to read this documentation carefully and, in case of questions, to contact developers. The authors would be happy to provide assistance--as far as possible--to potential users.

1.2 Organization of this Documentation

The main intentions while drafting this documentation have been: simplicity, clarity and expandability; therefore, it has been structured around two major parts containing:

- (1) an overall presentation, and
- (2) twenty appendices

It is believed that this documentation format allows:

- readers to clearly understand both the various scientific areas modeled (described in the appendices) and the SESOIL operations, and
- (2) users to efficiently apply SESOIL following knowledge gained after reading the entire documentation.

The overall presentation is covered in three main sections:

- Section 1.0 Introduction
- Section 2.0 SESOIL Description
- Section 3.0 User's Manual

The 20 appendices are self-contained, short documents and give both background information, and mathematics employed for the various areas of science (hydrology, sedimentation, chemistry, other) modeled via SESOIL. Each appendix is designated with two characteristic letters as shown in the Table of Contents of this documentation. References are given in each section or each appendix and are aggregared in Appendix RE. It is believed that this format permits expansion, improvement and substitution of the background and parts of the theory given in this documentation without affecting the overall model presentation. This loose-leaf binder version also provides the possibility of single-page substitutions in the near future. The latest version of a page is printed with a date next to the page number; if no date is given, the date of the first page of the section of appendix is assumed. The appendix format facilitates reading because the same text is not to be found in two different chapters; it also facilitates users who do not actually care for the background during an application.

A potential user, however, is advised to read all sections of the report, namely, from the Introduction to the last appendix (Appendix MI, Miscellaneous). A user who desires to only use a few aspects of the model operation (e.g., pollutant cycle) would have to refer only to the corresponding appendix (i.e., Appendix PT, Pollutant Transport Cycle). In the computer code, reference is made to the equations of individual appendices.

1.3 Raison d'Être of SESOIL

SESOIL is the acronym for a <u>SEasonal SOIL</u> compartment model, a development motivated by the individual needs of various technical and regulatory offices within the U.S. Environmental Protection Agency. SESOIL was designed to be an integrated package of a "user-friendly" tool for modeling hydrologic, sediment and pollutant cycles in soil "compartments." Many reasons supported this development as described below.

First, in reference to a soil compartment (see cover figure), we have today a variety of excellent watershed simulation (e.g., Johanson et al 1979) models, a variety of unsaturated soil zone numerical (e.g., Adams et al 1976) models, a variety of stochastic soil moisture models, or a number of watershed erosion and sediment transport models (e.g., Leytham et al 1979). However, we do not have an integrated, and developed from "scratch," soil compartment mathematical model, designed for long-term (defined below) environmental process simulations that can describe simultaneously water transport (quantity/quality), sediment transport (quantity/quality), pollutant fate (transport/transformation) and soil quality. SESOIL has been designed to fill this need.

Second, current regulations (e.g., the Resource Conservation Recovery Act) require that decision makers consider the environment as a continuum. Thus, pollutant fate must be modeled in this continuum-encompassing air, soil and water compartments--rather than in single medium. This request brought model users to the dilemma of "which model to interface with what model" in order to create a useful continuous package (Fiksel et al 1981). In many cases, data requirements and time resolutions of the various models were so different, that interfacing requirements necessitated the writing of complicated or lengthy data management computer programs. In addition, separate calibration procedures may have to be followed for different submodels (e.g., watershed submodel, unsaturated soil zone submodel) of one and the same environmental compartment (i.e., soil compartment), resulting in much duplication of effort.

In response to the above regulations, an immediate need for integrated modeling packages has emerged, leading to a boom in environmental modeling and the use of models as decision mechanisms. This immediate necessity has not left enough time to model developers to "sit back" and develop truely "integrated" approaches. The difficulties created by employing and interfacing incompatible submodels is analogous, for example, to the industry where in an attempt to quickly release a new product, manufacturers assemble--not always successfully--a new product with parts desinged for other similar situations. With SESOIL, an attempt is made to better integrate certain model categories and provide an efficient interfacing module between air and water compartmental models of the literature toward a formulation of an environmental continuum.

Third, a characteristic of the existing environmental models is the simulation time step. Most of the watershed soil models consist of a set of equations solved after each storm event. Therefore, hydrology, sedimentation and pollution mass transport at the end of a season (e.g., month, year) is estimated by summing up distribution estimates after each storm event. This necessitates lengthy data inputs of hydrologic records, a fact that makes use of models very time consuming, and may not necessarily lead to more accurate cycle (hydrology, sediment, pollutant) estimates. The SESOIL seasonality provides a different and flexible approach to this issue.

Principally, SESOIL is intended to be a model that:

- is seasonal--provision is also made for storm-by-storm simulations;
- (2) is independent from the size and the shape of the soil column, i.e., independent from the numerical discretization mathematical problems of some models;
- (3) is user-friendly and requires a minimum number of hydrologic and other input data;
- (4) can study the hydrologic cycle, the sediment cycle, pollutant fate and soil quality of the compartment in one integrated effort;
- (5) is operational at various "levels" depending on users' needs and data availability;
- (6) is operational either as a self-standing soil compartment model or as a model to be interfaced with an atmospheric and a fresh water body model toward the formation of a mathematical environmental continuum;

- (7) does not require calibration of coefficients that do not describe physical or chemical parameters, yet it might be calibrated via its basic parameters if field data are available;
- (8) a user can operate with data obtained from existing data bases (even on-line) or from handbooks;
- (9) is expandable in logic and capabilities; and
- (10) can be operated at minimum expense (time, cost) and by uses who may or may not exactly follow all the theoretical background of the various processes/subroutines modeled.

The attempt to accomplish the above 10 desires/needs is the "raison d'être" of SESOIL. It is hoped that it will become a valuable tool in environmental quality planning. The fact that SESOIL is user-friendly, comprehensive and inexpensive to run should stimulate users to take advantage of its benefits.

1.4 Acknowledgements

This report has been submitted by Arthur D. Little, Inc., in partial fulfillment of the requirements of EPA Contract No. 68-01-6271. The study was performed for the U.S. Environmental Protection Agency, Office of Toxic Substances, Washington, D.C. The authors would like to extend their graditude to all the contributors to this project.

Special thanks are extended to the Arthur D. Little, Inc., staff members who have contributed extensively to this report: Warren Lyman (Appendices HD, CE, CM), Diane Gilbert (Appendices SR, ID), Joo Hooi Ong (Appendix NV), Kate Scow (Appendix DE), Irene Rickabaugh (report coordination), Linda Nazaretian and Emma Wood (general support).

The authors extend their thanks to all scientists within EPA who have assisted them on this and the previous SESOIL contracts.

Dr. Annette Nold of the Office of Toxic Substances (OTS) has been the Project Officer of this first finalized model version (1981). She has assisted the authors in the improvement of the model over the past versions, she has been patient with developmental problems, and she was always ready to compile scientific and other information for us. We have appreciated our collaboration with her.

Dr. William Wood, at the head of OTS Modeling Group, has been the Project Officer and Mrs. Joan Leffler supervised in 1979 the EPA contract which lead to the conceptualization of SESOIL (Aravamudan et al 1979). The authors appreciate both the scientific advice and the support received for the SESOIL concept. Following its conceptualization in 1979, a model improvement and an application were sponsored in 1980 by the EPA, Monitoring and Data Support Division (MDSD). Mr. Michael Slimak, at the head of MDSM, has been the Project Officer and Mr. John Segna supervised that task. The authors appreciate the overall advice of Mike Slimak, the technical advice of John Segna and the ongoing support received from this office.

The authors would like to extend their appreciation to: Mr. Michael Callahan, head of the Exposure Assessment Branch, EPA/OTS; Dr. James Dragun, EPA/OTS, for his participation in the model development in 1979 and for reviewing the chemistry aspects of the model and insisting on valuable improvements; and Dr. Ronald Kickert, EPA/Corvallis Research Laboratory for requesting detailed information regarding the model and making substantial comments.

The authors express their graditude to Professor Peter S. Eagleson, Massachusetts Institute of Technology, Ralph M. Parsons Laboratory for Water Resources and Hydrodynamics, who has voluntarily reviewed and has made suggestions for the application of his annual water balance theory (Eagleson 1978) in respect to the long-term (averaged) estimates of pollutant concentrations in the unsaturated soil zone of SESOIL.

Appreciation is expressed to Professor George R. Foster, Purdue University, Agricultural Engineers Department, for his generosity in releasing his sedimentation theory (Foster et al 1980) and its computer code to the authors, although integration of his theory (Appendix SW) into the SESOIL framework has not yet been initiated.

Finally, we would like to thank Ms. Gayaneh Contos at Versar, Inc., and Dr. George Harris, at Arthur D. Little, Inc., who have managed this contract with the EPA; Dr. Alfred Wechsler, Dr. Alan Eschenroeder, Dr. Philip Levins; and Dr. Frank Feeley who have supported us in many different ways.

1.5 References

Adams, R.T. and F.M. Jurisu (1976). Simulation of pesticide, movement on small agricultural watersheds. U.S. EPA, Athens Environmental Research Laboratory, Georgia 30605. EPA-600/3-76-066, NTIS PB-259933.

Aravamudan, K.; M. Bonazountas; A. Eschenroeder; D. Gilbert; L. Nelken; K. Scow; R. Thomas; W. Tucker; and C. Unger (1979). An Environmental Partitioning Model for Risk Assessments of Chemical. Arthur D. Little, Inc., report, prepared for U.S. EPA, Office of Toxic Substances, under Contract No. 68-01-3857.

Eagleson, P.S. (1978). Climate, Soil and Vegetation (1-7). Water Resource Research, Vol. 14, No. 5, pp. 705-776.

Fiksel, J.; M. Bonazountas; H. Ojha; and K. Scow (1981). An Integrated Geographic Approach to Developing Toxic Substances Control Strategies. Report by Arthur D. Little, Inc., for U.S. EPA, Office of Policy and Resource Management, Contract No. 68-01-6160.

Foster, G.R.; L.J. Lane; J.D. Wowlin; J.M. Laflen; and R.A. Young (1980). A Model to Estimate Sediment Yield from Field-sized Areas: Development of Model. USDA, Purdue Agricultural Experiment Station. Purdue Journal Paper No. 7781.

Groundwater (1981). Journal of Groundwater Technology Division, National Water Well Association, p. 56 (quotation from Burma Shave, circa 1940).

Johanson, R.C.; J.C. Imhoff; and H.H. Davis (1979). Hydrologic Simulation Program-Fortran (HSPF). U.S. EPA, Athens Environmental Research Laboratory, Georgia 30605. EPA Grant No. R804971-01.

Leytham, K.M. and R.C. Johnson (1979). Watershed Erosion and Sediment Transport Model. U.S. EPA, Athens Environmental Research Laboratory, Georgia 30605. EPA-600/3-79-028. 2.0 THE SESOIL MODEL

2.0 THE "SESOIL" MODEL

This section contains only an executive summary of all appendices presented in this documentation, and as such it does <u>not</u> offer anything new to model readers or users.

The model developers have purposely kept this section brief because;

- (1) They feel that users of this first SESOIL version should read/consult--for their own benefit--the entire theory of each major area of science presented in each appendix separately, in order to appreciate both the capabilities of SESOIL and the limitations or assumptions supporting this model version.
- (2) SESOIL development has not yet been completed and since the model is limited to simulating the unsaturated soil zone of the soil compartment, the developers do not want to give the impression of having accomplished all their goals.

The executive summary of SESOIL is presented in the following pages. Please contact the authors with any questions regarding this model version and/or questions as to potential (future) capabilities of the model.

"SESOIL"

A SEASONAL SOIL COMPARTMENT MODEL

Bу

Marcos Bonazountas Janet Wagner

Arthur D. Little, Inc. Cambridge, Massachusetts 02140 617/864-5770

GENERAL INFORMATION

SESOIL is a newly developed "user-friendly" mathematical model for longterm environmental pollutant fate simulations that has been designed to describe:

- water transport (quality/quantity),
- sediment transport (quality/quantity),
- pollutant fate (transport/transformation), and
- soil quality

within a user specified soil column (designated as compartment) extending between the ground surface and the lower part of the saturated soil zone of a region. (See figure 2-1.)

SESOIL is designated as "seasonal" because it statistically estimates the pollutant distribution in the soil column after a season (e.g., year, month) "directly." It does not estimate pollutant distribution indirectly (i.e., by summing up pollutant distribution estimates in the soil column after each major storm event) as do existing models described in the literature.

SESOIL has been designed to become, in the long run: (1) a watershed model; (2) an unsaturated soil zone model; and (3) a groundwater model. However, the current SESOIL version can only simulate processes of an unsaturated soil zone of a compartment and can roughly account for certain watershed aspects of the compartment. The groundwater aspects of SESOIL are part of the long-range plans of the developers. As such, SESOIL is designed to simulate point or nonpoint pollution from major land use categories, and soil-column pollution originating on the watershed (future development), in the soil column (presently) and in groundwater (ultimate development).

SESOIL is designed as: (1) a self-standing soil compartment model, and (2) a compartment model to be interfaced with other atmospheric and water body models towards the formation of a mathematical environmental continuum (multi-media environmental modeling). The current version can



Figure 2-1 SCHEMATIC PRESENTATION OF THE SESOIL COMPARTMENT

be easily interfaced with a groundwater model toward the formulation of an unsaturated/saturated soil compartment model.

SESOIL simulations do not require the extensive and time consuming calibration procedures of other models, although it may be easily calibrated to agree with field records. The model employs theoretically derived equations driven by climatic, soil property, geometric and chemical compound property data. In addition, simulations are performed for the entire compartment (i.e., watershed, unsaturated and saturated soil zones) in one effort in order to circumvent the known calibration difficulties of simulation models. As such, SESOIL may be employed as a precalibration model for other simulation models.

There exist no artifically imposed limitations in: (1) timing and sizing the soil compartment (cell) and (2) the shape of the compartment per se. If the soil column is chosen small enough (i.e., finite approach), SESOIL encompasses the concept of the numerical models (e.g., finite difference/element models); if the soil column is chosen large enough (e.g., a river basin), SESOIL becomes a sophisticated one-compartment model. The unsaturated soil zone of the model can be discretized to account for more than one soil layer in order to best meet simulation needs.

SESOIL is designed to provide great flexibility to the user who can execute various "levels" of model operation, the criterion for a level selection being data availability and study objectives. The major advantage of SESOIL is that it can be executed with easily obtainable input data because this information can be compiled from existing data bases (e.g., NOAA) and known references for the pollutant and soil properties. A data management structure accompanies SESOIL. If the model is not linked to existing data bases, then the number of input data can be less than 50 as contrasted to the other numerical models which may require more than 500, because: (a) the model employs theoretically derived equations which may not require calibration, and (b) the statistical simulation does not take place after each major storm event.

Potential applications of SESOIL include long-term leaching studies from waste disposal sites, acid rain, pesticide and sediment transport on watersheds, contaminant exposure assessments, pre-calibration runs for other simulation models, hydrologic cycles of soil compartments, etc.

This model version has been developed on behalf of the U.S. Environmental Protection Agency, Office of Toxic Substances, Washington, D.C. A model application at industrial land treatment sites has been sponsored for a slightly different model version by the Monitoring and Support Data Division, EPA, Washington, D.C.

SIMULATION CYCLES

The simulation is structured around three cycles, each cycle being associated with a number of processes. These are the:

- Hydrologic cycle which takes account of: rainfall, infiltration, soil moisture, surface runoff, exfiltration, evapotranspiration, groundwater runoff, capillary rise, snow pack/melt (not operational in this version) and interception (not operational).
- Sediment cycle which takes account of: sediment resuspension (due to wind) and sediment washload (due to rain storms), not operational in this version.
- Pollutant cycle which can take account of: advection, diffusion, volatilization, adsorption/desorption, chemical degradation/decay, biological transformation and uptake, hydrolysis, photolysis, oxidation, cation exchange, complexation chemistry (metals by organics) and nutrient cycles (not operational).

The hydrologic cycle controls the sediment cycle, whereas both previous cycles control the pollutant cycle. Cycles, processes, mathematical modeling, application and validation issues are summarized in the following paragraphs.

(1) The hydrologic cycle is based on a statistical dynamic formulation of vertical water budget at a land-atmosphere interface (Eagleson 1978), adapted to account for monthly simulations. Uncertainty of the hydrologic cycle simulation is expressed via probability density functions of the independent climatic variables and yields derived probability distributions of the dependent water balance elements: surface runoff, evapotranspiration, and groundwater runoff. Some details of the hydrologic analysis are (Eagleson 1978):

Seasonal point precipitation is represented by Poisson arrivals of rectangular gamma distributed intensity pulses that have random depth and duration. Infiltration and exfiltration are described by the Philip equation (Philip 1969), which assumes the medium to be effectively semi-infinite, and the internal soil moisture at the beginning of each storm and inter-storm period to be uniform at its long-term space-time average. Gravitation and percolation to groundwater is assumed to be steady throughout the time step of a simulation and at a rate determined by the long-term space-time average seasonal soil moisture. Capillary rise from the water table is assumed to be steady throughout the season and to take place to a dry surface. Soil properties, soil moisture, climate and functional relationships derived by Brooks and Corey (1966) describe the wetting drying soil intrinsic permeability temporal variation.

Seasonal bare soil evaporation and vegetal transpiration are calculated for the interstorm periods as functions of properties of the climate, the storm sequence, the surface, the soil and the average rate of a derived potential evapotranspiration. The work of Penman (1963), Van den Honert (1948), Cowan (1965) is employed (Eagleson 1978). The distribution of surface runoff volume is derived from the distribution of rainstorm intensity and duration and the use of the previously discussed infiltration equation. Specific subroutines of the model have been validated in the literature. The annual dynamic water balance of the model has been validated by Eagleson (Eagleson 1978). The monthly dynamic water balance of the model has been applied; however, it was not validated as a hydrologic routine per se. Validation of the routine was undertaken in connection with pollutant migration in the unsaturated soil zone (Bonazountas and Wagner 1981).

(2) The sediment cycle accounts for both sediment washload due to precipitation and sediment (dust) resuspension due to wind. Two sediment washload routines are accounted by SESOIL: (a) an annual sediment yield equation based on the Universal Soil Loss Equation (USLE) as developed and documented by the U.S. Department of Agriculture (Wischmeier and Smith 1978) and employed by other watershed models of the literature; and (b) a monthly sediment washload routine based on theoretically derived equations and first physical principles (Foster et al 1980).

The theoretical monthly sediment routine can account for: (a) various sizes and shapes of watersheds (e.g., overland flow, channel flow, impoundment, pond); (b) detachment of soil particles, transport and deposition of soil particles, rill and inter-rill erosion on the watershed; (c) sediment characteristics and other fundamental relationships of precipitation energy and erosion sediment transport. The sediment washload model is based on the fundamental theoretical models of Yalin (1963), Foster et al (1980), and Cooley (1980). The sediment routine has not been validated with SESOIL's hydrologic cycle.

The dust resuspension routine estimates the losses from the surface of the SESOIL soil column of any pollutants associated with surface particles. The losses due to physical removal of the particles that have an associated pollutant load are calculated as a function of particle characteristics (chemical composition, diameter, etc.) and weather conditions. Variables such as soil moisture and wind speed are utilized; however, this routine has not been validated yet with SESOIL's hydrologic cycle.

(3) The pollutant cycle accounts for more than 12 chemical processes. (See previous section.) There exists no single equation that can optimally describe each of the pollutant processes under all all conditions, so some alternative simulation options are possible; for example, adsorption is modeled as sorption to soil particles, partitioning to soil organic carbon, or as an ion exchange process simply by varying the input parameters. Another example is volatilization from soil to air that can be modeled with more than one user specified equation (theoretical, experimental). The pollutant cycle is simulated in more than one soil subcompartments, each one consisting of three phases: soil-air, soilmoisture and soil-solids. The pollutant cycle routine has been rederived from a pollutant mass balance equation that can be expanded or modified easily in order to account for additional processes and model improvements.

MODEL VALIDATION

A pollutant transport application/validation study was undertaken to assess the long-term predictive pollutant pathway capabilities of SESOIL using field monitoring data and supporting background information already collected as part of another study (monitoring program). The model has been employed as (1) an upper unsaturated soil zone model at two industrial land treatment waste sites, and (2) as an exposure assessment model for fictitious environmental soil compartments. The behavior of two organic pollutants (napthalene, anthracene) and four inorganic pollutants (copper, chromium, nickel, sodium) at two sites was simulated and analyzed. Predicted concentrations and laboratory measured concentrations agreed within expected limits. Calibrated and noncalibrated model runs have been compared (Bonazountas et al 1981).

SESOIL has been also employed as a mathematical tool for exposure assessment studies for predicting the behavior of pollutants in soil compartments, and it proved to be an interesting application for screening, analyzing and prioritizing pollutant behaviors in soil systems (Bonazountas and Wagner 1982).

REFERENCES

Wagner, J. and M. Bonazountas (1982). Buried Halogenated Solvent Simulations via SESOIL. Arthur D. Little, Inc., study in progress, performed for U.S. EPA, Office of Toxic Substances. EPA Contract No. 68-01-6271.

Bonazountas, M.; J. Wagner; and B. Goodwin (1981). Evaluation of Seasonal Soil/Groundwater Pollutant Pathways. Arthur D. Little, Inc., Final Report, prepared for U.S. EPA, Monitoring and Data Support Division. EPA Contract No. 68-01-5949/9.

Bonazountas, M.; J. Wagner; and B. Goodwin (1981). Seasonal Cycles of Pollutants Originating from Land Treatment Practices. Proceedings Environmetrics '81 Conference, SIAM/SIMS, Virginia.

Brooks, R.H. and A.T. Corey (1966). Properties of Porous Media Affecting Fluid Flow. Proc. ASCE Journal of the Irrigation and Drainage Division, No. IR 2, Paper 4855, pp. 61-68.

Cooley, R.K. (1980). Erosivity "R" for Individual Rain Storms. (in Knisel, p. 386).

Cowan, J.R. (1965). Transport of Water in the Soil-Plant-Atmosphere System. J. App. Ecology, (2) pp. 221-239.

Eagleson, P.S. (1978). Climate, Soil, and Vegetation (1-7). Water Resources Research, Vol. 14, No. 5, pp. 705-776.

Foster, G.R.; L.J. Lane; J.D. Nowlin; J.M. Laflen; and R.A. Young (1980). A Model to Estimate Sediment Yield from Field-Sized Areas: Development of Model. Purdue Agricultural Experiment Station. Purdue Journal No. 7781.

Knisel, W.G. (1980). CREAMS: A Field Scale Model for Chemicals Runoff and Erosion for Agricultural Management Systems. USDA Conservation Research Report No. 26.

NOAA (National Oceanographic and Atmospheric Administration), Local Climatological data files, Monthly Climatologic data.

Penman, H.L. (1963). Natural Evaporation from Open Water, Bare Soil, and Grass. Proc. Roy. Soc. (London), Ser. A, Vol. 193, 1948, pp. 120-145.

Philip, J.R. (1969). Theory of Infiltration, in Advances in Hydroscience, Vol. 5, edited by V.T. Chow, pp. 215-296. Academic Press, New York.

Van den Honert, T.H. (1948). Water Transport in Plants as a Catenary Process. Discuss. Faraday Soc., (3), pp. 146-153 (cited fromk Eagleson 1978).

Yalin, Y.J. (1963). An Expression for Bedcover Transportation. Journal of the Hydraulics Division. Proceedings of the ASCE 89(HY3): 221-250.

Wischmeier, W.H. and D.D. Smith (1978). Predicting Rainfall Erosion Losses. U.S. Department of Agriculture, Agriculture Handbook No. 537.

3.0 USER'S MANUAL

SECTION 3.0

USER'S MANUAL

Arthur D Little, Inc

.

			Page
3.1	INTRODUCTION		3-3
	3.1.1	General Capabilities	3-3
	3.1.2	Phenomenology in the Soil Compartment	3-3
	3.1.3	Mathematical Modeling Issues	3-5
	3.1.4	Levels of SESOIL Operations/Capabilities	3-6
		3.1.4.1 General	3-6
		3.1.4.2 LEVELO	3-8
		3.1.4.3 LEVEL1	3-8
		3.1.4.4 LEVEL2	3-10
		3.1.4.5 LEVEL3	3-10
		3.1.4.6 Other Levels	3-10
	3.1.5	Problem Identification/Level Selection	3-12
	3.1.6	Canonical/Scenario's Chemical Fate Modeling	3-12
3.2	DATA STRUCTURE/MANAGEMENT		3-13
	3.2.1	General	3-13
	3.2.2	SESOIL Program Structure	3-13
	3.2.3	Model Execution Philosophy	3-15
	3.2.4	Input Data Files	3-15
		3.2.4.1 GE DATA File	3-16
		3.2.4.1.1 General	3-16
		3.2.4.1.2 Data Input	3-21
		3.2.4.2 LO DATA File	3-31
		3.2.4.2.1 General	3-31
		3.2.4.2.2 Data Input	3-31
		3.2.4.3 L1 DATA File	3-36
		3.2.4.3.1 General	3-36
		3.2.4.3.2 Data Input	3-36
		3.2.4.4 L2 DATA File	3-39
		3.2.4.4.1 General	3-39
		3.2.4.4.2 Data Input	3-39
		3.2.4.5 L3 DATA File	3-46
		3.2.4.6 EXEC DATA File	3-46
	3.2.5	Summary of Data Input	3-50
3.3	MODEL 1	EXECUTION	3-56
	3.3.1	Data Requirements	3-56
	3.3.2	Execution Statement	3-57
	3.3.3	Examples of Execution (Output)	3 - 57
3.4	MODEL '	"VALIDATION"	3-59
	3.4.1	General	3-59
	3.4.2	Model Application	3-59
	3.4.3	Model Calibration	3-61
	3.4.4	Model Validation	3-66

Page

	3.4.5 Model Sensitivity Analysis 3.4.6 Model Limitation 3.4.7 Discussion	3-67 3-68 3-69
3.5	REFERENCES	3-70

FIGURES

3-1	Environmental Pathways of Toxic Substances	3-4
3-2	Ultimate Developmental Features of SESOIL	3-7
3-3	Complete Disaggreation; LEVELO, LEVEL1, and LEVEL2 Operations	3-9
3-4	Conceptual Compartment Discretization for the LEVEL3 Operation	3-11
3-5	Schematic Presentation of SESOIL Operations	3-14
3-6	GE DATA File	3-17
3-7	LO DATA File	3-32
3-8	LL DATA File	3-37
3-9	L2 DATA File	3-40
3-10	L3 DATA File	3-47
3-11	EXEC DATA File	3-48
3-12	Schematic of Model Application/Calibration/ Validation	3-65

TABLES

3-1	GE DATA File	3-51
3-2	LO DATA File	3-52
3-3	L1 DATA File	3-53
3-4	L2 DATA File	3-54
3-5	L3 DATA File	3-55
3-6	Soil Modeling Major Input	3-60
	Parameter Categories	
3-7	References of Current Research in Calibration/	3-63
	Validation Procedures for Soil/Groundwater Models	

3.1 INTRODUCTION

3.1.1 General Capabilities

This version of SESOIL can be used as:

- a hydrologic cycle model for the watershed and the unsaturated soil zone of the compartment
- (2) as a hydrologic and pollutant cycle model for the unsaturated soil zone of the compartment.

SESOIL is still under development, therefore the following sections only guide a user through details and the input/output data of this version of the model. This section (User's Manual) might be expanded in the future to a self-contained document with guidelines for problem identification, problem structure, optimal compilation of input data, model validation and calibration procedures.

3.1.2 Phenomenology in the Soil Compartment

A soil "compartment" (or cell) is defined as a soil column extending between the ground surface and the bottom of the "upper" saturated soil zone. As such, the soil compartment interacts with the air and the water compartments of an environment as schematically shown in the cover figure of this document (also Figure 2-1). It is evident that the upper saturated soil zone might be underlain by impermeable soil layers and other saturated soil zones (or aquifers); however, these zones are not part of the soil compartment as previously defined.

Physical and chemical processes or phenomena of importance to the quality of a soil compartment are the hydrologic cycle, the sediment cycle, the biologic cycle, and the pollutant cycle. Processes important to each cycle are described in the appendices HY through PT of this documentation.

When released into the environment, pollutants move by a number of fate (transport/transformation) mechanisms. For some pollutants such as phosphorus, ammonia and certain pesticides, surface runoff, soil wash and dust particles might be the primary carriers to the final place of deposition. Other pollutants are directly applied to plants and reach the soil through drift, wash-off or when the plant decays. Many pollutants are transported through the hydrologic cycle or the hydrologic mechanisms of watersheds and have a final destination in the water compartment of an environment. The figure of the next page (Figure 3-1) is a generalized pathway diagram of toxic substances in the environment; from source-to-receiver. The SESOIL model deals with processes interacting and related to the soil compartment of the environment; the "elliptic" subcompartment in this figure.



Near/Far Field Paths

#1 Direct discharge pathways.

#2-4 Intermedia discharge pathways (primary, secondary, potential). Degradation of substances can take place in any compartment. Out-of and into basin transfers are not shown. <u>Source</u>: Fiksel et al (1981).

FIGURE 3-1: ENVIRONMENTAL PATHWAYS OF TOXIC SUBSTANCES

The manner in which pollutants enter the hydrologic cycle of the compartment depends on the characteristics of the pollutant source, such as location, time and physical or chemical form of pollutant. Gaseous, emulsified and dispersed airborne pollutants enter water by precipitation and/or dry fallout. Soluble pollutants and/or pollutants merely mixed in the water may then enter the soil. Relatively insoluble pollutants discharged to water or soil either are dispersed or are transported by stormwater runoff or are entrained by wind and subsequently redeposited. Pollutants are also adsorbed and desorbed by soil particles and then can be transported by the water cycle in either state.

A major characteristic of a soil compartment -- as contrasted to a water or an air compartment -- is that the temporal physical and the chemical behavior of the compartment is governed by both; out-compartmental forces such as precipitation, air temperature, solar radiation, and in-compartmental forces/features such as soil structure and biology. This characteristic is also one of the main reasons why soil compartmental mathematical modeling is much more complex than water or air modeling.

When dealing with mathematical modeling of pollutant transport in soil compartments, it is a natural way to study pollution migration in the hierarchical order: (1) hydrologic cycle, (2) sediment cycle, which is primarily governed by the first cycle, and (3) pollutant cycle, which is primarily governed by the two previous cycles. It has been, however, a frequent practice to model pollutant migration based upon soil-moisture migration, so that no soil moisture presence results in no pollutant migration. Although the latter is not the case in SESOIL (eg. appendix VO), the logical hierarchy hydrologic cycle, sediment cycle, pollutant cycle has been followed in this modeling effort.

3.1.3 Mathematical Modeling Issues

*

Environmental mathematical models can be classified in general into: deterministic models which describe the system as cause/effect relationships, and stochastic models which incorporate the concept of risk, probability or other measures of uncertainty. Deterministic and stochastic models may be developed from: observation, semi-empirical approaches, and theoretical approaches. In developing a model, scientists attempt to reach an optimal compromise among the above approaches given the level of detail justified by both the data availability and model objectives.

Deterministic models can be classified into simulation models which employ a well accepted empirical equation, that is forced via calibration coefficients, to describe a system, and analytic models in which the derived equation describes the physics/chemistry of a system. Both the simulation and the analytic models can employ numerical solution procedures for their equations. Although the above terminology is not standard in the literature, it has been used here as a means of outlining some of the concepts of modeling. This SESOIL version employs the stochastic approach for the hydrologic cycle, and the deterministic approach for the other physical or chemical processes. The sediment cycle of this model will also employ the stochastic approach. Parameters of simulation or empirical models are determined by calibrating the model output against large masses of observational data. This procedure involves curve-fitting and leastsquares analyses, and requires extensive field data information. In order to by-pass many calibration difficulties, SESOIL employs primarily theoretically developed stochastic or deterministic routines. Thus, model input variables describe physical or chemical parameters and can be determined or obtained independently either from laboratory analyses, field investigation, handbooks or data bases.

The choice of theoretical stochastic or analytic deterministic models does not imply -- of course -- that these models are always superior to empirical or simulation models. The desire for an alternate approach was of importance, therefore a "modular" structure has been employed for SESOIL, so that a substitution of a particular equation, theory, or subroutine (modules) can be undertaken at any time -- along the course of a model improvement -- and in a straightforward manner. To achieve effectiveness in the modular approach a new, efficient and chemistry strong concept has been employed for the pollutant transport cycle (appendix PT) of SESOIL.

3.1.4 Levels of SESOIL Operations/Capabilities

3.1.4.1 General

Discussions in the following sections are oriented towards the conceptual approaches employed in this methodology and are not intended to fully describe the fundamentals of all the cycles -- hydrologic, sediment, pollutant. For detailed information the reader is referred to the individual appendices.



SESOIL encompasses -- by design -- many features. It can be, for example, a watershed model, an unsaturated soil zone hydrologic model, a sediment transportation model. a soil chemistry model, etc. It can also be operated at annual or monthly time steps, for one-, two-, or three-soil

layers. In the future it may be expanded to incorporate the fate of second phase chemicals in the soil column, or nutrient cycles and sedimentation after each storm event. Many SESOIL potential features are schematically presented on the next page (Figure 3-2); however, only a few features are operational at the present time, and these features are focused around the unsaturated soil zone and are offered to users in "integrated" packages designed as "levels" of operation.

In this model version, the four different levels of operation are LEVELO, LEVEL1, LEVEL2 and LEVEL3. Each level is associated with certain temporal and spatial resolution characteristics, and each has different specific input requirements.



FIGURE 3-2 CURRENT AND POTENTIAL SESOIL RESOLUTIONS, FEATURES, AND CAPABILITIES

Arthur D Little, Inc

3-7

3.1.4.2 LEVELO

LEVELO is the simplest operational version of the model and does not require knowledge of the hydrology of the compartment (area modeled). SESOIL is employed as an unsaturated soil zone model interacting hydrologically (for mass balance purposes) with both the watershed and the groundwater table.

This level can be employed for pollutant transport simulations in "fictitious" compartments. The unsaturated soil zone compartment consists of two distinct layers, as shown in Figure 3-3, namely the upper unsaturated soil zone (or watershed zone), and the lower unsaturated soil zone. The saturated soil zone (groundwater) is not part of a level of operation of this model version.

The simulation is performed <u>annually</u> and for only one year. The user has to <u>input</u>: (1) the annual averaged values of <u>rainfall</u> depth, soil moisture (unsaturated soil zones), infiltration and groundwater recharge depths; (2) the total annual pollution load (pollutant mass) to the compartment; (3) chemical/compound related parameters; and (4) soil related parameters. The <u>output</u> from the model is: (1) the pollutant distribution (i.e. concentrations, mass distribution) in the compartment, and (2) the annual pollution contribution (transport) to other environmental compartments by means of surface runoff, volatilization, leaching, to groundwater, etc. Additional information for use of this level is provided in section PT-3.2. Details of input data formats are given in section 3-3.2. Details of the output are given in section 3-3.3.

LEVELO should be employed with care and only if "real" climatological and field data are available, because the hydrologic parameters (eg. rainfall vs. soil moisture content) are always correlated, though independency is assumed for these input data.

This level has specialized applications, for example: screening of a large numbers of chemicals that have to be compared for their environmental effects when released into non-site specific (fictitious) compartments. tertal MM pre-3.1.4.3 LEVEL1

LEVEL1 is philosophically like LEVELO; however, it has been designed for region <u>specific</u> simulations. The simulation is performed <u>annually</u> and for only one year and <u>requires</u> the knowledge of few annual <u>averaged</u> climatic and soil data for the area, in <u>order</u> for the model to <u>estimate</u> the <u>annual hydrologic</u> parameters relating to the compartment which have been a user input to LEVELO.

The user has to <u>input</u>: (1) climatic/storm parameters, (2) <u>soil para</u>meters which may vary in the two layers, (3) <u>chemi</u>cal parameters, and (4) <u>simulation specific parameters</u>. These data are readily available from the literature (eg. NOAA reports, handbooks). The <u>output</u> from the



FIGURE 3-3: COMPARTMENT DISAGGREGATION; LEVELO, LEVEL1, AND LEVEL2 OPERATIONS model is: (1) the hydrologic cycle components of the annual compartmental water balance, (2) the annual pollution distribution (eg. concentrations, mass to groundwater.)

As in LEVELO simulations are performed in two unsaturated soil zone layers (Figure 3-3) and are of use for specialized applications. Additional information for this level is provided in section PT-3.2. Details of input/output data formats are given in sections 3-3.2 and 3-3.3.

3.1.4.4 LEVEL2

LEVEL2 resembles LEVEL1 philosophically (Figure 3-3); however, operations are performed monthly. Up to 10 years can be simulated. In contrast to the input data required by LEVEL1, the user has to know: (1) the monthly distribution of rainfall depths and other climatic parameters in the area, and (2) the monthly distribution of input pollution (pollutant mass) to the compartment. Most other input data are similar to LEVEL1. The monthly output from the model resembles the LEVEL1 annual output.

Simulations performed with LEVEL2 may reflect site specificity, both in time (averages over month) and area (reflected in the compartment characteristics). Input data are easily compiled from existing data sources (eg. NOAA, handbook, this documentation); however, for site specific simulations, model output has to be compared to actual field data (for both hydrology and chemistry), and eventually calibrated. Validation of final output is essential.

Additional information for this level is provided in section PT-3.3. Details of input/output data formats are given in sections 3-3.0 and 3-3.4.4.

3.1.4.5 <u>LEVEL3</u>



LEVEL3 resembles LEVEL2 in philosophy (monthly simulations); however, operations are performed in three unsaturated soil zone layers. Therefore, in addition to the input data required by LEVEL2, the user has to input pollutant mass and soil characteristics for an additional layer. This fact increases spatial resolution, but also user's required effort. This level has been developed for the needs of this contract and can become a subset of the N-layered monthly SESOIL version (Figure 3-4).

Additional information for this level is provided in section PT-3.3. Details of input/output data formats are given in sections 3-3.2 and 3-3.3.

3.1.4.6 Other Levels

The authors intend to develop additional levels such as:

 a storm-by-storm temporal resolution and a N-layered compartment that may handle release
 of a second phase (insoluble) chemical spill on the soil surface, or



FIGURE 3-4: CONCEPTUAL COMPARTMENT DISCRETIZATION FOR THE LEVEL3 OPERATION
(2) a statistical version to handle pollutant (eg. pesticide) transport on the watershed.

Authors have conceptualized these versions but have not proceeded in ed any real developmental effort. Their main goal will always be a "user friendly" model accepting calibration of parameters describing physical or chemical rate processes (eg. intrinsic permeability of soil, cm²).

3.1.5 Problem Identification/Level Selection

This section will be expanded in the future to contain information regarding: (1) how to identify a problem suitable for modeling via SESOIL, (2) how to select the compartment's temporal and spatial resolution (i.e., level), and (3) how to proceed with the actual modeling (single medium, multimedia).

3.1.6 Canonical/Scenario's Chemical Fate Modeling

Recent concerns related to environmental quality require a methodology to relate sources and quantities of chemical releases into the environment to the actual amounts of these chemicals to which humans and other biota are exposed. SESOIL is extremely well suited for such environmental exposure studies, which are based on "canonical" environmental compartments and employ typical scenarios (single medium or multimedia). This section will discuss such issues in the future. It has to be emphasized that the selection/compilation of typical/canonical compartments is not an easy issue, and <u>has</u> to involve consideration of statistical techniques (eg. kriging, see section 3.4.3) to optimally/appropriately design the compartments.

3.2 DATA STRUCTURE/MANAGEMENT

3.2.1 General

For a user it is important to understand the model data structure and management, the model execution philosophy of the version accompanying this documentation, and the content of the input data files.

3.2.2 SESOIL Program Structure

A generalized flow chart of SESOIL data files and operation is shown on the next page (Figure 3-5). The executive program of operations, designated as SE81, calls the main program designated as SESOIL, and the latter calls the two basic data subroutines RFILE (read data file), PFILE (print data file), and one of the four operational basic subroutines LEVELO (level "O" operations), LEVEL1, LEVEL2 or LEVEL3. Consequently, each of the basic operational routines calls a number of secondary routines such as HYDROA (annual hydrologic cycle), HYDROM (monthly hydrologic cycle), TRANSA (annual pollutant transport routine), TRANSM (monthly pollutant transport routine; two soil layers) or TRANS3 (three soil layers). Finally, each of the secondary routines calls a variety of functional routines, as presented in appendix FC (FORTRAN Code).

A major emphasis is placed into the data management aspects and the easy input of data. Data are read by the model from 6 data files GE (general), LO (level 0), L1 (level 1), L2, L3 and EXEC (executive operation). In the <u>IBM system</u> data files are accessed via the file name and a "DATA" designation; therefore, reference is made in the following sections to the 5 files GE DATA, LO DATA, L1 DATA, L2 DATA, and EXEC DATA (Figure 3-5), which are all expandable in size.

GE DATA file (see section 3.2.4.1) contains: (1) climatologic data of regions, areas or cities, (2) soil data for various soil types, and (3) chemistry specific data and pollutant parameters.

LO DATA data file (see section 3.2.4.2) contains geometric and simulation related information for the LEVELO applications. Ll DATA contains information (see section 3.2.4.3) for using LEVEL1 of the model, and L2 DATA contains the LEVEL2 information/data (see section 3.2.4.4). Information data or parameters that are used for both LEVEL1 and LEVEL2 operations are given twice (double-input), one in each data file, in order to make each level of operation self-standing and self-contained.

EXEC DATA (see also section 3.2.4.6) contains executive simulation data and information for each actual execution of SESOIL, such as "what level of operation is desired?", "where is the area of simulation?", "what type of a soil or chemical compound is involved?".

Detailed information regarding the "loading" of these files is presented in section 3.2.4 and in appendix DF (Data Files).



FIGURE 3-5: SCHEMATIC PRESENTATION OF SESOIL OPERATIONS

•

3.2.3 Model Execution Philosophy

Model execution is accomplished with the following steps:

- The SESOIL user selects/decides for a level of operation (i.e. LEVEL#).
- (2) The user "edits" his basic data files (section 3.2.3.4) GE DATA, L# DATA and EXEC DATA
 - either through an interactive process via a screen terminal (eg. IBM VN-CMS system), or
 - by inserting or changing computer cards from his deck.

Editing of the basic data files involves either the input of non-existing values (eg. new climatological data) in the data files, or the updating of the previous information (eg. another region).

Above files have unlimited expansion capabilities so that data from previous simulations may be saved for future comparative runs.

(3) The user asks for program execution via the statement: <u>SE81</u>

3.2.4 Input Data Files

The following 6 sections (3.2.4.1-3.2.4.6) give the information contained in the 6 data files:

- GE DATA File of general information to be employed by all levels of operations
- LO DATA File containing data for LEVELO executions
- L1 DATA File containing data for LEVEL1 executions
- L2 DATA File containing data for LEVEL2 executions
- L3 DATA File containing data for LEVEL2 executions

EXEC DATA Executive data file of operations

At this point, the reader may find the program/data logic expressed previously to be somewhat confusing; however, this program/data/execution rationale will become clearer after reading the coming sections and paragraphs and the more detailed description of the data files; appendix DF.

3.2.4.1 GE DATA File

3.2.4.1.1 <u>General</u>

This file (Figure 3-6) contains information applicable to <u>all</u> levels of SESOIL operations: LEVELO, LEVEL1, LEVEL2 and LEVEL3. The file is permanent and self expandable with the insert of new data.

Information contained in the file includes:

- Regional Descriptions: climatic, storm data, for areas where the model might be applied,
- (2) Soil Classifications: soil, sediment data of typical or specific soil compartments, and
- (3) Chemistry Data: related to various pollutants whose fate might be simulated.

Regional Description data are <u>not</u> required for a LEVELO operation; however, soil and chemistry data are required.

The file is designed to be self explanatory and provision is made for non-readable (by the computer) statements in order to aid the user. The user can input into the file blocks of annual or monthly climatic data in <u>anv</u> sequence. The program can "spot" the correct areal climate via a user specified index (eg. "17" SITE A (KANSAS), Figure 3-6). This index does not have to be sequentially numbered. Only the actual input data sets (blocks) have to be correctly given to the file. Most of the other labels and text are designed for the user's aid. However, appropriate labeling and numbering is a good practice.

GE data is the largest file, and as such a new user may find this documentation to be intimidating. However, as the input procedures are easier to do than described, users will find that data input is easy once one becomes familiar with the use of the model.

Note: The SESOIL data files are formatted and thus it is important that data be entered in the appropriate columns, with decimals if real numbers, right justified if integer etc. The last page of Figure 3-6 presents some sample input data as coded for entry.

The model is delivered with a small input data file, so that section headings and labels of each file -- though described as inputs in the following pages -- do not have to be inserted again by users.

Section 3.2.5 presents a summary of all data entries for all files. Readers familiar with data entries may consult only that section.

	i : Lime*	\sim	5E . 1 N		1	.1		• • •	• . U .		·· (, ·	Actual _	Data		· ·,	,
	ile 1		1		<u>,</u>	<u>به د ۲۱۰</u>	• • • •	<u>, , 1·</u>	<u>, , , , , , , , , , , , , , , , , , , </u>	<u>AI.:</u>					. 1 4	
me	onsteen		•	- CPEA - MIAN - Gr (· · · · · · · · · · · · · · · · · · · ·	• 1	14 L I)		ريد . د 1 . د د د ک . ۱		2 101. 2 101.	ر کی رو ای از	· • (•)	•	د د	
Ξ.	رم : (<i>(</i> .	1	· · · · · · · · · · · · · · · · · · ·) (· e	1	<u>1.</u>)	<u></u>		<u></u>	<u>'ı</u>	<u>, , '</u>	<u>().</u> "	• • • •	
	١.,	typical	1	- (+1/) - //- + - // (·	• • • • • • • • • • • • • • • • • •	ль и • 41 л.с			1 3 • 0 0 1 • 74 1 • 74		، ري ر 1 ل ک م ^ر ل	ده معر المانية : - الوار	•••1		• 6 •	
	4 4 5	annual Set	, ,	(. 11 A (~		- 101 - 1	7 1 - 11 1	111-	رد	£ • .	• ذ ب	• • •	- • •	1	1.3-	
-+1				- Lelx 	•••••.• •••••• ••••	4 9 - 74 19 - 43 24 5	• • • •	[•]	12	د مد ا ان مد ان مد	2 119. 2 119.			1.4	ل ل م ا م	
<u>ا</u> و			4 .1	— А(-(-) 11. д. (-	ن - بر شور ((- ر د د د د	نون. جات 1		1	<i>ي د ه</i> د	->• I	č 14.	1 - 1 1	••• • \$	12	**•**1	
JUE	, .			- LIIS - VIJO - VIJO	••••••• -,1 .• *• }	がました。 (ま 11 (ない)	• 7 . • •	,,	.7.J .7.J 1.J.47		C 19 • C - 4 • 7 - 4 •	13 13	ده. باير و ر بر ۹۰ و آ	2.01	1	
10	:		,	.) ۱۰ – ره ۱ ۸ ۲ ۸ ۲)/ -u 	((11	;;:		. 12•3. , ,	د ا. ج	"·· ·	. 1	1	1 • V2	
า๊ โ	•				1	(* C.) (4 丁 (日本一)			ر ۹۰ و ۹۰ ز ان 3 و ۹۰ ز از ۹۰ و از	· · · · · · · · · · · · · · · · · · ·	ບ 3. ບ 3.	100 - 10 200 - 10	1.43	1.5 • 7 •	212.03	
	· . ,	- • .	11.0	$\frac{-\sin(4)}{11}$) (<u>)</u> 	1,	1110	11	<u>.</u> ,	(<u>)</u> ().	U years	of tollow	ing data s	d <u>, (, , , , , , , , , , , , , , , , , , </u>	
ຄ	· · · · · · · · · · · · · · · · · · ·	, ī	Х 4	12+20	ر د به ۲، د ۲ ۱۰ ه ۲۰	یر دلا ۱۱ و	1	12.50	12.00	1 _ = '> _ J = 3 &	12 - 170	1	ل ر و یک ۱٬۱۰۰	1	1 5	T
нт сл	`, , ,	V ž	_,*	· · · · · · · · · · · · · · · · · · ·	لی در . ۱۰ لم در ا	0.10 0.10) - L) - L) - L - L		3.10 0.00	0.10	0.10 (.00	1.10 1.10 0.00	انين ارمان بنان	0.10 0.10	(• 1 (• 1	-1
PAT	•	н 1	р., Т.	1 	· · 1	1 • 1 × 1 •	در اب و	1.00	1	ڑی ہ (ہ' ان ہ در مہ	ン・12 J・ビー		د ۲۰ م ۲۰ <u>۱</u> مه سه مه	11.024	ちゃら1 しゃごご りゃりし	ve a
ק .	. 7		Ţ	ب. ب. ل. ب. ب. ل	, , , , , , , , , , , , ,		ر و به د د ر به د د د ر	10.1	.).ju	3	10 - 11	<u>د بالا</u> بالا بالا ا	ل د و د د			Ļ
le	۔ رب	r sets	- 2 14 - 2 1 - 2	32.66	14.10	14-37	14.5	14.30	110	14.30	14.30	14	1	14.3.	14.3	+
	•	nthly , '.	-,2 -,2	• 5 5 • 6 9 • 1 6	• • • • • • • •	• / / • / /	•). •). •].	• 6, 3 • 1 v	- 1 U	• 10 • 10	• 0 G • 1 9	• 10	• • •	• ()	• 65	2
	• •	al/mo		0.00 1	0.00 +•.7 -1.5	ر ر ـ د. 5 ـ ا ـ د 1 ـ ا ـ) .) 3 . 4.	ر ۹۹ م ۱۰۰۰ م	()).(. 	0.00 2.12	0.00	12 + 1 1 + 4 + 1	0 • 50 • • - 1 • 1 •	U = EJ 3 14: = 17 = 1		9
			r-2	لية د ا براه ت	 	1.1	3).,	ں ں ال ر د ال 1	11.00	33.00	14.00	· • • • • • •	ل ل م ل ۲ م ۲ ک	17	د با د ۱۰ ≀ر د ۱۰	*
		peal 	- ; ; - ;	1.4.Ju		L + • / .	1403	1	1	14	1	1	1 7	14.20	14.3	
	, 2		.44 5	•	• • •	•	• •	• 10 • 1	• • • •	• U • '- U • 1	• 1		ر ۱۰ م مان مان	ر. د د .	• 1 • 1	
			, - , - •	• • • • • • •	· · ·		· · ·		· • J 5	L . L	· • · · ·		J.J.	6	J.U	,
		Us	AI d			ACTION	Jana		see la	ast pape	for input	format	\$	(contin	ued)	

3-17

Arthur D Little, Inc.

			FIGURE	3-6	GF	ATA Fi	le				(Contin	med	
		، سی 1 ، - 1 ، - 1 ، - 1 ، - 1 ، - 1 ، -		· i · · · · · · · · · · · · · · · · · ·	ذ . ا . ا . ا	1 • •	1 + 1 3 0 1 2 0	14 • 10 • 40 • 1- • 1 • 1	14.3.	1 ··	• • • • • • • • • • • • • • • • • • •	14 - 3., - 1 - 1 	1 -
		1572 - 1 17 - 1 17 - 1	1• • 1• 1 • • ↓ ↓ • ↓ 1 • • ↓ • ↓ • ↓	دي. در 1. پ در ۲. پ	• • •) ~ • •) ~	ام ه بابا ه در ۱۰ م بابا	۱۰۱ م رال ۱۰۵ ۱ راد ۱۰۱	۲۹ • ت ت • ک ۲۰ • ۱۹	• 1 5 1 • • • 1 • • •	د 1 . لري . ر ؟	• 1 .1 • . 1 0 5 •	• 1 3 17•00 47 • 1	+1+ ++0, 20++
ie.	· / · · · ·	x - 7 x - 7 x - 7 x - 7 x - 7	د	, .,, , .,, , ,,, , ,,, , ,,,	• • • • • • • • • • • • • • • • • • •	ن . ملاء مرد ما ملاد	۰، د ۱۰۰۰ - ۱۰ ۱۰۰۰ - ۱۰	•10 •10 0•30 ••••	• 1 · · · · · · · · · · · · · · · · · ·	• 1) • 1) • • • } • • • •	• • • • • • • • • • • • • • • • • • •	• 1 • 1 • 1 • 1 • 1 • • • 1	، د . . ا . ا . ا
ATH f	2 4 4 4	L-7 x - 7 x - 7	ري، دري 14•5 ، 4•5 • 5 •	u E 4 - 5 3 2 2	14.37	14.50	14 - 3	14.50	14.10	14.30	14.30	1 ·· • > ·) • 5 ·	14.1
6E)	, , , ,	4,2,3,4 - 2, ,17, 4 - 4, ,1,4 - 4, ,1,4 - 4, ,1,4 - 4,	1 4 - 24 - 5 - 5 - k k 1 3 - 65, - 5 - 5 3 66, - 5 - 5	۲ - ۱۶ د ۱۰ - ۱۶ د ۱۰ - ۱	ر ه ۲۰ ۲ ۲ ۲۰ ۲ ۲۰ ۲۰ ۲۰	۰.51 ۱.۱ ۱.۱۰۶۰ ۱.۱۰۶۰	10 - 54 - 10 13 - 09 53 - 50	_د . د <u>.</u> د ا (5 ل ∙ 5	12.32	1++1 +1- +00 	1 ، • ، • 1 • • بر • بر بر • • ب 1	14 • . • 1 > 1 / • C J 3 C • - U	يليك 1-1- 1-5-4- 1-5-4-4-
\mathbf{O}		ر - ر ۱۰ ر - ر ر - ن - ' - ر	• • • • • • • • • • • • • • • • • • •	د د می در ب د می از ز ب می د	ن و م رود م ر د ر ر د ر	ز ۲ ه ر د۰ه ۱ ۱ ه ر ۱، ه ژ	36. 90. 10. 10.	いい。 しつ。 しつ。 しつ。 しつ	• 2 4 • 3 5 • 1 0 • 1 6	رادی . با در ه ۱۰۰ م	・ 54 ・ 55 ・ 14 に・ 55	+ 1 + • 0 0 • 1 + • 1 +	د . بال . د ال .
(m)	• • • •	, 1 - , L - , [11 - ()	30+30 20+4 3++03 14+3 14+3	ز و میں۔ باڈ م ہ ۱		14.34	1 + • 5 •	14.10	14.3.	14	1	14	ر، ۱۰ مر) د ب ما ۹۰
5-6				// //	· · · · ·	د • ، ۱ د • ، ۱ د • ،	12.74	2010 2010 2010 2010 2010		I • • • •	· · · · · · · · · · · · · · · · · · ·	14.14 11 17.1	تين هاي ليري هاي بال هاي
405	?.	1 · - ·	د ۵۰ مالی با ۲۰۱۰ د ۵۰ مالی با ۲۰۱۰ د ۵۰ مالی ۱۰۱۰ با ۵۰ مالی ۱۰۱۰	1 • • • • •	• • •	14 • · · · · · · · · · · · · · · · · · ·	ل ۵۰۰۰ ا ال ۱۰۰۰ ۱۰۰۰ -	14 + 50 + 20 + 20 + 20		ر در . ر در . ر ار . سار .	ر، ۲۰ م ۲۰۱ ب کی م ۲۰ م ۲۰ م	1 % * * 3 u * 4 * 1	14.3 •5 •55 •15
ور اردا اردا اردا	(de)	·· (·· (·· (••••••••••••••••••••••••••••••••••••••	د ۲۰۰ روبه ۲۰۰ روبه ۲۰۰ م	- i - J - J	د ه ب ب ه ب د	د L • ۱ و • ا ر • ال ا	・レッ 3 + はに より・5 い	• 1 * • ()] (•) ,	• 4 - • • - • • • •	• 1 - 3 • 11 • • • • •	• 1 / 1 / • • 2 • • •	• L • بر • • ↓ • • با 5
han Pe	very v los	ب ہے ہے۔ ہے جو بر جی جو بر جو جو		· · · · · · · · · · · · · · · · · · ·	• • •	• 1'	• - 5 • 35 • 15 • 15 • 5 5	• 3(* • * 0 • 1 • • 0 • 0 0	• • • • • • • • • • • • • • • • • • •	• • • • 1• • • • • • • • •	• 44 • 10 • 10 • 10 • 11	• -1 • -1 •	د د د ر . د ل د . د .
		41-5 (- 4 X- 4	3	· , , , , , , , , , , , , , , , , ,	. ۰ . ۰ . ۱	11, • * 4 • • • • •	، ۲۰۰ ۱۰۰ ک	14.20	14.5	· • · ·	· · • ·		1.4.4.
	* '.	· · · · · · · · · · · · · · · · · · ·			•••	• •	1 • · · · · · · · · · · · · · · · · · ·	• • •	1 	••• •••	· · · · · · · · · · · · · · · · · · ·	17.	• • 1 • •

· is a second second second

Arthur D Little Inc

3-18

1..... **GE** (1

a calify da

GE 1. - 11 1 • • 1 1 1 A. . . . 1 ·· - / 1+.) 1. . 1 1 14. . . / 11 (- -1 1 ' - 1 .1 . .1. .14 . 1 • 1 - 1 . 1 .1 . 1 . .10 v 1- 1 1 1 . . . 1. 1... 1 > + + + + + 1.1 • . • , . . 11. -11-3 30.00 ز را و ب ال و م ر د うりゃしし うりゅうひ うじゅうひ きしゅうひ うりゅう 16.0 1... · · · · Sec. 2 . 1 1 1. L - -31.6. 1 - 1 1 1 . . . 1 1.030 14.3 1. 1 .. 11030 14032 14032 14020 1402 1 1 1 11-1 . ; , 10 . 1. . 1 . . 1 . - -• 1... • . . . • 3 , - / . . 1 • -'-• • • • • • · • . 4 • . • • .1. 1-1 1 . 1 . 1 / -1: • L C • 1 1 .. • i -1. - 1 - 1 2 1 - 2 4.16 - • J 1 6... • -.. . 1 1.000 12 - 2 14.14 3.13 3 . + - - 31 1 - - 34 2.52 1. 1.... . 1 3.00 . 4 **.** . *i* 14. . . .1 . - . .1 ' +1 1 .15 • 1 1 .. . 1 - 1 . 1 • 1 -1 · 4- 1 1. 1. 1. 1. 1.11 1 13.0 1. . . 3.60 1.3.05 1. . . 1. ı) . 17.000 4 1 - . 11.00 , 1 51.1. 30a - 11a Ju ر. و (ز 130 ۰. 33011 , L-1J 3 - 14.10 10.00 1 . . . 1 ł 1 + .) 1 + •) 14 . 3 . 1 % . . . 14 . V 14.5 14.3 × .- 1 .: . 1. • .iu . 5'-• • • • • • 2 . • . •~ 5-10 1 ... • 1 • . . • 2 , . 00 . 1 . .() .1.1 () · '341 • 2 : . - 1 .. . 1 . 1 3 · L J .1. .1. -1) .1. . . . • 1 • 1 · · * • • / 5 2.00 < -1. 1.40 1.0 . J = U U • • . . . 1 . . . - + + J 1.00 0.0. . . . U . U ø 1 *** 1 .. 3.1 ' + - 1 1 ... + 2.02 1...3. 1 3.02 1 1 1 31 1 4 • • ア .1 .. . 1 .5 .1.1 . 1 . 11-1-10 .11 • 1 -• 1 .1. . 1 .. . 1 .. . 1 . 1 ... • 1 - 4-1 -1.... ... 1...) • • • • ۱. . . 14.00 1.00 1.000 1.36 17.00 **،** ، 11-16 ۵.۰۰ د بروند بادول باده د بادول ، ز • ر ان از • 30.00 10.000 . うい・しい 30 . .. 3.470 . 1 ٠, ADAL CLAS, LILCALLESS, SILLOSSULMEST DATA : 1 1 2247 11.03 Heart Alt + Garater all + C. 1.30 1.6-10 11.6. 1 . . . 3. 34 1 .4 0.60 C. Sandalla ta ta ta . . . 2 CLAY-LLAS 4 4 . A Catch a Calling Color 1.12 2.51-10 را **دا د ل** . . ! 11.00 4.31 1.00 typical sail set CECE LUESLOESE 11.00 3.63 0.t : 0.30 1 . . 1 4 J JAC YELUN . . . 1.32 1.20-02 12.111 1.0. 3.0. 6.00 . . . C. Lasloa Isasla 1..... しゃしょ 0.60 1 , A .) Y - L !, . . 4.00 . 11 . . - • L J 4.04 1 C.C. M. L. J. M. L.Y. M. L. 0.0. 0.00 12.07 1.00 . 1 1 · Lulu mala J 1 -1 1.1-1 ICILOR RELIANCE 1100.00 100.00 3 ... 3 . / 31. - 3 The Lation and Additions of the second secon 5.04 . 1 - Moleccicontrak 1. Com 6.03 0.00 0.00 -) • C C 0.01 . who and the 2. . . . -1 . . . • . L 5.00 e -12104 1 1 1 THE PROPERTY AND A SEC 6.00 اد م ا . د. ۲ 1. 6.1. Nonlaw is an itsek offer in a U . G. 1 0.00 U ...) 1.01 1.002 • • • • • • • • ر ب و د 6.04 J. U. ٨ 3' Actual Data (continued) User hid Line * typical chemistry set mot ou screek (see next pope) FIGURE 3-6 (cm¹'d) GE DATA FILE

FIGURE 3-6 (milid) GE DATA file

3-19

Arthur D Little Inc



Arthur D Little Inc

3-20

3.2.4.1.2 Data Input

The following data information is entered into the GE file in the following order in the file:

- (1) Climatologic Data
 - (1.1) Annual
 - (1.2) Monthly
- (2) Soil Data
- (3) Chemistry Data
- (4) End of File

Note: Annual and monthly data sets can be interspersed.

In this documentation the following symbols are used:

***** indicates repetition of data lines, or data sets
!!!!! indicates important comments
"..." indicates an emphasis in the word or sentence in quotation

Input parameters for each line, data format, units, and sample input for each line follow.

The line numbers used below (refer to Figure 3-6) are employed for demonstration purposes only and do not appear in the data file.

- (1) Climatologic Data
- Line 1 FORMAT(I1,5X,12A4)

contains the heading

1 REGIONAL DESCRIPTIONS; CLIMATIC STORM DATA;

The index "1" <u>must</u> appear in the first column (#1, see circled number Figure 3-6) of this statement, because it controls the reading of the climatic data.

The user is not concerned with such an entry, since it is delivered with the SESOIL-code.

!!!!!! Following the above statement, either annual or monthly, climatic sets can be given to the file.

(1.1) Annual Climatic Data Inserts

Only one year of data is contained in an annual data set (lines 2-6) below, since the annual data are used for LEVELS 0 and 1 which run for one year only.

Line 2 FORMAT(2X, I3, 1X, 12A4)

contains the heading of an area where the model may be applied

1 CLINTON, MA (SUB-HUMID)

The above is not an executable statement. The number "1" before the area <u>has</u> to appear in the statement.

Line 3 FORMAT(38,6F7.2)

contains the numerical data sets of

- <u>either</u>
 - L [°N] latitude of the area (eg. 42.50)
 - TA [°C] temperature of area at surface (eg. 8.40)
 - NN [fraction] fraction of sky covered by clouds (eg. 0.35)
 - S [fraction] relative humidity of the area (eg. 0.70)
 - A [-] shortwave albedo of the surface (eg. 0.30)
- <u>or</u>

REP [cm/day] evapotranspiration rate of the area (eg. 0.15)

• <u>or</u>

"both" above sets if the user desires so; however, the program will assume the evapotranspiration rates in the area as "known" and will "not" estimate it from the first data set L, TA, NN, S, A.

Note:

- Above data is stored in array CLIMA1(6); see appendix DF

- Only the temperature is required for a LEVELO model execution; however:
- If line 2 above is inserted, then this data (line 3) must be given even with 0.00 values.

Line 4 FORMAT(38X,4F7.2)

contains the numerical values of:

MPA[cm] annual depth of rain (eg. 94.10)

- MTR[day] annual mean storm duration (eg. 0.32)
- MN[-] number of storm events per year (eg. 109.00)
- MT[days] mean length of rainy season (eg. 365.00)

Note:

- Above data is stored in the array spaces CLIMA1(6); see appendix DF, figure DF-2.
- This data is <u>not</u> required for a LEVELO execution.
- If line 2 is inserted, then this data (line 4) must be given, even with 0.00 values.
- Line 5 FORMAT(38X,6F7.2)

contains the numerical values of:

MPM[cm] mean monthly (M) depth of rain of the <u>first</u> six months of the year (October through March) starting with the month of October (M=1); eg. 10.36, 8.96, etc.

Note:

- Above data is stored in the array spaces CLIMA3(1-6).
- This data set is <u>not</u> required for the LEVELO and LEVEL1 executions; however:
- If line 2 is inserted, then this data set (line 5) <u>must</u> be given, even with 0.00 values (required by LEVELO and LEVEL1).

Line 6 FORMAT(38X,6F7.2)

contains the numerical value of:

MPM[cm] mean monthly (M) depth of rain of the remaining six months of the year (April through September) starting with the month of April (M=7); eg. 8.38, 7.82, etc.

Note:

- Above data is stored in the array spaces CLIMA3(7-12).
- This data set is <u>not</u> required for the LEVELO and LEVEL1 executions; however:
- If line 4 is inserted, then this data set <u>must</u> be given, even with 0.00 values (for LEVELO).
- !!!!! Data in lines 5 and 6 are not used in this form and are usually centered as zero. These lines have been left in this version for future use, and to be compatible with previous versions.
- ***** Above data entries (lines 2-6) can be repeated as necessary for multiple sites in this permanent and expandable data file (eg. 2 SITE B (MONTANA),..., etc).

(1.2) Monthly Climatic Data Inputs

Note: It is not necessary to input annual climatic data sets before monthly sets; the two types of data sets may be interspersed.

> The line numbers used are from Figure 3-6. In the actual file these line numbers may change, although their contents will not. The line numbers are for reference only, they are <u>not</u> used by the code.

Line 27 FORMAT(2X,13,1X,12A4,15)

contains the heading of an area/region where the model may be applied

17 SITE A (KANSAS) Oct. '79-Sept '80 10

This is not an executable statement. However, the numbers "17" and "10" have to appear in the statement and at the correct place in the file. The "17" is

the index of the site. The "10" is the index of how many years of data follow this first statement of SITE A (KANSAS) Oct. '79-Sept. '80 (i.e. 10 years).

Following the above statement numerical input data for the climatic parameters for each month and year can be inserted.

Line 28 FORMAT(8X,1F6.2)

This line contains the numerical value of:

L[°N] latitude of area (eg. 39.00)

Lines 29

to 38 FORMAT(8X,12F6.2) for each line

These lines contain the monthly values of the parameter, starting with the October value in column 1 and ending with the following September value in column 12.

- 29 TA[°C] temperature of the area (eg. 12.80)
- 30 NN[fraction] fraction of sky covered by clouds (eg. 0.30)
- 31 S[fraction] relative humidity of area (eg. 0.60)
- 32 A[-] shortwave albedo of the surface (eg. 0.10)
- 33 REP[cm/day] daily evapotranspiration or 0.0 (see Line 3)
- 34 MPM[cm] monthly precipitation (eg. 0.91)
- 35 MTR[days] mean time of rain (eg. 0.22)
- 36 MN[#] mean number of storm events (eg. 1.00)
- 37 MT[days] mean length of rain season (i.e. days in a month). If it rains almost every 3-4 days in a week during the entire month, then MT=365/12=30.50 (eg. 30.50).
- 38 --- This is an empty line for visual purposes. It indicates end of year.

***** Above lines (28-38) can be repeated for the same area
for up to 10 years (by indexing; eg. "10");ic L is given 10-times.
***** The above data set (lines 27-38) can be repeated for
any number of areas (by indexing; eg. "17").

Above data is stored in arrays CLIMM1(6,12,10) and CLIMM2(6,12,10); see appendix DF, figure DF-2.

(2) Soil Data Inserts

Following the climatological input entries, the soil data must be given. In Figure 3-6 climatological entries end in line 137. For this documentation, soil data entry starts therefore with line 138 (circled "2"). Note: Lines 137 and 138 are not the real line #s. They only correspond to Figure 3-6.

Line 138 FORMAT(11,5X,12A4)

This line contains the indexed alphanumeric statement

2 SOIL CLASSIFICATIONS; SOIL, SEDIMENT DATA:

The index "2" has to appear in the first column of the statement. Following this statement, numerical input data for the various soil types are given.

Line 139 FORMAT(2X, I3, 1X, 12A4)

contains the description of the indexed soil type whose data follow; eg.

1 CLAY

Note:

- The index should always appear (eg. "1")
- This type of data is required by <u>all</u> levels of operation (LEVELO - LEVEL3); therefore, <u>at least</u> one soil-type data set has to be given to activate the program.
- The alphanumeric title is stored in the array spaces TITLES (5,12); see appendix DF.

Line 140 FORMAT(38X,6F7.2)

contains the numerical values of:

RS[g/cm²] soil density (eg. 1.32)

- Kl[cm²] soil intrinsic permeability (eg. 1.00 X 10⁻¹⁰)
- C[-] soil disconnectedness index (see appendix HY) (eg. 12.00)
- N[-] effective soil porosity (eg. 0.45)
- OC[%oc] organic content of soil (eg. 1.46)
- CC[%cc] clay content of soil (eg. 3.0)

Note:

- Above values are stored in the array spaces SOIL1(6)
- If line 140 is inserted in this file, then this line must be given, even with 0.00 values.
- Line 141 FORMAT(38X,4F7.2)

contains the numerical values of:

CEC[me/100 g soil] soil cation exchange capacity (eg. 15.00) K1U[cm²] intrinsic permeabilities of upper soil layer K1M[cm²] intrinsic permeabilities of middle soil layer K1L[cm²] intrinsic permeabilities of lower soil layer

Note:

- KlU, KlM, KlL are used for LEVEL3; therefore, they do not have to be inserted for other levels.
 If both Kl and the set of KlU, KlM and KlL are given, the program will ignore the later values and will use the value of Kl for all layers.
- This data is stored in array SOIL2(6); see appendix DF, figure DF-3
- If line 140 is inserted in this file, then this line must be input (even with 0.00 values.
- **** Lines 139-141 can be inserted for an "unlimited"
 number of soils. As such they create the soil-part
 of the GE DATA base. Note: "indexing" (eg. "1" CLAY)
 is necessary.
- !!!!! Assume that the soil entries have reached line 150, figure 3-6, 3rd page. Chemistry data inputs will follow in line 151.

(3) Chemistry Data Inserts

Following the soil entries, the chemistry data must be given. In Figure 3-6 soil entries end with line 150. For this documentation chemistry data entry starts with line 151.

Line 151 FORMAT(I1,3X,12A4)

contains the headings of the next category of input data.

3 CHEMISTRY DATA:

Following this statement, chemical descriptions and numerical input data are given. The "3" has to appear in the first column of this line.

Line 152 FORMAT(2X, I3, 1X, 12A4)

contains the name of the compound whose data follow (lines 153-155), eg.

1 1,1-TRICHLOROETHANE

Note:

- The index of the compound (eg. "1") should always appear in this statement.
- This type of input is required for all levels of operation, LEVELO-LEVEL3; therefore, at least one chemical data set should be given into the GE DATA base (even with 0.0 values).
- The name of the chemical is stored in the spaces of the alphanumeric array TITLE(5,12).
- Line 153 FORMAT(38X,6F7.2)

contains the index and the name of the chemical compound and numerical values of parameters associated with it. These parameters are:

SL[ug/mL]	compound solubility in water (eg. 1100)
KOC[(ug/g∝)/(ug/mL)]	adsorption coefficient of the compound on organic carbon (eg. 180.00)
DA[cm ² /sec]	diffusion coefficient in air (eg. 0.04)
KDE[day ⁻¹]	biodegradation rate of the compound (eg. 0.00)

H[m ³ ·atm/°K·mol]	Henry's law constant (eg. 3.93E-3)
K[(ug/g)/(ug/mL)]	averaged adsorption coefficient for the compound on the soil (eg. 0.0)

Note:

- Above data are stored in the array CHEM(18).
- If K is given (different from 0.0) then the program uses as an adsorption coefficient this K, otherwise it uses the KOC.
- Line 154 FORMAT(38X,5F7.2)

This line contains the numerical values of:

MWT[g/mol] molecular weight of compound

VAL[-]	valence	of	compound
--------	---------	----	----------

KNH[day⁻¹] neutral hydrolysis constant

KBH[L/mol·day] base hydrolysis constant

KAH[L/mol day] acid hydrolysis constant

All above entries in Figure 3-6 are 0.0. This data is stored in array CHEM1(18); see appendix DF, figure DF-3

Line 155 FORMAT(38X,3F7.2)

This line contains the numerical value of:

- SK[-] stability constant of compoundligand complex
 - B[#] number of moles of ligand per mole of compound complexed

MWTLIG[g/mol] molecular weight of ligand

Above entries in Figure 3-6 have 0.0 values. This data is stored in array CHEM1(18).

***** Lines 152-155 can be repeated for an "unlimited" number of chemicals. Indexing (eg. "1") is essential. For example, lines 156-159 contain "2"; COPPER values. (4) End of File

Following creation of the chemistry section of the GE DATA file, we designate the "End of File" with (Figure 3-6, 3rd page).

Line 160 FORMAT(I1,5X,12A4) (End of File) 9 END FILE

> This is the last line of the GE DATA file, and should be identified by the number 9 in the first column (#1, Figure 3-6, 4th page).

A summary of all previously discussed data entries is given in one table in section 3.2.5 (Table 3-1).

3.2.4.2 LO DATA File

3.2.4.2.1 <u>General</u>

This file (Figure 3-7) contains information applicable to the LEVELO operations. This file is used in conjunction with the GE DATA file and contains:

- (1) Geometric and other parameters related to a region
- (2) Information relating components of the hydrologic cycle of the area and pollutant transport of the LEVELO simulation.
- (3) Pollutant (and other chemical) loadings.

Because the LO DATA file is used in conjunction with the GE file, it is assumed that the user is familiar with the presentation of the input data of the previous section 3.2.4.1 (GE DATA File), and therefore input data descriptions are only briefly presented. A reminder: the SESOIL code is delivered with a basic data base that facilitates additional data entry in terms of format.

As was done for the GE DATA file, a summary of all inputs of the LO DATA file is presented in section 3.2.5 of the user's manual. Users familiar with the input data concepts of SESOIL may consult only the summary section.

3.2.4.2.2 Data Input

Spacing and format of data are shown in the 2nd page of Figure 3-7. The content of each line is described below. For clarity, input data to this file are described with reference to the figure; line 1 below does not have to be the first line in the file.

Line 1 FORMAT(2X, I3, 1X, 12A4)

contains the heading of the first region of the file, eg.

1 TEST LOCATION

Note:

- The index of the region (eg. "1") must appear.
- The title is stored in the first line of the array TITLES(5,12); see appendix DF.



See next page for spacing format

FIGURE 3-7 LO DATA file

Arthur D Little, Inc



3-3 -33

Arthur D Little, Inc

"Number of forms par ped may vary slightly

Line 2 FORMAT(38X,5F7.2)

contains the numerical data of:

- AR[cm²] surface area of the compartment (eg. 1.00)
 - Z[m] depth to groundwater table for this application (eg. 50.00 meters)
- DU[cm] depth of upper unsaturated soil zone for this application (eg. 0.7576 centimeters)
- PH[-] pH of the upper unsaturated soil zone (eg. 8.00)
- APH[-] ratio of pH lower/upper unsaturated soil zone (eg. 0.875)

Note:

- Above data is stored in the array spaces GEOM(20); see appendix DF.
- Line 3 FORMAT(38X,6F7.2)

contains the numerical data of:

AKDE[-]	ratio: biodegradation rate of compound in lower soil zone to upper unsaturated soil zone (eg. 1.10)
AOC[-]	ratio: organic carbon content in soil in lower soil zone to upper unsaturated soil zone (eg. 1.26)
ACC[-]	ratio: clay content in soil in lower soil zone to upper unsatu- rated soil zone (eg. 1.30)
ISRA[-]	index of surface runoff participation in pollutant transport
	<pre>ISRA = 0 no participation ISRA ≠ 0, any participation ISRA = 1 pollutant in surface runoff</pre>
ASL[-]	ratio of pollutant concentration in rain to maximum solubility in water (eg. 0.40)
ACEC [-]	ratio of CEC, lower/upper soil zone (eg. 0.001)

Note:

- Above data is stored in array spaces of GEOM(20).

Line 4 FORMAT(38X,4F7.2)

contains the numerical data:

POLINU[ug/cm ²]	total pollution load (mass) per unit area (cm ²) per year, entering the compartment in the upper zone (eg. 200.00)
POLINL[ug/cm ²]	total pollution load (mass) per unit area (cm ²) per year, entering the compartment in the lower zone (eg. 1000.0)
LIGU[ug/cm ²]	ligand input mass to upper zone (ug/cm ²) (eg. 10.00)
LIGL[ug/cm ²]	ligand input mass to lower zone (ug/cm ²) (eg. 2000.0)

Note:

- This data is stored in array LOAD(6); see appendix DF.

Line 5 FORMAT(38X,4F7.2)

contains the numerical data:

THA[-] soil moisture content (%) (eg. 9.76)

IA[cm] infiltration (eg. 67.43)

RGA[cm] groundwater recharge (eg. 19.03)

RSA[cm] surface runoff (eg. 35.12)

Note:

- Above data is stored in array RUNLO(6); see appendix DF.
- ***** Lines 1-5 can be repeated for an unlimited number of entries by "indexing" each time the new region (eg. "4" TEST LOCATION)

Line 21 FORMAT(I1,5X,12A4) (End File)

This is the last statement of the LO DATA file, given as

·· 9 END FILE

3.2.4.3 L1 DATA File

3.2.4.3.1 General

This file (Figure 3-8) contains information required to perform a LEVELL simulation. This file is used in conjunction with the GE DATA file, and may contain information existing in other files (eg. GE DATA), in order to make this level of operation a self-contained section for the user. As such, L1 DATA contains:

- (1) Geometric and other parameters related to a region.
- (2) Pollutant (and other chemical) loadings.

It has been assumed again that the user is familiar at this point with the data entry presentation of the previous section; therefore, only brief statements are given below.

3.2.4.3.2 Data Input

Spacing (format) of data and parameter descriptions (Figure 3-8) is described below. For clarity, input data to this file is described with reference to the figure. Line 1 below does not have to be necessarily the first line of the file.

Line 1 FORMAT(2X, I3, 1X, 12A4)

contains the heading:

1 CLINTON, MASS

Note:

- The index (eg. "1") must appear.
- The title is stored in alphanumeric array spaces TITLES(5,12); see appendix DF.

Line 2 FORMAT(38X,5F7.2)

contains the numerical data of:

- $AR[cm^2]$ surface area of the compartment (eg. 1.00 cm²)
 - Z[m] depth to groundwater table for this application (eg. 100.00 meters)
- DU[cm] depth of upper unsaturated soil zone for this application (eg. 15.00 centimeters)



•

A CHARLE D. LITTLE. INC. CLAPUTAILUM CENTER

--

	(1) CLEATCA, AAA (203-40.010):- AK62.00.004.4901- AK62.405.004.1534.0004.00- POLEAJ.POLE10.004.000	1.J) - 1.J 1.J.J	100.0 -10 5.00	$15.00 \\ 0.10 \\ 10.00$	3+6+ 1+6+ 24+63	•075 6.40	0.1
. /	- AF-2-DJ-170-5PH	1.00	50.0	15.00	7.00	1.00	
, (· AFEL + AUL + SCI + I + EN + NOL + SC -	0.00	1.00	0.00	L • U J	U . UI.	
, \	- POLIGU + PULIAL + LIGJ + LIGE	3.24	0.00	0.05	0.63		
	J KANSAS-CUPPEL-TIST CCC						
	- Ak+2+DU+HIT+ AHIT	1.00	50.00	.7570	さ。しい	. 0/5	
301	- ALLERDER/LERESAAAASLARLE	1.00	1.00	1.00	ر ت و ن	0.00	1.00
,	- PULINU PULINE + ISUALISE	100.0	0.00	10.00	5000.0		
	4 TINT LOCATION						
•	- AP . L . UU . Phy APM	10.00	50.00	.7570	3.60	. 75	
	- AKULAACCAACCAISTAAAALAACTC	1.10	1.20	1.30	1.40	6.5	• U C 1
	- PELIQUAPULI ALALIGCALIUC	200.00	1000.0	5.00	0.03		
r (.)	END OF FILL						
, ()	END OF FILL						

This Line # not lu fle

3-37

Figure 3-8. L1 DATA FILE

- PH[-] pH of the upper soil zone (eg. 8.00)
- APH[-] ratio of pH, lower/upper soil zone (eg. 0.875)

Note:

- This data is stored in spaces of array GEOM(20); see appendix DF.
- Line 3 FORMAT(38X,6F7.2)

contains the numerical data of:

- AKDE[-] ratio: biodegradation rate of compound in lower soil zone to upper unsaturated soil zone (eg. 0.10)
- AOC[-] ratio: organic carbon content of soil in lower soil zone to upper unsaturated soil zone (eg. 0.10)
- ACC[-] ratio: clay content of soil in lower soil zone to upper unsaturated soil zone (eg. 0.10)
- ISRA[-] index for pollutant participation in surface runoff

ISRA = 0.00; no surface runoff participation ISRA ≠ 0.00; any participation (eg. 1.4) ISRA = 1.00; surface runoff participation

- ASL[-] ratio of pollutant concentration in rain to maximum pollutant solubility in water
- ACEC[-] ratio: lower/upper cation exchange capacity of soil (eg. 0.01)

Note:

- Above data is stored in the array spaces GEOM(20)

Line 4 FORMAT(38X,4F7.2)

contains the numerical data of:

POLINU[ug/cm²] total pollution load (mass) per unit area (cm²) per year, entering the compartment in the upper zone (eg. 10.00)

POLINL[ug/cm²] total pollution load (mass) per unit area (cm²) per year, entering the compartment in the lower zone (eg. 5.00)

	LIGU[ug/cm ²]	pollutant input upper zone (eg.	mass per uni 10)	t are in	
	LIGL[ug/cm ²]	pollutant input lower zone (eg.	mass per uni 20)	t are in	
	Note:				
	- This data is appendix DF.	stored in array	LOAD(6); see		
****	Lines 1-4 can b of data sets. CLINTON, MA).	oe inserted for a Area has to be i	n unlimited ndexed (eg.	number "1"	
Line 17 (End of	FORMAT(11,5X,124	44)			
File)	This is the last	statement of th	is file, giv	en as	
	9 END OF FILE				
A summary o	of all data entri	ies is presented	in Table 3-3	of section 3.2.	5.

3.2.4.4 L2 DATA File

3.2.4.4.1 General

This monthly data file (Figure 3-9) contains information required to perform a LEVEL2 simulation. This file is used in conjunction with the GE DATA file. L2 DATA contains:

- Soil quality and soil moisture quality data (input) for the months of a year, and
- (2) Pollutant input/transformation data for each month of a year.

In the following sections it is assumed that the user is familiar with data entries in the previous files, GE DATA, LO DATA, L1 DATA; therefore, data entries are described only briefly.

3.2.4.4.2 Data Input

Spacing and format of data are shown in Figure 3-9 and is described below. For clarity, data input is described by means of an example; line 1 does not need to be the first line of the file.

FILL L2 DATA A ATLUS J. ITTER INC. CO-UTATION CENTER

1			5 A 5 - C 3		L):1.	173-35	· • •				Jean	rs followin	ig.		
-	· /	<u> </u>	ANOL	+	. 46.14			1.00	: 0.0	15.	<u>5</u> 7	1.00	× 15		
	1	••••	41. D.	• × 2 < • /	201-21	l		0.00	1.0	υ.	05	1.00	6.6 ~		
	- 1	FLUM	0.00	0.00	1.00	1.00	0.00	0.00	0.00	6.00	0.00	0.00	5.66	V	╞
,		1.1	3.00	0.00	2.00	1.30	3.UC	ა. ას	0.00	0.00	0.00		0.00	UN:	
3	_ x	BUJLINU	J.JU	しょうし	1.00	1.10	0.00	0.50	3064	0.00	0.00	9 · 00	0.00	0.0	
1	۵	JULIL	6.00	ر ا م ا	ر ر. و ز	ى ئى مى ر	1.00	3.00	し・ひょ	6.69	0.00	しょうき	6.00	0.0	1
~	ש	8 1544	U.LC	00	J •80	1.00	し・じょ	J.00	0.00	u. 00	0.06	0.00	6.60	0.5	
:	で	き ふっし	3.60	7.37	J • • • •).)J	2.00	0.00	L.Û.J	0.00	0.05	しょうひ	C.30	0	-1
6 2	-	8 1211.50	0.00	ປະປະ	7.47	1.00	4.0C	ມູ່ມີເ	0.00	00.0	0.00	0.00	U .00	U • J	2 \
11	\$	T T MASE	0.00	1.0)	رۍ د ر	9.5	しして	3.00	6.30	0.00	0.00		J.(J	U•)	je)
1_	۵ I	SJIKU	ບ•ບີປ	ປະປະ	2.20	3.30	J.UU	0.00	3.00	しょじい	0.00	0.00	0.06	0.0	175
1 1	2	ia ji vKL	0.03	J. G. G. G.	1.33	しょうし	ショウレ	2.30	J. 0ù	0.00	0.00		じょしい	C	
1 4	8	f .1.0	0.00	0.00).))	しょうし	0.00	0.00	0.00	5.06	0.06	0.30	6.00	6.0	1
1 >	Ŭ	LISL	0.00		きょうひ	1.00	6.30	0.00	J-06	ししい	じょうい	しょうし	0.00	U • 🤆	
1 .		-curr	0.00	0.00	1.00	1.00	1-01	0.00	0.00	0.00	0.00		0.00	0.0	ト
. i	-	CL 1.2	0.11).3.	زر . ر).jŭ	2.00	7.70	0.00	0.00	じょじし	0.00	6.00	0.0	
1,	0	- JCT 405	J.CC	0.00	5.05	0.30	J.U.	0.00	0.00	C.00	0.00	3.00	0.00	U. • C	
ι,	~	-20-1-6-2	0.60	J.J'	2.01	0.0U	0.00	0.00	J.00	6.63	Ú • • U	6.00	しゅんじ	J • J	
- 1	-5	12435	0.00	1.00	0.00	ົງເປັດ	J.UU	0.00	0.00	0.00	0.00	6.96	L.00	L . I)	'. /
- 1	- 8	الم الم	0.03	0.0.	3.33	ე.ბი	5.00	0.00	J.00	0.00	0.00	ι.Οι	0.00	0.0	*
-1	Ĩ	TRA LUUZ	0.60	2.22)	1.01	1.0	0.00	00	0.00	0.00	6.00	6.00	6.0	5
•	3	TRANSL2	3.00	0.04	1.00	1.00	.) . 00	0.00	0.00	0.00	0.00	0.00	6.00	し・ご	.0
<u>.</u> •	.ÿ.	51 4KU2	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.03	0.00	0.00	e • 1	2
-	7	JINKLL .	0 • 5 C	C • 34	1.00	しょうに	4.00	9.00	3.00	0.00	C - C D	5.55	0.00	0.1	
	-ī-1		2.01	0.50	1.10	v • Jv	2.00	5.20	3.00	0.00	0.00	0.00	6.00	0.13	
			0.06	<u> </u>	<u> </u>	3.00	<u> </u>	3.00	0.00		<u></u>	0.00	0.00	<u> </u>	_لد
23		Z KAN	545-50			13-25	P 141	• • • •	<pre>c ^ `</pre>	• *	<u>U</u>		<i>c u z</i>		•
•	<u> </u>			• J J • • •	• \ ••••			1.00		12.	0.5		0.00		
	"	·	AKUL D. CD	+ 102+1	CC PRA		1 11-	0.00	2 0.0	· · · ·	00 00	1.00	0.00		
	_ z		0.00		0.00		0.000	3.30	J.00	0.00	0.00		0.00	0.0	
11	-4 8		0.00	0.00	1.00	0.00	0.00	J • JU	0.00	0.00	0.00		0.00	0.0	
	24		0.00					0.000	.0423	0.00	0.00		0.00	0.0	
J +	_B,		0.00	0.000 5. 0	1.03	0.00	0.00	0.00	0.00		0.00		0.00	0.0	
· .	ġ 8					0.00	0.05	0.00		0.00			0.00	L • L	
12	0 2		0.00	0.00	0.01	0.00	0.00		1.00	0.00	0.00	0.00			
	శా	1 1 1 1 1 1 1 1	0.00						3.00	0.00	0.00			U • J	
; .	2	A JUNI		0.00	1.01	0.00	0.00		3.30	0.00	0.00	0.00	6.00	0.0	
	5				0.07		1.00		0.00	0.00	0.00	0.00	0.00	0.0	
	-i-'	J. T. STIERE	0.00	0.00		1.01	0.00		.00	0.00	0.00		0.00		
••				0.0	1.00			1.10		0.00			(• 00	0.5	
						0.00				0.00	0.00	0.00	0.00		

FIGURE 3-9 L2 DATA file

3-40

Arthur D Little, Inc.

Line 1 FORMAT(2X, I3, 1X, 12A4, I5) contains the heading related to the region/application for which data follow, eg. 1 KANSAS-COPPER (OCT '78-SEP '80) 2 Note: - the index "1" of the region (or description) must be given - "2" represents the number of years for which a simulation will be performed (2=IYRS). This number may be 1-10. - the title is stored in the alphanumeric array TITLES(5,12); see appendix DF. Line 2 FORMAT(38X,5F7.2) contains the numerical geometric and other data of: $AR[cm^2]$ surface area of the compartment (eg. 1.00) depth to groundwater table for this Z[m] application (eg. 100.00 meters) DU[cm] depth of upper unsaturated soil zone for this application (eg. 15.00 centimeters) pH of upper soil zone layer(eg. 8.00) PH[-] ratio of pH for lower/upper soil layer(eg. 0.875) APH[-] Note: - This data is stored in the array spaces GEOM(20); see appendix DF. Line 3 FORMAT(38X,5F7.2) contains the numerical data of: ratio: biodegradation rate of compound AKDE[-] in lower soil zone to upper unsaturated soil zone (eg. 0.00) ratio: organic carbon content of soil AOC[-] in lower soil zone to upper unsaturated

soil zone (eg. 1.00)

- ACC[-] ratio: clay content of soil in lower soil zone to upper unsaturated soil zone (eg. 0.00)
- FRN[-] Freundlich equation exponent (eg. 1.00)
- ACEC[-] ratio: lower/upper soil zone cation exchange capacity (eg. 0.00)

Note:

- Above data are stored in the array GEOM(20); Appendix DF

Line 4 FORMAT(8X,12F6.2)

contains the numerical data:

CUM(Oct-Sept)[ug/mL] concentration of pollutant in soil moisture of upper zone. If an application is to start with an already polluted column, this concentration should be entered in the month before any loading is specified.

Note:

- These data are stored in the 1st line of the array RUNM1(10,12); see appendix DF.
- Line 5 FORMAT(8X,12F6.2)

contains the numerical data:

CLM(Oct-Sept)[ug/mL] concentration of pollutant in soil moisture of lower zone. If an application is to start with an already polluted column, this concentration should be entered in the month before any loading is specified.

Note:

- These data are stored in the 2nd line of the array RUNM1(10,12); see appendix DF.

Line 6 FORMAT(8X,12F6.2)

contains the numerical data:

	POLINU(Oct-Sept)[ug/cm ²]; monthly pollution load (mass) per unit area (cm ²) entering the upper soil zone.
	Note:
	 These data are stored in the 4th line of the array RUNM1(10,12); see appendix DF.
Line 7	FORMAT(8X,12F6.2)
	contains the numerical data:
	POLINL(Oct-Sept)[ug/cm ²]; monthly pollution load (mass) per unit area (cm ²) entering the lower soil zone.
	Note:
	 These data are stored in the 6th line of the array RUNM1(10,12); see appendix DF.
Line 8	FORMAT(8X,12F6.2)
	contains the numerical data:
	ISRM(Oct-Sept)[ug/cm ²]; monthly index for pollutant appearance in surface runoff.
	ISRM=0 no surface runoff participation
	ISRM≠0 any runoff participation
	ISRM=1 pollutant in surface runoff
	Note:
	 These data are stored in the 7th line of the array RUNM1(10,12); see appendix DF.
Line 9	FORMAT(8X,12F6.2)
	contains the numerical data:
	ASL(Oct-Sept)[-] monthly ratio: concentration of pollutant in rain to maximum solubility in water.
	Note:
	- These data are stored in the 1st line of the array

Line 10 FORMAT(8X,12F6.2)

contains the numerical data:

TRANSU(Oct-Sept)[ug/cm²] monthly amount of pollutant transformed (chemically, biologically or other) in upper soil zone, and not accounted by individually existing model processes.

Note:

- These data are stored in the 2nd line of the array RUNM2(10,12); see appendix DF.

Line 11 FORMAT(8X,12F6.2)

contains the numerical data:

TRANSL(Oct-Sept)[ug/cm²] monthly amount of pollutant transformed (chemically, biologically or other) in lower soil zone, and not accounted individually.

Note:

- These data are stored in the 4th line of the array RUNM2(10,12); see appendix DF.

Line 12 FORMAT(8X,12F6.2)

contains the numerical data:

Note:

- These data are stored in the 5th line of the array RUNM2(10,12); see appendix DF.

Line 13 FORMAT(8X,12F6.2)

contains the numerical data:

Note:

- These data are stored in the 7th line of the array RUNM2(10,12); see appendix DF.

Line 14 FORMAT(8X,12F6.2)

contains the numerical data:

LIGU(Oct-Sept)[ug/cm²] ligand mass input to the upper soil zone

Note:

- These data is stored in the 8th line of the array RUNM2(10,12); see appendix DF.
- Line 15 FORMAT(8X,12F6.2)

contains the numerical data:

LIGL(Oct-Sept)[ug/cm²] ligand mass input to the lower soil zone

Note:

- These data is stored in the 10th line of the array RUNM2(10,12); see appendix DF.
- ***** Lines 4-15 can be repeated up to 10 times (10 years)
 for multi-annual applications of the same site. The
 number of sets of lines 4-15 should be specified as IYRS
 in line 1 (eg. IYRS=2).

Lines 1-15 can be repeated for an unlimited number of siteapplications by "indexing" (eg. 2 KANSAS, SODIUM) the region/application).

Line 43 FORMAT(11,5X,12A4) (End of file) This is the last statement of this file (line 43, figure 3-9)

9 END OF FILE

..

The previous information is summarized also in Table 3-4 of section 3.2.5.

Arthur D Little, Inc.

3.2.4.5 L3 DATA File

This monthly data file contains (Figure 3-10) information required to perform a LEVEL3 application. This file is used in conjunction with the GE DATA file. This file is in structure almost identical to L2 DATA with the exception of additional entries for the third (middle) soil layer.

A user who desires to employ both levels 2 and 3 should load both files, or should load file L3 DATA, copy it into L2 DATA, and consequently eliminate from the L2 DATA the lines he doesn't need for the level 2 simulation.

Because data files L2 and L3 are almost identical, only a file output is presented in this section (Figure 3-10). The user is referred to section 3.2.4.5 for the units of the input parameters or to section 3.2.5 where input information for all files is summarized.

3.2.4.6 EXEC DATA File

This file is used with <u>all</u> levels of operation and controls the execution of the program, as well as the reading of the various data files. As such, EXEC DATA is employed in conjunction with one or more of the previously described data files.

Each line of the EXEC DATA file corresponds to <u>one</u> run of SESOIL. An unlimited number of runs can be specified. Each line of the file contains 8 integer (control) numbers (Fiugre 3-11). Format and description of these parameters are as follows:

Line 1 FORMAT(815)

controls the control variables:

- JRUN[-] incremental number of the run (i.e. 1,2,...)
- LEVEL[-] SESOIL level of operation (i.e. 0-3)

 - JSO[-] soil type (eg. 8, i.e. CLAY-LOAM)
 - JCH[-] chemical compound (eg. 20, i.e. TCE)
- JAPPL[-] application area (eg. 21, i.e. CLINTON, MASS '79-'80)

JYRS[-] number of years to be simulated (eg. 1)

Arth
n D
Little
e Inc

3-47

∓i Gure 3-10 L3 DATA file

: م ب		- (;) r	237 -01 *A<+2+DU *24+2254	• UN • Fr	t.			1.00	50.0	0 .75	., .,	, / u	1.4(
- 			- A 2KUL	K (2 A 2 K (2	60+200	• *	ACC	1.1)	1 • 2	ύ ι.	30 I 55 A	.40 .(c	1.00	1.50
:			0.62	دان و و	د د د	0.00	3096	0.00	ن ، رې	6.61	1000	6.00	C.00	6.7
)		3.00	0.00	0.00				$(\cdot \cdot \cdot \cdot)$	
		POLIS		0.00			1.00	1.00	3.10	1.60	3.03			6.6
1		PULLIN	430.63		J.J.	1.20	3.00	0.00	6.03	0.00	0.00	L.00		0.0
11		123211.1	1000-0	. د. ه. ت.	5.1)		5.00) • (, J	0.50	L. U.U	J.UJ	6.00	し・レッ	U . 4
١c		13-11-	1.00	1.90	1-77	1.00	1.00	1.00	1.00	C.CO	0.00	0.00	C.UU	L.0
		S ASL	C.I	3.00	1.0.1	2.00	J = 10	1.12	0.25	5.16	ပ်နှင့်ပ	· • • •	y.y.	ų. į – 1
ι		2 1 1 1 1 1 1	Jely	0.00	U C		0.00	0.00	0.00		0.00	0.00	6.60	U J 3
1.5		S ISAUSI	0.20		J • JU J • JJ		5 • C 5 6 • C 6	9.00	0.00	4.60	5.00	0.00		
		JULINU	6.4.	L. J.	1.16		0.00	J.00	6.00	6.00	0.00	5.00	0.00	v. P
1 >	3	1.141.6	0.00	6.00	3.00	·	0.00	ບ.ບບັ	0.00	0.00	0.60	0.00	6.60	0.0
Ľì	ð	31466	9.00	0-30	נו 🗤 נ).	6.00	00.00	0.00	0.00	0.35	J.J.L	し・じし	U . J
	의	LICU	10.00	6.00	1.10	0.00	0.00	J.00	0.00	0.00	0.00	L.00	6.60	6.7
1	. 101		20.03	J • . U	ار ر م را	0.25	6.00	0.00	0.00	6.60	0.00	0.00	1.0	Ú • ·)
	Ч							J.00	0.00	0.00	0.00	6.50	0.00	
14	-	C 111 - 2	ŭ.ŭú	6.00	0.00	J.00	0.00	3.Ŭŭ	0.00	C. 00	0.00	6.66	6.00	c.o
	3	JL 1- Z	0.CO	6.30	زَ رَ قَ لَ	0.00	0.00	0.00	0.00	0.00	0.30		0.05	0.0
	ם '	POLINU	e 0.Cu	0.00	3.03	ט 3 🗤 נ	0.00	0.00	0.00	0.00	U.CO	0.00	し・レリ	0.0
1	0	100 1 11	(با و ا	S • 0 2	(ر، ر	2. 34	0.09	0.00	ງ. ທູທ	0.00	ل ي . ب	0.00	0.00	G.) !
20	T	PJLI VL.	2 0.00	0.00	ວ. ແບ	0.00	0.00	2.00	0.60	0.00	0.00	6.00	0.00	C • J
	~	1 3 (S. 14) T 42 (15) 1 1 2		J • J J		0.00	9.00	0.00	0.00	0.00	J.00	0.00	6.00	2.38
	3.	194650	2 6.00				0.00	J • 69	0.00	0.00	0.00		0.00	
52	- 2	TRANSM	2 0.00	3.30	0.00	J.JO	0.00	0.00	0.00	6.00	0.00	0.00	0.00	0.5 Q
3 1	귀	TENISL	2 0.00	V.J.	0.15	0.35).00	5.00	0.00	0.66	3.00	C. C.	0.05	U.U ~~
5.4	+	SLAKUZ	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	L.00	u•ù ľ
		51.1K 42	رل• Ú	ن و ب ب	3.00		J.00	0.30	0.00	0.00	0.00	6.00	0.00	0.0
		51 (KL2		J • 0 0		0.00	J • 0 0	0.00	0.05	0.00	5.00	0.00	0.00	0.0
		119645		3.20	1 • 1 • 1	3.00	0.00	0.00		0.00	0.00	6.00	Č. 00	
		LISCLE	0.00	ŭ. 3 Č	1.00	Ū.Ū.	3.00	J. 00		0.00	2.02	0.00	3.00	0.0
• • •		LU1-3	0.00	3.00	0.00	3.30	0.00	0.00	0.00	0.00	0.00	0.00	0.00	C.J .
• 1	- 1	211-3		0.00).Ju	1.00	Ĵ.UC	ა. მა	4.60	6.00	0.00	L. JU	6.66	6.0
• • -		CL 1- J	0.00	0.30	J.)J	0.10	ب ن .و	0.00	0.00	0.60	0.00	0.00	6.00	0.0
•••		20.1.0			1.J.J.J.		0.00	0.00	0.00				0.00	
	1		3 0.00			1.33	0.30	0.00	6.64	0.00		6.00	6.00	<i>i</i>
• •		1-11-1			J. 1 J.	0.00	6.02	3.00	3.00	6.00	0.C 0	6.30	6.63	U. U.
. 7		ハッレージ	0.61	1.30	0.00	0.00	0.30	2.03	0.00	0.00	0. 00	0.00	0.00	℃ •∪‴)
• =		TRAHSU.	: 0.00	J . K	0.00	1.00	1.00	0.30	0.00	0.00	0.00	0.00	C.00	4.0 J
• •		1 (4 1 : 5 -	5 C.CC	1.1	ر . (0.20	3.00	3.03		0.00	0.00	6.00	K •3 8
				1.1	1.00	0.00	0.00			0.00	0.00	5.00	6.00	113 A
• •		STAK J.	0.00	(),)		6.30	9.60	3.40	0.00	0.00	0.00	10	6.00	0.1
3		SI INL.	6.5	0.00	J.J.	5.52	0.00	6.0C	U.00	0.00	ů.Ju	ί.ίΰ	0.00	Ū.J
۰ د		LISCUS	0.01	1.11	0.00	0.33	3.00	5.00	9.00	6.00	J. 96	1.00	0.00	0.0
	N	CISC#3	0.03	0:00	0.00	3.36	0.01	J.00	0.00	0.00	0.00	し、しい	し・しつ	
	u u				۰.	•		1	5				4 . 1 . 1	


(Expandable & Permanent) **FIGURE** 3-11

File FXEC JATA

**** Multiple runs are specified with multiple run/line entries.

END OF FILE The number 999 [FORMAT(I5)] indicates end of file.

3.2.5 Summary of Data Input

Table 3-1 summarizes all input data of the GE DATA file.

Table 3-2 summarizes all input data of the LO DATA file.

Table 3-3 summarizes all input data of the Ll DATA file.

Table 3-4 summarizes all input data of the L2 DATA file.

Table 3-5 summarizes all input data of the L3 DATA file.

Note: Only 1 region, 1 soil type, 1 chemical and 1 application area are shown in the following tables. Following his familiarization with SESOIL, the user may consider consulting only the following tables to load data.

GE DATA FILE

Line		I1	nput Parameter/Variable	FORTRAN Format
Section 3.2.4.1.2 1		$\left(1\right)$	Regional Data	11,5X,12A4
2	۳L	\bigcup_{1}	Regional Title	2X,13,1X,12A4
3	Dat		L TA NN S A (REP) ¹⁾	38X,6F7.2
4	al		MPA MTR MN MT	38X,4F7.2
5	nuu		MPM(OCT) MPM(MAR)	38X,6F7.2
6	۹ L		MPM(APR) MPM(SEP)	38X,6F7.2
27	Г	, 2	Regional Title(IYRS)	2X,13,1X,12A4,15
28		Г	L	8X,1F6.2
29			TA(OCT) TA(SEP)	8X,12F6.2
30		/	NN (OCT) NN (SEP)	8X,12F6.2
31			S(OCT) S(SEP)	8X,12F6.2
32	ta	1	A(OCT) A(SEP)	8X,12F6.2
33	Da	IYRS	REP(OCT) ¹ REP(SEP)	8X,12F6.2
34	hly	(Sets)	MPM(OCT) MPM(SEP)	8X,12F6.2
35	ont		MTR(OCT) MTR(SEP)	8X,12F6.2
36	Σ		MN(OCT) MN(SEP)	8X,12F6.2
37	L	L	MT(OCT) MT(SEP)	8X,12F6.2
38		~	Empty Line	· · · · . ¹⁾
138		(2)	Soil Data	I1,5X,12A4
139		\smile_1	Soil Title	2X,I3,1X,12A4
140			RS K1 C N OC CC	38X,6F7.2
141			CEC KIU KIM KIL	38X,4F7.2
151		(3)	Chemical Data	11,5X,12A4
152		\bigcup_{1}	Chemical Title	2X,I3,1X,12A4
153			SL KOC DA KDE H K	38X,6F7.2
154			MWT VAL KNH KBH KAH	38X,5F7.2
155		\frown	SK B MWTLIG	38X,3F7.2
160		(9)	End of File	11,5X,12A4

1) see section 3.2.4.1.2.

3-51

,

LO DATA file

Line		<u>Input Parameter/Variable</u>	FORTRAN Format
Section 3.2.4.2.2			
1	٦Ū	Application Area / Title	2X,13,1X,12A4
2		AR Z DU PH APH	38X, 5F7.2
3	Set	AKDE AOC ACC ISRA ASL ACEC	38X,6F7.2
4	Area	POLINU POLINL LIGU LIGL	38X,4F7.2
5		THA INF RGA RSA	38X,4F7.2
21	9	End of File	I1,5X,12A4

Ll DATA file



L2 DATA file

Line			Input'Parameter/Variable	FORTRAN Format	
Section 3.2	2.4.4	. 2			
1		Γ (1)	Application Area / Title IYRS	2X,13,1X,12A4,15	
2			AR Z DU PH APH	38X,5F7.2	
3			AKDE AOC ACC FRN ACEC	38X,5F7.2	
4			CUM(OCT) CUM(SEP)	8X,12F6.2	
5			CLM(OCT) CLM(SEP)	8X,12F6.2	
6			POLINU(OCT)POLINU(SEP)	8X,12F6.2	
7			POLINL(OCT) POLINL(SEP)	8X,12F6.2	
8			ISRM(OCT ISRM(SEP)	8X,12F6.2	
9		(RS ets)	ASL(OCT) ASL(SEP)	8X,12F6.2	
10	Set	I) (se	TRANSU(OCT) TRANSU(SEP)	8X,12F6.2	
11	63		TRANSL(OCT)TRANSL(SEP)	8X,12F6.2	
12	Are		SINKU(OCT) SINKU(SEP)	8X,12F6.2	
13			SINKL(OCT) SINKL(SEP)	8X,12F6.2	
14			LIGU(OCT) LIGU(SEP)	8X,12F6.2	
15			LIGL(OCT) LIGL(SEP)	8X,12F6.2	
43		9	End of File	I1,5X,12A4	

•

3-54

L3 DATA file

Line		Input Parameter/Variable	FORTRAN Format
Figure 3-10	-	Application Area (Title and	
1		Application Area/litte lyrs	2X, 13, 1X, 12A4, 15
2		AR Z DU DM FRN	38X,5F7.2
3		рн А2рн Арн	38,3F7.2
4		A2KDE AKDE A2OC AOC A2CC ACC	38X,6F7.2
5		A2CEC ACEC	38X,2F7.2
6		CUM	8X,12F6.2
7		CMM	11
8		CLM	11
9		POLINU	
10		POLINM	81
11		POLINL	11
12		ISRM	11
3	s)	ASL	
14 6	set 1	TRANSU	11
15	KS (TRANSM	n
16 🐇	I AI	TRANSL	"
17		SINKU	"
18		SINKM	"
19		SINKL	11
20		LIGCU	
21		LIGCM	
22		LIGCL	11
	~		

57

9 End of File

I1,5X,12A4

3.3 MODEL EXECUTION

3.3.1 Data Requirements

Input of data to the model consists of "adding to" and "editing" the various data files previously described. The following data files and corresponding lines (blocks of data) have to be included for a simulation run (see Tables 3-1 through 3-5).

LEVELO Operations

• GE DATA

Lines 1, 138, 151, 160 -- Titles Lines 2, 3(only TA), and 4, 5, 6 with 0.00 -- Climate Lines 139, 140, 141 -- Soil Lines 152-155 (B=1) -- Chemistry

• LO DATA

at least 1 set (5 lines)

• EXEC DATA

at least 1 line

<u>LEVEL1</u> Operations

• GE DATA

Lines 1,138, 151, 160 -- Titles Lines 2, 3, 4, and 5, 6(with 0.0) -- Climate Lines 139, 140, 141 -- Soil Lines 152-155 (B=1) -- Chemistry

• L1 DATA

at least 1 set (4 lines)

• EXEC DATA

at least 1 line

LEVEL2 Operations

• GE DATA

Lines 1, 27, 138, 151, 160 -- Titles and Lines 28-38 -- Climate Lines 139, 140, 141 -- Soil Lines 152-155 -- Chemistry

o L2 DATA

at least 1 set (15 lines)

Arthur D Little, Inc.

• EXEC DATA at least 1 line

LEVEL3 Operations

• GE DATA

as in LEVEL2 operations

• L3 DATA

at least one set (22 lines)

• EXEC DATA

at least one line

3.3.2 Execution Statement

The user has to give the statement SE81, via a terminal or a card to start the model execution.

3.3.3 Examples of Execution (Output)

The output is intended to be self explanatory and presently it provides information on

- all simulation input data (application, climatic, chemical, soil, other) employed in a particular run/execution
- (2) the hydrologic cycle components (estimated or assumed in LEVELO), and the monthly or annual pollutant cycle in two or three zones of the compartment, depending upon the level of operation
- (3) pollutant concentrations (in soil-moisture, soilair and adsorbed on soil) in the various soil zones modeled
- (4) pollutant masses in the various phases and zones within the soil compartment and released to air and groundwater

(5) averaged and/or totaled behavior of above cycles over a year of monthly simulations via LEVEL2 or LEVEL3.

The SESOIL code is released (upon request) with four pre-programmed runs (see editing of data file EXEC). The user can give the statement SE81 and receive an output for levels 0,1,2 and 3. Periodically developers undertake aesthetic improvements of the output. If necessary they will notify receivers of their tape, and/or provide updated model versions.

A typical input/output is presented in appendix AP (Applications)

3.4 MODEL "VALIDATION"

3.4.1 General

Model output validation is essential to any modeling effort. However, model "validation" is a very broad term and may include model verification, application, calibration, validation and frequently sensitivity and model capabilities. The definition of these terms and the procedures needed to accomplish these objectives is discussed in this section.

Model verification is defined as "the action during which model computer code is run to extremes and model equations are applied to boundary conditions to assure proper code operation" under all potential climatic, soil and other input parameters. Such an action has been undertaken by the developer (see also section 1.1) and users should not be extremely concerned with it. Therefore, only the remaining issues are discussed below.

3.4.2 Model Application

Once a verified model has been obtained, data have to be compiled and input to the model for the "first" application (i.e. model application). Input data can be compiled from:

- site specific investigations and analyses (eg. leaching rates of pollutants, soil permeability);
- national data bases (eg. climatological data from the NOAA); and
- other sources (eg. diffusion rate of pollutants from handbooks).

Compilation of input data for site specific computer runs are model specific, geohydrology and chemistry specific. Some data categories are pollutant source data, climatological data, geographic data, particulate transport data and biological data. Table 3-6 presents some parameters associated with each category.

SOIL MODELING MAJOR INPUT PARAMETER CATEGORIES

CLIMATE:

Evapotranspiration Temperature Latitude Sunlight Plant Cover Humidity Cloud Cover Wind Precipitation

SOIL:

Porosity Density Hydraulic Conductivity Permeability Adsorption Capacity Organic Carbon Content Clay Content

GEOGRAPHY:

Slope Surface Storage Terrain Area Coordinates

SOURCES:

Leaching Rates Release Mechanisms Patterns of Operation (continuous, batch) Locations Compilation of input data can be relatively straightforward for SESOIL, since SESOIL employs parameters with a physical/chemical meaning. However, time and spatial resolution input data are user-decision input parameters and have to be determined with a previous understanding of the hydrogeology, soil and pollutant characteristics.

Expected outputs from the SESOIL model are:

- temporal and spatial pollutant concentration distributions in soil-air, soil-moisture;
- temporal and spatial pollutant concentration distributions on soil particles; and
- leachate (pollutant mass) migration from the unsaturated soil zone to groundwater.

However, the first application of SESOIL can not be expected to match monitoring records. The common procedure prior to seeking "final" model output is the performance of a number of model runs associated with model applications, model calibrations and a final model validation. This is an iterative operational procedure as discussed in the following sections.

3.4.3 Model Calibration

The calibration, or identification, of a model is the process in which the various model parameters (and that may also include its geometry, inputs, etc.) are redetermined -- although knowledge of them is available from the application stage -- or verified (if such information is available).

The calibration is based on data obtained from observation of the behavior of the simulated "regime" (eg. water balance of basin) in the past. Such data usually include:

- soil moisture, soil infiltration or percolation rates; and
- water levels at gaging station of the basin.

As discussed in Appendix PT, unsaturated models are mathematically structured by:

- developing a flow (moisture movement) submodel;
- developing a quality (pollutant transport) submodel; and
- interfacing the above two submodels.

Calibration procedure, in the context of SESOIL, applies to the flow model part of a code and, therefore, is <u>defined</u> as "the effort (parameter estimation) towards a historical matching of the climate, soil and water balance of a basin."

The calibration procedure is often referred to as the "inverse problem." Methods of solving this problem are discussed in the literature. Table 3-7 lists a number of current research efforts that are oriented towards calibrating mainly groundwater models. The same techniques might be applied for unsaturated soil zone models, although unsaturated soil zone modeling is complicated and no single mathematical method can optimally be applied. The following general discussion presents the concept of calibration as applicable to the SESOIL model.

When performing simulations, two different systems are being compared: (1) soil column and (2) the (conceptual) model. Data are taken from the first system, say, on basin annual yield (Eagleson 1978), in order to calibrate the latter. Roughly speaking, the calibration procedure for the model consists of finding a parameter set (intrinsic permeability, porosity) that minimizes deviations between observed and calculated values of annual yields. Least square's deviation is one of the methods employed in the literature. Other methods are linear programming, quadratic programming, and dynamic optimization. The least square's criterion may be written as:

> Minimize $\sum_{i=1}^{n} [Y(x,y,t) - Y(x,y,t)]^2$ observed calculated i

where i = 1, . . . n, and n is the number of observed yield values.

The statistical analysis of parameter estimates and model predictions are very promising areas of current research. The mathematics involved, unfortunately, tend to be rather advanced and may be beyond the scope or needs of this modeling effort. Many of the parameter estimation techniques require both initial estimates of the soil cell parameters and their statistical properties. This has stimulated an interest in obtaining the statistical properties directly from field data. One of the more promising procedures is "kriging," which is a stochastic interpolation technique (Delhomme 1979), the developers are planning to use.

Calibration is not a single process, neither is it a process that can be designed step by step <u>a priori</u>. As more data become available, the calibration process should be repeated leading to improved model parameters. A schematic of the proposed calibration procedure is shown in Figure 3-12.

Table 3-7

References of Current Research in Calibration/Validation Procedures for Soil/Groundwater Models

Aguado, E.; N. Sitar; and I. Remson (1977). Sensitivity Analysis in Aquifer Studies. Journal of Geophysical Research, Vol. 13, No. 4, p. 733.

Bachmat, Y. and A. Dax (1979). An Iterative Method for Calibrating a Hulticell Aquifer Model. Nater Resources Research.

Brakensiek, D.L. and C.A. Onstad (1977). Parameter Estimation of the Green and Ampt Infiltration Equation. Water Resources Research, Vol. 13, No. 6, p. 1009.

Cooley, R.L. (1977). A Method of Estimating Parameters and Assessing Reliability for Models of Steady State Groundwater Flow. 1. Theory and Numerical Properties. Water Resources Research, Vol. 13, No. 2, pp. 318-324.

Cooley, R.L. (1979). A Method of Estimating Parameters and Assessing Reliability for Models of Steady State Groundwater Flow. 2. Application of Statistical Analysis. Water Resources Research, Vol. 15, No. 3, pp. 603-617.

Cooley, R.L. and P.J. Sinclair (1976). Uniqueness of a Model of Steady-State Groundwater Flow. Journal of Hydrology, Vol. 31, pp. 245-269.

Delhomme, J.P. (1979). Spatial Variability and Uncertainty in Groundwater Flow Parameters: A Geostatistical Approach. Water Resources Research, Vol. 15, No. 2, pp. 269-280.

Delhomme, J.P. (1978). Kriging in the Hydrosciences. Advances in Water Resources, Vol. 1, No. 5, p. 251-266.

Dettinger, M.D. and J.L. Wilson (1981). First Order Analysis of Uncertainty in Numerical Models of Groundwater Flow. 1. Mathematical Development. Water Resources Research, Vol. 17, No. 1, pp. 149-161.

Gambolati, G. and G. Volpi (1979). A Conceptual Deterministic Analysis of the Kriging Technique in Hydrology. Water Resources Research, Vol. 15, No. 3, pp. 625-629.

Haverkamp, R. and M. Vauclin (1979). A Note on Estimating Finite Difference Interblock Hydraulic Conductivity Values for Transient Unsaturated Flow Problems. Water Resources Research, Vol. 15, No. 1, p. 181.

Table 3-7 (continued)

Hayhoe, H.N. (1978). Study of the Relative Efficiency of Finite Difference and Galerkin Techniques for Modeling Soil-Water Transfer. Water Resources Research, Vol. 14, No. 1, p. 97.

Hefez, E.; U. Shamir; and J. Bear (1975). Identifying the Parameters of an Aquifer Cell Model. Water Resources Research, Vol. 11, No. 6, p. 993.

Kohberger, R.C.; D. Scavia; and J.W. Wilkinson (1978). A Method for Parameter Sensitivity Analysis in Differencial Equation Models. Water Resources Research, Vol. 14, No. 1, p. 25.

McElwee, C.D. and M.A. Yukler (1978). Sensitivity of Groundwater Models with Respect to Variations in Transmissivity and Storage. Water Resources Research, Vol. 14, No. 3, pp. 451-459.

Murty, V.V.N. and V.H. Scott (1977). Determination of Transport Model Parameters in Groundwater Aquifers. Water Resources Research, Vol. 13, No. 6, p. 941.

Navarro, A. (1977). A Modified Optimization Method of Estimating Aquifer Parameters. Water Resources Research, Vol. 13, No. 6, p. 935.

Nutbrown, D.A. (1975). Identification of Parameters in a Linear Equation of Groundwater Flow. Water Resources Research, Vol. 11, No. 4, p. 581.

Sagar, B.; S. Yakowitz; and L. Duckstein (1975). A Direct Method for the Identification of the Parameters of Dynamic Nonhomogeneous Aquifers. Water Resources Research, Vol. 11, No. 4, p. 563.





.

Calibration of unsaturated soil zone models can be uncertain and difficult because climate, soil moisture, soil infiltration and percolation are strongly interrelated parameters that can be difficult and/or expensive to measure in the field. Therefore, calibration of unsaturated soil zone models is frequently associated with a model validation (described in the next section).

For the available version of SESOIL that employs Eagleson's (1978) annual water balance theory (as expanded to monthly simulations and with moisture transfer budget in the course of the months), authors recommend model calibration by varying the intrinsic permeability (k), the pore disconnected-ness index (c) and the porosity (n) of the soil; all input parameters to the model (see appendix ID, section 3.0).

The outlining of steps for the calibration procedure of SESOIL is a feasible task; however such a task would require a thoughtful elaboration of this issue by the authors prior to outlining it in this document.

3.4.4 Model Validation

Frequently, the definitions of calibration and validation are synonymously employed in the literature because of the large number of nonvalidated but calibrated groundwater models and the limited number of noncalibrated but validated unsaturated soil zone models. For SESOIL , model validation is defined as "the process which analyzes the validity of final model output." In SESOIL, the validity of the predicted pollutant concentrations would be compared to available knowledge of measured pollutant concentrations from monitoring data (field sampling).

A disagreement in absolute levels of concentration (predicted versus measured) does not necessarily indicate that either method of obtaining data (modeling, field sampling) is incorrect or that either data set needs revision. Field sampling approaches and modeling approaches rely on two different perspectives of the same situation.

Field data give concentrations at points in time and space, models predict "average" concentrations for a particular assumed set of conditions. Thus, field and model results may differ and still both be correct. Some possible reasons for a discrepancy are:

 The field sample was taken from a spot with atypical concentrations (eg. a water sample may be close to an unidentified confounding source, and so give abnormally high readings).

- The sample was taken under a typical conditions (eg. on the one day/month that it rained); model results were calculated for average conditions, which may rarely occur (eg. at the "average" soil moisture content which occurred only for a short period).
- The sample contained interactive compounds (eg. a water sample contained some sodium that may have resulted in increased soil permeabilities).
- The extraction procedure for the sample was under or over efficient (eg. it not only extracted all organic pollutants from a soil sample but also dissolved the soil).

Mathematically, the available model validation procedures and techniques are similar to those presented in the previous section. Following a validation procedure with good field data, "no better model predictions" can be made. This will be the "best possible" output.

For SESOIL, the approach -- at the present time -- would be to:

- apply simplified mathematical techniques (as described above for calibration);
- exercise the professional experience, gained from original model application work to refine the results; and
- document the validation logic for the SESOIL level employed.

A schematic figure of the previously discussed processes is shown in Figure 3-12.

3.4.5 Model Sensitivity Analysis

It is frequently worthwhile to perform sensitivity analyses to determine the effect on the predicted concentrations caused by a change in the input parameters. These sensitivity analyses are particularly important when data gaps or uncertain input values exist. It may also be useful to rerun SESOIL to estimate the impact of various site management or design strategies on pollutant distribution and concentrations in the environment. Two main techniques are widely used to perform sensitivity analyses:

- model simulations; and
- analytic techniques.

Model simulations are performed by running and rerunning the model, simultaneously varying the value of one or more parameters following a "scenario" logic. Model concentration predictions may be compared to monitoring data as described in the previous section.

Analytic techniques of linear systems theory (Dooge 1973) and optimization theory (Haimes 1977) may correlate sensitivity of model input (eg. leaching quantity) to model output (eg. soil concentrations) "without" performing multiple model reruns (simulations). An example is given below for an analytic technique (Fiksel et al 1981).

Assume a SESOIL column receiving pollutant input quantifies I in all layers (cells). Assume SESOIL accounting for a linear Freundlich adsorption coefficient and assume that average predicted adsorbed concentrations in the N cells are the c(N). Then the following matrix, linear response function F(N) can be written:

 $\underline{I(N)} = \underline{F(N)} \cdot \underline{c(N)}$

where F(N) a low triangular (f,0) matrix:

$$\underline{F}(N) = \int f \bigvee_{i=1}^{O} f$$

From the first relation, we have:

 $\underline{F}(N) = \underline{I}(N) \cdot \underline{c}(N)^{-1}$

from which all elements N of F can be estimated. Given a possible linearity of c versus I, we are at a position now to vary input sources of the model and estimate concentration "without" having to run and rerun SESOIL. This approach has not been exercised by the developers; however, they may consider it in the future for certain processes and parameters.

3.4.6 Model Limitation

Any mathematical model is "as good as its weakest link"; therefore, limitations of the model are correlated with the limitations of each of the routines and processes coded (see also section 1.1). In general, limitations can be due to:

- data availability,
- inoptimal single medium model application to a particular site,
- omission of important chemistry reactions, and

lack of appropriate model validation opportunity

Data availability refers to environmental data (eg. climatic, soil, water resource), source data (eg. pollutant leaching from the site), chemistry data (eg. chemical properties of pollutants) and monitoring data (eg. ambient concentrations). It is not appropriate, for example, to drive any soil model without a proper set of weather records and then attempt to validate the model output with available monitoring data.

Omitting certain important pathways or chemistry reactions or processes (eg. volatilization) because of lack of data can become an issue of concern. Such pathways should be evaluated outside the model or, at certain times, another modeled via SESOIL as far as possible. A model sensitivity analysis of omitted (or to be omitted) processes is essential.

3.4.7 Discussion

During the SESOIL development a number of verifications, calibration. validation and sensitivity steps have been performed to one degree or another.

The model code has been verified by extensive testing and under extreme conditions of input data. Each level of operation has been run multiple times and the results have been compared and rectified with sample hand calculations and by other models.

On an earlier contract to the EPA, SESOIL has been applied to two actual land treatment sites at which considerable monitoring data were available. Good agreement was obtained between monitoring data and model results. This study provided to model developer the only validation of SESOIL. These sites were also used for a sample calibration effort where the soil parameters of intrinsic permeability and adsorption coefficients (K,K_{OC}) were calibrated/validated to field records (Bonazountas et al 1981).

One short study for metals has been performed (Bonazountas et al 1981) and another short study for halogenated solvents is underway (Wagner & Bonazountas 1982), both aimed to evaluate the overall fate of pollutants in the soil compartment, including losses to the air and to groundwater. In both studies, canonical/scenario environments were designed by combining a range of climates, soils and pollutants, in order to trace sensitivity of various parameters upon the long-term overall pollutant fate.

In the future model developers hope to extend these efforts, particularly to include increased calibrations and validations with actual field data, an essential task for model improvement and validity.

3.5 References

Aguado, E.; N. Sitar; and I. Remson (1977). Sensitivity Analysis in Aquifer Studies. Journal of Geophysical Research, Vol. 13, No. 4, p. 733.

Bachmat, Y. and A. Dax (1979). An Iterative Method for Calibrating a Multicell Aquifer Model. Water Resources Research.

Bonazountas, M.; J. Wagner; and B. Goodwin (1981). Evaluation of Seasonal Soil/Groundwater Pollutant Pathways. U.S. EPA, Data and Monitoring Support Division, EPA Contract No. 68-01-5949, Task 9, Arthur D. Little, Inc., report.

Brakensiek, D.L. and C.A. Onstad (1977). Parameter Estimation of the Green and Ampt Infiltration Equation. Water Resources Research, Vol. 13, No. 6, p. 1009.

Cooley, R.L. (1977). A Method of Estimating Parameters and Assessing Reliability for Models of Steady State Groundwater Flow. 1. Theory and Numerical Properties. Water Resources Research, Vol. 13, No. 2, pp. 318-324.

Cooley, R.L. (1979). A Method of Estimating Parameters and Assessing Reliability for Models of Steady State Groundwater Flow. 2. Application of Statistical Analysis. Water Resources Research, Vol. 15, No. 3, pp. 603-617.

Cooley, R.L. and P.J. Sinclair (1976). Uniqueness of a Model of Steady-State Groundwater Flow. Journal of Hydrology, Vol. 31, pp. 245-269.

Delhomme, J.P. (1979). Spatial Variability and Uncertainty in Groundwater Flow Parameters: A Geostatistical Approach. Water Resources Research, Vol. 15, No. 2, pp. 269-280.

Delhomme, J.P. (1978). Kriging in the Hydrosciences. Advances in Water Resources, Vol. 1, No. 5, p. 251-266.

Dettinger, M.D. and J.L. Wilson (1981). First Order Analysis of Uncertainty in Numerical Models of Groundwater Flow. 1. Mathematical Development. Water Resources Research, Vol. 17, No. 1, pp. 149-161.

Dooge, J.C. (1973). Linear Theory of Hydrologic Systems. U.S. Department of Agriculture, Washington, D.C. 20402, Technical Bulletin No. 1468.

Eagleson, P.S. (1978). Climate, Soil, and Vegetation (1-7). Water Resources Research, Vol. 14, No. 5, pp. 705-776. Fiksel, J.; M. Bonazountas, H. Ojha; and K. Scow (1981). An Integrated Geographic Approach to Developing Toxic Substance Control Strategies. Arthur D. Little, Inc., Report for the U.S. Environmental Protection Agency, Office of Policy and Resource Management. EPA Contract No. 68-01-6160.

Haimes, Y.Y. (1977). Hierarchiacal Analyses of Water Resource Systems. McGraw Hill International Book Company, New York. ISBN 007-025507-5.

Gambolati, G. and G. Volpi (1979). A Conceptual Deterministic Analysis of the Kriging Technique in Hydrology. Water Resources Research, Vol. 15, No. 3, pp. 625-629.

Haverkamp, R. and M. Vauclin (1979). A Note on Estimating Finite Difference Interblock Hydraulic Conductivity Values for Transient Unsaturated Flow Problems. Water Resources Research, Vol. 15, No. 1, p. 181.

Hayhoe, H.N. (1978). Study of the Relative Efficiency of Finite Difference and Galerkin Techniques for Modeling Soil-Water Transfer. Water Resources Research, Vol. 14, No. 1, p. 97.

Hefez, E.; U. Shamir; and J. Bear (1975). Identifying the Parameters of an Aquifer Cell Model. Water Resources Research, Vol. 11, No. 6, p. 993.

Kohberger, R.C.; D. Scavia; and J.W. Wilkinson (1978). A Method for Parameter Sensitivity Analysis in Differencial Equation Models. Water Resources Research, Vol. 14, No. 1, p. 25.

McElwee, C.D. and M.A. Yukler (1978). Sensitivity of Groundwater Models with Respect to Variations in Transmissivity and Storage. Water Resources Research, Vol. 14, No. 3, pp. 451-459.

Murty, V.V.N. and V.H. Scott (1977). Determination of Transport Model Parameters in Groundwater Aquifers. Water Resources Research, Vol. 13, No. 6, p. 941.

Navarro, A. (1977). A Modified Optimization Method of Estimating Aquifer Parameters. Water Resources Research, Vol. 13, No. 6, p. 935.

Nutbrown, D.A. (1975). Identification of Parameters in a Linear Equation of Groundwater Flow. Water Resources Research, Vol. 11, No. 4, p. 581.

Sagar, B.; S. Yakowitz; and L. Duckstein (1975). A Direct Method for the Identification of the Parameters of Dynamic Nonhomogeneous Aquifers. Water Resources Research, Vol. 11, No. 4, p. 563.

Wagner, J. and M. Bonazountas (1982). Buried Halogenated Solvent Simulations via "SESOIL." U.S. EPA, Office of Toxic Substances, EPA Contract No. 68-01-6271, Arthur D. Little, Inc., report.

HY - hydrologic cycle

APPENDIX HY

HYDROLOGIC CYCLE

To the Reader/User:

Information contained in this appendix is not well documented at all, for reasons explained below, and authors fully recognize this fact.

For the authors -- although the hydrologic cycle governs the model operations -- the full and accurate hydrologic cycle "documentation" has not been of primary importance because:

- Hydrologic cycle development of SESOIL has not been accomplished yet.
- The hydrocycle of SESOIL is primarily based on Eagleson's annual "Climate, Soil, and Vegetation" theory, which is excellently documented in the literature (Eagleson 1978; other publications). Therefore, the author would have spent appreciable effort in representing Eagleson's work.
- Budget constraints of this contract (see section 1.1) led the authors to prioritize documentation of chemistry related and other issues of SESOIL, i.e. original information generated for SESOIL.
- This SESOIL documentation is not to be released in the public domain, therefore, a tentative draft regarding the hydrologic cycle would be sufficient for a reader/user to understand the basic "concept" adapted for the hydrologic cycle; i.e. Eagleson's theory adaptation.
- The hydrologic cycle documentation is of secondary importance to a reader or a user, contrasted to the documentation of other processes, because of the simplicity of input data (and here lies the sophistication and elegancy of Eagleson's theory) required to drive this part (hydrology) of SESOIL.

However: Eagleson's annual theory has been adapted to SESOIL needs by

expanding the work into monthly hydrocycles,

- accounting of moisture storage transfer from month-to-month in the entire compartment,
- not fully employing the vegetational aspects of the basic theory, because the watershed aspects of SESOIL have not been developed yet,
- accounting for zero rain (depth, number of storm events, etc)
- accounting for a "smooth" heterogeneity along the soil column (soil type stratification)
- omitting vegetational and soil surface moisture retention for reasons related to SESOIL needs under the current contract (i.e. a model for overall fate of pollutants) at non-vegetated areas rather than a model for basin-specific pollutant transport on the watershed.
- presenting (model output) only expected values of the statistical distributions of the hydrologic processes model.

The above adaptation issues are not always clearly documented in this appendix. The authors <u>are</u> aware of the deficiencies of this document that was drafted in 1980 (see page HY-1) when SESOIL was conceptualized; however, they intend to improve it when possible.

The SESOIL authors appreciate that Eagleson's work might have been frequently misquoted, paraphrased and mis-duplicated. The issue is not so much of credit given to Eagleson by citing his work, but rather it is one of possible misunderstanding and misuse of his theory. Therefore, readers or users are advised to consult Eagleson's original theory.

					A Contraction of the second se	CON	CEPT	UAL
					APPENDIX HY	JK	AF	1
					HYDROLOGIC CYCLE	(see	section	1.1)
							Page	
	1.0	INTR	ODUCTIO	N			HY-2	
		1 1	Genera	1			HY-2	
		1.2	Hydrol	- ogic Proc	esses Involved		ну-2	
		1.3	Modeli	ng Backgr	ound		HY-5	
	2.0	LEVE	LO - AN	NUALLY "K	NOWN" HYDROLOGIC COMPONENTS	5	HY-7	
		20.0						<i>.</i> .
	3.0	LEVE	1.1 – AN	NUALLY "E	STIMATED" HYDROLOGIC CYCLE		HY-8	
		3.1	Genera	1			HY-8	
		3.2	Defini	tions			HY-8	
		3.3	Annual	Mathemat	ical Analysis		HY-10	
			3.3.1	Assumpti	ons		HY-10	
			3.3.2	Theoreti	cal Overall Approach		HY-11	
				3.3.2.1	The Water Balance Equation	L	HY-11	
				3.3.2.2	Precipitation		HY-14	
				3.3.2.3 3.3.2.4	Soil Infiltration Annual Infiltration and		HY-17	
					Surface Runoff		HY-20	
				3.3.2.5	Potential Evaportranspirat	ion	HY-23	
				3.3.2.6	Annual Evaportranspiration	1	HY-25	
				3.3.2.7	Groundwater Runoff		HY-26	
				3.3.2.8	Annual Hydrologic Water Balance		HY-27	
				3.3.2.9	Depth Dependent Infiltrati	.on	HY-28.1	
		3.4	Model	Variables	/Parameters		HY-29	
		3.5	Water	Balance S	ensitivity		HY-37	
		3.6	Subrou	tine HYDRO	0A		HY-41	
			3.6.1	Equation	Summary		HY-41	
			3.6.2	Input/Out	tput Variables		HY-49	
			3.6.3	Step-by-S	Step Calculation Procedure		HY-51	
	4.0	LEVE	L2 – MOI	NTHLY "EST	TIMATED" HYDROLOGIC CYCLES		HY-52	
		4.1	Genera	1			HY-52	
		4.2	Definit	tions			HY-52	
		4.3	Monthly	y Matheman	tical Analysis		HY-52	
1			4.3.1	Principal	1 Assumptions		HY-52	
\sim			4.3.2	Theroeth	ical Overall Approach		HY-52	
4		4.4	Input/0	Output Var	riables		HY-54	

HY-1

Dec. 80

5.0	MODEL EXTENSION					
	5.1 5.2 5.3	General Snow Pack/Melt Interception	HY-55 HY-56 HY-57			
6.0	REFE	RENCES	HY-58			

.

1.0 INTRODUCTION

1.1 General

The hydrology of the soil compartment can be:

- (1) Assumed annually known (LEVELO),
- (2) Simulated annually (LEVEL1), or
- (3) Simulated monthly (LEVEL2, LEVEL3).

Simulations^{*} (LEVEL1, LEVEL2, LEVEL3) are performed via two hydrologic cycle subroutines, designed as HYDROA (Annually) and HYDROM (Monthly), the latter being an extention of the first.

1.2 Hydrologic Processes Involved

The two hydrologic subroutines (HYDROA and HYDROM) simulate the atmosphere, surface and subsurface hydrologic processes shown in Figure HY-1. The hydrologic processes are the main governing factors of pollutant movement in the soil compartment.

Precipitation encompasses rainfall and snow. Snowpack and snowmelt affect pollutant movement first by reducing erosion and secondly by causing less polluted runoff than the corresponding rain runoff.

Infiltration is the movement of water through the soil surface into the soil column. Infiltration rates are variable and change with the moisture content of the soil profile. During a storm event, the rate of infiltration decreases as the soil voids become filled. Usually more than half of the water which infiltrates is retained in the soil until it is returned to the atmosphere by evapotranspiration. Some infiltrated water may move laterally through the upper soil through the stream channel as interflow, and some may enter temporary storages and be later discharged into the stream channel as base or groundwater flow. The infiltration capacity is a function of the plant cover and of variable hydrogeologic characteristics, primarily soil moisture content.

^{*}In this appendix the word simulation is equivalent to modeling.



- (1) Upper Unsaturated Soil Zone
- (2) Lower Unsaturated Soil Zone
- (3) Saturated Soil Zone



Atmosphere, Land, Surface, Subsurface,

Biotic and Pollutant/Processes

Evapotranspiration is the transfer of water from land, vegetative cover and water bodies to the atmosphere. The term involves two distinct processes, exfiltration and transpiration. The volume of water leaving the watershed through evapotranspiration is greater than the total contribution to the base streamflow of most systems. All surfaces that are exposed to precipitation are considered to have a potential for evapotranspiration. Transpiration is a function of a vapor pressure gradient between air and leaf cells and occurs when leaf pores are stimulated by light. Deeply rooted plants continue to transpire even in periods of infrequent rainfall.

Interception is the amount of precipitation remaining on leaves, branches, and stems. This volume may or may not return to the atmosphere through transpiration. Intercepted water quantities during a single storm are relatively minor. However, they can have a significant effect on long term surface runoff volumes. Interception is a function of the type and extent of vegetation, land and meteorologic characteristics of the area (wind, temperature, solar radiation, precipitation, etc.)

Percolation results in groundwater runoff. Percolation rates depend mainly upon the infiltration rate, the moisture storage in the unsaturated soil zone and the depth to groundwater. During dry seasons percolation may become negative (upward) due to capilarity.

HY-4

1.3 Modeling Background

Previous efforts to model combined hydrologic, soil and vegetational systems of an area have been in two noteworthy directions (Eagleson, 1979; p. 924):

- (1) Empirical studies that provide validated interrelationships among the principle variables but that, due to their weak physical basis, lack both the generality and the parametric incorporation of climate, soil and vegetal properties that are necessary for general insight into soil processes. Prominent among these studies are the works of Lettan (1973) and Thornthwaite (1948); and
- (2) Numerical studies--that utilize detailed formulations of the physics at the microprocess scale but that, due to their complexity, impose infeasible validation data requirements and impede the generation of overall behavioral insight. Prominent among these studies are the works of Adams and Jurisa (1976), Donigian, et al (1977), and Novotny, et al (1978).

It is beyond the scope of this study to review the literature and describe the physics and mathematics of the previous studies. SESOIL does not employ either an empirical or a numerical hydrologic routine; instead it employs the statistical analytic "annual water balance" model of Eagleson (1978), which couples atmosphere, soil and vegetation systems. In SESOIL, however, Eagleson's model is modified to perform both annual and monthly simulations. The scientific background of the annual model has been presented and discussed in various journals since 1978. Therefore, this appendix (HY) is intended to presently only an "extracted" outline of the model as previously given in Eagleson's publications. The following paragraphs intend only to inform the reader about the nature of the hydrologic routine employed and <u>not</u> to give a thorough background of it. To maintain a consistent approach, Eagleson's notation is followed. Readers interested in the derivation of the annual equations presented

HY-5

in the following sections, are referred to the original publications.

The work of Eagleson presents a "generalized" annual water balance model based upon simplified physics of the component processes. The model is detailed enough to capture the "essential" system dynamics yet simple enough to permit analytical (as opposed to numerical) solution. It produces valuable insights into the role of soil moisture in environmental compartments, of which moisture is one of the most important factors governing pollutant transport and decay in the soil cell. Eagleson's model has a unique statistical approach in coupling systems and represents the state of the art in environmental modeling. The model is easy to use because of the limited number of input parameters required. The latter is fully justified by the sophisticated mathematical approach developed by Eagleson.

Section 2.0 of this appendix provides the hydrologic cycle background for the LEVELO model operation, dealing with "known" hydrologic components of the soil compartment. The following two main sections (Sections 3.0 and 4.0) outline the theoretical background for the "annual" hydrologic cycle subroutine (HYDROA) and the "Monthly" hydrologic cycle subroutine (HYDROM), the latter being developed based upon the theory of the first. 2.0 LEVELO - ANNUALLY "KNOWN" HYDROLOGIC COMPONENTS

2.1 Input Data

The following parameters must be known (input) for the LEVELO operation of SESOIL.

- I_A = annual infiltration; (cm)
 R_{sA} = annual surface runoff; (cm)
 R_{gA} = annual groundwater runoff; (cm)
 - Θ = mean annual soil moisture content; (mL/mL)

2.2 Discussion on Soil Water Models

Recent developments of soil water models based on column mass balance provide an alternative to directly or indirectly measuring soil moisture in the field. Figure HY-1 is a schematic diagram of the physical system and the driving forces that must be considered in modeling the system.

Based upon conservation of mass, the soil moisture in the system at any time can be determined using the relationship:

$$P_{A} = E_{TA}^{+} \Delta^{s}_{o} + R_{gA}^{+} + R_{sA}^{-}$$
(HY-0)

where:

 $P_A = total annual precipitation$ $R_{sA} = total annual surface runoff$ $E_{TA} = total annual evapotranspiration$ $\Delta s_o = s_o(t)-s_o(t-1)$; annual change in the soil moisture storage. $R_{gA} = total annual groundwater runoff$ Four of the above variables are user supplied input, the fifth is estimated. A standard practice in annual soil water budget modeling is to estimate Δs_0 from the above equation, using site-specific estimates of the other parameters. Consequently $s_0 = s_0(t)$ is estimated from the relationship $s_0(t) = \Delta s_0 - s_0(t-1)$, given the historical moisture storage $s_0(t-1)$.

Various references (e.g., National Water Atlas, Gerapty & Miller, Inc.) can be consulted for the annual averages of P_A , E_{TA} , R_{GA} and R_{SA} , for various locations in the U.S.
3.0 LEVEL1 - ANNUALLY "ESTIMATED" HYDROLOGIC CYCLE

3.1 General

The theory for the "annual" hydrologic cycle routine is presented by Peter S. Eagleson, in <u>Water Resources Research</u> (WRR), Volume 14, October 1978, Number 5, pages 705 through 776 and in a number of other publications. In this report, reference is mainly made to the above publication by indicating the page number and the equation number of the equations employed.

3.2 Definitions

It is the writer's feeling that the reader of this documentation and of Eagleson's publications might be confused with the definitions: annual, seasonal, monthly, long-term, etc., unless they are thoroughly acquainted with the theoretical background of the hydrologic cycles presented in this documentation. Therefore, it is worthwhile to clarify at this point the following expressions applicable to both the "annual" (Section 3.0) and the "monthly" (Section 4.0) hydrologic cycle routines.

- (1) For the "annual" hydrologic cycle routine the "simulation period" or simulation time (or time step) equals a period of "one year" (i.e. 12 months). Within this year we have a rainy "season" which can be shorter or equal to one year (12 months) depending on the climate of the area.
- (2) For the "monthly" hydrologic cycle routine, the "simulation period" or simulation time equals a period of "one month". Within this month we have a rainy "season" which can be shorter or equal to one month (1 month) depending on the climate of the area.

Eagleson employs both expressions "annual/seasonal" while presenting his "annual" water balance theory [e.g.,WRR,p.749,eq.(1)]. This is fully justified from his perspective, however, to avoid confusion (e.g. seasonal for annual vs. seasonal for monthly simulation) in the following section we will standardize our definitions as discussed above, and will explicitly define the terms.

3.3 Annual Mathematical Analysis

3.3.1 Assumption

The principal assumptions made are [Eagleson, 1977]:

- (1) No consideration of heavy snow or ice precipitation
- (2) Consideration of vegetation only as it affects surface albedo and roughness
- (3) One-dimensional analysis, involving vertical processes only
- (4) All processes are stationary in their long-term (annual/ seasonal) average
- (5) First-order analysis, namely, long-term averaged behavior, is used to represent relationships between seasonal averages.

3.3.2 Theoretical Overall Approach

3.3.2.1 The Water Balance Equation

The analysis of subroutine HYDROA is based upon:

 The volumetric water balance equation per unit area of the soil column over time [WRR,p.706,eq.(1)]:

$$\int_{0}^{t} [i(t)-e_{T}(t)-v_{g}(t)]dt = \int_{0}^{t} [r_{g}(t)+r_{g}(t)]dt = \int_{0}^{t} y(t)dt \quad (HY-1)$$
where:

* t = time

* t = time

* s = soil moisture concentration

i(t) = precipitation

e_{T}(t) = potential evapotranspiration rate

v_{g}(t) = rate of moisture storage in soil,

vegetation, snow, ice, lakes, etc.

r_{g}(t) = groundwater runoff rate

y(t) = yield rate

* (HY-1)

*

(2) The surface infiltration conservation equation over time
[WRR,p.706,eq.(3)]:

$$\int_{0}^{t} i(t)dt - \int_{0}^{t} v_{ss}(t)dt - \int_{0}^{t} f_{i}(t)dt = \int_{0}^{t} r_{s}(t)dt$$
(HY-2)

•

where:

v = rate of capture of precipitation in surface storage (i.e., on the soil and vegetal surface)

Assuming all evapotranspiration comes from soil moisture and considering only systems which are steady-state in long-term average, Eagleson [WRR,p.706,eq.(2)] developed the water balance equation:

$$E[P_A] - E[E_{TA}] = E[R_{SA}] + E[R_{QA}] = E[Y_A]$$
(HY-3)

where:

 $E[x] = m_{[x]}, \text{ expected value or mean of a variable x}$ $P_A = \text{ annual precipitation; (depth, cm)}$ $R_{sA} = \text{ annual surface runoff; (depth, cm)}$ $R_{gA} = \text{ annual groundwater runoff; (depth, cm)}$ $E_{TA} = \text{ annual total evaporation; (depth, cm)}$ $Y_A = \text{ annual yield; (depth, cm)}$

Rearranging the above equation and by omitting the E[] designations [WRR,p.707,eq.(4)]:

$$I_{A}(P_{A}) = P_{A} - R_{sA}(P_{A})$$
(HY-4)
Infiltration

$$= \underbrace{E_{TA}(P_A)}_{Evapotranspiration} + \underbrace{R_{gA}(P_A)}_{Groundwater}$$

$$\underbrace{Runoff}_{(Recharge and Loss)}$$

The following sections present a summary of the mathematical expressions developed by Eagleson for the various terms of the above equation. This equation is designated later on as "soil moisture (s_0) " equation because factors become soil-moisture dependent. It must also be noted that the above equation involves the implicit assumption of constant water storage over the given water season. This is only an approximation of reality; it is closest in nature to an arid, seasonal climate with ephemeral streams because the end-of-year moisture storage is there only at its annual minimum, and therefore very small.

Significant snowfall may have large interception losses, theoretically invalidating the above. However, the authors believe that the employment of Eagleson's model -- a discussion for snow pack/melt and interception is made in a later section -- is strongly desired because of its advantages in its formulation that is based on a relatively few physical parameters and very few input data.

3.3.2.2 Precipitation

The storm sequence is represented in the model by Poisson arrivals of rectangular pulses, as shown in Figure HY-2.

The cumulative distributive function for normalized annual point precipitation is [WRR,p.715,eq.(36)]:

$$\operatorname{Prob}\left[\frac{P_{A}}{m_{PA}} < z\right] = e^{-\omega m_{\tau}} \left\{ \begin{array}{l} \sum_{\substack{n=1\\ \nu = 1}}^{\infty} \frac{(\omega m_{\tau})^{\nu}}{\nu!} P[\nu \kappa, \omega m_{\tau} \kappa z] \right\}$$
(HY-5)
where:

PA	=	total seasonal precipitation (cm)
^m PA	=	average seasonal precipitation (cm)
ω	=	storm arrival rate (days ⁻¹)
^m τ	=	average length of rainy season (days)
ĸ	=	shape parameter of Gamma distribution of
		storm depth (h)
ν	=	number of storms
P[]	=	Pearson's incomplete Gamma function
Z	=	p/m _{PA}
P	=	value of annual rainfall (taken on by the random variables p _A).

The Gamma distribution of storm depth is given by [WRR,p.714,eq.(15)]:

$$f_{\rm H}(h) = G(\kappa, \lambda) = \frac{\lambda(\lambda h)}{\Gamma(\kappa)} e^{-\lambda h}$$
 (HY-6)

with mean [WRR,p.714,eq.(16)]:

$$m_{\rm H} = \kappa / \lambda = m_{\rm PA} / \omega \cdot m_{\tau} \tag{HY-7}$$





Precipitation Event Model

Source: [WRR,p.707,Figure 5]

and variance [WRR,p.714,eq.(17)]:

$$\sigma_{\rm H}^2 = \kappa / (\lambda)^2 \tag{HY-8}$$

in which:

h = storm depth (cm) λ = parameter of Gamma distribution of storm depths, equal to $\kappa/m_{\rm H}$ (cm⁻¹)

Equation HY-6 is shown to accurately reproduce the observed annual precipitation probability relationship in applications to both humid and arid-seasonal climates using only a few years (e.g., five) of storm data.

3.3.2.3 Soil Infiltration

The following assumptions were made for deriving statistical equations for capillary rise from the water table (dry seasons), infiltration, exfi.tration (i.e., desorption against gravity), and percolation to groundwater table [P.Eagleson, 1977]:

- (1) Homogeneous soil
- (2) No vegetation, snow or ice presence
- (3) Movement of water vapor negligible
- (4) Soil column is effectively semi-infinite concerning surface processes of infiltration and exfiltration

 (i.e., the water table or other boundary is deeper
 than the penetration depth of the surface processes)
- (5) Soil moisture is spatially uniform at the beginning of each storm and interstorm period with a value s_o, given by the long-term temporal and spatial average
- (6) Infiltration processes (infiltration, exfiltration, gravitational percolation and capillary rise) are considered as separable superimposable processes
- (7) Infiltration is described by the Phillip equation.

The derived soil moisture velocity equations are: Capillary Rise from the Water Table [WRR,p.728,eq.(59)]:

$$w = \left[\frac{mc + 1/2}{mc - 1}\right] \quad K(1) \quad \left[\frac{\Psi(1)}{Z}\right]^{mc} \tag{HY-9}$$

$$\Psi(1) = \frac{\sigma_{\rm w}}{\gamma_{\rm w}} \left[\frac{n}{k(1)\phi(c)} \right]^{1/2}$$
(HY-10)

in which:

= apparent velocity of capillary rise; (cm/sec) W = pore size distribution index; (-) m = effective medium porosity; (cm³/cm³) n = pore disconnectedness index; (-) С K(1) = saturated effective hydraulic conductivity (cm/sec), $\equiv \frac{\gamma_{w} \cdot k(1)}{\mu_{m}}$ k(1) = spatial average effective soil permeability at saturation; (cm²) $\Psi(1)$ = saturated matrix potential; (cm) Ζ = depth to groundwater table; (cm) = surface tension of pore liquid; (dynes/cm) σ = pore shape parameter; (-) ¢ = specific weight of liquid; (dynes/cm³) Υ_w = dynamic viscosity of pore fluid; (poises) μ

Infiltration [WRR,p.726,eq.(42);p.723,eq.(16)]:

$$f_{i}(t,s_{o}) = (1-s_{o}) \left[\frac{5nK(1)\Psi(1)\phi_{i}(d,s_{o})}{3\pi m t} \right]^{1/2} + \frac{1}{2}K(1)[1 + s_{o}^{c}] - w \quad (HY-11)$$

$$\phi_i = 10^{(0.66 + 0.55/m + 0.14/m^2)}$$
 (HY-12)

where:

Exfiltration [WRR,p.727,eq.(44)], simplified for no vegetation:

$$f_{e}(t,s_{o}) = s_{o}^{1+d/2} \left[\frac{nK(1)\Psi(1)\phi_{e}(d)}{\pi mt} \right]^{1/2} + w$$
(HY-13)

where:

- f = apparent exfiltration velocity; (cm/sec)
- ϕ_e = dimensionless exfiltration diffusivity; (-)

Percolation to Water Table [WRR,p.729,eq.(62)]:

$$v(s_0) = K(1)s_0^{C} - w$$
 (HY-14)

where:

v = apparent percolation velocity (cm/sec)

3.3.2.4 Annual Infiltration and Surface Runoff

The principal assumptions made are [Eagleson, 1977]:

- (1) No evaporation from surface storage at any time
- (2) No infiltration from surface storage following cessation of precipitation
- (3) No surface inflows from outside region
- (4) Soil moisture is uniform at s at the beginning of each storm
- (5) Precipitation intensity, i, and duration, t_r, are statistically independent.

The probability density function of storm surface runoff is determined and gives the:

Frequency of Flood Volume [WRR,p.746,eq.(72)]:

$$\frac{1}{T_E} = 2m_v z^{1/2} e^{-G-2\sigma} K_1(2z^{1/2}) \Gamma(\sigma + 1)/\sigma^{\sigma}$$
(HY-15)

Annual Average Surface Runoff, E[R_{sA}]; [WRR,p.746,eq.(68)]:

$$\frac{E[R_{sA}]}{m_{PA}} = e^{-G-2\sigma} \Gamma(\sigma + 1)/\sigma^{\sigma}$$
(HY-16)

in which [WRR,p.746,eq.(69)]:

$$T_{E} = \text{recurrence interval of flood of depth, } R_{sj}$$

$$z = R_{sj}/m_{H}$$

$$\sigma = \left[\frac{5n\eta^{2}K(1)\Psi(1)(1-s_{o})^{2}\phi_{i}(d,s_{o})}{6\pi\delta m}\right]^{1/3}$$
(HY-17)

and [WRR, p. 746, eq. (70)]:

$$G = \frac{\alpha K(1)}{2} [1 + s_o^c] - \alpha w$$
 (HY-18)

where:

and

$$n^{-1}$$
 = mean storm depth, m_{H} ; (cm)
 δ^{-1} = mean storm duration, m_{tr} ; (sec)

$$\alpha^{-1}$$
 = mean storm intensity = m_H/m_{tr} ; (cm/sec)

For representative soil properties Equation HY-16 illustrates the range of observed surface runoff values. A graphical presentation of the surface runoff function is shown in Figure HY-3.

Net Infiltration

Based upon Equation HY-16 and because [WRR,p.747,eq.(74)]:

$$E[I_A] = m_{PA} - E[R_{SA}]$$
(HY-19)

the expected seasonal net infiltration as a function of precipitation equals [above equation and WRR,p.746,eq.(68)]:

$$\frac{E[I_A]}{E[P_A]} = 1 - e^{-G - 2\sigma} \Gamma(\sigma + 1) / \sigma^{\sigma} = 1 - \xi$$
 (HY-20)

This also represents the fraction of all storms which do not produce surface runoff.



Source: [WRR,p.746, Figure 6]

Figure HY-3

Surface Runoff Function [Eagleson, 1979]

3.3.2.5 Potential Evaportranspiration

The "potential" evapotranspiration of an area can be estimated based on the following assumptions [Eagleson, 1977]:

- The energy balance equation may be time averaged by replacing variables by their time averages
- (2) The average rates of energy advection and storage are zero
- (3) Difference between surface and atmospheric temperatures may be neglected in estimating net outgoing longwave radiation.

The modified Penman energy balance equation can be used to estimate the average rate of potential evapotranspiration.

$$\overline{e}_{p} = \frac{\overline{q}_{1}(1 - A) - \overline{q}_{b} + H}{\rho_{e}L_{e}(1 + \gamma/\Delta)}$$
(HY-21)

in which:

$$\overline{q}_i$$
 = average rate of insolation (ly/min)
 \overline{q}_b = average rate of net outgoing longwave
radiation (ly/min)
H = average sensible heat flux residual (ly/min)
A = shortwave albedo of surface
 ρ_e = mass density of evaporating water (g/cm³)
L_e = latent heat of vaporization (cal/g)
 γ/Δ = atmospheric parameter (a function of temperature)

Empirical values of $\overline{q_i}$, $\overline{q_b}$ and H are presented by Eagleson (1977) in a form suitable for practical use. (Table HY-1). The potential evapotranspiration can be <u>either</u> an input variable to SESOIL <u>or</u> can be estimated using the above equation.

Observed Values of Annual Potential Evapotranspiration

		Observe	<u>d</u>						Calculated
Location	Ref.	E ŀ in/yr	: 1 • <u>N</u>	<u>،</u> ۲	T _a 3 °C	к ^с	\$ ⁵	u ⁵ n/sec	ra X 10 min/cm
Mesilla, N.M.	50	34.0	32.27	0.26	14.6	0.18	0.53	4.2	5.5
Pecos, N.M.	50	35.3	35.57	0.26	13.8	0.18	0.45	4.1	4.4
Sangamon R., 111.	50	29.2	40.02	0.26	11.5	0.34	0.70	3.8	4.2
Green R., Ky.	50	31.4	37.90	0.26	13.1	0.46	0.73	3.0	6.4
Tallapuosa R, Ga.	50	33.0	32.50	0.26	15.4	0.36	0.77	2.3	8.7
Mad R., Ohio	50	25.8	40.00	0.26	10.8	0.40	0.72	3.0	8.6
Skunk R., Iowa	50	27.0	40.70	0.26	9.4	0.34	0.70	4.0	4.3
W.Fork, White R., Mo.	50	31.0	37.00	0.26	13.4	0.33	0.72	4.0	4.6
N. Platte R., Neb.	50	23.8	42.00	0.26	9.0	0.30	0.60	4.2	7.0
Black R., Wis.	5Ô	22.2	43.93	0.26	8.0	0.40	0.74	3.5	7.9
Cyprus Crk., Tex.	50	36.2	32.00	0.26	17.9	0.20	0.75	5.0	3.5
Wagon Wheel Gap. Col	. 50	15.6	37.77	0.40	5.3	0.25	0.60	3.8	7.2
Merrimac R., Ma.	50	21.5	43.20	0.30	7.6	0.47	0.78	2.5	7.0
West R., Vt.	50	21.5	42.98	0.30	7.6	0.47	0.78	2.5	7.0
*Lake Cochituate Ma.	50	23.2	42.50	0.30	10.7	0.33	0.70	4.9	3.7
Swift R., Ma.	50	23.1	42.50	0.26	8.4	0.33	0.70	3.9	5.9
*Phoenix, Arix.	45	71.6	33.43	0.05	21.3	0.20	0.43	2.7	4.8
*Davis, Cal.	45	50.0	38.37	0.05	15.7	0.30	0.52	3.9	5.2
*Fresno, Cal.	45	60.0	36.77	0.05	16.8	0.25	0.43	2.8	3.4
*Grand Junction,Col	45	35.6	39.12	0.05	11.5	0.30	0.34	3.4	5.0
*Boise, Idaho	45	33.4	43.57	0.05	10.5	0.35	0.37	3.8	3.7
*Dodge C:ty, Kans.	:5	62.6	37.77	0.05	12.7	0.27	0.50	6.0	1.8
*Glascow, Mon.	5	38 0	48.22	0.05	5.3	0.25	0.54	4.8	3.0
*Great Falls, Mon	45	35.0	47.48	U 05	7.2	0.25	0.50	5.9	3.9
*Ely, Nev.	_ 5	46.0	39.28	0.05	6.7	0.28	0.36	4.7	5.4
*Bismarch, N.Dak.	45	34.1	48.77	0.05	5.2	0.30	0.63	4 4	3.2
*Stillwater, Ok.	45	57.0	36.13	0.05	15.6	0.35	0.50	5.5	1.8
*Astoria, Ore.	45	20.0	46.15	0.05	10.3	0.50	0.80	4.0	•
*Medford, Ore.	45	32.0	42.37	0.05	11.7	0.50	0.66	2.2	-
*Rapid Citv, S.D.	45	40.6	44.03	0.05	8.1	0.35	0.56	4.6	3.7
*Brownsville, Tex.	-5	58.0	25.90	0.05	23.2	0.40	0.77	5.2	3.9
Fort Worth, lex.	45	57.2	32.82	0.05	18.6	0.40	0.63	4.8	4.6
*Midland, Tex.	45	713	31.93	0.05	17.7	0.25	0.50	4.5	2.0
*Spokane, Wash.	نہ	38.4	47.62	0.05	8.5	0.40	0.42	3.9	4.5
*Lander, Wyo.	45	35.0	42.80	0.05	6.9	0.25	0.42	3.2	14.3

* Water surface

-

1 Estimated from stans

A = 0.26 chosen for vegetal surfaces (slightly greater for seasonal snow or ice) A = 0.05 chosen for water surfaces

- ³ Average annual due taken from Sacional Weather Service Climatological Summary at nearest station
- From Climatological Surmary at nearest station
- 5 Average from Clicatelosical Summary at prarest station

Source: P. Eagleson (1977)

3.3.2.6 Annual Evapotranspiration

The previous background and the following assumptions lead to the estimation of the "expected annual" evapotranspiration $E[E_{TA}]$:

- (1) No evaporation except from water which has first infiltrated
- (2) No effect of vegetation in bringing soil moisture to surface
- (3) Soil moisture is uniform at s at the beginning of each interstorm period
- (4) Variance of average annual rate of potential evapotranspiration is negligible
- (5) e >> w.

The annual average evapotranspiration $E[E_{TA}]$ is given by [WRR,p.736, eq.(45)]:

$$\frac{E[E_{TA}]}{m_{PA}} = J(E) = 1 - [1 + 2^{1/2}E]e^{-E} + (2E)^{1/2}\Gamma[\frac{3}{2},E]$$
(HY-22)

in which [WRR,p.762,eq.(70),(71)]:

$$E = \frac{2\beta n K(1) \Psi(1) \phi_{e}(d)}{\pi m (\overline{e}_{p} - w)^{2}} s_{o}^{d+2}$$
(HY-23)

where:

$$\beta^{-1}$$
 = mean time between storms, m_{tb} , (sec)
E = exfiltration parameter

3.3.2.7 Groundwater Runoff

The annual groundwater runoff is defined as the "net" groundwater recharge from the unsaturated soil zone, namely the percolation flow reduced by the capillary rise flow.

The following assumptions are associated with the potential seasonal groundwater runoff estimate:

- Percolation to water table is steady throughout the wet season at value determined by the average soil moisture, s_a, and is zero during the dry season
- (2) Capillary rise from the water table is steady throughout the entire year at the rate given by a dry surface
- (3) Water table elevation, z = Z, is constant.

The annual non-dimensional average groundwater runoff, $E[R_{gA}]$ is given by [WRR,p.751,eq.(20)]:

$$\frac{E[R_{gA}]}{m_{PA}} = \frac{m_{T}K(1)}{m_{PA}} s_{O}^{C} - \frac{Tw}{m_{PA}}$$
(HY-24)

Assuming that no groundwater storage occurs within a season, the total groundwater runoff will recharge adjacent surface waters.

3.3.2.8 Annual Hydrologic Water Balance

The annual water budget (HY-4) is given by:

$$E[I_{A}(s_{o})] = E[E_{TA}(s_{o})] + E[R_{gA}(s_{o})]$$
 (HY-25)

Equations HY-20, HY-22 and HY-24 contribute to the first-order dimensionless water budget equation [WRR,p.766,eq.(6)]:

	Groundwater:
Infiltration	Recharge Loss
$[1 - e^{-G-2\sigma} \Gamma(\sigma + 1)\sigma^{-\sigma}] = \frac{E_p}{P_A}$	$\frac{A}{P_{A}} J(E) + \frac{m_{\tau} K(1)}{P_{A}} s_{0}^{c} - \frac{T_{W}}{P_{A}} $ (HY-26)
Precip. Surface Runoff Eva spi	potran- Groundwater Runoff ration

A graphic presentation of function J(E), equation HY-22, is shown in Figure HY-4 [WRR,p.737, Figure 5].

The above equation is used to define the dependent variable, s_o, which can be used, in turn, to define the separate terms of the water budget in terms of the independent climate and soil variables and paraparameters.

The annual water yield, Y_A , is determined in this way as

$$Y_{A}(P_{A}) = R_{sA}(s_{o}(P_{A})) + R_{gA}(s_{o}(P_{A})) = g(P_{A})$$
 (HY-27)

and is used to transform the CDF of annual precipitation $[Y_A(P_A)]$ into the CDF of annual yield according to

$$\operatorname{Prob}\left[\frac{Y_{A}}{m_{PA}} < z\right] = e^{-\omega m_{\tau}} \left\{ \begin{array}{c} \sum_{\substack{\alpha = 1 \\ 1+\sum \\ \nu=1 \end{array}}^{\infty} \frac{(\omega m_{\tau})^{\nu}}{\nu!} P[\nu \kappa, \omega m_{\tau} g^{-1}(z)] \right\}$$
(HY-28)



Figure HY-4 Base Soil Evaporation Function $(w/\bar{e}_p <<1)$

Source: Eagleson (1978); WRR,p.737, Figure 5

3.3.2.9 Depth Dependent Infiltration

In the previous sections the infiltration (I) is defined as the total depth of rainfall infiltrating the ground surface (see Figure HY-4.1). Consequently the groundwater runoff (R_g) is defined as the excess of water in the soil column percolating the ground, namely reaching the saturated soil zone.

SESOIL is designed to estimate pollution distribution in the upper, middle and lower unsaturated soil zone of the compartment. It is, therefore, required to have a seasonal averaged estimate of the infiltration at a depth z of the soil column as shown in Figure HY-4.1. This estimate (I_z) is required for both the annual and the monthly simulations of the pollutant transport.

Based upon the geometry of the soil compartment we may make the assumption that the annual (A) -- section 3.3 of this appendix -- and monthly (M) -- section 3.4 of this appendix -- variations are given by:

$$I_z(A,M) = R_o(A,M) + d_1 \cdot \tan \alpha =$$

=
$$R_g(A,M) + \frac{d_1}{d_u + d_1} [I(A,M) - R_g(A,M)]$$
 (HY-28.1)

where d_u and d_1 the depths of the upper and the lower unsaturated soil zones respectively. The I_z values estimated by the above equation are employed by the pollutant transport routines, Appendix PT, equations PT-6, PT-13, PT-28 and PT-31.

The layered averaged intrinsic permeability of the compartment is approximated to

$$k_{z} = (d_{u} + d_{M} + d_{L})/(d_{u}/k_{u} + d_{M}/k_{M} + d_{L}/k_{L})$$
 (HY-28.2)

and each infiltrating quantity (depth) is given by

$$I(A,M) = I_{2}(A,M)$$
 (HY-28.2)

where

$$\overline{k_1} = k_1, \overline{k_2} = (d_u + d_M) / (d_u / k_u + d_M / k_M), \overline{k_3} = k_z$$
 (HY-28.2)



Figure HY-4.1

Depth Dependent Infiltration in the Soil Column

3.4 Model Variables/Parameters

The full set of twenty (20) parameters and variables governing the previous equations are:

^m PA	=	mean annual precipitation; (cm)
E[E _{PA}]	=	mean annual potential evapotranspiration; (cm)
ēp	=	mean annual rate of potential evapotranspiration; (cm)
^m tr	=	δ^{-1} = mean storm duration; (days)
^m tb	=	β^{-1} = mean time between storms; (days)
^m H	=	η^{-1} = mean storm depth; (cm)
^m i	=	α^{-1} = mean storm average intensity; (cm/sec)
m _τ	=	mean length of rainy season; (days)
Т	=	duration of capillary rise from watertable ; (days)
k(1)	=	spatial average effective soil permeability at
		saturation; (cm ²)
Ta	8	normal annual temperature of surface soil moisturn; (°C)
с	=	soil conductivity index; (-)
m	=	soil matrix potential index; (-)
d	=	soil diffusivity index; (-)
n	=	effective soil porosity; (cm ³ /cm ³)
Ψ(1)	=	spatial average soil matrix potential at saturation;
		(cm)
Z	=	depth to watertable; (cm)
^ф е	=	exfiltration diffusivity function: (-)
Φ _i	=	infiltration diffusivity function; (-)
so	=	spatial and temporal average soil moisture within the
		soil boundary layer; (-)

HY- 29

Introducing a twenty-first parameter,

 m_v = mean number of storms per season, (#) we have ten supplementary relations in addition to the water budget equation (HY-26):

$$\begin{split} \mathbf{m}_{\mathrm{H}} &= \mathbf{m}_{\mathrm{PA}}/\mathbf{m}_{\mathrm{v}} \quad (\text{definition}) \\ \mathbf{m}_{\mathrm{T}} &= \mathbf{m}_{\mathrm{v}}(\mathbf{m}_{\mathrm{tb}} + \mathbf{m}_{\mathrm{tr}}), \ \mathbf{m}_{\mathrm{v}} \geq 1 \quad (\text{definition}) \\ \mathbf{m}_{\mathrm{i}} &= \mathbf{m}_{\mathrm{v}}/\mathbf{m}_{\mathrm{tr}} \quad (\text{assuming independence of i and tr}) \\ \mathbf{E}[\mathbf{E}_{\mathrm{PA}}] &= \mathbf{m}_{\mathrm{v}}\mathbf{m}_{\mathrm{tb}}\mathbf{e}_{\mathrm{p}} \quad (\text{definition}) \\ \mathbf{d} &= (\mathbf{c} + 1)/2 \quad (\text{semi-empirical}) \quad (\mathrm{HY-29}) \\ \mathbf{m} &= 2/(\mathbf{c}-3) \quad (\text{semi-empirical}) \\ \mathbf{T} &= \mathbf{cne} \quad \text{year} \quad (\text{definition}) \\ \mathbf{\Psi}(1) &= \quad \Psi(\mathbf{n},\mathbf{k}(1), \ \mathbf{T}_{\mathrm{a}}); \quad [\mathrm{WRR}, \mathbf{p}.724, \mathrm{eq}.(17)] \\ \mathbf{\phi}_{\mathrm{i}} &= \quad \mathbf{\phi}_{\mathrm{i}}(\mathbf{d},\mathbf{s}_{\mathrm{o}}) \quad (\mathrm{Figure} \quad \mathrm{HY-5}) \\ \mathbf{\phi}_{\mathrm{e}} &= \quad \mathbf{\phi}_{\mathrm{e}}(\mathbf{d}) \quad (\mathrm{Figure} \quad \mathrm{HY-6}) \end{split}$$

To solve the water budget relation for the dependent variable, s_0 , we must therefore specify the values of ten parameters (i.e., 21 variables minus 11 equations). Thus:

(1) <u>INPUT</u> parameters to the model are:

Soil system:

<u>Climate System</u>: Five independent parameters may be chosen from the set of six:

$$m_{PA}, \overline{e}_{p}, m_{H}, m_{tr}, m_{v}, m_{\tau}$$
 (HY-31)



Figure HY-5

Dimensionless Infiltration Diffusivity

Source: [Eagleson, WRR, p. 727, Figure 9]



Figure HY-6

Dimensionless Exfiltration Diffusivity

Source: [Eagleson, WRR,p.727, Figure 10]

(2) OUTPUT parameters from the model are:

An example of input and other climatic data for sub-humid (Clinton, Massachusetts) and an arid (Santa Paula, California) climate-soil system is shown in Table HY-2.

Representative independent soil properties nominal values covering a range of observations (Eagleson 1977) are given in Table HY-3. It is important to notice that there is <u>no</u> unique association of the particular c and n values with the tabulated value of k(1) for each soil. Derived (Eagleson 1979) climate-soi¹ parameters for indicated climatic and soil input are given, as an example, in Table HY-4. The latter values are derived using Table HY-3 values.

<u>Note</u>: Model users should validate their model output based upon "water balance" data from a given site, and they should never rely upon the derived parameters of Tables HY-3 and HY-4. These tables should not give the impression, for example, that they contain <u>all</u> one needs to know about soil. That is, if the soil is clay it has the properties of the first column of Table HY-3. This is <u>not</u> the case of course, since soil stratification properties are of paramount importance and may drastically alter the k(1), n and c values of a site. The soil properties are critical to the moisture fluxes and are tremendously variable spatially. Use of point measured soil properties can yield results of only local (and hence not really averaged) character.

Independent Soil and Climate Parameters

for a Sub-Humid and an Arid Climate Soil System

		Location				
Parameter	Units	Clinton, Mass.	Santa Paula, Ca.			
n	-	0.35	0.35			
k(1)	cm ²	2.8X10 ⁻¹⁰	1.2x10 ⁻⁹			
с	-	10	5 ¹⁾			
^m Pa	CM	94.1	54.4			
e _p	cm/day	0.15	0.27			
^m tr	days	0.32	1.4			
m _r	days	365	212			
m _v	-	109	15.7			
mH	cm	1.0	3.0			
Z	m	œ	œ			
T _a	°c	8.4	13.8			
^m tb	days	3.0	10.4			
ĸ	-	0.50	0.25			
ho	cm	0.1	0.1			
k _v	-	1	1			
ω	day ⁻¹	0.30	0.084			
Source: Eagleson [WRR,p.717, Table 1]						
¹⁾ See comment on Table HY-3.						

INDEPENDENT SOIL PROPERTIES¹⁾ FOR VARIOUS SOIL TYPES

Property	Soil Type ²)				
(Variable)	Clay	Clay-Loam	Silty-Loam	Sandy-Loam	
k(1) [cm ²]	1X10 ⁻¹⁰	2.8X10 ⁻¹⁰	1.2X10 ⁻⁹	2.5x10 ⁻⁹	
n	0.45	0.35	0.35	0.25	
с	12	10	5 ³⁾	4	

Source: Eagleson 1977, p. 256.

```
    See limitation discussion in previous pageS.
    Derived by using Table HY-4 values and the corresponding environments.
    Personal communication with Eagleson. In his publications c(silty-loam) = 6; however, later investigations indicated c = 4.5-5.5. An average value c = 5 is given here.
```

DERIVED CLIMATE-SOIL PARAMETERS FOR A CLIMATE AND RELATED SOIL TYPES (OF TABLE HY-2)

Climate:
$$\alpha = 1.5 \times 10^4 \text{ sec/m}$$
; $\delta = 10^{-1} \text{ hr}^{-1}$
 $\beta = 7 \times 10^{-3} \text{ hr}^{-1}$; $m_{\text{H}} = 2.54 \text{ cm}$
 $m_{\tau} = 365 \text{ days}$; $m_{v} = 75 \text{ events}$
 $\overline{T}_{2} = 15 \text{ °C}$

Derived Climate-Soil Parameters

Derived Parameter	Clay	Clay-Loam	Silty-Loam	Sandy-Loam
m	0.222	0.286	0.667	2
d	6.5	5.5	3.5	2.5
Ψ(1), cm	25	19	166	200
K(1), cm/sec	8.2×10^{-6}	2.32×10^{-5}	9.94×10^{-5}	2.08×10^{-4}
φ _e (d)	0.0385	0.0494	0.0920	0.1430
φ _i (d,0)	0.122	0.140	0.194	0.240
φ ₁ (d,1)	0.6	0.6	0.6	0.6
G(0)	0.0621	0.174	0.746	1.560
G(1)	0.124	0.348	1.490	3.120
σ(0)	0.432	0.482	1.340	1.220
σ(1)	0	0	0	0
$\Gamma(\alpha(0) \pm 1)$	0 886	0 886	1 200	1 110
$\Gamma(0(0) = 1)$	1	1	1	1
Γ(σ (Ι) + Ι)	T	I	I	1
σ(0) ^{-σ(0)}	1.44	1.42	0.68	0.79
$S(0) = \frac{1}{2}$	1.04×10^{-2}	1.27×10^{-2}	5.97×10^{-2}	5.15×10^{-2}
$s_{i}(0), cm/sec$ $s_{i}(1), cm/sec^{1/2}$	0	0	0	0
1				
$S_{e}(0), cm/sec^{1/2}$	0	0	0	0
$S_{e}(1), cm/sec^{1/2}$	4.54×10^{-3}	5.82×10^{-3}	3.19×10^{-2}	3.08×10^{-2}
Source: Eagleson	[1978, 1979]			

Nov. 80

.

HY-36

3.5 Water Balance Sensitivity

Equation HY-26 and its component equations were used by Eagleson (1979) in studying the sensitivity of the model to variation of the climatic or soil parameters. For the climatic parameters of Table HY-2 the sensitivity of the average annual water budget components is presented as a function of the soil permeability and soil porosity in Figures HY-7 and HY-8 for both locations, Clinton, Massachusetts and Santa Paula, California.

According to Eagleson, by comparing the two columns of Figure HY-7 we see contrasting behavior only in evapotranspiration and soil moisture. Beginning with the former, we see insensitivity of E_{TA} to soil properties in the sub-humid climate except when the soil gets very permeable. For the arid climate, however, E_{TA} is sensitive to the soil properties over their full range.

In the humid case, the supply of water is adequate and the soil moisture will be largest where the permeability readily admits water (and holds it against gravity). This requires a small σ which occurs for small k and large m (i.e., small c).

In the arid case where the evapotranspiration is controlled by the moisture supply to the surface, s_0 will be largest where the moisture movement to the surface, as given by E, is smallest. This will occur for small k(1) and large d (i.e., large d).

The runoff behavior is qualitatively the same in both climates. For small k(l), the total yield is predominantly surface runoff because the water cannot enter the soil. This component increases with c due

HY-37





Sensitivity of Annual Water Budget to Changes in Soil Parameters (M = 0, $h_0 = 0, w/\overline{e}_p <<1$)

Source: Eagleson (1979)



Figure HY-8

Effect on Annual Budget Due to Decreasing Mean Annual Precipitation

Source: Eagleson (1979)

to decreasing permeability and it decreases with increasing k(1) due to increasing permeability. The groundwater component also increases with k(1). The "saddle" in the Santa Paula groundwater component with increasing c results from the behavior of the factor s_0^c where s_0 is less than one and is increasing with c.

For additional information regarding the sensitivity analysis the reader is referred to the original publication of P. Eagleson [WRR, p.749].

3.6 Subroutine HYDROA

3.6.1 Equation Summary

The water balance model, as presented previously, has two distinctive working-steps:

- Based upon climatic data, the soil moisture, s_o, is determined;
- (2) Based upon the soil moisture value, s_o, any desired seasonal water balance component is obtained.

Therefore:

(1) The soil moisture, s_0 , is estimated by the first order conservation equation

Infiltration

$$\begin{bmatrix}
1 - e^{-G-2\sigma} \Gamma(\sigma + 1)\sigma^{-\sigma}
\end{bmatrix} = \frac{E[E_{PA}]}{m_{PA}} J(E) + \frac{m_{T}K(1)}{m_{PA}} s_{0}^{C} - \frac{Tw}{m_{PA}} \quad (HY-32)$$
Precip. Surface Runoff Evapotran-spiration

in which:

$$G = \frac{\alpha}{2} K(1) [1 + s_0^c] - \alpha w$$
 (HY-33)

0-----

$$\sigma = \left[\frac{5n\eta^2 K(1)\Psi(1)(1 - s_o^2)\phi_i(d, s_o)}{6\pi\delta m}\right]^{1/3}$$
(HY-34)
$$J(E) \equiv 1 - (+\sqrt{2} \cdot E) \cdot e^{-E} + \sqrt{2E} \cdot \Gamma[3/2, E]$$
 (HY-35)

w = K(1)
$$[1 + \frac{3}{2}(mc-1)] [\Psi(1)/Z]^{mc}$$
 (HY-36)

$$E = \frac{2\beta nK(1)\Psi(1)\phi_{e}(d)}{\pi m(\bar{e}_{p}-w)^{2}} s_{o}^{d+2}$$
(HY-37)

The values of any desired seasonal water balance component are estimated by substituting a value for s in Equations H-32 through H-36, namely for:

Infiltration

$$I_A/P_A = 1 - e^{-G-2\sigma} \Gamma(\sigma + 1)\sigma^{-\sigma}$$
 (HY-38)

Surface Runoff

$$R_{sA}/P_{A} = e^{-G-2\sigma} \Gamma(\sigma + 1)\sigma^{-\sigma}$$
(HY-39)

Evapotranspiration

$$E_{TA}/P_{A} = \frac{E_{PA}}{P_{A}} J(E)$$
(HY-40)

Groundwater Runoff

$$R_{gA}/P_{A} = \frac{m_{\tau}K(1)}{P_{A}} s_{o}^{c} - \frac{Tw}{P_{A}}$$
 (HY-41)

Annual Yield

$$Y_{A}/P_{A} = \frac{P_{SA} + R_{gA}}{P_{A}} = 1 - E_{PA}/P_{A}$$
 (HY-42)

Precipitation

$$P_{A} = \frac{E[E_{PA}] J(E) + m_{\tau} K(1) s_{0}^{c} - T \cdot w}{1 - e^{-G} \cdot \xi}$$
(HY-43)

where:
$$\xi = e^{-2\sigma} \cdot \frac{\Gamma(\sigma+1)}{\sigma}$$

Equation HY-43 is another form of the water balance equation HY-32, that is employed in the step-by-step calculation procedure, section 3.6.3.

.

where:

E[]	=	expected value of []
г()	=	Gamma function of () (see Table HY-5)
E[E _{PA}]	=	long-term expected (average) annual
		potential evapotranspiration (cm)
^m PA	=	average annual precipitation (cm)
G	=	gravitational infiltration parameter
		(Equation HY-33)
σ	8	capillary infiltration parameter (Equation HY-34)
J()	=	evaporation function (Equation HY-35)
Е	=	evaporation parameter (Equation HY-37)
W	=	apparent velocity of capillary rise from
		water table (cm/sec)
с	=	<pre>pore disconnectedness index = ln[K(s_0)/K(l)]/lns_0</pre>
Т	=	time (year)
Ξτ	8	long-term average length of annual rainy season (days)
K(1)	=	saturated effective hydraulic conductivity
		(cm/sec)
s o	8	long-term average effective soil moisture
		concentration in the unsaturated soil zone
α	=	reciprocal of mean storm intensity $\equiv m_i^{-1}$
		(sec/cm)

Table HY-5

Values of the Gamma Function

<u>x</u>	[(x)	<u>x</u>	Γ(x)	<u>x</u>	Γ(λ)
1.00	1.000	1.34	0.892	1.68	0.905
1.01	0.994	1.35	0.891	1.69	0.907
1.02	0.989	1.36	0.890	1.70	0.909
1.03	0.984	1.37	0.889	1.71	0.911
1.04	0.978	1.38	0.889	1.72	0.913
1.05	0.974	1.39	0.888	1.73	0.915
1.06	0.969	1.40	0.887	1.74	0.917
1.07	0.964	1.41	0.887	1.75	0.919
1.08	0.960	1.42	0.886	1.76	0.921
1.09	0.955	1.43	0.886	1.77	0.924
1.10	0.951	1.44	0.886	1.78	0.926
1.11	0.947	1.45	0.886	1.79	0.929
1.12	0.944	1.46	0.886	1.80	0.931
1.13	0.940	1.47	0.886	1.81	0.934
1.14	0.936	1.48	0.886	1.82	0.937
1.15	0.933	1.49	0.886	1.83	0.940
1.16	0.930	1.50	0.886	1.84	0.943
1.17	0.927	1.51	0.887	1.85	0.946
1.18	0.924	1.52	0.887	1.86	0.949
1.19	0.921	1.53	0.888	1.87	0.952
1.20	0.918	1.54	0.888	1.88	0.955
1.21	0.916	1.55	0.889	1.89	0.958
1.22	0.913	1.56	0.890	1.90	0.962
1.23	0.911	1.57	0.890	1.91	0.965
1.24	0.909	1.58	0.891	1.92	0.969
1.25	0.906	1.59	0.892	1.93	0.972
1.26	0.904	1.60	0.894	1.94	0.976
1.27	0.903	1.61	0.895	1.95	0.980
1.28	0.901	1.62	0.896	1.96	0.984
1.29	0.899	1.63	0.897	1.97	0.988
1.30	0.897	1.64	0.899	1.98	0.992
1.31	0.896	1.65	0.900	1.99	0.996
1.32	0.895	1.66	0.902	2.00	1.000
1.33	0.893	1.67	0.903		

Source: Eagleson, 1977

n = effective soil porosity = volume of active voids/total volume

 η = reciprocal of mean storm depth = m_{H}^{-1} (cm⁻¹)

$$= \frac{\sigma_{w}}{\gamma_{w}} \left[\frac{n}{k(1)}\right]^{1/2} \left(10^{-0.66 - 0.55/m - 0.14/m^{2}}\right)^{1/2}$$

σw	=	surface tension of pore water (dynes/cm)
Υ _w	=	specific weight of pore water (dynes/cm ³)
k(1)	=	spatial average saturated effective intrinsic
		permeability of soil (cm) = $K(1)\mu_w/\gamma_w$ (see
		Table HY-3)

$$\mu_{W}$$
 = dynamic viscosity of pore fluid (poises)
 $\phi_{1}(d, \hat{s}_{0})$ = dimensionless infiltration diffusivity (see
Figure HY-5)

$$\beta$$
 = reciprocal of mean interstorm period = m_{tr}^{-1}

d = (c + 1)/2 = diffusivity index $\xi(\sigma) = e^{-2\sigma} \cdot \Gamma(\sigma + 1)\sigma^{-\sigma}$; surface runoff function (see Figure HY-3)

With Equations HY-32 through HY-42 the average seasonal water balance can be displayed graphically in a variety of ways, one of which is illustrated in Figure HY-10 for an annual water cycle.





Climatic Influence of Annual Water Balance

Source: Eagleson (1979)

3.6.2 Input/Output Variables

INPUT date to HYDROA are categorized into the three groups of:

- (1) <u>Climatic</u> data: either: \overline{e}_p (cm/day) or: the data set L (°N) \overline{T}_a (°C) \overline{S} (fractional) A (-) NN (fractional)
- (2) Storm data:
 - T = 365 (days) $m_{PA} (cm)$ $m_{tr} (day)$ $m_{v} (\#)$ $m_{\tau} (days)$
- (3) <u>Soil</u> data: k(1) (cm²) c (-) n (-)

Above variables are stored in arrays CLIMA1, CLIMA2 and SOIL1 of the SESOIL Data Base; see Appendix DF, Data Files.

OUTPUT data from HYDROA is the data set.

s _o	(-)
P _A ≡ m _{PA}	(cm)
I _A	(cm)
^E ta	(cm)
R _s a	(cm)
^R gA	(cm)
YA	(cm)

3.6.3 General Calculation Procedure.

The calculation of the annual water balance has the following major steps.

- (1) Estimation of initial parameters
 - 1.1 Estimation of climate parameters
 - 1.2 Estimation of soil parameters
 - 1.3 Estimation of potential evapotranspiration
 - 1.4 Estimation (or input) of annual evapotranspiration
- (2) Solution of the annual water balance equation HY-32
 - 2.1 Solution of the water balance equation HY-32 (i.e. estimation of s_0) is best accomplished by employing an iterative procedure for s_0 increments, that is by assuming an s_0 and consequently estimating P_A via eq. HY-43.
 - 2.2 Estimation of water balance components for the assumed s_o value.
 - 2.3 Comparison of estimated ${\rm P}_{\rm A}$ versus assumed ${\rm m}_{\rm PA}$ to obtain the solution.

4.0 LEVEL2 - MONTHLY "ESTIMATED" HYDROLOGIC CYCLES

4.1 General

Subroutine HYDROM estimates the monthly (M) hydrologic cycle components of the soil compartment. HYDROM is based upon the theory of HYDROA discussed in the previous section.

4.2 Definitions

See Section 3.2.

4.3 Monthly Mathematical Analysis

4.3.1 Principal Assumptions

The principal assumptions made are:

- The assumptions made for the annual warer balance (see Section HY-3.3.1); and
- (2) The response of the environment (eg. moisture content) at the end of a month with "constant" and "continuous" rain MPA is "similar" to the response of this environment at the end of a year with constant and continuous rain of the 12 (MPA). Mathematical Linearity of Processes (Dooge 1973) is assumed.

4.3.2 Theoretical Overall Approach

To reduce the averaging time of a simulation from a year to a month or shorter, traditionally modelers develop a numerical, finitedifference solution to the basic equations, thus "scaling down" the temporal resolution of equations.

To by-pass the numerical discretization difficulties of the literature, in SESOIL the temporal resolution of the equations is "scaled up." That is, the monthly subroutine HYDROM employs subroutine HYDROA which is now run 12 times in a year for 12 "typical" years. Each typical year has, for example, an annual precipitation that equals 12 times the precipitation of the month to be simulated. At the end of the "typical" year simulation, the annual output variables P_{A, I_A} , E_{TA} , R_{SA} , $R_{\sigma A}$ are divided by 12 in order to estimate the monthly values of the month under consideration, while the annual output value s_0 (moisture content) is kept at the typical annual year output value. However, because of the new time resolution issue, the change in soil moisture storage from month-to-month becomes important. To account for this moisture budget transfer (from time t-1 to time t) in the entire column, a moisture storage term $\Delta s_0 = Z \cdot n(s_0(t) - s_0(t-1))$ has been added to the denominator of equation (HY-43) to balance (via precipitation) the deficit or surplus of moisture in the course of the months. The mathematical derivation of this logic can be traced from equation (HY-3) in conjunction with equation (HY-0) and will be documented in the future in this section.

In addition to the soil moisture storage issue, the authors had to retrieve solutions of equation HY-43 when $m_{PA}=0$ (no rain) because in the original Eagleson theory, when $m_{PA}=0$, $m_{p}=0$ and in that case many parameters (eg. α , δ , σ , G) tend to ∞ . The designed scheme will be presented in the revised documentation, but principally when $m_{pa}=0$, the $\sigma(t-1)$, G(t-1) function of a previous time step (t-1) have been used for time step (t).

Finally the authors have accounted for a "smooth" soil anisotropy along the soil column by employing the theoretical background given in Freeze & Cherry (1979, p. 32). Documentation will also be presented in the near future.

Averaged annual estimates of the soil moisture content (s_0) calculated with both the subroutines HYDROA and HYDROM, gave an excellent correlation, thus leaving the authors to believe that the accuracy obtained from HYDROM is satisfactory.

HY-53

4.4 Input/Output Variables

INPUT data to HYDROM are the:

- (1) Input data to HYDROA, and
- (2) The 12 monthly storm depths $m_{p}(M)$; (cm)

Above variables are stored in arrays CLIMA1, CLIMA2 and SOILI of the SESOIL Data Base; see Appendix DF (Data Files).

OUTPUT data from HYDROM are 12 data sets (one for each month)

 $s_{o}(M)$, P(M), I(M), R_s(M), R_g(M), Y(M)

M = 1 through 12; October through September

This data is stored in array HYDBAL.

5.0 MODEL EXTENSION

5.1 General

The hydrologic cycle routines (HYDROA, HYDROM) are based upon on analytic solution (versus numerical) approach and can be operated for monthly and annual simulations. The routines employ Eagleson's annual water balance model, the state of the art in analytic hydrologic research, which provides excellent theoretical insight to the coupling of the water balance components. This model is very easy to operate with a minimum of inputs. In addition the hydrologic cycle routines provide the feature of not requiring calibration procedure of non-physically based parameters (i.e. coefficients). Their hydrologic routines are suitable, for the time being, for:

- Seasonal (i.e. annual or monthly) simulations;
- Omission of snow or ice phenomena;
- Omission of energy and surface moisture storage in soil processes;
- Omission of vegetal influence on soil moisture movement;
- Linear superposition of moisture phenomena.

To remove the above constraints and reduce the seasonal period to less than a month (eg. storm-by-storm event), theoretically, it is necessary to formulate a finite difference solution to the basic equations. However, authors have a different approach to this issue, encompassing the use of the developed monthly routine (HYDROM) and the use of a finite difference moisture movement model. The latter will be "selfcalibrational" based upon input information received from HYDROM.

For either vegetated or bare soils (but particularly the former), the effect of surface retained water and of moisture fluctuations to the fluctuations of the groundwater table 1s of importance and should be included in this model development. In certain climates, such as the Pacific Northwest where the rain is intermittent drizzle with broken spells of sunshine, this fluctuation can be quite a significant item in the annual/monthly water balance estimates (Personal communication with Eagleson).

The phenomena of snow pack/melt and interception might be handled by assuming (in the future, however), in a semi-theoretical way, an increase (snow melt) or decrease (snow pack, interception) in precipitation. These water quantities can be separately estimated (see subsequent sections) and added or subtracted from the unit-term of equation HY-32. The snow pack/melt phenomenon might be also treated mathematically as increased "soil permeability," but additional thinking is required for this approach. A short discussion regarding above processes is given in the following sections.

5.2 Snow Pack/Melt

Two well known methods for modeling snow pack/melt phenomena are: (1) the Heat Balance Method; and (2) the Temperature Index, or Degree-Day Method. The first was developed by the U.S. Army Corps of Engineers and has been successfully applied in numerous cases. The Heat Balance Method requires extensive calculations, taking account of phenomena such as radiation, melt, condensation-convection melt, rain melt, snow density and compaction parameters, areal coverage, snow evaporation, snow pack heat and snow pack liquid water storage. The Temperature Index Method is extremely simple, but it has been reported to provide estimates which are of the same accuracy as those of the detailed Heat Balance Method (Novotny, 1976). SESOIL might employ, for example, the Temperature Index Method, adopted, however, to SESOIL'S theoretical needs:

$$S_m = abs(k_s * \overline{T}_a)$$
 (HY-44)

Arthur D Little Inc.

in which:

When the average daily temperature is below freezing temperature (i.e., $T \leq 0$ °C), precipitation becomes snowfall and accumulates as snow pack providing a minimum runoff. Conversely, when the average daily temperature is above freezing (i.e., T > 0), the snowmelt quantity is added to rainfall runoff (if any) for that period.

A simplified calculation procedure might be as follows:

Subtract (when $T \leq 0$) or add (when T > 0) the quantity $S_m^* = S_m \cdot P_A / \Delta t$

from the unit term on the left-hand side of Equation HY-32 (where Δt the simulation time step); and use the hydrologic cycle subroutines as previously described.

5.3 Interception

Interception is a function of the type and extent of vegetation, land use and meteorologic characteristics (wind, temperature, solar radiation, precipitation, etc.). In nature, all precipitation is assumed to enter interception storage until it is filled to capacity. Water is removed from the interception storage by transpiration, which may occur even during rain. For, soil interception is modeled as a "one way" phenomenon, namely as a precipitation volume retained by vegetation, which for a particular season equals (Novotny et al, 1976);

$$I_{(A,M)} = 2.54 (a + b \cdot (P/2.54))^{m} = c \cdot P(A,M)/2.54$$
 (HY-45)

HY-57

where:

```
Ir (A,M) = intercepted rainfall; annual, monthly (cm)
P=P(A,M) = average seasonal precipitation (cm);
a,b,m = constants (Table HY-6)
c = % coefficient of average interception (Table HY-6)
```

A simplified calculation procedure might be as follows:

Subtract the quantity c of the above equation from the unit precipitation term on the left-hand side of Equation HY-32 and use the hydrologic routines (annual, monthly) as described in a previous section.

6.0 REFERENCES

References of this section only are given in appendix RE.

TABLE HY-6

Constants a, b, m of the Interception Equation (H-52)

Vegetal cover	٠	٥٩٢	D
Orchards	0.04	0.18	1.00
Ash, in wouds	0.02	0.18	1.00
Beech, in woods	.0.04	0.18	1.00
Oak, in voods	0.05	0.18	1.00
Maple, in woods	0.04	0.18	1.90
Willow, shrubs	0.02	0.40	1.00
Hemlock and pine woods	° 0.05	0.20	0.50 📮 i.e.
Beans, potatoes, cabbage and other small hilled	*	*	
crops	0.02h	0.15h	1.00
Clover and meadow grass Forage, alfalfa, vetch,	0.005h	0.08h	1.00
millet, etc.	0.01h	0.10h	1.00
Small grains, rye, wheat,			
barley	0.005h	0.05h	1.00
Corn	0.005h	0.005h	1.00

* This approximation is reasonable for the sole purpose of using the equation (HY-45); $I_r = c.P(A,M)/2.54$

** The symbol h refers to the height of the plant (h in m)

Source: Novotny, et al, 1978.

Arthur D Little, Inc.

SW - soil washload

•

APPENDIX SW

TERRESTRIAL SOIL WASHLOAD SIMULATION*

			Page
1.0	INTE	RODUCTION	SW-2
2.0	''ANN	NUAL" SEDIMENT YIELD SIMULATIONS	SW-4
	2.1	General	SW-4
	2.2	Universal Soil Loss Equation (USLE)	SW-5
		2.2.1 General	SW-5
		2.2.2 Rainfall and Runoff Factor (R)	SW-6
		2.2.3 Soil Erodibility Factor (K)	SW-9
		2.2.4 Topographic Factor (LS)	SW-13
		2.2.5 Cover and Management Factor (C)	SW-17
		2.2.6 Support Practice Factor (P)	SW-31
		2.2.7 Sediment Delivery Factor (D)	SW-33
	2.3	Subroutine SEDIMA	SW-35
		2.3.1 General	SW-35
		2.3.2 Input/Output Parameters	SW-35
		2.3.3 Parameter Units	SW-35
	2.4	Numerical Example	SW-37
	2.5	Discussion	SW-39
3.0	"MON	VTHLY" WASHLOAD SIMULATIONS**	SW-41
	3.1	Objective	SW-41
	3.2	Background/Acknowledgments	SW-41
	3.3	Overview of the Monthly Washload Model	SW-43
	3.4	Model Mathematics	SW-47
		3.4.1 Basic Concepts and Equations	SW-47
		3.4.2 Modeling Issues	SW-52
		3.4.3 Sediment Characteristics	SW-53
		3.4.4 Overland Flow Element	SW-58
		3.4.5 Channel Element	SW-64
		3.4.6 Impoundment Element	SW-68
		3.4.7 Discussion	SW-68
	3.5	Subroutine SEDIMM	SW-70
	3.6	Sensitivity Analysis of SEDIMM	SW-70
	3.7	Conclusions	SW-70
4.0	REFE	ERENCES	SW-71

* This subroutine is not operational in this version; therefore, little emphasis has been placed in its accurate documentation.

**
 Information abstracted from Foster et al (1980). The authors appreciate
 his contribution (see sections 1.4 Acknowledgements and SW-3.2 of this
 appendix).

LIST OF FIGURES

Figure No.		Page
SW-1	Average Annual Rainfall Erosion (R) Values	SW-7
SW-2	Annual Rainfall Erosion (R) Values, Hawaii	SW-8
SW-3	K-Values	SW-12
SW-4	LS-Values	SW-14
SW-5	Delivery Ratio Relationship to the Watershed Size	SW-34
SW-6	Relation of Sediment Delivery Ratio to Storm Characteristics	SW-34
SW-7	Typical Field Elements	sw-40
SW-8	Schematic Representation of Typical Field Systems in the Field-Scale Erosion/Sediment Yield Model	SW-45
SW-9	Flow Chart for Detachment-Transport-Deposition Computations Within a Segment of Overland Flow or Concentrated Flow Elements	SW-48
SW-10	Schematic Representation of Convex Slope Profile for Overland Flow	SW-63

•

LIST OF TABLES

Table No.		Page
SW-1	Computed K Values for Soils on Erosion Research Stations	SW-10
SW-2	Approximate Values of the Soil Erodibility Factor, K, for 10 Benchmark Soils in Hawaii	SW-11
SW-3	Values of the Topographic Factor, LS, for Specific Combinations of Slope Length and Steepness	SW-15
SW-4	Estimated Relative Soil Losses from Successive Equal-Length Segments of a Uniform Slope	SW-15a
SW-5	Adjustment of LS-Factor, Irregular Slopes	SW-16
SW-6	Ratio of Soil Loss from Cropland to Corresponding Loss from Continuous Fallow	SW-18,19,20
SW-7	Approximate Soil Loss Ratios for Cotton	SW-21
SW-8	Soil Loss Ratios for Conditions not Evaluated in Table SW-5	SW-22
SW-9	Soil Loss Ratios (Percent) for Cropstage 4 When Stalks are Chopped and Distributed Without Soil Tillage	SW-23
SW-10	Factors to Credit Residual Effects of Turned Sod	SW-24
SW-11	Percentage of the Average Annual El which Normally Occurs Between January 1 and the Indicated Dates	SW-25
SW-12	Mulch Factors and Length Limits for Construction Slopes	SW-26
SW-13	Factor C for Permanent Pasture, Range, and Idle Land	SW-27
SW-14	Factor C for Undisturbed Forest Land	SW-29
SW-15	Factor C for Mechanically Prepared Woodland Sites	SW-30
SW-16	P Values and Slope-Length Limits for Contouring	SW-32
SW-17	P Values, Maximum Strip Widths, and Slope-Length Limits for Contour Stripcropping	SW-32

.

LIST OF TABLES (Continued)

Table No.		Page
SW-18	P Values for Contour-Farmed Terraced Fields	SW-32
SW-19	Summary of Available Sediment Transport Formulas	SW-42
SW-20	Possible Elements and Their Calling Sequence Used to Represent Field-Sized Area	SW-46
SW-21	Sediment Characteristics Assumed for Detached Sediment Before Deposition; Assumed Typical of Many Midwestern Silt Loam Soils	SW-54
SW-22	Equations Employed to Describe Particle Size Distribution	SW-55
SW-23	Assumed Typical Diameters of Particle Sizes	
SW-24	, Equations Employed for Particle Composition of Sediment Load	SW-56
SW-25	Overland Flow Element Equations	SW-59,60
SW-26	Dimensionless Sediment Transport Capacity Equation of S. Yalin [1963]	SW-61,62

1.0 INTRODUCTION

The SESOIL terrestrial washload simulation provides a quantitative seasonal prediction of soil sediment transport because of rainfall erosion.

Soil washload is the overland sediment transport of fine particles carried by surface runoff. Nutrients and pollutants can be adsorbed readily on fine soil particles and carried to receiving water bodies. In addition, sediment itself is a serious pollutant of surface water resources. The washload magnitude can be related to the available supply of solid particles in a watershed.

Washload is usually caused by land erosion and is defined by the American Geophysical Union (Konrad, <u>et al</u>, 1978) as the part of the sediment load composed of particles smaller than those found in appreciable quantities in the shifting portion of the streambed. The bedload portion is composed mostly of larger particles--sand and gravel-which originate from gulley and river bank erosion. It does not possess the high adsorptive capacity characteristic of clay and fine soil particles and may not be a significant nutrient or pollutant carrier.

Increased awareness of the ecological and financial consequences of severe erosion and resulting sedimentation on both urban and agricultural lands has increased the need for better methods of estimation deposition and sediment yield. Section 208 of Public Law 92-500 requires planning Agencies to develop plans for evaluating and controlling pollution, including sediment from nonpoint sources. Prediction equations for sediment yields are desirable in all these plans. [Neibling, W.H. and G.R. Foster, 1977.]

The mechanics of washload are very comples and it is impossible to formulate a realistic "all purpose" mathematical model at the microlevel, that will account for all variables describing the physics of transport. However, numerous mathematical algorithms for estimating sediment yield are available in the literature. The choice of an algorithm depends on the watershed characteristics, input data and objectives of the modeling effort, but in general, sediment washload can be estimated using: (1) empirical models; or (2) theoretically developed models.

May 81

SW-2

Well known and widely used empirical models, formulated employing statistical techniques to measured sediment transport yields, are the Universal Soil Loss Equation (USLE) developed by the US Department of Agriculture [Wischmeier, W.H. and D.D. Smith, 1978], and the Raiting Curves Sediment Method (RCSM) [Novotny <u>et al.</u>]. Excellent discussions of such models have been presented in the literature by Novotny [Novotny, V., 1980] and Foster [Foster, G.R., 198].

Theoretically-developed sediment yield models can be categorized into stochastic yield models [Murota and Hashino, 1969; Woolhister and Todonivic, 1974], models using kinetic wave theory [Madsen and Grant, 1976] and models using the continuity mass transport equation [Foster and Meyer, 1972, and Adams, <u>et al.</u>, 1976]. It is beyond the scope of this appendix to present a review of all models, rather an effort is made to shortly document employed models to performing "annual" and "monthly" sediment simulation. SESOIL employs: (1) USLE for annual simulations; and (2) two theoretically developed sediment yield models for monthly simulations.

2.0 ANNUAL SEDIMENT YIELD SIMULATIONS

2.1 General

Subroutine SEDIMA (Sediment Annual) of the model simulates annual sediment (soil) losses for a particular area. SEDIMA is employed by both LEVELO and LEVEL1 SESOIL simulations. Simulations are performed via the Universal Soil Loss Equation (USLE) as developed and documented by the US Department of Agriculture. [Wischmeier and Smith, 1978.]

The USLE initially developed for areas (regions) east of the Rocky Mountains has been applied to the entire United States for both urban and agricultural areas. The USLE enables planners to predict the average rate of soil erosion for each feasible alternative combination of crop systems and management practices in association with a specified soil type, rainfall pattern, and topography. The equation has been also applied to construction sites and other non-agricultural uses.

The USLE does not predict deposition, does not estimate sediment yields from gully, streambank and streambed erosion; it is applicable only for annual sediment transport predictions mainly originating from small watersheds subject to sheet and rill erosion. In case the USLE has to be applied to specific storm events or time periods, less than a year, two recent reports [Foster, G.R. <u>et al.</u>, 1977; Oustand, C.A. <u>et al.</u>, 1975] are recommended by the equation developers [Wischmeier and Smith, 1978]. SESOIL, however, does not employ for monthly or after-eachrainfall-event sediment simulations the USLE; therefore, above issue is not of concern.

In the following sections a summary of the USLE theory is presented in order to make this appendix a self-contained document for SESOIL users. Additional details regarding the USLE are to be found in the original publication. 2.2 Universal Soil Loss Equation (USLE)

2.2.1 General

The USLE is

A = R K L S C P (SW-1)

Several studies [Roehl, 1962; Denfro, 1975; Novotny, 1980] have shown that the upland erosion estimated either by an erosion model or extrapolated from measurements on small plots, does not equal the sediment nor pollutant yield measured at the watershed outlet. This fact is applicable to the USLE theory as well. To overpass these differences, a sediment delivery ratio factor D, was introduced by Novotny [Novotny, <u>et al.</u>, 1978] to account for resettling of particulate matter after or during the overland flow. Thus, the USLE equations is formulated as:

$$SYA = R K L S C P D$$
 (SW-2)

where

S	Ϋ́Α	=	annual sediment yield of basin
A		=	estimated soil loss by the USLE
R		=	rainfall and runoff factor
K		=	soil erodibility factor
L		=	slope-length factor
S		=	slope-steepness factor
С		=	cover and management factor
P		=	support practice factor
D		-	sediment delivery factor

Above factors, their units, and their numerical values for practical application (related to LEVELO and LEVEL1 simulations) are discussed in the following sections. A numerical example for sediment yield delivered by an agricultural small watershed in Clinton, Massachusetts is presented in section 2.4.

2.2.2 Rainfall and Runoff Factor (R)

The factor R encompasses a rainfall erosion index unit factor and a factor for runoff from snowmelt or applied water where such runoff is significant.

Data have shown that the rainfall factor used to estimate the average annual soil loss must include the cumulative effects of many moderate-sized storms as well as the effects of the occasional severe ones. The latter ones are represented by a rainfall erosion index (EI) theoretically presented for "single rainfall events" by the equation:

$$R_{-} = EI/100$$
 (SW-3)

with

$$EI = 210 + 89 \log_{10} I_{30}$$
 (SW-4)

where

However, for long-term and annual simulations the local value of the above index generally equals R for the USLE. R-values have been compiled by the equation developers, and can be obtained for use for both LEVELO and LEVEL1 simulations from the isoerodent map [Wischmeier and Smith, 1978, p. 7] presented in figures SW-1 and SW-2.





2.2.3 Soil Erodibility Factor (K)

The K factor in the USLE represents the soil loss rate per erosion index unit for a specified soil as measured on a unit plot, which is defined as a 21.8 m (72.6 ft) length of uniform 9% slope continuously in clean-tilled fallow.

The K factor in the USLE is a quantitative value experimentally determined on a "unit" plot arbitrarily defined. Representative values of K for various soil types and texture classes can be obtained from tables prepared by soil scientists using the latest available research information and data. Table SW-1 and SW-2 are two examples.

For soils containing less than 70% silt and very fine sand, the following regression relationship has been derived by the USLE developer:

$$K = 2.1 \times 10^{-2} M^{-1.14} (10^{-4}) (12-a) + 3.25 (b-2) + 2.5 (c-3) (SW-5)$$

where:

M = particle size parameter [mm] a = % of soil organic matter [-] b = soil structure code used in soil classification c = profile permeability class

Above equation is presented for practical application and is in figure SW-3. In tests against measured K values ranging from 0.03 to 0.69, 65% of the nomograph solutions differed from the measured K values by less than 0.02 and 95% of them by less than 0.04.

sw-9

<u>TABLE SW-1</u> Computed K Values for soils on Ercosian Research Stations

Soil	Source of data	Computed K
Dunkirk silt Ico n	Geneva, NY	'0 69
Keene sitt loam	Zanesville, Ohio	48
Shelby loam	Bethany, Mo	41
Lodi loam	Blacksburg, Va	39
Fayette silt loc n	LaCrosse, Wis	' 38
Cecl sandy day loan	Watkinsville, Ga	36
Marshall silt la im	Clarinda, Iowa	33
ida silt loam	Castana, Iowa	33
Mansic clay to im	Hays, Kans	32
Hacerstawn sil / clay loom	State College, Pc	31
Ausin clay	Temple, Tex	29
Mexico silt loc n	McCredie, Mo	28
Honeoyo silt Icam	Marcellus, NY	28
Cecil sandy loi m	Clemson, SC.	28
Ontorio loam	Geneva, NY	27
Cecil clay loom	Watkinsville, Ga	26
Boswell find sondy Icam	Tyler, Tex	25
Cecil sandy Ican	Watkinsville, Ga	23
Zaneis fine sar dy locim	Guthrie, Okla	22
Tifton loamy sind	Tifion, Go	10
Freshold loam sand	Marlboro, NJ	08
Bath flaggy sil: loam with surface stones > 2 inches removed	Arnot, NY	' 05
Albia gravelly loam	Beemarville, NJ	03

TABLE 1 --- Computed K values for soils on erosion research stations

* Evaluated rom continuous follow All others were computed from rowcrop cata

•

Souce : Evishmeier & Suck, 19997

TABLE SW-2

Toke sw-2

TABLE --- Approximate values of the soil erodibility factor, K, for 10 benchmark soils in Hawaii

Order	uborder	Grea' group	Subgroup	Family	Series	ĸ
Ult sols	Homults	Tropohumulis	Humoxic Tropohumul .	Clayey, Foolinitic, isohyperthermic	Waikane	0.10
Oxisols	I. reas	Torrox	Typic Torrox	Clayey, kaolinitic, isohyperthermic	Malakai	.24
Oxisols	U-lox	Eutrustox	Tropeptic Eutrustos	Clayey, kaolinitic, isohyperthermic	Wahiawa	.17
Vertisols	U terts	Chromusterts	Typic Chromusterts	Ver, fine, montmorillonitic, isohyperthermic	Lualualei	28
					Kawaihae	32
Aridisols	C' thids	Comborthids	Ustallic Camborthids	Medial, isohyperthermic	(Extremely stony phase)	
Inceptisols	A idepts	Dystrondepts	Hydric Dystrondepts	Thisotropic, isothermic	Kukaiau	.17
Inceptisols	A idepts	Eutrandepts	Typic Eutrondepts	Menial isohyperthermic	Naolehu (Variant)	.20
Inceptisols	A idepts	Eutrandepts	Entic Eutrondepts	Mi dial, isohyperthermic	Pakini	.49
Inceptisols	Andepty	Hydrondepis	Typic Hydrandepts	Theratropic, isohyperthermic	Hilo	10
Inceptisols	T crept.	Ustropepts	Vertic Ustropepts	Very fine kaolinitic, isohyperthermic	Woipohu	20

SOURCE | El Swoif and Dangler (9)



FIGURE 8.--The sail-eradibility nomograph. Where the silt fraction does not exceed 70 percent, the equation is 100 K = 2.1 M^{1,14} (10⁴) (12 - o) + 3.25 (b - 2) + 2.5 (c - 3) where M = (percent si + vfs) (100 - percent c), a = percent organic matter, b = structure code, and c = profile permitability class.

SW-12

2.2.4 Topographic Factor (LS)

Both the length and the steepness of the land slope affect the rate of soil erosion by rain. The two effects have been evaluated separately in research and are presented in the USLE by:

- L, the slope length factor, which is the ratio of soil loss from the field slope length to that from a 21.8 m (72.6 ft) length under identical conditions; and
- S, the slope-steepness factor, which is the ratio of soil loss from the field slope gradient to that from a 9% slope under otherwise identical conditions.

LS, therefore, is the expected ratio of soil loss per unti area from a field slope that from a 21.8 m (72.6 ft) length of uniform (9% slope under otherwise identical conditions. The following equation was derived [Wischmeier and Smith; 1978].

LS =
$$(0.045_{\lambda})^{m}$$
 (65.41 sin²0 + 4.56 sin0 + 0.065) (SW-6)

where

λ = slope length [m] Θ = angle of slope m = 0.5 for 4.5% < 0 < 4.5% 0.4 for 3.5% < 0 0.3 for % < 0 < 3.5% 0.2 for Θ < 1%</pre>

A graphical presentation of equation (SW-6) is presented in figure SW-4. Those who prefer a table may use table SW-3. For irregular slopes, the LS values (figure SW-4, table SW-3) have to be adjusted [Wischmeier and Smith, 1978, p. 16]:

- (1) by using table SW-4, and
- (2) as shown in table SW-5





gradients \sim 1 percent, 0.3 for 1 to 3 percent slopes, 0.4 for 3.5 to 4.5 percent slopes, and 0.5 for slopes of 5 percent ar steeper-
<u>TABLE SW-3</u> LS values

TABLE Values of the tapographic factor, LS, for specific combinations of slope length and steepness!

		Slope length (feet)											
Fercent slope	25	50	75	100	150	200	300	400	500	600	800	1,000	
0 :	0 060	0 069	0 07 5	0 080	6 086	0 072	0 099	0 105	0 1 1 0	0 1 1 4	0 121	0 126	
01	073	.083	090	096	104	110	119	126	132	137	145	152	
3 0	086	078	107	113	123	130	141	149	156	162	171	179	
2	133	163	185	201	∠27	248	280	305	326	344	376	402	
3	.190	.233	264	287	·_25	354	400	437	466	492	536	573	
4	.230	303	357	400	71	528	621	697	762	820	920	101	
5	268	379	464	536	656	758	928	1 07	1 20	1 31	1 52	1 69	
¢	336	476	583	.673	824	95 2	1 17	1 35	1 50	1 65	1.90	2 13	
8	496	701	859	992	1 21	141	172	1 98	2 22	2 43	2.81	3 14	
10	.685	968	1 19	1 37	1 68	194	2 37	274	3 05	3 36	387	4 33	
12	903	1 28	1 56	1 80	2 2 1	2 55	3 1 3	3 61	4 04	4 42	5 11	571	
1.	1 15	1 62	1 99	2 30	2 81	3 25	3 98	4 59	5 13	5 62	6 49	7 26	
10	1 42	2 01	2 46	284	3 48	4 01	4 92	5 68	6 35	6 95	8 03	8 98	
16	1 72	2 43	2 97	3 43	4 2 1	3 86	5 95	6 87	7 68	8 41	971	109	
2(2 04	2 88	3 53	4 08	5 00	5 77	7 07	8 16	912	10 0	115	129	

"15 = 1 72 6)^m (65 41 sin (1 + 4 56 sin (1 -- 0 065) where slope length in feet, m 0.2 for cracients < 1 percent, 03 for 1 to 3 percent lopes, 04 for 35 to 45 percent slopes, 05 for 5 percent slopis and steeper, and li - angle of slope (For other combinations of length and gradient, interpolate between adjacent values or see fig - Sur).

•

	Sequence number	Fract	ion of soil	loss
Nuntier of segments	of segment	m - ⁻ 0 5	m = 0 4	m = 0.3
2	1	0 35	0.38	0.41
	2	65	.62	.59
3	1	19	22	24
	2	35	35	.35
	3	46	43	.41
4	1	12	14	17
	2	23	24	.24
	3	30	29	.28
	4	.35	.33	.31
5	1	09	.11	12
	2	.16	.17	18
	3	21	21	21
	4	25	24	23
	5	28	27	.25

TABLE *K*-- Estimated relative soil losses from successive equal-length segments of a uniform slope¹

Derived by the formula Soil loss fraction $\Rightarrow \frac{m+1}{1-(j+1)}$ m+1N

where 1 segment sequence number, $m = slope length exponent (0.5 for slopes <math>\geq 5$ percent, 0.4 for 4 percent slopes, and 0.3 for 3 percent or less), and N = number of equal-length segments into which the slope was divided

-

Source : [wischneier & Smith, 1987]

Segment to	Table 3	Tobis 4	к	Proauct
1	1C ⁻	0 ° 9	0 27	0 055
2	274	35	32	307
3	512	46	37	871
			KLS .	1 233

Seyment 1 7 3	Pircent slope 5 10 15	Table 3 1 07 2 74 5 12	Table 4 0 19 35 46	Product 0 203 959 2 355
			LS :	= 3 517

.

2.2.5 Cover and Management Factor (C)

The factor C represents the ratio of soil loss from an area with specified cover and management to that from an identical area in tilled continuous fallow.

This factor represents the combined effect of all the interrelated cover and management variables. Deriving the appropriate C values for a given locality requires knowledge of how the erosive rainfall in that locality is likely to be distributed through the 12 months of the year and how mud erosion protects the growing plants, crop residues and selected management practices will provide at the time when erosive rains are most likely to occur.

For an optimal derivation of C values, the reader is referred to the corresponding chapter of the original research. [Wischmeier and Smith, 1978, p. 17.] The estimation of the C-factor is the most time consuming effort when employing the USLE. C-values can be estimated for: (1) Agricultural areas, (2) Construction areas, (3) Pasture, Range and Idle Land, and (4) Woodland. Some information is presented below.

Agricultural Areas

Tables SW-6 through SW-11 provide details needed by a trained agronomist to develop simple handbook tables of C values for conditions in specific climatic areas. The tables are self-explanatory and within their broad limits of accuracy these tables can supply the research data needed to complete the estimation of C. The procedure is not that straightforward for site specific applications; however, it is well explained in a "problem-procedure" on page 29 of the original work.

Construction Areas

Applied mulches immediately restore protective cover on denuded areas and drastically reduce C that has a maximum value C=1. Soil loss ratios for various percentages of cover are presented in table SW-12.

Pasture, Range, and Idle Land

Factor C values for a specific combination of cover conditions on these types of land may be obtained from table SW-13.

1	Cover clop sequence,	Spring	Cover		So	il loss period	l and	o' to: cano	(10) 07 (0)	stage er '		Line	Cuver, crop sequence,	Spring	Cover		50	period	rutio Land	0 10r	(10) (6)	er er	
NU,	and management	residue.	, 'e v'	Ŧ	58	;	2	3 80	90	74	-i	N0	and management.	residue	r'~~'	;	.	:	2	3 80	50	75	41
	COPN AFTER C, GS, G OR C IN MEADOWLESS SYSTEM	16 01 5	Pet	Pct	Pet	Pet	Pet	Pet	Per	Pel	Pel		CORN AFTER WC OF RYEGRAS OR WHEAT SEEDED IN	16 15	Pct	Pet	Pet	PcI	Pct	Pel	Pel	Pet	Pc
1	Roll sorg TP	4 500		31	55	48	38			20	23		WC reaches slemming slage										
2		3,400		36	69	52	41		24	20	30	79	No till pl in killed WC	4,000			7	7	7		7	6	"
J		2,600		43	64	56	43	32	25	21	37	80		3,000			11	11	п	11	Y	'	
4		2,000		51	68	60	45	33	25	27	47	81		2,000			15	15	14	14	11		
5	RdL, full TP	HP'		44	65	53	38			20		82		1,500			70	10	18	18	14	11	
6		GP		49	70	57	41		24	20			Strip till one-fourth row sp	ace								_	
7		fP		57	74	61	43	32	25	21		83	Rows U.'D slope	4,000			13	12	11		11	. 9	"
9		(P		٥5	78	65	45	32	26	22		84		3,000			16	17	16	10	13	10	
•	RdR, sprg TP	HP		66	74	65	47			22	56	65		2,000			23	22	20	22	13	14	
10		GP		67	75	65	47		27	23	62		• • • •	1,500			40	20	24	~ ~			
12		FP 10		68	76	67	48	35	27		69	87	Rows on contour"	4,000			10	10	10		10	8	· (· ·
				07	"	08	49	35			/4	80		3,000			20	20	10	10	15	12	
13	RdR, foli TP	HP		76	82	70	47			22		90		1,500			25	24	23	22	17	14	
14		GP 50		79	83	71	50	76	27	23			TP and andhed	4 000		74	40	62			2.4	20	
10		10		70	R4	73	52	35	"			02	ir, (0n/ Seedbed	3,000		41	64	56	41	u	25	21	`
17		4 500		•••			2.			10		93		2.000		51	68	60	45	33	26	22	
1.8	Wheele dek pi, koi, re	3,400			16	12	30		77	18	73	\$4		1 500		61	73	64	47	35	27	23	
10		2.600			43	36	32	29	23	19	37		WC surrulent blades only										
20		2 000			51	43	36	31	24	20	47	95	No-till pl in killed WC	3.000			11	11	17	23	18	16	(''
21	Deep off set disk or	4 500	10		45	38	34	_		20	23	96		2,000			15	15	20	25	20	17	
72	disk plow	3 400	10		52	43	37		24	20	30	97		1,500			20	20	23	26	21	18	
~3	•	2 600	5		57	48	40	32	25	21	37	98		1,000			26	25	27	27	22	17	
24		2 000			61	51	42	33	26	22	47	99	Strip till one-fourth row space	e 3,000			18	18	21	25	20	17	1
25	No till plant in crop residuat	6,000	95		2	2	2			2	14	100		2 000			23	23	25	27	21	18	
75		6 000	90		3	3	3			3	14	101		1,500			28	26	28	23	22	19	
27		4 500	80		5	5	5			5	15	ICZ		1,000			33	33	31	25	23	20	
2.		3,400	70		2	8	8		8	6	19		CORN IN SOD-BASED SYSTEM	S									
7		3 400	03		12	12	12	12		8	23		No till pl in killed sod										
11		3 400	20		15	10	14	14			27	103	3 to 5 tons hay yid				1	2		-	;	2	
17		2 600	30		26	24	22	21	17	14	30	104	TO 2 TONS HOY YID				4	•	4	•	•	•	
	Charles I allowed at the	1 000			••	•••	••	• •	.,				Strip till, 3 5 ton M				-	-	•		-	•	
	fid cult as only "illage											105	20 percent cover, tilled strip 20 percent cover, tilled strip	25 25			3	3	3		Ĵ	3	
33	On moderate slopes	6,000	70		8	8	7			7	17		Strip fill, 1-2 ton M										
34			60		10	9	8			8	17	107	40 percent cover, tilled strip	5			4	4	4	4	4	- 4	
35			50		13		10			. 9	18	108	20 percent cover, tilled strip	5			5	5	5	5	5	5	2
30			40		15	13				10	19		Other tillage after sod			(P)	e'')	(11)	C ')	(14)	(*)	('')	
10			20		27	20	10			14	20		CORN AFTER SOUREANS										
30					~ ~							109	Sore TP. conv till	нр		40	72	60	48			25	21
35	Do	4,500	/0		12	10					18	110		GP		47	78	65	51		30	25	3
40			50		14	17	, i				10	111		FP		56	83	70	54	40	31	26	4
42			40		17	15	13			10	20	112	Fall TP, conv till	нр		47	75	60	48			25	
43			30		21	18	15			13	21	113		GP		53	81	65	51		30	25	
1.4			20		25	22	10			14	77	114		FP		62	84	70	54	40	31	26	

TABLE 5.- Ratio of soil loss from cropland to corresponding loss from continuous fallow

SW-18

4A 5D 1a 3 12 12 7 2 11 (1) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) </th <th>45</th> <th>Uo</th> <th>3,400</th> <th>60</th> <th>1.</th> <th>11</th> <th>19</th> <th></th> <th>10</th> <th>8</th> <th>20</th> <th>115</th> <th>Fall & sprg chise! or cult</th> <th>HP</th> <th>' 30</th> <th></th> <th>10</th> <th>15</th> <th>27</th> <th></th> <th></th> <th>23</th> <th>79</th>	45	Uo	3,400	60	1.	11	19		10	8	20	115	Fall & sprg chise! or cult	HP	' 30		10	15	27			23	79						
27	46			50	ie	- 13	12		12	;	24	116	· •	CP	25		4"	-6	ינ		27	73	37						
48	27			40	19	17	16		14	11	25	117		GP	20		51	44	39	34	27	23	37						
47 57 57 71 16 77 17 16 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 77 <td< td=""><td>48</td><td></td><td></td><td>30</td><td>23</td><td>21</td><td>19</td><td></td><td>17</td><td>14</td><td>26</td><td>118</td><td></td><td>FP</td><td>15</td><td></td><td>58</td><td>51</td><td>44</td><td>25</td><td>28</td><td>23</td><td>44</td></td<>	48			30	23	21	19		17	14	26	118		FP	15		58	51	44	25	28	23	44						
91 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 <th< td=""><td>4?</td><td></td><td></td><td>20</td><td>29</td><td>25</td><td>23</td><td></td><td>21</td><td>16</td><td>27</td><td>119</td><td></td><td>LP</td><td>10</td><td></td><td>67</td><td>57</td><td>48</td><td>36</td><td>28</td><td>23</td><td>54</td></th<>	4?			20	29	25	23		21	16	27	119		LP	10		67	57	48	36	28	23	54						
51 Do 2.60 50 17 6 13 13 10 24 10 12 10 12 10 10 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 <t< td=""><td>50</td><td></td><td></td><td>10</td><td>36</td><td>32</td><td>29</td><td></td><td>24</td><td>20</td><td>30</td><td>120</td><td>No till of in crou ies d</td><td>КР</td><td>' 10</td><td></td><td>25</td><td>20</td><td>19</td><td></td><td>14</td><td>11</td><td>26</td></t<>	50			10	36	32	29		24	20	30	120	No till of in crou ies d	КР	' 10		25	20	19		14	11	26						
32 is is <th< td=""><td>51</td><td>Do</td><td>2,600</td><td>50</td><td>17</td><td>16</td><td>15</td><td>15</td><td>13</td><td>10</td><td>29</td><td>121</td><td></td><td>Gr</td><td>ວບໍ</td><td></td><td></td><td>1,</td><td></td><td></td><td>.2</td><td>14</td><td></td></th<>	51	Do	2,600	50	17	16	15	15	13	10	29	121		Gr	ວບໍ			1,			.2	14							
	52		-,	40	21	20	19	19	15	12	30	122		FP	2Û		44	38	32	27	23	18	40						
14 20 22 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 29 20 27 27 17 13 11 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 <th< td=""><td>53</td><td></td><td></td><td>30</td><td>25</td><td>23</td><td>22</td><td>27</td><td>10</td><td>14</td><td>32</td><td></td><td>REAMS WHILE CORN</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	53			30	25	23	22	27	10	14	32		REAMS WHILE CORN																
35 10 41 36 34 36 34 36 34 36 34 36 34 36 34 36 36 36 36 36 36 36 36 36 36 36 36 36 36 36 36 36 36 36 36 36 36 37 38 37 36 77 16 17 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 17 18 18 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 11 10 10 10 <t< td=""><td>51</td><td></td><td></td><td>20</td><td>32</td><td>29</td><td>28</td><td>27</td><td>22</td><td>17</td><td>34</td><td>121</td><td>See TR Bdl soon till</td><td>ND</td><td></td><td>22</td><td>40</td><td>52</td><td>78</td><td></td><td>20</td><td>17</td><td>/! • •</td></t<>	51			20	32	29	28	27	22	17	34	121	See TR Bdl soon till	ND		22	40	52	78		20	17	/! • •						
56 De 2,000 40 22 21 20 20 75 77 72 77 72 77 72 77 73 77 74 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 75 76 76 76 76 76 76 76 76 76 76 76 76 76 76 76 76 76 76 76	55			10	4	36	34	32	25	21	37	124	spire ir, kat, tony nu	C.P		33	64	54	41		21	19	()						
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	56	Da	2 000	40	27	21	20	20	15	12	17	125		60		45	48	- CC	41	20	27	10							
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	57	00	2,000	20	2.	25	24	20	10	15	30																		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	58			20	1	22	20	22	22	19	42	126	Fall IP, RdL, conv till	HP		45	67	57	39		20	17	(")						
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	59			10		47	79	11	24	22	47	127		GP		57	73	01	41	••	21	18							
	•	On slopes 12 percent					30		20	~~	-/	128		+P		59	77	65	41	27	22								
BEAMS AFTER BEAMS γ $(\gamma$	60	Luns 33 59 t mer forto			1		11	10	10	10	10		Chusel or fid cult			(¹)	(°)	1	<i>t</i> r*	(")	(°)	(")	(')						
GAINN AFTER C, G, GS, COT** GAINN AFTER C, G, GS, COT** Intens 33 9° Imme Tactor of GAINN AFTER C, G, GS, COT** In disked residues 3,400 60 16 14 12 1 7 4 2 (°) GAINN AFTER C, G, GS, COT** In disked residues 3,400 60 16 14 12 2 1 1 4 2 (°) GAINN AFTER C, G, GS, COT** In disked residues 3,400 60 16 14 12 2 1 1 4 2 (°) GAINN AFTER C, G, GS, COT** In disked residues 3,400 60 16 14 12 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	•••				•		• •	••		••	•••		BEANS AFTER BEANS			٦	()	1	(,	()	(¹)	(')	(**)						
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		abund on Ord suit											GRAIN AFTER C. G. GS. COT																
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		Lunes 23 59 tumos factos of										129	In disked residues	4.500	70		12	12	11	7	4	2	('')						
Consider of the probability of the second of the secon	A1	On moderate slepp			,			10			10	130		3,400	60		16	14	12	7	4	2							
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	42				;			10	10	10	10	131			50		22	18	14	8	5	3							
nings plon! 33 59 intres factor of 133 59 intres factor of 20 38 20 11 10 6 3 c.i Rows on contour 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 </td <td>~</td> <td></td> <td>6111</td> <td></td> <td></td> <td></td> <td>• • •</td> <td></td> <td></td> <td></td> <td></td> <td>132</td> <td></td> <td></td> <td>40</td> <td></td> <td>27</td> <td>21</td> <td>16</td> <td>2</td> <td>5</td> <td>3</td> <td></td>	~		6111				• • •					132			40		27	21	16	2	5	3							
Lines 31 39 intres factor of information informating information information information inform		Ridge plant										133			30		32	25	18	9	6	3							
C1 Nows on contour 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7		Lines 33 39 limes 'de'or e'					-	_	_	_	-	134			20		38	20	21	10	6	3							
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	4	Rows on contour				<u> </u>				.,		135	Pa.	2.600	40		29	74	19	•	6	3							
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	04	Kows U/D slope - 12 per	rcent					10	10	10	10	136	05	2,000	20		43	34	74	ú	7								
Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant Init plant <th colspan="6" init<="" td=""><td>63</td><td>Rows U/D stope > 12 per</td><td>rcent</td><td></td><td></td><td>y ,</td><td>10</td><td>10</td><td>10</td><td>10</td><td>19</td><td>137</td><td></td><td></td><td>10</td><td></td><td>52</td><td>37</td><td>27</td><td>12</td><td>7</td><td>4</td><td></td></th>	<td>63</td> <td>Rows U/D stope > 12 per</td> <td>rcent</td> <td></td> <td></td> <td>y ,</td> <td>10</td> <td>10</td> <td>10</td> <td>10</td> <td>19</td> <td>137</td> <td></td> <td></td> <td>10</td> <td></td> <td>52</td> <td>37</td> <td>27</td> <td>12</td> <td>7</td> <td>4</td> <td></td>						63	Rows U/D stope > 12 per	rcent			y ,	10	10	10	10	19	137			10		52	37	27	12	7	4	
Lines 33 39 trines foctor of 7 85 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10		Tili plant										170	D-	2 000	10		70	20		11	2								
Cold Rows on contour!1 7 85 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 10 <th1< td=""><td></td><td>Lines 33 59 times factor of</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>130</td><td>08</td><td>2,000</td><td>20</td><td></td><td>30</td><td>36</td><td>25</td><td>12</td><td>÷</td><td>2</td><td>.,</td></th1<>		Lines 33 59 times factor of										130	08	2,000	20		30	36	25	12	÷	2	.,						
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	66	Rows on contour"			_	78	5 10	10	10	10	10	140			10		54	47	20	12	á	2							
Strip till one faurth of row specing 77 67 78 79 67 71 10 67 71 7 57 11 67 7 57 77 7 57 67 77 5 71 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 7 5 7 7 5 7 7 5 7 7 5 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	•••	Kows U'D slope < 7 perc	ent		1	0 10	10	10	10	10	10				10				30										
n^{e} Rows on contour!! 4,500 "60 12 10 9 8 23 142 Winter G after fall TP, RdL HP 31 55 48 31 12 7 5 7 60 3,400 50 16 14 12 11 10 27 143 GP 36 60 52 33 13 8 5 70 2,600 40 22 19 17 17 14 12 0 144 GP 43 64 50 36 60 38 15 10 6 71 2,000 30 27 23 21 70 16 13 36 145 IP - 53 68 60 38 15 10 6 72 Rows UD stope 4,500 60 16 13 11 27 146 With grain residues 200 10 70 55 43 18 13 11 1 11 14 12 71 14 12		Strip till one fourth of row sp	oacınıs									141	In disked stubble, RdR				79	62	42	17		6	1						
69 3,400 50 16 14 12 11 10 27 143 GP 36 60 52 3J 13 8 5 70 2,600 40 22 19 17 17 14 12 30 144 FP 43 64 56 34 14 9 5 71 2,000 30 72 23 21 70 14 12 11 20 145 C FP 43 64 56 34 18 13 11 9 73 72 Ro vs U D s'ope 4,500 40 26 22 19 17 14 12 11 27 146 With groun residues 200 10 70 55 43 18 13 11 1 17 14 12 30 147 100 50 20 10 70 55 43 18 13 11 1 17 74 100 10 20 10 18 10	4°C	Rows on contour"	4,500	'^60	12	10	9			8	23	142	Winter G after fall TP, RdL	HP		31	55	48	31	12	7	5	')						
70 2,600 40 22 19 17 14 12 30 144 (FP) 43 64 56 34 14 9 5 71 2,000 30 27 23 21 70 16 13 36 144 (P) 53 68 60 38 15 10 6 72 Ro vs. U D s'ope 4,500 60 16 13 11 9 23 GRAIN AFTER SUMMER FALLOW 70 55 43 18 13 11 1 9 72 144 9 5 70 50 30 34 22 13 10 8 74 2,600 40 26 22 19 17 14 12 30 147 144 750 40 34 22 13 10 8 7 7 74 2,600 40 13 12 11 12 20 144 13 12 16 14 14 13 12 16	69		3,400	50	16	14	12		11	10	27	143		GP		36	60	52	31	13	8	5							
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	70		2,600	40	23	19	17	17	14	12	30	144		FP		43	64	56	34	14	9	5							
72 Rovs U D s'ope 4,500 ' 60 16 13 11 9 23 GRAIN AFTER SUMMER FALLOW 73 3 400 50 20 17 14 12 11 27 146 With grain residues 200 10 70 55 43 18 13 11 1 74 2,600 40 26 22 19 17 14 12 30 147 500 30 13 27 18 10 7 7 75 2,600 40 26 22 19 17 14 12 30 147 500 30 43 42 13 10 8 75 400 30 40 13 12 11 22 150 1,500 60 20 16 12 7 5 5 76 Rows on contour ¹¹ 3,400 40 13 12 11 11 22 150 1,500 60 16 12 7 5 5 5 77 <td>71</td> <td></td> <td>2,000</td> <td>30</td> <td>27</td> <td>23</td> <td>21</td> <td>20</td> <td>16</td> <td>13</td> <td>36</td> <td>145</td> <td></td> <td>LP</td> <td>•</td> <td>53</td> <td>68</td> <td>60</td> <td>38</td> <td>15</td> <td>10</td> <td>6</td> <td></td>	71		2,000	30	27	23	21	20	16	13	36	145		LP	•	53	68	60	38	15	10	6							
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	72	Rows U D stope	4,500	¹ 60	16	13	11			9	23		GRAIN AFTER SUMMER FALLOW																
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	73		3 400	50	20	17	14		12	11	27	146	With groin residues	200	10		70	55	43	18	13	11	(1)						
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	74		2,600	40	26	22	19	17	14	12	30	147	-	500	30		43	34	23	13	10	8							
Vori-full 149 1,000 50 26 21 15 8 7 6 76 Rows on contour!! 3,400 40 13 12 11 11 22 150 1,500 60 20 16 12 7 5 5 77 3,400 30 16 15 14 14 13 12 26 151 2,000 70 - 14 11 9 7 5 5 79 2,600 20 - 21 19 19 19 16 14 34 152 With row crop residues 300 5 82 63 44 19 14 12 (²¹) 79 2,600 20 - 21 19 19 16 14 34 152 With row crop residues 300 15 - 62 49 35 17 13 11 154 750 23 - 50 40 31 24 13 10 8 7 <td>75</td> <td></td> <td>2,000</td> <td>30</td> <td>31</td> <td>26</td> <td>23</td> <td>20</td> <td>16</td> <td>13</td> <td>36</td> <td>148</td> <td></td> <td>750</td> <td>40</td> <td></td> <td>34</td> <td>27</td> <td>18</td> <td>10</td> <td>7</td> <td>7</td> <td></td>	75		2,000	30	31	26	23	20	16	13	36	148		750	40		34	27	18	10	7	7							
76 Rows on contouril $3,400$ 40 13 12 11 11 22 150 $1,500$ 60 20 16 12 7 5 5 77 $3,400$ 30 16 15 14 14 13 12 26 151 $2,000$ 70 $ 14$ 11 9 7 5 5 79 $2,600$ 20 $ 21$ 19 19 16 14 34 152 With row crop residues 300 5 82 65 44 19 14 12 $(^{11})$ 79 $2,600$ 20 $ 21$ 19 19 16 14 34 152 With row crop residues 300 15 -62 49 35 17 13 11 154 75 23 $ 50$ 40 29 14 13 10 8 155 $1,500$ 45 <td></td> <td>Vari-till</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>149</td> <td></td> <td>1,000</td> <td>50</td> <td></td> <td>26</td> <td>21</td> <td>15</td> <td>8</td> <td>7</td> <td>6</td> <td></td>		Vari-till										149		1,000	50		26	21	15	8	7	6							
77 3,400 30 16 15 14 14 13 12 26 151 2,000 70 - 14 11 9 7 5 5 79 2,600 20 - 21 19 19 19 14 34 152 With row crop residues 300 5 82 65 44 19 14 12 (²¹) 79 2,600 20 - 21 19 19 16 14 34 152 With row crop residues 300 5 82 65 44 19 14 12 (²¹) 153 153 500 15 - 62 49 35 17 13 11 154 750 23 - 50 40 29 14 11 9 155 1,000 30 - 40 31 24 13 10 8 156 1,500 45 31 24 18 7 5 15	76	Rows on contour!!	3.400	40	13	12	11			11	22	150		1,500	60		20	16	12	7	5	5							
79 2,600 20 - 21 19 19 16 14 34 152 With row crop residues 300 5 82 65 44 19 14 12 (21) 153 500 15 - 62 49 35 17 13 11 154 750 23 - 50 40 29 14 11 9 154 750 23 - 50 40 29 14 11 9 155 1,000 30 - 40 31 24 13 10 8 156 1,500 45 31 24 18 10 8 7 157 2,000 55 23 19 14 8 7 5 158 2,500 65 17 1 12 7 5 4	77		3,400	30	14	15	14	14	13	12	26	151		2,000	70	-	14	11	9	7	5	5							
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	79		2,600	20	- 21	19	19	19	16	14	34	152	With row crop residues	300	5		82	65	44	19	14	12	(**)						
$\begin{array}{cccccccccccccccccccccccccccccccccccc$			-,		-	.,	.,			• •		157		500	15	-	62	49	35	17	13	11							
155 1,000 30 - 40 31 24 13 10 8 156 1,500 45 31 24 18 10 8 7 157 2,000 55 23 19 14 8 7 5 158 2,500 65 17 1 12 7 5 4												154		750	23	-	50	40	29	14	11	9							
156 1,500 45 31 24 18 10 8 7 157 2,000 55 23 19 14 8 7 5 158 2,500 65 17 1 12 7 5 4												155		1.000	30	-	40	31	24	13	10	8							
157 2,000 55 23 19 14 8 7 5 158 2,500 65 17 1 12 7 5 4												156		1.500	45		31	24	18	10	8	7							
158 2,500 65 17 1 12 7 5 4												157		2.000	55		23	19	14	8	7	5							
												158		2,500	65		17	11	12	7	5	4							

159

160

POTATOES

Contoured rows, ridged when

Rows with slope

43 64 56 36 26 19 16

43 64 56 18 13 10 8

TABLE

Sw -6

(contrinued)

_{sw-19}

footnotes for table 5

¹ Symbols B, saybeans, C, corn, conv till plaw, disk and harrow for seedbed, cat, cottan, F raigh rollow Nd cult, field cultivatur, G small grain, GS grain sorghum, M, grass and lingume minidaw, at least 1 full year, pl, plant, RdL, crap residues left an field, RdR crap residues removed, SB seedbed period, sprg spring, TP plawed with moldboard, WC, winter cover crap. — insignificant or an unlikely combination of variables

Dry weight per acre after winter loss and reductions by grazing ar partial remaval 4,500 lbs represents 100 to 125 bu corn, 3,400 lbs, 75 to 99 bu, 2,600 lbs, 60 to 74 bu, and 2 000 lbs, 40 to 59 bu, with normal 30 percent winter loss For RdR or fall-blow practices, these four productivity levels are indicated by HP, GP, FP and LP, respectively thinh goad fair and law productivity) In lines 79 to 102, this column indicates dry vight of the winter-cover crop

Pricentage of sail surface covered by plant residue mulch after crap seeding. The diffurence between spring residue and that on the surface after crap seeding is reflected in the sail loss ratios as residues mixed with the topsail.

The soil loss ratios given as percentages, assume that the indicated crop sequence and practices are followed consistently. One year deviations from normal practices do not have the effect of a permonent change Linear interpolation between lines is recommended when justified by field conditions.

C-opstage periods are as defined on p=18. The three columns for crapstage 3 are for R0 90, and 96 to 100 percent conopy cover at maturity.

Column 4L is for all residues left an field. Corn stalks partially standing as left by some mechanical pickers. If stalks are shredded and spread by picker, select ratio from table 5C. When residues are reduced by grazing, take ratio from lawer spring-residue line.

Period 4 values in lines 9 to 12 are for corn stubble (stover removed)

`Inversion plawed, no secondary tillage. For this practice, residues must be left and incorporated

"Soil surface and chopped residues of matured preceding crop undistuibed except in narrow slots in which seeds are planted

""Top of old row ridge sliced off, throwing residues and some sail into furrow areas Reridging assumed to accur near end of cropstage 1

"Where lower soil loss ratios are listed for rows on the contour, this reduction is in addition to the standard field contauring credit. The P value for contauring is used with these reduced loss ratios.

¹⁻ Field-average percent cover, probably about three-fourths of percent cover on un disturbed strips

¹ If again seeded to WC crop in corn stubble, evaluate winter period as a winter aroin seeding (lines 132 to 148). Otherwise, see table 5 C

"Select the appropriate line for the crop, tillage, and productivity level and multiply the listed soil loss ratios by sod residual factors from table 5 D

"Spring residue may include corryover from prior corn crop

"See toble 5 C

¹ Use volues from lines 33 to 62 with appropriate dates and lengths of cropstage pariods for beans in the locality

"Values in lines 109 to 122 are best available estimates, but planting dates and lengths of cropstages may differ

"When meadow is seeded with the grain, its effect will be reflected through higher percentages of cover in cropstages 3 and 4

²⁰ Ratio depends on percent cover. See table 5-C

" See item 12, table 58.

TABLE S

5 w-	7
-----------------	---

TABLE	5-A Approximate soil loss	ratios	for	cotton
Expected	final + anopy percent cover	65	80	95
Estimate	d initial percent cover from defoliction - is down	30	45	¢0
Proutice Number	Tiliciae operation,s	Se	n loss	rntio'
COITON	ANTIUALLY		Perce	r'
1	None			
	Defo ation to Dec. 31	36	24	15
	Jin 1 o icu or Mar tillage			
	Cu Fd only	52	41	32
	Pet & 2C percent cover vol veg	32	26	20
	Rc & 30 percent cover vol veg	26	20	14
,	Ch. et slaw soon after cat harvest			
	Chaeling to Dec 31	40	31	24
	Jan i to sprg tillage	56	47	40
3	Fall di k atter chisel			
	Diking to Dec 31	53	45	37
	Jo 1 to sprg tillage	62	54	47
4	Chise' alow teb Mar, no prior tillege			••
		50	42	35
	Ko & 20 percent vol veg	39	33	28
	KO & JU percent vol veg	34	29	25
5	Eed (ip ' leb Mar no priar tillage		• •	
		100	84	70
	KC & 20 percent vol veg	78	00	20
	KC & 30 percent vol veg	68	28	50
	Split ridges 6 plant after hip, or			
	Ca Del selv	43		
		51	24	4/
	RC & 20 percent vol vcg	33	4/	41
		30	**	30
		67		
	Ro & 20 percent vol von	37	43	43
	Rr & 30 percent vol veg	47	41	30
	Cror tope 2	45	10	30
		40	27	17
\$	Bed thay after 1 prior tillanc		•'	
	Cet Rd only	110	94	84
	Rt & 2C percent ven	94	82	72
	Rr & 3(percent yea	90	78	68
	Split ridge: after hip 'SB'			•••
	Co Rd only	66	61	52
	Rri & 20 to 30 percent yes	61	55	49
	Croj Juge 1			
	Cr Rd cnly	60	5c	49
	Rr & 20 to 30 percent veg	56	51	46
	Cro; Jage 2	47	44	38
	Crop lage 3	42	30	19
7	hip of er 2 prior tillages			
	Cri Ra only	116	105	78
	Ro & 20 30 percent veg	108	9:	88
	Split ridges ofter hip (SB	67	62	57
8	Hip alter 3 or more fillinges	120	110	102
	Split ridge, after hip (SB)	68	٤·	59
?	Conventional moldboard plaw and disk			
	Fallow period	42	39	36
	Seensed pariod	68	6	59
	Croj stage 1	63	57	55
	Crop stops 2	49	46	43
	Cro, stegi 3	44	3:	2.
	Cropstage 4 See practices 1 2 and	3		

COTTON AFTER SOD CROP

For the first or second crop after a grass or grass and legime multiply values given in the loss five lines above by soil residual factors from table 5 D

COTTON AFTER SOVEEANS

Select volu s from obose and multiply by 1.25

Source : []

Footwote

¹ Alternate procedure for estimating the soil loss ratios

The rotios given above for cotton are based on estimates for roductions in percent cover through normal winter loss and by the successive tillage aperations. Research is underway in Mississippe to obtain more accurate residue data in relation to tillage practices. This research should provide more accurate soil loss ratios for cotton within a few years.

Where the reductions in percent cover by winter loss and tillage operations are small, the following procedure may be used to compute soil loss ratios for the preplant and seedbed periods. Enter figure 6 with the percentage of the field surface covered by residue mulch, move vertically to the upper curve and read the mulch factor on the scole at the left Multiply this factor by a factor selected from the following tabul tion to multiply this factor by a factor selected from the following tabul tion to mult for effects of land use residual, surface roughness and points the

Productive ** .	No	Rough	Smoothea
level	tillage	surface	surface
High	0 66	0.50	0 56
Mudium	71	54	61
fao	75	59	65

 List for the predictor private private states of less to mail preservations of the private of the class computed above for rough surfaces.

* Pill crop residuel volver, volunteer vegolation

Sw-B

TABLE 58.—Soil loss ratios for conditions not evaluated in table 5

COTTON See table 5-A CROPSTAGE 4 FOR RUWCROPS Stalks proken and partially standing. Use col 4L Stalks standing after hand picking. Col. 4L times 1.15 Stalks ihredded without soil tillage. See table 5 C tall chise! Select values from lines 33-62, seedbed column CROPSTAGE 4 FOR SMALL GRAIN-Sec tot le 5.C COUBLE CROPPING Derive annual C v lue by selecting from table 5 the soil loss per centages for it's successive cropstage periods of each crop ESTABLISHED MEADO'Y, FULL-YEAR PERCENTAGES-Grass and legume nix, 3 to 5 t hay 04 Do. 2 to 3 t hay .6 10 Do 1 t hoy Serices, after second year 1.0 Red clover 15 Alfalfa lespedeza, and second-year sericea 2.0 2.5 Sweetciover MEADOW SEEDING WITHOUT NURSE CROP-Determine appropriate lengths of cropstage periods SB, 1, and 2 and apply values given for small groin seeding PEANUTS Comparison with surbeans is suggested PINEAPPLES. Direct data not available. Tentative values derived analytically as available from the SCS in Hawaii or the Western Technical Ser vice Center ct Portland, Oreg (Reference 5) SORGHUM beloct values given for carn, on the basis of expected grop residues and conopy cover SUGARBEI TS. Direct data not av ilc'ble Probably most nearly comparable to pr tations, without the ridging credit. SUGARCANE. entative values availably from sources given for pineapples SUMMER FALLOW IN LOW-RAINFALL AREAS, USE GRAIN OR ROW CROF RESIDUES the opproximate soil loss percentage after each successive tillage operation may be obtained from the following tabulation by estimoting the perioni surface cover after that fillogs and selecting the column for the appropriate amount of initial residue. The given values condit bonefits of the residue mulch, residues mixed with soil by tillage, and the grop system residual Percen cover initial residue (lbs/A) by nulch ~4 000 3,000 2,000 1,500 9(4 -------80 B 18 ۲ 12 114 13 60 16 17 **18** 119 5(20 '25 22 24 :(25 27 30 32

For crain residue only

30

20

1(

WINTER COVER SEEDING IN ROW CROP STUBBLE OF RESIDUES Define cropstage periods pased on the cover seeding date and apply values from 1: es 12th to 145

33

39

55

37

44

63

39

46

58

29

35

47

TABLE S.C. Soil loss ratios (percent) for cropstage 4 when stalks are chopped and distributed without soil tillage

	Corn or	Sorghum	Soyl	beans	
Mulch cover ¹	Tilled	No uli	Tilled seedbed-	No till in corn rd ³	Grain Stubble'
20	48	34	60	42	48
30	37	26	46	32	37
40	30	21	38	26	30
50	22	15	28	19	22
60	17	12	21	16	17
70	12	8	15	10	12
80	7	5	9	6	7
90	4	з		-	4
95	3	2			3

Part of a field surface directly covered by pieces of residue mulch This column applies for all systems other than no till

Cover after bean horvest may include an appreciable number of stalks carried over from the prior corn crop ' for grain with meadow seeding, include meadow growth in percent

' for grain with meadow seeding, include meadow growth in percent cover and limit grain period 4 to 2 ma. Thereafter, classify as established meadow

Source: []

TABLE	5	D	••	Factors to	credit	residual	efiects	of	turned
					'hos				

6	No	ige peri	bd			
Crop	HOY YIELD	Ŧ	SB and 1	2	3	4
	Tons					
First year after mead						
Row crop or grain	35	0 2 5	0 40	0 45	0 50	0 60
	2-3	30	45	.50	55	65
	1-2	35	50	55	60	70
Second year after mea	Ŀ					
Row crop	35	70	80	.85	90	.95
	23	75	.85	90	95	10
	1-2	80	.90	95	1.0	10
Spring groun	35		75	80	.85	.95
	23		80	85	90	10
	1.2		.85	90	95	10
Winter grain	3 5		60	70	85	95
	2.3		65	75	90	10
	1.2	-	70	85	95	1.0

¹ These factors are to be multiplied by the appropriate soil loss percentages selected from table 5 They are directly applicable for sod forming meadows of at least 1 full year duration, plowed not more than 1 month before final seedbed preparation

When sod is fall plawed for spring planting the listed values far all cropstage periods are increased by adding 0.02 for each additional month by which the plawing precedes spring seedbed preparation. For example, September plawing would precede May disking by 8 months and 0.02(8 1), or 0.14, would be added to each value in the table For nonsad furming meadows, like sweetclaver or lespedeza, multiply the factors by 1.2 When the computed value is greater than 1.0, use as 1.0

TABLE SW-11

Area	 	ŗ	cr.		Feb.		Mar		Ap.	^	.04	, ,	une	Ju	ly	A	ug	Se	-pt	0	Oct	N	.	D	ec.
No		ī	1.	1	15	ī	15	1	15	1	15	ī	15	1	15	1	15	1	15	1	15	1	15	T	15
1 .		0	(0	0	0	0	1	2	3	6	11	23	36	49	63	77	90	95	98	99	100	100	100	100
2.		0	(0	0	1	1	2	3	6	10	17	29	43	5 0	67	77	85	91	96	98	99	100	100	100
3		Ċ	(0	0	1	۱	2	3	6	13	23	37	51	ć	69	78	85	91	94	96	98	99	99	100
4		0	,	ì	•	2	3	4	7	12	18	27	38	48	51	62	6.9	7.5	83	9 0	94	97	98	99	100
	 	C		2	3	4	ć	8	13	21	2≎	37	4Ô	54	63	65	69	74	81	87	92	95	9 7	96	۶¢
£.		0	(C	0	۱	1	۱	2	6	16	25	39	46	53	60	67	74	81	88	95	99	99	100	100
2		0		1	2	3	4	6	3	13	25	40	45	56	٤2	67	72	76	80	85	91	97	98	99	99
ŧ		0		3	5	7	10	14	20	28	3.7	48	56	61	64	68	72	77	81	86	89	92	95	98	99
5		0	:	4	6	ò	11	17	23	30	37	43	49	54	58	62	66	70	74	78	82	86	90	94	97
10		С		2	4	6	e	10	15	21	29	38	47	53	57	61	65	70	76	83	88	91	94	96	98
11		0		3	5	7	ç	11	14	18	27	35	41	46	51	57	62	68	73	79	84	89	93	96	98
12		C		C	0	1	۱	2	3	5	9	15	27	38	50	62	74	84	91	95	97	96	99	99	100
13		0	£	C	1	1	2	3	5	7	12	19	33	48	57	65	74	82	88	93	96	98	99	100	100
1		C	٢	c	1	2	3	4	6	9	14	20	28	39	52	63	72	80	87	91	94	97	98	99	100
15.	 	0	ι	۱	2	3	4	6	8	11	15	22	31	40	49	59	69	78	85	91	94	96	98	99	100
10 .		0	,	2	3	4	6	8	10	14	18	25	34	45	56	64	72	79	84	89	92	95	97	98	99
17 .	 	0	7	2	3	4	5	6	8	11	15	20	28	41	54	65	74	82	87	92	94	96	97	98	99
18 .	 	0		2	2 4	6	8	10	13	19	26	34	42	50	58	63	68	74	79	84	89	93	95	97	99
19		0		3	6	9	12	16	21	26	31	37	43	50	57	64	71	77	81	85	88	91	93	95	97
20		0		3	5	7	10	13	16	19	23	27	34	44	54	63	72	80	85	89	91	93	95	96	98
21		0		é	10	12	10	19	23	26	29	33	39	47	58	68	75	80	63	86	88	90	92	95	97
22		0		é	9	13	17	21	27	33	38	44	49	55	61	67	71	75	78	81	84	86	90	94	97
23 .	 	0	1	5	7	10	14	18	23	27	31	35	39	- 45	53	60	67	74	80	84	86	88	90	93	95
24		Э	ł	ć	, 9	12	16	20	24	28	33	38	43	50	59	69	75	80	84	87	90	92	94	96	۶.
25		0		3	5	7	10	13	17	21	24	27	33	40	45	53	61	69	78	<u>`</u> 9	92	94	95	97	98
25		0	1		6	8	12	16	20	25	30	35	41	47	56	67	75	81	85	67	89	91	93	95	97
27		0		2	3	5	7	10	14	18	22	27	32	37	46	58	69	80	89	93	94	95	96	97	99
23	• •	J		3	5	7	9	12	15	18	21	25	29	36	45	56	68	77	83	88	91	93	95	9 7	99
29	• • •	0		2	3	4	5	7	۶	11	14	17	22	31	42	54	65	74	ъ3	89	92	95	97	98	99
30.		0		2	3	4	5	6	8	10	14	اد	26	34	45	56	66	76	82	86	90	93	95	97	99
1 ز.		0	3	C) "	2	3	4	5	7	12	17	24	33	42	55	67	76	83	89	92	94	96	98	99
32		Э	Ι	2	3	4	5	6	8	10	13	17	22	31	42	52	60	68	75	80	85	89	92	96	98
33		0	1	2	2 4	6	3	11	:3	15	18	י 2	25	32	38	46	55	64	71	77	81	55	89	23	97

TABLE 6 -- Percentage of the average chnual El which normally occurs between January 1 and the indicated dates.¹ Computed for the geographic areas shown in figure & Velaw

For rates not listed in the table, interpolate between adjust traduc-



Soure: []

JUE	SW-	12

Type of mulch	Mulch Rate	Land Slope	Factor C	Length lumit ²
	Tons per acre	Perceni		Feet
None	0	oll	1.0	
Straw or hay	10	15	0 20	200
tied down by	10	6-10	20	100
anchoring and				
tocking	1.5	1.5	12	300
equipment	15	6-10	12	150
Do	20	15	06	400
	20	6 10	06	200
	20	11 15	07	150
	20	16 20	11	100
	20	21 25	14	75
	20	26 33	17	50
	20	34 50	20	35
Crushed stone	135	<16	05	200
'1 to 1 · m	135	16 20	05	150
	135	21 33	05	100
	135	34 50	05	75
Do	240	<21	02	300
	240	21 33	02	200
	240	34 50	02	150
Wood chips	7	- 16	08	75
	7	16 20	08	50
Do	12	16	05	150
	12	16 20	05	100
	12	21 33	05	75
De	25	- 16	02	200
	25	16 20	02	150
	25	21 33	02	100
	25	34 50	02	75

TABLE 9 ---- Mulch factors and length limits for construction slopes'

From Meyer and Forts (24) Developed by an intriagency workby arous on the lasss of field experience and limited research d."a

Assuming Jope length for which the specified mulch rate is considered effective. When this light is exceeded either a nigher opplication rate or mechanical shortening of the effective slope length is required

When the straw or hay mulch is not anthored to the soil C sul in on must rate or storp clupes of soils having K values greater that 0.30 should be taken at double the values given in this table

Table Sw-13

Vigetative con	Cover that contacts the soil surface								
Type and	Percent			Pe	rcent	ground	COVER		
height	cover	Type'	0	20	40	60	80	95+	
No appreciable		G	0 45	0 20	0.10	0 0 4 2	0 013	0 003	
canopy		w	45	24	15	.091	.043	011	
Tall weeds or	25	G	36	17	09	038	.013	003	
shart brush with aver age		w	36	20	13	083	041	011	
drop fall heigh	1 50	G	26	13	07	035	012	.003	
of 20 in		w	26	16	11	076	039	011	
	75	G	17	10	06	032	011	003	
		w	17	12	09	.068	038	011	
Appreciable brust	25	G	40	18	09	040	013	003	
or bushes with average drop fo	1	w	40	22	14	087	042	011	
height of 612 f	r 50	G	34	.16	08	.038	.012	003	
		w	34	19	13	082	.041	011	
	75	G	28	14	08	036	012	003	
		w	28	17	12	078	040	011	
Trees, hut no	25	G	42	19	10	041	013	003	
appreciable low brush Average	,	w	42	23	14	089	042	011	
drap fall neigh	1 50	G	39	18	09	040	013	003	
of 13 ft		w	39	21	14	087	042	011	
	75	G	36	17	09	039	012	003	
		٣	36	20	13	084	041	011	

TABLE 10 ---- Factor C for permanent pasture, range, and idle land

' The listed C values assume that the vegetation and mulch are randomly distributed over the entire area

Canopy height is measured as the average fall height of water drop, falling from the canopy to the ground Canopy effect is inversely proportional to drop fall height and is negligible if fall height exceeds 33 It

Portion of total area surface that would be hidden from view by comply in a striked projection (a bird's eye view)

- G color at surface is grass grasslike plants decaying com period dult or litter at least 2 in deep
- W cover at surface is mostly broadleaf herboreous plants (as words with little lateral root network near the surface) or undicay, d residues or both

Woodland Areas

Three categories of woodland can be can be considered separately: (1) undisturbed forest land; (2) woodland that is grazed, burned, or selectively harvested; and (3) forest lands that have site specific preparation treatments for re-establishment after harvest.

Factor C values for undisturbed forest land may be obtained from table SW-14. Factor C-values for mechanically prepared woodland sites can be obtained from table SW-15.

. . .

TABLE - Fuctor C for undisturbed forest land

Perient of area covered by caropy of trees and undergrawth	Percent of area covered by cluff at least 2 in deep	Fortor C
100-75	100-90	0001 001
70-45	85-75	007 004
40-20	70-40	003 009

"Where effective litter cover is less than 40 percent or compy cover is less than 20 percent, use table 6 Also use table 6 where woodlands are bring grazed, harvested, or burned

The ranges in listed C values are caused by the ranges in the specified forest liter and canopy covers and by variations in effective canopy heighs.

Table SW-15

TABLE 12.—Fa tor C for mechanically prepared woodland sites

		Soil condition and weed cover'										
Site	Mulch cover ¹	Exce	llent	Ga	od	Fe	pir i	Po	01			
		NC	VYC	NC	wc	NC	wc	NC	wc			
	Percint											
Disked, raked,												
or beaded'	None) 52	U 20	072	0 27	0 8 5	0 32	C 94	0 26			
	10	33	15	46	20	54	24	60	26			
	20	24	12	34	17	40	20	44	22			
	40	17	11	23	14	27	17	30	15			
	60	11	08	15	11	18	14	20	15			
	80	65	04	07	.06	.09	08	10	09			
Burned	None	.25	10	26	10	31	12	45	17			
	10	23	10	24	10	26	11	36	16			
	20	19	10	19	10	21	11	27	14			
	40	14	09	14	09	15	09	17	11			
	60	08	06	09	07	10	08	11	08			
	80	04	04	05	04	05	04	06	05			
Druin chopped	None	15	07	17	07	20	.08	29	11			
	10	15	07	16	07	17	08	23	10			
	20	12	06	12	0ć	14	07	18	00			
	40	09	06	.09	05	10	06	11	07			
	60	00	05	06	05	07	05	07	05			
	BC	03	03	03	03	03	03	04	01			

'Percentage of surface covered by residue in contact with the soil

Excellent soil condition—Highly stable soil aggregates in topsoil with fine tree racts and litter mixed in

Good—Moderately stable soil aggregates in topsail or highly stable aggregates in subsail (topsail removed during raking), only traces of little mixed in

Fair—Highly unstable soil aggregates in topsoil or moderately stable aggregates in subsoil, no litter mixed in

Paar—Na topsoil, highly eradible soil aggregates in subsoil, no litter mixed in

"NC-No live vegetation

WC-75 percent cover of grass and weeds having an average drop fall height of 20 in For intermediate percent ages of cover, interpolate between columns

"Modify the listed C values as follows to account for effects of surface roughness and aging

First year after treatment multiply listed C values by 0.40 for rough surface (depressions >6 in), by 0.65 for moderately rough and by 0.90 for smooth (depressions <2 in)

For 1 to 4 years after treatment multiply listed factors by 0.7 For 4 - 1c 8 years use table 6

More than 8 years use table 7

For first 3 years use C values as listed

For 3-; to 8 years after treatment use table 6

More than 8 years after treatment use table 7

2.2.6 <u>Support Practice Factor</u> (P)

The P factor is the ratio of soil loss within a support practice like contouring, stripcropping, or terracing to that with straight-row farming up and down the slope.

The P factor is related to the C factor and to practices that can slow the runoff water. The most important of these supporting cropland practices are contour tillage, stripcropping on the contour and terrace systems.

Current recommendation for contouring are presented in table SW-16. Effects of contour stripcropping are shown in table SW-17. Terracing effects are presented in table SW-18.

Table Sw-16

TABLE T-P	values	and	slope-length	limits	1.31

	Confeiring	
Long sloph Pricent	P voive	Moximum length
to 2 to 5 to 8 to 12 to 12 to 16 17 to 20 21 to 25	0 60 50 50 60 70 80 80	Fer + 400 200 200 120 80 60
		* 2

Limit may be increased by 25 percent if residue cover at 21 crop seedlings will i gularly exceed 50 percent

Holle Sw-17

TABLE P values maximum strip widths, and slopelength limits for contour stripcropping

Lund slope			P	values		5		
percent		A	B	c	arrip widin-	Maximum length		
						Feet	Feel	
1	10	2	0 30	0 45	0 ó 0	130	800	
3	to	5	25	38	50	100	600	
6	to	8	25	38	50	100	400	
9	10	12	30	45	60	80	240	
13	to	16	35	52	70	80	160	
17	to	20	40	60	80	60	120	
21	10	25	45	68	90	50	100	

P values

A For 4 star rotation of row crop small grain with meadow seeding: and 2 years of meadow A second row crop can re place the small grain if meadow is established in it

B for 4 year rotation of 2 years row crop winter grain with microlow sciencing and 1 year meadow

C For obternate stript of row crop and small grain

Adjust strip width limit generally dawnward to accommodate widths of farm eavipment

Table Su-18

TABLE Jo --- P values for contour-farmed terraced fields'

Land slope (percent)					Computing sediment yield ³					
		nt)	Farm	lonning	Graded channels	Steep backslope				
			factor	factor	sod outlets	underground outlets				
1	to	2	0 60	0 30	0 12	0 05				
3	to	8	50	25	10	05				
9	to	12	60	30	12	05				
13	to	16	70	35	14	05				
17	to	20	03	40	16	06				
21	10	25	90	45	18	06				

Slope length is the horizontal terroce interval. The listed values are for contain furning. No additional contairing factor is used in the computation

Use these values for control of interferrace erosion within specified soil loss toi rances

These values include entropment efficiency and are used for control of $c^{\frac{1}{2}}$ with sediment within limits and for estimating the field's contribution to watershed sediment yield

2.2.7 <u>Sediment Delivery Factor</u> (D)

The D factor is the ratio of the watershed sediment yield (SY) versus the upland erosion potential (A); D - SY/A.

Many factors and processes contribute to its estimation, such as redeposition of the particulates in the surface water runoff, storage, trapping of the sediment by vegetation and its residues, local scouring and redeposition in rills and channels, and possibly other yet unidentified. [Novotny, V., 1980.]

For D, the following formula has been proposed in the literature [Williams, 1975]:

$$-bt(d_{50})^{1/2}$$

D = e (SW-7)

where:

D = sediment delivery factor

b = decay constant (or routine coefficient)

t = travel time between two sections of a channel

 d_{50} = mean particle diameter of sediment.

A graphical presentation of sediment delivery ratio as a function of the watershed size is shown in figure SW-5 [Roehl, 1972]. The statistical relationships relying on the morphological characteristics of the watershed have limited applicability for estimating long term (annual or more) deliveries. Furthermore, their reliability for "event" deliveries is almost nil as demonstrated by Berkowitz. [Berkowitz, 1979.] Another relation of D to the storm characteristics (EI) and the runoff volume (V) is shown in figure SW-6 [Berkowitz, 1979]. An interesting discussion regarding D is presented by Novotny. [Novotny, 1980.]



sw-34

2.3 Subroutine SEDIMA

2.3.1 General

Subroutine SEDIMA (Sediment Annual) estimates the annual sediment washload of the soil compartment due to rain, based upon the USLE-theory previously outlined. It is "important" to keep in mind that all the sediment of the compartment will reach an adjacent receiver that has to be at a distance not exceeding 330 meters (1000 ft), as mandate by the assumption of the USLE. If the receiver happens to be at a greater distance, then it might be assumed that only the 330 m area will contribute sediment.

2.3.2 <u>Input/Output Parameters</u>

Input parameters and their associated units (metric system) for subroutine SEDIMA are:

$R_{m} [cm/hr] = R \times 10^{-2}; fi$	gure SW-1 for R index
K _m [t/ha/EI unit] = 1.292	K; figure SW-3 for K
$(LS)_{m} = LS$; figure SW-4 for LS
$C_m = C$; see section 2.2.5 for C
$P_m = P$; see section 2.2.6 for P
$D_{m} = D$; see section 2.2.7 for D.

Output from SEDIMA is:

SYA [tons/ha]

2.3.3 Parameter Units

Metric equivalents were not included in the general procedures and tables presented in the original USLE documentation [Wischmeier and Smith, 1978]. Metric untils can then be selected so that each factor will have a counterpart whose values will be expressed in numbers that are easy to handle and to combine in computations.

It is recommended, however, by the USLE designers rather to converting into metric individual empirically derived parameters (especially R), to converting into metric the USLE as a whole. The overall converting formula (1 t/ha = 2.242 tons per acre) for the metric (m) system is:

SYA_m = 0.446 SYA (SW-8)

$$A_m = t/ha/yr = 0.1 \text{ kg/m}^2/yr$$

A = tons/acre as estimated by the USLE.

-

2.4 Numerical Example

Annual soil loss from a particular field area is estimated by LEVELO and LEVELI SESOIL operations by inputing the values of R, K, LS, C, P and D. These variables describe both the average climate of the area and the agricultural or field conditions, and can be obtained from the tables of the previous sections, as demonstrated by the following numerical example [see also Wischmeier and Smith, p.40].

Assume, for example, a field on Russell silt loam soil in the Topeka area, Kansas. The dominant slope is assumed 8% with a length of 200 ft. Fertility and crop management on this field are such that crop yields are rarely less than 85 bu corn, 40 bu wheat, or 4 t alfalfa-brome hay. The probability of meadow failure is slight.

The USLE equation factors are obtained as follows:

- (1) Factor R is taken from the isoerodent map of figure SW-1. Topeka, Kansas, in north-east Kansas, lies between isoerodents 200 and 250. By linear (graphical) interpolation R = 205.
- (2) Factor K can be obtained: (a) from a table (SW-1, SW-2) of K values derived either by direct research measurements, or (b) by use of the soil erodibility nomograph (figure SW-3). For the Russell silt loom soil, K=0.37 (figure SW-3).
- (3) Factor LS is obtained from figure SW-4, the slope-effect chart, where an 8% slope along a 200 ft distance gives LS = 1.41.

If the field was continuously in clean tilled fallow and the delivery factor was assumed D = 1, the average annual soil loss from the dominant slope would equal (equation SW-8):

$$SYA = 0.446 (205)(0.37)(1.41) = 47.69 t/ha/yr$$

However, for the present agricultural area we need to account for the effects of the cropping and management system and support practices existing in the field, this effect being represented by factors C, P and D as follows:

- (4) Factor C for the field may be: (a) either derived by the procedures described in the original USLE theory [Wischmeier and Smith, p. 28] using data of tables 5 and 6; or (b) obtained from centrally prepared C value tables available from the SCS (Soil Classification System). Let us assume for the present example that C = 0.085 [Wischmeier and Smith, p. 40].
- (5) Factor P = 1.0, because rows and tillage are in the direction of the land slope. However, if farming were on the contour, the average P value would have been P = 0.5 (see section 2.2.6).
- (6) Finally the sediment delivery factor can be assumedD = 80% (see section 2.2.7) without having any particular justification for its value in this illustrative example.

Thus, total annual sediment yield of the field is estimated to (equation SW-8) •

SYA = 0.446 (205)(0.37)(1.41)(0.085)(1.0)(0.80) = 3.24 t/ha/yr

The USLE may also be used to compute the average soil loss for each crop in the rotation or for a particular cropstage period, during annual simulations. For such additional information regarding C, and P values, the reader is referred to the original work. [Wischmeier and Smith, p. 41.]

2.5 Discussion

The USLE is designed to predict longtime-average soil losses for specified conditions. Best predictions are "averaged-annual" losses from small watersheds, because the USLE factors are more difficult to evaluate for large mixed watersheds. Under "small" watersheds it is meant watersheds with adjacent receivers at a distance not exceeding 330 m (1000 ft), as shown in figure SW-7. For larger watershed simulations it can be assumed that only one portion of the watershed delivers sediment to the receiver, because upland erosion will be deposited within the watershed, thus, not contributing to the sediment yield of the basin. For large watersheds, factor D (delivery factor) becomes of paramount importance, and USLE predictions should be calibrated or validated with field data.

The USLE does not consider the basic processes of soil detachment, transport and deposition separately and does not account for various basin forms as schematically shown in figure SW-7. Therefore, above equation is employed only for LEVELO and LEVEL1 (annual simulation) of SESOIL. For more site specific, accurate and monthly sediment simulations, the user has to employ LEVEL2 of SESOIL, whose sediment routine is described in the following section.

sw-39



sw-40

3.0 "MONTHLY" WASHLOAD SIMULATIONS

3.1 OBJECTIVE

The objective of this section is to select and document the most appropriate sediment transport routine to meet the following criteria; the routine should: (1) represent the state-of-the-art, (2) be physically based and not require calibration, (3) be driven by a limited number of input parameters, (4) simulate sediment detachment, transport and deposition, (5) account for various basin shapes, (6) be applicable to an entire watershed and to discrete portions of the watershed as well, and (7) account for both long-term (monthly) and short-term (peaks within a month) simulations.

3.2 BACKGROUND/ACKNOWLEDGMENTS

Numerous factors and processes have been reported in the literature as providing statistically significant correlations to the attenuation of sediment and particulate pollutants from nonpoint sources. These factors include: (1) the effect of rainfall energy, that detaches the soil particles from small rills and keeps them in movement as long as the overland flow persists, (3) the effect of vegetation, that slows down the flow and filters out the particles during shallow flow conditions, (4) infiltration, which filters out the particles from the overland flow, (5) small depressions and surface roughnesses in which particles can settle due to reduction of velocity, and (6) change of slope of the overland flow. [Novotny, V.; 1980.]

The number of available sediment transport formulas in the literature is extremely large. Some of these formulas have not received extensive application, others are too complicated or require knowledge of the concentration of the suspended load and, therefore, have not been suitable for hydrologic simulations. A comprehensive analysis and evaluation of sediment transport theories have been conducted by Alonso [Alonso, C.V.; 1980] with reference to flume and silt data (Table SW-19).

Theories examined range from simplified formulas to sophisticated modeling packages accounting for the micromechanics of sediment movement. Following the testing of the formulas presented in Table SW-19, Foster, G.R. <u>et al</u> [1980] developed a model to estimate sediment yield from field-size areas. The model summarizes the state-of-the-art in erosion and sediment yield modeling with appropriate simplifications required to couple the governing equations.

SESOIL employs the sediment yield model as developed by Foster <u>et al</u>, however, adapted to the statistical needs of SESOIL for monthly yield and maximum yield of individual rainstorms within a month. Interested readers are advised to the original work of Foster. [Foster, G.R. <u>et al</u>; 1980, Knisel, W.G.; 1980], since the following sections have been mainly

TABLE SW-19. SUMMARY OF AVAILABLE SEDIMENT TRANSPORT FORMULAS

туре	Predicted load	Author(s) and reference	Date
Deterministic	6ed	DuBoys, (<u>11</u>)	1879
Deterministic	8ed	Schoklitsh, (26)	1934
Deterministic	8ed	Meyer-Peter, (22)	1934
Deterministic	Bed	Straub, (29)	1935
•		Waterways Experiment	
Deterministic	Bed	Station, (31)	1935
Deterministic	કન્વ	Shields, (<u>28</u>)	1936
Deterministic	Bed	Schoklitsh, (<u>27</u>)	1943
Deterministic	Bed	kalinske, (19)	1947
Empirical	Total	inglis, (<u>18</u>)	1947
Deterministic	Bed	Meyer-Peter and Muller, (23)	1948
Stochastic	Bed	Einstein, (13)	1950
Stochastic	Total	Einstein, $(\overline{13})$	1 9 50
Stochastic	Bed	Einstein and Brown, (7)	1950
Stochastic	Total	Colby and Hembree, (10)	1955
Deterministic	Bed	Bagnold, (2)	1956
Deterministic	Total	Egiazaroff, (12)	1957
Deterministic	Total	Bogardi, (6)	1958
Deterministic	Total	Laursen, (21)	1958
Deterministic	Bed	Rottner, (25)	1959
Deterministic	Bed	Yalin, (37)	1963
Empirical	Total	Blench, (5)	1964
Stochastic	Total	Colby, (9)	1964
		Bishop, Simons and	
Stochastic	Total	Richardson, (<u>4</u>)	1965
Deterministic	Total	Bagnold, (<u>3</u>)	1966
Stochast ic	Total	Wilson, (<u>36</u>)	1966
		Chang, Simons and	
Deterministic	Total	Richardson, $(\underline{8})$	1967
Deterministic	Total	Engelund and Hansen, $(\underline{14})$	1967
Deterministic	Fot al	Graf.(25)	1968
Deterministic	Total	Toffaleti, (36)	1968
Deterministic	Total	Ackers and White, (1)	1973
Deterministic	Total	Yang, (<u>39</u>)	1973

Source: [Alonso, C.V.; 1980]

.

"abstracted" from his work. The author of this section gratefully acknowledges the assistance received by Professor Foster (Agricultural Engineering Department, Purdue University, West Lafayette, Indiana 47907) while working out the adaptation for SESOIL of his sediment yield theory.

3.3 OVERVIEW OF THE MONTHLY WASHLOAD MODEL

Erosion of soil particulates and their transport can be broken down into four processes [Foster and Meyer; 1972]: (1) detachment by rainfall, (2) detachment by overland flow, (3) transport by rainfall. and (4) transport by overland flow. On a given field, either detachment or sediment transport capacity may limit sediment yield depending on topography, soil characteristics, cover, and rainfall/runoff rate and amounts.

Control of sediment yield by detachment or transport can change from season to season, from storm to storm, and even within a storm. The relationship for detachment is different from the one for transport so that they cannot be lumped together into a single equation. Furthermore, the interrelation between detachment and transport is nonlinear and interactive for each storm, or each storm category, which prevents using separate equations to linearly accumulate the amount of detached sediment transport capacity over several storms. Therefore, to simulate erosion and sediment yield and to satisfy the need for a continuous simulation model, a rather fundamental approach was selected by Foster <u>et al</u> [1980] where separate equations are used for soil detachment and sediment transport.

Every model is a representation and a simplification of a real environmental situation. Various techniques, including plains and channels, square grids, converging sections, and stream cubes have been used to represent subsections of an area. [Foster <u>et al</u>; 1980.] Most erosion/ sediment yield models have adequate degrees of freedom to fit observed data. Some models, depending on their representation scheme, distort parameter values more than others do. Distortion of parameter values greatly reduces the transfer ability of parameter values from one area to another. An objective, therefore, in Foster's model, was the development of a theory representing the field in a way minimizing parameter distortion. In addition, a minimum number of input parameters have to be compiled by the user, the simulation being performed with the aid of theoretically derived equations rather than the employment of massive input data sets and calibration coefficients to account for the processes previously described.

Regarding the possible shape of elements and the calling sequence used to represent field-size areas, Foster <u>et</u> <u>al</u> distinguish between overland flow, channel flow, and impoundment (pond) elements as shown in Figure SW-8.

The model user selects the best combination of elements and enters the appropriate sequence number according to Table SW-20. Computation

starts in the uppermost element, which is <u>always</u> an overland flow element and proceeds downslope. Sediment concentration (for each particle type) is the output from each element which becomes the input to the next element in the sequence.



OVERLAND FLOW SLOPE REPRESENTATION

ALOIN SLUPE --

. (0,7)

Source: [Foster, G.R. et al; 1980]

OVERLAND FLOW

FIGURE SW-8. SCHEMATIC REPRESENTATION OF TYPICAL FIELD SYSTEMS IN THE FIELD-SCALE EROSION/SEDIMENT YIELD MODEL

.

TABLE SW-20. POSSIBLE ELEMENTS AND THEIR CALLING SEQUENCE USED TO REPRESENT FIELD-SIZED AREA

Sequence number	Elements and their sequence
1	Overland
2	Overland-Pond
3	Overland-Channel
4	Overland-Channel-Channel
5	Overland-Channel-Pond
6	Overland-Channel-Channel-Pond

Source: [Foster, G.R. <u>et al;</u> 1980]

3.4 MODEL MATHEMATICS

3.4.1 Basic Concepts and Equations

Sediment load is a function of the sediment quantity available: (1) after detachment by precipitation energy, and (2) by the transport capacity of overland flow. Quasi-steady state conditions can be assumed, so that a single rainfall and runoff rate characteristics of each storm (or storm series) can be used in computational procedures. [Foster and Meier; 1975.] The major sequence of computation is shown in Figure SW-9.

<u>Sediment transport</u> downslope of an area can be described with the steady state equation of sediment mass continuity [Foster, G.R. et al; 1980]:

$$lq_{s}/dx = D_{L} + D_{F}$$
 (SW-10)

where:

q_s = sediment load per unit width per unit time
x = distance (location)
D_L = lateral sediment inflow (mass/unit area/unit time)
D_F = sediment detachment or deposition by overland flow
 (mass/unit area/unit time)

Lateral sediment inflow to a watershed segment may originate from interill erosion on overland flow elements, or from overland flow (or a channel, if two channel segments are in a sequence) for the channel elements. Flow in rills on overland flow areas or in channels, transports the sediment load downslope. Lateral sediment inflow can be independently assumed of whether the flow is detaching or depositing.

During simulations of a watershed segment (overland flow element or in a channel), the initial "potential sediment load" is estimated. This load equals the sum of the sediment load from the: (1) immediate upslope segment and (2) the lateral inflow. If:

- (1) the initial potential sediment load is less than the "transport capacity" of the overland flow, detachment takes place at a rate: "equal or less" the detachment capacity of the flow. When this detachment takes place, it adds particles having the particle size distribution for detached sediments. No sorting is assumed to take place during detachment;
- (2) the initial potential sediment load is greater than the transport capacity of the overland flow, deposition is assumed to take place.



Source: Foster, G.R. et al; 1980]

.

FIGURE SW-9. FLOW CHART FOR DETACHMENT-TRANSPORT-DEPOSITION COMPUTATIONS WITHIN A SEGMENT OF OVERLAND FLOW OR CONCENTRATED FLOW ELEMENTS Sediment deposition within a segment is described in the simulation by:

$$D = \alpha(T_c - q_s)$$
, and
 $\alpha = \epsilon V_s/q_w$ (SW-11)

where:

D = sediment deposition rate (mass/unit area/unit time) α = first order reaction coefficient (length⁻¹) T_C= transport capacity (mass/unit width/unit time) ε = 0.5 for overland flow, and 1.0 for channel flow V_S= soil particle fall velocity qw=q_L . x ; discharge rate of runoff per unit width (volume/width/time).

<u>Sediment detachment and deposition</u> by flow is simulated in a segment in four independent cases (processes):

- (1) deposition over the entire segment
- (2) detachment in the upper boundary and deposition in the lower boundary of the segment
- (3) deposition on the upper end and detachment in the lower end of the segment
- (4) detachment all along the segment.

Above cases are described mathematically [Foster, G.R. et al; 1981] as follows:

Case (1): takes place when:

 $T_c < q_s$ all along the segment (SW-12)

where:

$$D = [\phi/(1+\phi)] (dT_c/dx-D_L)[1-(x_u/x)^{1+\phi}]+D_u(x_u/x)^{1+\phi}$$

$$\phi = \epsilon V_s/q_L$$

$$dT_c/dx = \text{constant over segment} \qquad (SW-13)$$

$$D_u = \alpha(T_{cu} - q_{su})$$

$$q_s = T_c - (D/\alpha)$$
in which:

 $T_{c} = transport capacity$ $q_{s} = sediment load at distance x$ D = deposition rate $\phi = depositing coefficient$ $V_{s} = soil particle fall velocity$ $q_{L} = discharge rate$ $D_{L} = lateral sediment inflow$ $x_{u} = distance (location)$ $D_{u} = deposition rate at x_{u}$ $T_{cu} = transport capacity at x_{u}$ $q_{su} = sediment load at x_{u}$ $\alpha = first order reaction coefficient$

Case (2) takes place when:

 $T_c < q_s$ within the segment $T_{cu} > q_{su}$ and (SW-14) $dT_c/dx < 0$

If dT /dx < 0 for a segment where $T_{cu} > q_{su}$, T_c may decrease below q_s within the segment. The point location where $q_s = T_c$ is determined (defined) as x_{db} (x_u in equation SW-13), with $D_u = 0$. Deposition and sediment load are estimated from equations SW-13.1, SW-13.2 and SW-13.5 (equations group SW-13).

<u>Case (3)</u> takes place when: $T_c > q_s$ within the segment $T_{cu} < q_{su}$ and (SW-14) $dT_c/dx > 0$ At a point (location) x_{de} "deposition" may end. In this case $D_u = 0$, $T_c = q_s$. Downslope, detachment and "sediment load" take place.

Deposition ends at:

$$x_{de} = x_u \left\{ 1 - \left[(1+\phi)/\phi \right] \left[D_u/(dT_c/2x - D_L) \right] \right\}^{1/(1+\phi)}$$
 (SW-15)

Sediment load is given by:

$$q_{s} = (D_{Fu} + D_{Lu} + D_{FL} + D_{LL}) \Delta x/2 + q_{su}$$
 (SW-16)

where:

^x de = lo	cation where deposition ends
x _u = di	stance
φ = de	position coefficient
D _u = de	position rate at x_u ; see equation (SW-13.4)
T _c = tr	ansport capacity
D _L = 1a	ateral sediment inflow
q _s = se	diment load
u,L ≃ se	gment subscripts; u = upper, L = lower
$D_F = D_F$	r_u , D_{FL} ; sediment detachment or deposition
$D_L = D_L$	$_u$, D_{LL} ; lateral sediment inflow
∆x = le	ength of segment where detachment occurs
q _{su} = se	diment load from upper segment

in which:

 Δx is from x_{de} to the lower end of segment

 q_{su} is at x_{de} , which is T_c at x_{de} ; ie $q_{su}(x_{de}) = T_c$

Case (4) takes place when:

 $T_{C} > q_{S}$ over the entire segment. Sediment load is estimated with equation SW-16.

SW-51

3.4.2 Modeling Issues

Eroded sediment is a mixture of particles having various sizes and densities. Simulations are performed for each particle type.

Equations describing: (a) sediment characteristics, (b) flow detachment capacity, (c) rainfall erosivity, (d) effects of overland slope, (e) sediment transport capacity, and (f) other parameters are presented [Foster, G.R.; 1980] in the following sections for the:

- (1) overland flow element
- (2) channel element
- (3) impoundment element

The following four sections describe sedimentation characteristic issues, and issues relating to the modeling of the three elements; overland, channel and impoundment.

3.4.3 Sediment Characteristics

Eroded sediment is a mixture of primary particles and aggregates of various sizes. Size distribution is either an input to the model, or can be estimated by the model analytically if distribution is not given. In the latter case the model assumes 5 particle size distributors derived from surveys of existing data as described in the following paragraph.

Typical sediment characteristics assumed for detached sediment before disposition, typical for midwestern silt loam soil are presented in Table SW-21. Equations employed to estimate particle size distributions are presented in Table SW-22. Particle sizes assumed to derive the equations for the particle size distributions are presented in Table SW-23. Primary particle composition of the sediment load is estimated for small and large aggregates with the equations presented in Table SW-24.

TABLE SW-21. SEDIMENT CHARACTERISTICS ASSUMED FOR DETACHED SEDIMENT BEFORE DEPOSITION; ASSUMED TYPICAL OF MANY MIDWESTERN SILT LOAM SOILS

Particle Type	Diameter	Specific Gravity	Fraction of Total Amount (mass basis)
	(mm)	(g/cm^3)	
Primary clay	2ر .0	2.60	0.05
Primary silt	.010	2.65	.08
Small aggregate	.030	1.80	.50
Large aggregate	.500	1.60	.31
Primary sand	. 200	2.65	.06

Source: [Foster, <u>et al</u>.; 1980]

TABLE SW-22. EQUATIONS EMPLOYED TO DESCRIBE PARTICLE SIZE DISTRIBUTION

PSA = $(1.0 - ORCL)^{2.49}$ ORSA PSI = 0.13 ORSI PCL = 0.2 ORCL (SW-14) SAG = $\begin{cases} 2 \text{ ORCL} & ORCL < 0.25 \\ 0.28(ORCL - 0.25) + 0.5 & 0.25 \le ORCL \le 0.50 \\ 0.57 & 0.5 < ORCL \end{cases}$

LAG = 1.0 - PSA - PSI - PCL - SAG

where:

ORCL, ORSI, ORSA: Fractions for primary clay, silt, sand in the original soil mass
PCL, PSI, PSA, SAG, LAG: Fractions for primary clay, silt, sand, small and large aggregates in the detached sediment
LAG ≥ 0.0: In case LAG < 0.0 multiply all other parameters with a coefficient to make LAG = 0.0

Source: [Foster, et al.; 1980]

TABLE SW-23. ASSUMED TYPICAL DIAMETERS OF PARTICLE SIZES

DPCL = 0.002 mm DPSI = 0.010 mm DPSA = 0.20 mm $DSAG = \begin{cases} 0.03 \text{ mm} & ORCL < 0.25 \\ 0.20(ORCL - 0.25) + 0.03 \text{ mm} & 0.25 \le ORCL \le 0.60 \\ 0.1 \text{ mm} & 0.60 < ORCL \end{cases}$ DLAG = 2(ORCL) mm

where DPCL, DPSI, DPSA, DSAG, and DLAG are, respectively, the diameters of the primary clay, silt, and sand, and the small and large aggregates in sediment. The assumed specific gravities are shown in Table SW-21.

Source: [Foster, G.R. et al; 1980]

TABLE SW-24. EQUATIONS EMPLOYED FOR PARTICLE COMPOSITION OF SEDIMENT LOAD*

Small aggregates: CLSAG = SAG · ORCL/(ORCL + ORSI) SISAG = SAG · ORSI/(ORCL + ORSI) SASAG = 0.0 Large aggregates:** CLLAG = ORCL - PCL - CLSAG SILAG = ORSI - PSI - SIAG (SW-17) SALAG = ORSA - PSA

where CLSAG, SISAG, and SASAG = gractions of the total for the sediment of, respectively, primary clay, silt, and sand in the small aggregates in the sediment load, and CLLAG, SILAG, and SALAG are corresponding fractions for the large aggregates.

"The text of this table was "quoted" from Foster, G.R, et al. [1980].

**If the clay in the large aggregate expressed as a fraction for that particle alone is less than 0.5 times ORCL, the distribution of the particle types is recomputed so that this constraint can be met. A sum, SUMPRI, is computed whereby:

SUMPRI = PCL + PSI + PSA.

The fractions PSA, PSI, and PCL are not changed. The new SAG is:

SAG = (0.3 + 0.5 SUMPRI)(ORCL + ORSI)/[1 - 0.5 (ORCL + ORSI)].

Above equation is derived given previously determined values for PCL, PSI, and PSA; the sum of primary clay fractions for the total sediment equals the clay fraction in the original soil, and the assumption that the fraction of primary clay in LAG equals half of the primary clay in the original soil.

The model also computes an enrichment ratio using specific surface areas for organic matter, clay, silt, and sand. Organic matter is distributed among the particle types based on the proportion of primary clay in each type. Enrichment ratio is the ratio of the total specific surface area for the sediment to that for the original soil.

Although these relationships are approximations to the data found in the literature (Young, R.A.; 1978), they represent the general trends.

3.4.4 Overland Flow Element

Detachment on interrill and rill areas and transport and deposition by rill flow are the erosion-transport processes on the overland flow element. Detachment equations are presented in Table SW-25 (Equation 18.1 and 18.2). Mathematical expressions for the storm erosivity (EI) and slope length exponent (m) of above equations are also presented in Table SW-25 (Equation 18.3 through 18.4).

Sediment transport capacity is described by the Yalin equation [Yalin, S.; 1963]; however, it has been modified by Foster and Meyer [1972] to account for various particle sizes and types. All equations employed are presented in Table SW-26. Foster, G.R., et al. [1980], presented six computational steps to redistribute the transport capacity when excess and deficits of sediment occur.

Regarding the computational procedure the authors [Foster, <u>et al.</u>; 1980] established a sequential simulation starting with the upper end of a slope and routing sediment downslope, as in most discretized sediment models. Computations take place for each particle size type. Concentration multiplied by the runoff volume and overland flow area represented by the overland flow profile gives the sediment yield for the stone on the overland area of the field.

The overland flow is represented by a typical land profile selected from possible overland flow paths. Its shape may be uniform, convex, concave, or a combination of these shapes. Inputs are total slope length, average steepness, the slope at the upper end of the profile, the slope at the lower end of the profile and location of end points of a miduniform section.

Given the above information, the model establishes segments along the profile. The procedure is illustrated by the convex shape shown in Figure SW-10. Coordinates of points A, C, and D are given, as are slopes S_b and S_m . A quadratic curve will pass through point C tangent to the line CD and through point E tangent to line AB. The location of point E is the intersection of a line having a slope equal to the average of S_b and S_m with line AB. If x_2 is less than x_1 , x_3 is shifted downslope so that $x_1 = x_2$. [Foster, et <u>al</u>.; 1980].

$$\frac{\text{TABLE SW-25}}{\text{(Equation category SW-18; for notation see next page)}}$$

$$D_{\text{Li} = 4.57 \text{(EI)(s + 0.014) KCP } \sigma_{\text{p}}/V_{\text{u}} \text{(SW-18.1)}$$

$$D_{\text{Fr}} = (6.86 \times 10^{6}) \text{m} V_{\text{u}} \sigma_{\text{p}}^{1/3} (\text{x}/22.1)^{\text{m}-1} \text{s}^{2} \text{KCP}(\sigma_{\text{p}}/V_{\text{u}}) \text{(SW-18.2)}$$

where:

$$EI \begin{cases} = 0.103 V_R^{1.51} & \text{or} \\ = 0.0276 V_R I & \text{or} & (SW-18.3) \\ = e=11.9 + 8.73 (log_{10}i) & (if hyetograph is available) \end{cases}$$
$$m \begin{cases} = 1.0 + 3.912/\ln(x); x>50m & (SW-18.4) \\ = 2 & ; x \le 50m \end{cases}$$

where:

TABLE SW-25. OVERLAND FLOW ELEMENT EQUATIONS (Continued)

= soil loss ratio of the USLE cover-management factor С = USLE contouring factor^{3/} Ρ = peak runoff rate expressed as volume/area/time (m/s) σъ = runoff volume/area (m) V₁₁ = distance downslope (m) х = volume of rainfall $(mm)^{4/}$ VR = maximum 30-minute intensity (mm/h) 1 = rainfall energy per unit of rainfall $(J/m^2/mm \text{ of rain})^{5/2}$ e = rainfall intensity (mm/h) i

1/EI[English]*1.702 = EI[metric; N/h]

2/Units of K must be carefully noted. K[English]*131.7 = K[metric; gh/Nm²] 3/Only the contouring part of P is used. 4/Equation SW-18.3a was derived from 2700 data points (r²=0.56)

^{5/}EI≃e [Foster, et al.; 1980]

Source: [Foster, G.R. et al; 1980]

Single Particle Equation:

$$P_{s} = \frac{W_{s}}{(S_{g})g\rho_{w}dV_{*}} = 0.635 \ \delta[1 - \frac{1}{\sigma} \ln (1 + \sigma)] = P_{s}$$

where:

$$\sigma = \delta A$$

$$\delta = \frac{Y}{Y_{cr}} - 1 \text{ (when } Y < Y_{cr}, \ \delta = 0\text{)}$$

$$A = 2.45(S_g)^{-0.4}(Y_{cr})^{1/2}$$

$$Y = \frac{V^2 *}{(S_g - 1.0) \text{ gd}}$$

$$V_* = (gRS_f)^{1/2}$$

(SW-19)

Modified Multiparticle Supporting Equations:

•
$$T = \sum_{i=1}^{n_s} \delta_i$$

•
$$(N_e)_i = N_i (\delta_i/T)$$

•
$$(P_e)_i = P_i \delta_i / T$$

•
$$(W_{si}) = (P_e)_i (S_g)_i \rho_w g d_i V_*$$

where:

$$\tau_{soil} = \gamma ys (n_{bov}/n_{cov})^{0.9}$$
$$y = \left[q_w n_{bov}/s^{0.5}\right]^{0.6}$$

July 1981

(sw-20)

TABLE SW-26. DIMENSIONLESS SEDIMENT TRANSPORT CAPACITY EQUATION OF S. YALIN [1963] (Continued)

where:

Ps	=	nondimensional transport
W _s	=	transport capacity (mass/unit time/unit flow width)
Sg	=	particle specific gravity
8	=	acceleration due to gravity (g=9.81 m/sec ²)
ρw	=	mass density of fluid (water)
۷.	=	sheer velocity = $(\tau/\rho_w)^{1/2}$
τ	=	sheer stress
0.635	=	constant from Shield's diagram
σ,Α,δ	=	defined dimensionless expressions
Y	=	actual lift force given by Yalin
Ycr	=	critical lift force given by Shield's diagram as a
		function of the particle Reynolds number
d	=	particle diameter
R	=	hydraulic radius
Sf	=	slope of energy gradeline
i	=	sediment particle type
Т	=	total value of δ 's in the mixture
n _s	=	number of particle types in the mixture
(Ne) _i	=	number of transported particles of type i in a mixture
Ni	=	number of particles transported in sediment of uniform
		type i for a δ _i .
(Pe) _i	=	effective P for particle type i in a mixture
(Ps) _i	=	the P _s calculated for uniform material i
(W _s) _i	=	transport capacity of each particle type in a mixture
[⊤] soil	=	sheer stress acting on soil
γ	=	weight density of water
у	=	flow depth for bare smooth soil
n _{bov}	=	Manning's coefficient (n) for bare soil; =0.01 for
		overland flow; =0.03 for channel flow
ncor	=	total Manning's (n) for rough surfaces or soil covered
		by mulch or vegetation
٩w	=	discharge rate per unit width

July 1981

Arthur D Little, Inc



Source: [Foster, et al.; 1980]

FIGURE SW-10. SCHEMATIC REPRESENTATION OF CONVEX SLOPE PROFILE FOR OVERLAND FLOW

3.4.5 Channel Element

The channel element (Figure SW-8) is used to represent flow in terrace channels, diversions, major flow concentrations whose topography has caused overland flow to converge, grass waterways, row middles or graded rows, etc. This element does not describe gully or large channel erosion [Foster, G.R.; et al; 1980].

The spatially varied flow equation of the channel element is given in Table SW-27. Equation system (SW-21) is solved for a range of typical values C_1 , C_2 , C_3 for subcritical flow, and regression curves are derived for the components of the normalized friction slope of the channels (SSF). Curves are fitted to the solutions in order to reduce computation time.

The equation for: (1) the detachment capacity (D_{FC}) by flow over a loosely tilled seedbed; (2) the erosion rate in the channel (E_{Ch}) ; (3) the width of the channel (W) at any time after the channel has eroded to the nonerodible layer; (4) the final width of the channel (W_f) ; (5) the hydraulic radius due to soil (R_{SOII}) ; (6) the shear stress acting in the soil (τ_{SOII}) ; and (7) the shear stress acting on the soil cover (τ_{COV}) are given in Table SW-28 (equations SW-22 to SW-28).

TABLE SW-27

SPATIALLY VARIED FLOW EQUATION; CHANNEL ELEMENT

$$dy_{*}/dx_{*} = [S_{*} - C_{2} x_{*}^{2}/y_{*}^{16/3} - C_{3} x_{*}/y_{*}^{4}]/[1 - C_{3} x_{*}^{2}/y_{*}^{5}]$$

$$c_{1} = [z^{5/2}/2(z^{2}+1)^{1/2}]^{2/3}$$

$$c_{2} = [Q_{e} n L_{eff}/C_{1}y_{e}^{19/6}]^{2}$$

$$c_{3} = 2 \beta Q_{e}^{2}/g z^{2} y_{e}^{5}$$
(SW-21)

where:

 $y = y/y_e$ y = flow depth y_{p} = flow depth at the end of the channel $S_* = sL_{eff}/y_e$ s = channel slopex = distance along channel $x_* = x/L_{eff}$ L_{eff} = effective channel length (i.e., the length of the channel if it is extended upslope to where discharge would be zero with the given lateral inflow rate C_1 , C_2 , C_3 = constants m = Manning coefficient z = side slope of channel Q_c = discharge at end of channel β = energy coefficient (app. 1.56) g = 9.81 m/sec²; acceleration of gravity

TABLE SW-28

OTHER EQUATIONS DESCRIBING SEDIMENT TRANSPORT

$$D_{Fc} = K_{ch} (1.35 \ \bar{\tau} - \tau_{cr})^{1.05}$$
(SW-22)

$$E_{ch} = W_{ac} K_{ch} (1.35 \bar{\tau} - \tau_{cr})^{1.05}$$
(SW-23)

$$(dW/dt)_{i} = 2 k_{ch} (\tau_{b} - \tau_{cr})^{1.05/\rho}$$
 soil (SW-24)

$$W_{f} = [Qn/S_{f}^{1/2}]^{3/8} [(1 - 2 \times cf) / x_{cf}^{5/3}]$$
(SW-25)

$$R_{soil} = (Vn_{bch}/S_{f}^{1/2})^{3/2}$$
(SW-26)

$$\tau_{soil} = \gamma R_{soil} S_{f}$$
(SW-27)

$$\tau_{\rm cov} = \gamma \left[V(n_{\rm t} - n_{\rm bch}) \ S_{\rm f}^{1/2} \right]^{3/2}$$
(SW-28)

where:

Arthur D Little, Inc.

```
\tau_b = shear stress in a channel at a
nonerodible boundary
\rho_{soil} = man density of soil
W_f = final channel width
Q = discharge rate (m<sup>3</sup>/s)
n = Manning friction coefficient
S_f = Friction slope for flow hydraulics
in a channel
x_{cf} = Normalized distance around wetted perimeter
where \tau=\tau_{cr} at nonerodible boundary
R_{soil} = hydraulic radius due to soil
\tau_{cov} = shear stress at which the cover starts to move
n_t = total Manning coefficient
n_{bch} = Manning coefficient for a base channel.
```

ŧ.

3.4.6 Impoundment (Pond) Element

The impoundment or pond element (Figure SW-8) describes deposition behind impoundments (including parallel tile outlet terraces) that drain after each storm. The pond element (receiver) is the last element of an element series.

The equations for: (1) the sediment fraction (F_{pi}) deposed in an impoundment and (2) the runoff volume (V_{out}) out of an impoundment are presented in Table SW-29.

3.4.7 Discussion

Discussion regarding use of the model will be presented later.

TABLE SW-29

SUPPORTING EQUATIONS; IMPOUNDMENT ELEMENT

$$F_{pi} = A_{1} [exp(B_{1}d_{u}) - exp(B_{1}d_{1})]/(B \Delta d)$$
(SW-29)
$$V_{out} = 0.95 V_{in} exp(Z_{r})$$
(SW-39)

where:

 $A_1 = 1.136 \exp (Z_S)$ $B_1 = -0.152 \exp(Y_S)$ $Z_s = Z_s (f_a, C_{or}, V_{ro}, I_p)$ $Y_s = Y_s (f_a, C_{or}, V_{ro}, I_p)$ $C_{or} = 0.15 d_{or}^2 = (7.02 \times 10^4) Q_p / y_d^{1/2}$ $Z_r = Z_r (f_a, C_{or}, V_{ro}, I_p)$ in which: F_{pi} = fraction passed for partiçle i $A_1, B_1 = coefficients$ $d_u = equivalent$ sand diameter of upper end of a sediment particle class; (mm) d, = equivalent sand diameter of lower end of a sediment particle class: (mm) Δd = width of a particle class; (mm) V = volume of runoff discharged (out of impoundment); (m³/storm) V_{in} = runoff volume into impoundment; (m³/storm) Z_r = exponent in equation for runoff reduction by an impoundment; (-) $Z_s = exponent in equation for deposition in an impoundment; (-)$ Y_S = exponent in deposition equation of an impoundment; (-) f_a = coefficient in surface area-depth relationship for impoundment V_{ro} = runoff/volume (m³/storms)

(SW-39)

- 3.5 Subroutine SEDIMM
 - 3.5.1 General
 - 3.5.2 Input/Output Parameters
 - 3.5.3 Data Files
 - 3.5.4 Discussion

3.6 Sensitivity Analysis of SEDIMM

3.6.1 General

.

- 3.6.2 Hydrologic Parameters
- 3.6.3 Sediment Yield Parameters
- 3.6.4 Discussion
- 3.7 Conclusions

Sections 3.5 and 3.6 will be presented in the draft report of this contract. Section 3.7 will be presented at a later time.

.

4.0 REFERENCES

Adams, R.T.; Jurisu, F.M. Simulation of pesticide movement on small agricultural watersheds. Report No. EPA-600/3-76-066. Athens, GA: U.S. Environmental Protection Agency Research Laboratory; 1976. Available from: NTIS, Springfield, VA; PB-259933.

Beasley, D.B.; Monke, E.J.; Huggins, L.F. The ANSWERS model: A planning tool for watershed research. Paper No. 77-2532. St. Joseph, MI: American Society of Agricultural Engineers; 1977.

Chow, V.T. Open-Channel Hydraulics. New York, NY: McGraw-Hill Book Company; 1959.

Crawford, N.H.; Donigian, A.S., Jr. Pesticide transport and runoff model for agricultural lands. Report No. EPA-660/2-74-013. Washington, DC: U.S. Environmental Protection Agency; 1973.

Daniel, H.A.; Elwell, H.M.; Fox, M.B. Investigation in erosion control and reclamation of eroded land at the Red Plains Conservation Experiment Station, Guthrie, Oklahoma, 1930-40. USDA Technical Bulletin No. 837. Washington, DC: United States Department of Agriculture; 1943.

David, W.P.; Beer, C.E. Simulation of Soil Erosion--Part I. Development of a mathematical erosion model, Transactions ASAE 18:126-1; 1975.

Daniel, T.C.; McGuire, P.E.; Stoffel, D.; Miller, B. Sediment and nutrient yield from residential construction sites. Journal of Environmental Quality 8:304-308; 1979.

Donigian, A.S., Jr., <u>et al</u>. Agricultural Runoff Management (ARM) model version II. Report No. EPA-600/3-77-098. Athens, GA: U.S. Environmental Protection Agency Research Laboratory; 1977.

Donigian, A.S., Jr.; Crawford, N.H. Modeling pesticides and nutrients on agricultural lands. Report No. EPA-600/2-7-76-043. Athens, GA: U.S. Environmental Protection Agency Research Laboratory; 1976.

Donigian, A.S., Jr.; Crawford, N.H. Modeling nonpoint source pollution from the land surface. EPA-600/376-083. Washington, DC: United States Environmental Protection Agency; 1976.

Foster, G.R. Sediment yield from farm fields: The Universal Soil Loss Equation and Onfarm 208 plan implementation. Chapter 3. Universal Soil Loss Equation: Past, Present, and Future. Madison, WI: Soil Science Society of America; 1979. Foster, G.R.; Huggins, L.F. Deposition of sediment by overland flow on concave slopes. Soil Erosion Prediction and Control, Special Publication No. 21. Ankeny, IA: Soil Conservation Society of America; 1977.

Foster, G.R.; Lane, L.J. Simulation of erosion and sediment yield from field sized areas. Proceeding of the International Conference on Watershed Management and Land Development in the Tropics. London: John Wiley. Forthcoming; 1980.

Foster, G.R.; Lane, L.J.; Nowlin, J.D. A model to estimate sediment yield from field-sized areas: selection of parameter values. CREAMS--A Field Scale Model for Chemicals, Runoff, and Erosion from Agricultural Management Systems, Volume II: User Manual. Washington, DC: United States Department of Agriculture, Science and Education Administration. Forthcoming; 1980.

Foster, G.R.; Lane, L.J.; Nowlin, J.D.; Laflen, J.M.; Young, R.A. A model to estimate sediment yield from field-sized areas: development of model. Report No. CP-80-10. Laxenburg, Austria: International Institute for Applied Systems Analysis; 1980. 40 p.

Foster, G.R.; Meyer, L.D. Transport of soil particles by shallow flow. Transactions of the American Society of Agricultural Engineers 15(1):99-102; 1972.

Foster, G.R.; Meyer, L.D. Mathematical simulation of upland erosion by fundamental erosion mechanics. Present and Prospective Technology for Predicting Sediment Yields and Sources. ARS-S-40. Washington, DC: United States Department of Agriculture, Science and Education Administration; 1975: 190-207.

Foster, G.R.; Meyer, L.D.; Onstad, C.A. A runoff erosivity factor and variable slope length exponents for soil loss estimates. Transactions of the American Society of Agricultural Engineers 20(4):683-687; 1977.

Graf, W.H. Hydraulics of Sediment Transport. New York, NY: McGraw-Hill Book Company; 1971.

Knisel, W.G. A system of models for evaluating nonpoint source pollution--An overview. CP-78-11, Laxenburg, Austria: International Institute for Applied Systems Analysis; 1978.

Lane, L.J.; Foster, G.R. Concentrated flow relationships-CREAMS--A Field Scale Model for Chemicals, Runoff and Erosion from Agricultural Management Systems, Volume III: Supporting Documentation. Washington, DC: United States Department of Agriculture, Science and Education Administration. Forthcoming; 1980. Lane, L.J.; Woolhiser, D.A.; Yevjevich, V. Influence of simplification in watershed geometry in simulation of surface runoff. Hydrology Paper No. 81. Fort Collins, CO: Colorado State University; 1975.

Langdale, G.W.; Barnett, A.P.; Leonard, R.A.; Fleming, W.G. Reduction of soil erosion by no-till systems in the southern Piedmont. Transactions of the American Society of Agricultural Engineers 22(1):82-86, 92; 1979.

Lombardi, F. University Soil Loss Equation (USLE), runoff erosivity factor, slope length exponent, and slope steepness exponent for individual storms. Ph.D. thesis. West Lafayette, IN: Purdue University; 1979.

Neibling, W.H.; Foster, G.R. Sediment transport capacity of overland flow. CREAMS--A Field Scale Model for Chemicals, Runoff, and Erosion from Agricultural Management Systems, Volume III: Supporting Documentation. Washington, DC: United States Department of Agriculture, Science and Education Administration. Forthcoming; 1980.

Novotny, V. Delivery of suspended sediment and pollutants from nonpoint sources during overland flow. Water Resources Bulletin 16(6):1057-1065; 1980.

Onstad, C.A.; Foster, G.R. Erosion modeling on a watershed. Transactions of the American Society of Agricultural Engineers 18(2):288-292; 1975.

Schwab, G.O.; Frevert, R.K.; Edminster, T.W.; Barnes, K.K. Soil and Water Conservation Engineering. New York, NY: John Wiley and Sons, Inc.; 1966.

Shields, A. <u>Anwendung der Ahnlichkeits--mechanik und der Turbu-</u> lenzforschung anf die Geschiebebewegung, Preuss. Versuchanstalt fur Wasserbau and Schiffbau. Berlin; 1936.

Springer, D.L.; Breinig, C.G.; Springer, M.E. Predicting soil losses in Tennessee. Journal of Soil and Water Conservation 18(4):157-158; 1963.

U.S. Department of Agriculture. Sediment sources, yields and delivery ratios. In: National Engineering Handbook, Section 3, Sedimentation. U.S. Department of Agriculture, Soil Conservation Service; 1971.

Williams, J.R. Sediment routing for agricultural watersheds. Water Resources Bulletin 11:965-974; 1975. Wischmeier, W.H.; Smith, D.D. Predicting Rainfall Erosion Losses. Agricultural Handbook Number 537. Washington, DC: United States Department of Agriculture, Science and Education Administration; 1978.

Wischmeier, W.H.; Johnson, C.B.; Cross, B.V. A soil erodibility nomograph for farmland and construction sites. Journal of Soil and Water Conservation 26(5):189-193; 1971.

Wischmeier, W.H.; Mannering, J.V. Relation of soil properties to its erodibility. Soil Science Society of America Proceedings 33:131-137; 1969.

Woolhiser, D.G. Simulation of unsteady overland flow. In: Unsteady Flow in Open Channels. Fort Collins, CO: Water Resources Publications, Volume II, Chapter 12; 1975; 485-508.

Yalin, Y.S. An expression for bedload transportation. Journal of the Hydraulics Division, Proceedings of the American Society of Civil Engineers 89(HY3):221-250; 1963.

SR - soil resuspension

APPENDIX SR

SOIL RESUSPENSION*

1.0	INTRODUCTION	SR-3
2.0	MATHEMATICAL MODELING	SR-4
	2.1 General2.2 Mathematical Approach2.3 The SESOIL Subroutine2.4 Subroutine Parameters	SR-4 SR-4 SR-10 SR-13
3.0	EXAMPLES OF APPLICATIONS	SR-15
	3.1 Waste Disposal 3.2 Agricultural Application of Wastes	SR-15 SR-15
4.0	REFERENCES	SR-16

TABLE

SR-1	SOIL ERODIBILITY INDEX (I)	SR-6
FIGURE	5	
SR-1:	THE SOIL RESUSPENSION MODULE STRUCTURE	SR-5
SR-2:	RELATIONSHIP OF SOIL RIDGE ROUGHNESS FACTOR K, FROM HEIGHT OF SOIL RIDGES	SR-7
SR-3:	RELATIONSHIP BETWEEN L, 1, K, C AND E	SR-9
SR-4:	RELATIONSHIP BETWEEN I, K, C, L, U AND E	SR-11
SR-5:	SIMPLIFIED FLOWCHART FOR SR SUBROUTINES	SR-12

*Contribution from Diane Gilbert.

Dec 81

Page

1.0 INTRODUCTION

The transport by wind of ground surface particles is called soil resuspension or wind erosion. In many geographical areas the amount of material (fugitive dust) involved is significant; therefore, sediment associated pollutant load entrained by the air can also be significant. "Fugitive dust processes" involve mechanical disturbances on the ground surface causing atmospheric pollution dishcarges. The physics of soil resuspension are complex, with several dependent variables.

This appendix is <u>not</u> intended to thoroughly describe the fugitive dust (soil resuspension) mechanics; rather it provides generalized background information on the nature of the process, the assumptions made for the mathematical modeling developed, and examples of the soil-resuspension routine of SESOIL. Alternative modeling approaches are possible. This routine is not operational in the 1981 version of SESOIL; therefore no special attention is given to the drafting of this appendix. However, in case of interest, a potential model user should contact the SESOIL model developer (Dr. M. Bonazountas, (617) 864-5770 x5871), since coding of this routine is a minor task.

2.0 MATHEMATICAL MODELING

2.1 General

The Soil Resuspension model (SR) estimates the amount of soil lost from the surface layer due to wind erosion. The amount of pollutant carried with the resuspended particles is proportional to the soil loss and the concentration of adsorbed pollutant on soil particles. Depending upon data availability, or the desired degree of accuracy, the model output provides estimates of: (1) the monthly soil loss, (2) the average annual loss, or (3) the annual loss as the sum of monthly losses.

Figure SR-1 shows the overall module structure of SESOIL. The module accounts internally for a check for average wind speeds below a "critical velocity" and passes on to the next time step if the critical condition is not met. A similar check is carried out for frozen or snow-covered ground conditions; this check is carried out differently for the annual as opposed to monthly time step analysis.

2.2 Mathematical Approach

Soil resuspension has been described and analyzed by Chepil and Woodruff (1963), Woodruff and Siddoway (1965) and Evans and Cooper (1980) among others. This subroutine is based primarily on the work of Chepil, Woodruff, and Siddoway in the development of an erosion equation and upon the work of Evans and Cooper in application of the equation to estimate particulate emissions from various open sources.

The amount of soil eroded by wind, E, as estimated by Chepil and Woodruff (1963) is:

$$E = f(I, K, C, L, V)$$

The variables are:

 I: soil erodibility index (tons/acre/season) This parameter is based upon the percentage of soil fractions larger than 0.84 mm (A) as determined by dry sieving.

> The value for I, as used in equation SR-1, however, is given in tons/acre/month (or year). For flat sites, values of I are provided in Table SR-1 for various soil types covering the range of particle sizes.

 K: soil ridge roughness factor
 K is approximated from the soil ridge height (in inches) using Figure SR-2 or Equation SR-2.

```
K = ah^2 + bh + 1  (SR-2)
```

(SR-1)



FIGURE SR-1: THE SOIL RESUSPENSION MODULE STRUCTURE

TABLE SR-1

SOIL ERODIBILITY INDEX (I)

	I	
Soil Types	(Tons/Acre/Year)	t/A/mo
Sand	436	36
Loam	207	17
Loamy Sand	180	15
Sandy Loam	156	13
Sandy-clay-loam	129	11
Silt Loam	91	7.6
Clay, Silty-clay	60	5
Sitly-clay-loam	59	4.9
Clay Loam	44	3.7
Sandy-clay	31	2.6
Silt	11	0.92

Source: Evans and Cooper 1980.



FIGURE SR-2: RELATIONSHIP OF SOIL RIDGE ROUGHNESS FACTOR K, FROM HEIGHT OF SOIL RIDGES

where h is the height (in inches of the soil ridges and must be less than 10. Parameters "a" and "b" will be determined for the operational version of SESOIL, which will not require any user interaction.

 C: climatic factor The relationship defining C is based upon work by Thornthwaite (1931) in establishing a P-E (precipitation-evaporation) index. The general equation for C is:

$$c = \frac{0.0026 \left(\frac{30}{z}\right)^{3/2} \cdot v_z^8}{\left((P/T) - 10\right)^{2.222}}$$
(SR-3)

where

z = height of mean wind velocity measurement (ft)
v_z = mean wind speed at elevation z (ft/sec)
P = mean precipitation (inches)
T = mean temperature (°F)

This relationship was derived from the equation correcting for wind speeds not measured at 30 ft elevation (e.g. SR-4), the equation for determining P-E (e.g. SR-5), and the original relationship for C (e.g. SR-6).

$$\overline{v}_{30} = \left(\frac{30}{z}\right)^{1/2} \cdot \frac{1}{v_z}$$
(SR-4)

$$P-E = 11((P/T)-10)^{1.111}$$
 (SR-5)

$$C = 34.483 \left(\frac{V_{30}}{(P-E)^2} \right)^3$$
(SR-6)

• L: field length factor

This parameter is dependent upon the values of I, K, and C already determined and upon dimensions of the site. These latter are used to find the equivalent (or unsheltered) field length, L. Graphs provided by Woodruff and Siddoway (1965) (Figure SR-3) are



FIGURE SR-3: RELATIONSHIP BETWEEN L, 1, K, C AND E

used to determine an intermediate value for E as f(I, K, C, L).¹

The field length factor is used to account for the shielding effect of barriers around the site. A boundary barrier, such as wind break or building can reduce wind speed in the upwind direction and may even cause an adjacent dead spot, i.e. $\bar{v}_0 = 0$. Within the site itself, due to the presence of barriers, effective length of field exposed to the wind may, therefore, be decreased.

• V: vegetative cover factor

The presence of vegetation reduces erosion through the combined action of three mechanisms: increased soil moisture held in root zone, physical presence of the roots, and raising of the mean aerodynamic surface above the ground surface. The latter phenomenon reduces the movement of air (i.e. wind velocity) at the ground surface.

This parameter is a function of I, K, C, and L, found graphically from a family of curves representing different values of V (Figure SR-4).¹

2.3 The SESOIL Subroutine

A simplified flowchart for the SR simulation subroutine is shown on the next page (Figure SR-5). In order to keep this figure simple, the details of time step decisions and summation of monthly erosion are omitted.

The SR subroutine requires input data for parameters which are not used by other subroutines. These parameters are:

- SOI = the soil erodability index, I; (tons/acre/month)
- WVZ = mean wind velocity at elevation z; (ft/sec)
- WEZ elevation, z, of wind velocity (WVZ) measurement; (ft)
- SRH soil ridge height: (inches)
- DFG = days the ground is frozen per month; (days)
- APW = angle of prevailing wind

¹In the final, fully automated version of SR, this graph will not be necessary. It is shown here to demonstrate the relationships between the variables.


FIGURE SR-4: RELATIONSHIP BETWEEN I, K, C, L, U AND E



FIGURE SR-5: SIMPLIFIED FLOWCHART FOR SR SUBROUTINES

- DTS = distance across field to site under analysis (ft)
- BBF = boundary barrier height (ft)

The subroutine requires data for two parameters already incorporated into SESOIL: MPM, the mean monthly precipitation (MPM) and the mean monthly temperature (TA). The subroutines which use MPM and TA as inputs require them in units of cm and °C, respectively, Therefore, the SR subroutine will read the appropriate values from the data files and automatically convert MPM from cm to inches and TA from °C to °F as part of calculating the climatic index C.

The subroutine, as currently assembled, requires user inputs to find the final eroded amount: EAF. Those inputs are necessary because a single equation expressing the eroded amount, as a function of <u>all</u> the variables, has not yet been derived. The relationship between \overline{E} and Vis of the form $E = f(e)^V$, while that between \overline{E} and L is of the form E = f(1 - b). The vegetative cover factor (V) is called VCF and the field length factor (L) is named FLF in the SR subroutine.

The FLF is a function of I, K, and C as well as the physical dimensions of the site/field relative to the prevailing wind direction. Therefore, the model outputs an intermediate value for E, defined as EAII, and a value for the unsheltered wind distance or equivalent field length (EFL). These two are applied to Figure SR-3 to determine a second intermediate value for E, called EAI2.

The final value for E, defined as EAF, is also user determined as the relation for the VCF is handled graphically based upon EAI2. The user must then input EAF so it can be integrated into the pollutant cycle subroutine. Figure SR-4 is used for this purpose.

The time steps used in the SR subroutine correspond to the four levels of simulation used by SESOIL: monthly - time-specific, monthly - general, annual - time-specific, and annual - general.

- Note: The final version of the SR subroutine will not require user input as we will determine the mathematical relationships involved to allow SR to run unaided.
- 2.4 Subroutine Parameters

Input Parameters

• Climatic Parameters

MPM = mean precipitation (inches)
TA = mean temperature (°F)

Soll Flosion Falameters SOI = soil erodability (I) (tons/acre/month) WVZ = mean wind velocity at elevation z (ft/sec) WEZ = elevation z at which WVZ was measured (ft) SRH = soil ridge height (inches) DFG = days of frozen or snow-covered ground per month DTS = perpendicular distance across field to site of analysis (ft) BBH = boundary barrier height (ft) APW = angle of prevailing wind (°) NDM = number of days in the current month User Specified Parameters EAI2 = second intermediate value for eroded amount

EAF = eroded amount -- final value

• Program Parameters (not user inputs)

CLI = climatic index, C

- SRF = soil ridge height factor
- EFL = equivalent field length
- EAIl = first intermediate value for eroded amount

• Soil Erosion Parameters

3.0 EXAMPLES OF APPLICATIONS

3.1 Waste Disposal

For the simple and probably commonly-encountered situation of a waste disposal site, the following assumptions may be made:

- 1. Soil types will vary greatly, thus relying upon the specification of a value for 1 for each site.
- 2. The site will be flat and have no ridges, thus SRH = 1.
- 3. C will be determined by the data input for the site.
- 4. The site will be of sufficient size to eliminate any reductions in the equivalent field length. Thus F(L) = 100% or a factor of one.
- 5. No vegetation will be present at the site of recent pollutant (waste) application. Thus f(V) = 100% or a factor of one.

Thus equation SR-1 reduces in this case to

$$E = \frac{0.43 (z)^{-3/2} (\bar{v}_z)^3}{((P/T) - 10)^{2.222}}$$

Assumption 4 is based upon the fact that EFL = DTS if not the analytical location is at least 6000 feet from the site's boundary along the direction of the wind and if I > 30; if I > 130, and the location of analysis is 2000 feet from the edge, then boundary barrier effects do not reduce EFL.

3.2 Agricultural Application of Wastes

Whether liquid or sludge wastes have been applied to agricultural lands, there will likely be a residual pollutant load in the surface soil layer. While some amount of the residues may be taken up by the crop, the possibility of erosion exists. In this case, a value for SRH and determination of the VCF is necessary. The user must specify from Figure SR-2 (or equation SR-2) a value for SRH during data initialization and use Figure SR-3 during the program. Similarly, if the site does not meet the size limitations described in application 3.1 above, the procedure for EAI2 is necessary -- also to be input by the user during the program. Because this function is particularly difficult to utilize with any degree of accuracy from Figure SR-4, it is recommended that assumption 4 always be made.

4.0 REFERENCES

Evans, J.S., Cooper, D.W. "An Inventory of Particulate Emissions from Open Sources" J.A.P.C.A. December 1980, 30(12):1298.

Thornthwaite, C.W. "Climates of North America According to a New Classification" Geographic Review 21:633, 1931.

Woodruff, N.P., Siddoway, F.H. "A Wind Erosion Equation" Soil Science Society Proceedings, 1965, p. 602. VO - volatilization

APPENDIX VO

DIFFUSION AND VOLATILIZATION

1.0 IN	TRODUCTION	VO-3
2.0 BA 2. 2. 2. 2.	CKGROUND 1 General 2 Diffusion/Volatilization 3 Partitioning/Distribution 4 Factors Affecting Volatilization	V0-4 V0-4 V0-5 V0-8
3.0 MA 3. 3. 3. 3. 3. 3. 3. 3.	THEMATICAL MODELING 1 General 2 Diffusion Coefficients 3 Concentrations of Compound 4 Theoretical Models 5 Experimental Models 6 The SESOIL Diffusion/Volatilization Models 7 Numerical Examples	VO-9 VO-10 VO-12 VO-14 VO-17 VO-19 VO-22
4.0 DI	SCUSSION	VO-23
5.0 NO	MENCLATURE	VO-24
6.0 RE	FERENCES	VO-27
Table V	0-1 The Farmer et al Model in SESOIL	v0-20
Figure	VO-1 Schematic Presentation of Diffusion and Volatilization	V0-6
Figure	VO-2 Soil Matrix Partitioning	VO-7

Page

•

1.0 INTRODUCTION

Diffusion and dispersion (terminology is given in Section 2.0) are two processes by which molecules of a compound in a region of high concentration move into a region of lower concentration. Diffusion occurs most readily in gases, less so in liquids, and least in solids. Volatilization is a form of diffusion that occurs when a compound moves from the soil environment into the atmosphere. For many pollutants, volatilization is an important mechanism for their loss from the soil.

The rate at which a chemical volatilizes from the soil is affected by many factors, such as soil properties, chemical properties, and environmental conditions. The magnitude of these factors and the complexity of their interactions are such that assumptions must be made in order to develop volatilization mathematical models. Many models are available in the literature, and some of these models can be applied only to specific environmental situations and only for the chemicals for which they were developed. Obviously, all models do not provide the same numerical results when employed to provide answers to a particular problem.

This appendix is <u>not</u> intended to thoroughly describe the dispersion, diffusion and volatilization processes of chemical species in soils; rather, it provides: background information on the nature of these processes, a short discussion on the physicochemical parameters affecting volatilization, a short discussion of available volatilization models and a presentation of the volatilization models employed in SESOIL. Additional information regarding diffusion and volatilization is given by Freeze and Cherry (1979) and Thomas (1981).

2.0 BACKGROUND

2.1 General

In order to select a volatilization model, it is important to understand the mechanism of movement of chemicals in the soil matrix and from the soil to the atmosphere. An elucidation of this mechanism will also aid in enumerating the factors affecting volatilization and the complexity of their interaction.

It is known that flow regimes of soils have the ability to transport dissolved substances known as solutes. Solutes are transported by advection, at an average rate equal to the average linear velocity of the flow regime. In addition, there is a tendency for the solute to spread out from its path both longitudinally and transversally. This phenomenum is called mechanical dispersion, or dispersion. Diffusion is a dispersion process of importance only at low flow regime velocities. Throughout this appendix only the terminology of diffusion will be used.

2.2 Diffusion/Volatilization

Substantial information in this section has been obtained from Freeze and Cherry (1979).

Diffusion in solution is the process whereby ionic or molecular constituents move under the influence of their kinetic activity in the direction of their concentration gradient. Diffusion occurs in the absence of any bulk hydraulic movement of the solution. If the solution is flowing, diffusion is a mechanism, along with mechanical dispersion, that causes mixing of ionic or molecular constituents. Diffusion ceases only when concentration gradients become nonexistent. The process of diffusion is often referred to self-diffusion, molecular diffusion, or ionic diffusion.

The mass of diffusing substance passing through a given cross section per unit time is proportional to the concentration gradient (Fick's first law), or

$$F = -D(dc/dx)$$
 (VO-1)

where

F	=	solute mass flux along x; (ug/cm ² ·s)
D	-	diffusion coefficient of pollutant in aqueous solu- tion; (cm ² /s)
с	=	solute concentration of pollutant; (ug/mL)

dc/dx= concentration gradient; (-)

x = direction

The diffusion coefficients for electrolytes, for example, in aqueous solutions are well known. The major ions in saturated soil layer (groundwater) of NA⁺, K⁺, Mg²⁺, Ca²⁺, Cl⁻, HCO₃⁻ have diffusion coefficients in the range of 1×10^{-9} to 2×10^{-9} m²/s at 25°C. The coefficients are temperature-dependent and at 5°C, for example, they are about 50% smaller. In unsaturated soil zones, estimation of the overall diffusion coefficient (aqueous, vapor phase) is more complicated as will be discussed in a later section.

In <u>summary</u>, it is important to realize that both diffusion and volatilization refer to the movement of pollutants from a region of high concentration towards a region of lower concentration (minus sign in equation VO-1). Thus, in soils, when a "slug" of highly concentrated pollutant is introduced into a volume of soil (soil, soil moisture, air), it will "spread" out (diffuse) and will occupy a greater volume and at a lower concentration. (Figure VO-1.) Within the soil compartment, this spreading is called <u>diffusion</u>. When the pollutant spreads from the soil column to the atmosphere, the process is called <u>volatilization</u>. Therefore, diffusion and volatilization are the two processes contributing to the continuous movement of a pollutant from its point of release into the soil compartment to the atmosphere.

2.3 Partitioning/Distribution

A soil environment consists of three media: air, water and soil. (Figure VO-2.) Therefore, a compound incorporated into a soil matrix will be partitioned, and the pollutant will be present in all three phases: (1) mixed in soil air, (2) dissolved in soil moisture, and (3) sorbed on soil particles. The concentrations of the compound in each medium can be related to equilibrium partitioning coefficients (frequently constant parameters or isotherms) as discussed in Appendix AD (Adsorption) and in Appendix PT (Pollutant Transport).

The three main distribution pathways involved in the diffusion/volatilization process of a compound incorporated in a soil matrix can be summarized as:

•	Compound	on	soil	particles	↔	compound	in	solution
---	----------	----	------	-----------	---	----------	----	----------

- Compound in solution ↔ compound in vapor phase (in soil air)
- Compound in vapor phase →→ compound into atmosphere (volatilization)



FIGURE VO-1

SCHEMATIC PRESENTATION OF DIFFUSION AND VOLATILIZATION

Total matrix volume: $V_{\tau} = V_{a} + V_{w} + V_{s}$ moisture content : $\Theta = V_{w} / V_{\tau}$ air-soil content (air-porosity): $m_{air} = V_{a} / V_{\tau}$ porosity (total) : $m_{air} = (V_{a} + V_{w}) / V_{\tau}$



FIGURE VO-2



Diffusion occurs in all three media of the soil environment; however, it takes place most rapidly in air and least rapidly in solids. Diffusion in the vapor phase (air) occurs 10^4 times faster than in water. Diffusion in the solid phase is extremely slow compared to other pollutant transport processes, therefore, it is neglected by most pollutant models.

Volatilization at the soil surface can also occur from all three phases. However, volatilization from every phase is much faster than diffusion. Thus, diffusion is the rate-controlling process in the movement of chemicals from a soil layer, to a soil surface, and then into the atmosphere.

Due to the interaction of diffusion within a phase with partitioning among phases, the rate of volatilization depends on both diffusion rates and partitioning behavior. Therefore, any factor which causes a small change in the distribution of a compound among the soil, soil-water, soil-air, and atmosphere can have a large effect on the rate of volatilization of that compound. For example, when a pollutant is in a soil to which it is strongly adsorbed, very little pollutant will be in the soil-air or the soil-water. Since diffusion in solids is slow, only small amounts of pollutant will be available to volatilize. When this (same) pollutant is in a slightly less adsorbant soil, soil-air concentrations can become significant, and the faster air diffusion process will contribute large amounts of pollutant to the soil surface. In the latter case, volatilization releases can be quite high.

2.4 Factors Affecting Volatilization

Chemical, soil and general environmental properties affect volatilization. Some of the physicochemical properties of a compound which can affect volatilization are vapor pressure, solubility in water, adsorption behavior, and diffusion coefficients in air and water. Some soil properties which influence volatilization behavior include moisture content, density, porosity, organic carbon content, clay content, and soil diffusion characteristics. Typical environmental factors impacting volatilization can be wind speed at the surface, humidity, temperature, pH, surface cover, and hydrology at the site (e.g., infiltration, capillarity).

The impact on volatilization rates of many of the above factors is not quantifiable yet. Therefore, to formulate mathematical models, assumptions have been made by researchers by limiting the factors included in their equations. As of now, no one general model is available to estimate volatilization rates in all situations, but effective and promising research is underway. The SESOIL model is designed to employ existing volatilization models; therefore, as more knowledge and information become available regarding the process, refined volatilization models might be implemented by SESOIL.

3.0 MATHEMATICAL MODELING

3.1 General

Volatilization modeling encompasses the

- Selection of a model most applicable to a user's needs (e.g., problem to be simulated, data availability);
- (2) Estimation of the diffusion and other coefficients for the model selected.

Generally speaking, there exist two types of models for estimating volatilization rates

- Theoretical models based upon Fick's first and second laws;
- (2) Experimental models based upon laboratory or field experimental data and statistically derived equations.

Fick's first law describes the steady state mass flux of a pollutant due to diffusion. (See Section 2.2.) Fick's second law is obtained from Fick's first law and the equation of continuity and describes the nonsteady state mass flux of a pollutant due to diffusion. Fick's second law is also known in the literature as the diffusion equation of solutes in porous media (air, water, soil).

Fick's first law in one direction z is expressed as

$$F = -D (dc/dz)$$
 (VO-2)

Fick's second law is expressed as

$$dc/dt = D^* (d^2c/dz^2)$$
 (VO-3)

where

 $\mathbf{D}^{\star} = \mathbf{k} \cdot \mathbf{D} \tag{VO-4}$

in which

F = mass flux across a surface; (ug/cm²·s)
D = aqueous diffusion coefficient of compound; (cm²/s)
c = solute concentration of compound in soil moisture; (ug/mL)
z = distance (depth) normal to diffusing direction; (cm)

Arthur D Little, Inc.

- t = time; (s)
- D* = apparent diffusion coefficient of compound in soil
 matrix; (cm²/s)
- k = proportionality coefficient, to correct for soil
 matrix effects; (-)

Diffusion coefficient (D, D*) definitions and other issues are discussed in the following section.

No scientist can argue that theoretical models are better or worse than experimental models since each type has its own strengths. In the first category are the models of Mayer et al (1973), Jury et al (1979), and Farmer et al (1980). In the second category are the models of Hartley (1969), Hamaker (1972), and Dow (1979). Each model cannot be applied to all situations. One model might be appropriate for pollutant applied on surface, another model might be more appropriate for a chemical incorporated into an upper soil layer, and a third one the most appropriate for a buried compound. Some of these models are presented in subsequent sections.

3.2 Diffusion Coefficients

Fick's law applies to all environmental media--air, water, and soil. Therefore, for a particular compound, we may have its diffusion coefficients in air (D_a) , in water (D_w) , in soil (D_s) , in soil-air (D_{sa}) , in soil-moisture (D_{sm}) , and overall apparent coefficients (D^*) in all above media.

In the second Fick's law, the diffusion coefficient of ions is characterized as an "apparent" diffusion coefficient, D*, which in soil systems is smaller than that in water bodies, because in porous media the ions follow longer paths of diffusion caused by the presence of the particles in the solid matrix and because of adsorption on the solids. In laboratory studies of diffusion of nonadsorbed ions in porous materials, k values between 0.5 and 0.01 are commonly observed (Freeze and Cherry 1979). No specific definition of an apparent coefficient can be made. Estimation of this apparent diffusion coefficient D* is equation- and chemical-specific, and such issues are discussed in the following sections with the work of each individual researcher.

Most of the volatilization equations require as input a chemicalspecific diffusion coefficient, primarily in air or water, in units of cm/s. It is worth mentioning at this point a few ways of obtaining this parameter since this information will facilitate readers to better understand the <u>correct</u> way of applying the various volatilization equations or models that will be presented in the subsequent sections.

Diffusion coefficients in the air (D_a) are available in the literature for some chemicals (e.g., Weast et al 1978). Diffusion coefficients in

water (D_w) are also available in the literature or have been estimated by researchers individually (Bode et al 1973). Nelken (1981) gives some estimation techniques for both air and water diffusion coefficients.

3.2.1 Molecular Weight Effects

Air diffusion coefficients of two similar compounds 1 and 2 can be related by the expression

$$D_{a_1}/D_{a_2} = (M_2/M_1)^{1/2}$$
(VO-5)

where

 $D_a = vapor phase (air) diffusion coefficient; (cm²/s)$

M = molecular weight of compound; (g/mol)

3.2.2 Porosity Effects

Millington and Quirk (1961) have proposed a correcting relationship for relating the apparent diffusion coefficient of a compound in the soilair (D_{sa}^{*}) to the diffusion coefficient of the same compound in the air to (D_{a}) .

$$D_{sa}^{*} = D_a \left(n_{air}^{10/3} / n^2 \right)$$
 (VO-6)

where

- D'sa = apparent diffusion coefficient of compound in soilair; (cm²/s)
- Da = diffusion coefficient of compound in vapor (air); (cm²/s)

nair = soil air-filled porosity; (cm³/cm³); (mL/mL)

n = soil (total) porosity; (mL/mL)

Frequently, it has been assumed in model applications that $D_{sa}^* = D_{sa}$; where, D_{sa}^* , the apparent diffusion coefficient of the compound in the soil-air and D_{sa} the "real" coefficient of the compound is the soil-air. This is done because of the difficulty in defining the meaning of the "apparent" diffusion coefficient.

The soil air-filled porosity in equation VO-6 can be estimated from

(a) either the (total) soil porosity (n) and the soil moisture content (θ)

$$n_{air} = n - \theta \tag{(10-7)}$$

(b) or the soil bulk density (ρ_b) and the soil particle density (ρ)

$$n = 1 - \rho_b / \rho \tag{V0-8}$$

where

 ρ_b = soil bulk density; (g/cm³) ρ = particle density; (g/cm³)

The particle density can be measured, but for most soil minerals, materials it usually equals 2.65 g/cm³.

3.2.3 Temperature Effects

Temperature affects values of an air diffusion coefficient according to (Farmer et al 1980)

$$D_{a2} = D_{a1} (T_2/T_1)^{1/2}$$
(VO-9)

where

 $D_{a2}, D_{a1} =$ air diffusion coefficients of a compound at T_2 and T_1 ; (cm²/s)

 $T_2, T_1 = temperatures; (OK)$

According to Hamaker (1972), vapor-phase (air) diffusion coefficients of the same compound and at different temperatures may be related by

$$D_{a1}/D_{a2} = p_2/p_1 (T_1/T_2)^m$$
 (VO-10)

where

p = ambient total pressure; (consistent units)
T = temperature; (°K)
m = experimental coefficient; (-)

3.3 Concentrations of Compound

Volatilization equations have to be employed with <u>care</u> because of diffusion and the concentration definitions. Some equations, for

example, account for the total concentration of the pollutant in the soil c_0 , other equations account only for the solute concentration c and others only for the pollutant concentration in the soil-air c_{sa} .

The total concentration of a chemical in the soil matrix can be expressed (Jury et al 1980) as

$$c_{0} = (\rho_{b} \cdot s + \theta \cdot c + n_{air} \cdot c_{sa})$$
(VO-10)

where

- co = overall (total) concentration in soil matrix; (ug/cm³ soil)
- ρ_b = soil bulk density; (g/cm³)
- s = adsorbed concentration on soil particles (ug/g of soil)
- θ = volumetric soil moisture content; (mL/mL)
- c = solute concentration (in liquid phase) of compound; (ug/mL)
- nair = soil-air content or air-filled porosity; (mL/mL)
- c_{sa} = concentration of compound in the soil-air; (ug/mL)

The solute concentration c of a compound can be related to its soil-air concentration c_{sa} via Henry's law.

$$c_{sa} = c \cdot H/R(T+273)$$
 (VO-12)

where

csa	=	concentration in soil-air; (ug/mL)
с	=	concentration in soil moisture; (ug/mL)
н	=	Henry's law constant; (m ³ ·atm/mol)
R	=	gas constant; (8.2x10 ⁻⁵ m ³ ·atm/mol· ^o K)
Т	=	temperature; (^o C)
o _K	=	°C+273

The solute concentration c of a compound can be related to its adsorbed concentration s on soil particles via adsorption isotherms, the latter

being described in Appendix AD (Adsorption). Two well known isotherms are Freundlich and Langmuir.

3.4 Theoretical Models

The following sections present only a general overview of available mathematical volatilization models. Discussions are tailored to the needs of this appendix.

3.4.1 Farmer, Yang, Letey

A simple volatilization model to study the steady state rate of volatilization of hexachlorobenzene (HCB) wastes in soil through a soil cover is developed by Farmer et al (1980).

This model is based on a discretized version of Fick's first law (equation VO-2) over space, and assumes vapor phase diffusion being the rate controlling processes. The volatilization rate of HCB from the landfill, equals

$$P = -D_{sa}^{*} (c_{atm} - c_{sa})/L$$
 (VO-13)

where

P = pollultant flux across soil cover; (ug/cm²·s)

- D^{*}_{sa} = apparent steady diffusion coefficient in soil-air; (cm²/s)
- catm = concentration of the volatilizing compound in the atmosphere at the surface of the soil layer; (ug/mL)
- c_{sa} = concentration of compound in the soil-air; (ug/mL)
- L = depth of covering soil layer; (cm)

Expression VO-3 is known also as the Stephen equation. Farmer et al employed equations VO-6 and VO-7 for the estimation of the apparent diffusion coefficient. When employing this model, the concentration of a compound in the soil-air can be estimated from the concentration of the compound in the soil moisture via equation VO-11 and vice versa. The resulting equation is

$$P = D_{a} \cdot [(n-\theta)^{10/3}/n^{2}] [H \cdot c/R(T+273) \cdot L]$$
 (VO-14)

and

$$P_0 = P \cdot \Delta t$$

where

Ρ	=	pollutant flux across soil surface; (ug/cm ² ·s)
Po	=	total pollutant flux/loss across soil surface; (ug/cm ²)
Da	=	diffusion coefficient of compound in air; (cm^2/s)
н	=	Henry's law constant; (m ³ ·atm/mol)
R	=	gas constant; $(8.2 \cdot 10^{-5} \text{ m}^{3} \cdot \text{atm/mol} \cdot ^{\circ}\text{K})$
T	=	temperature; (^O C)
с	=	concentration of compound in soil moisture; (ug/mL)
n	=	soil porosity; (fraction)
θ	=	soil moisture; (fraction)
∆t	=	length of simulation time step; (s)

This Farmer et al (1980) model is employed by the current version of SESOIL.

3.4.2 Jury, Grover, Spencer and Farmer

Jury et al (1980) account in their work for the total concentration of the chemical in the soil matrix (equation VO-11). By employing Fick's second law, Henry's law, linear adsorption isotherms and the moisture continuity equation in porous media, they derived the expression

$$P_{o} = 2(c_{o}-g)(D^{*} \cdot t/\pi \cdot e)^{1/2}$$
 (VO-15)

where

- P₀ = volatilized (total) chemical mass per unit area after time t; (ug/cm²)
- co = total initial concentration of compound in soil; (ug/g soil)
- D^* = apparent diffusion coefficient; (cm²/s)
- $\pi = 3.14$
- e = derived coefficient by Jury et al; (mL/mL)

g = derived coefficient by Jury et al; (ug/cm³)

The above information is obtained from Thomas (1981); therefore, interested readers should refer to the original publication. The Jury et al model is not incorporated (coded) in the present version of SESOIL since its employment has not be mandated by a particular application.

3.4.3 Mayer, Letey and Farmer

Mayer et al (1974) applied Fick's second law (equation VO-3) to the mathematical description of the movement chemicals <u>in soils</u>. Diffusion is assumed to be the only mechanism transporting chemicals to the soil surface; therefore, this model may underestimate volatilization rates. (Thomas 1981.)

Depending upon the boundary conditions employed to solve the diffusion equation, Mayer et al presented five different solutions (models) of the form

$$c(z,t) = c(c_{to},D^*,t,L,R_0)$$
 (V0-16)
 $P_0 = f[c(z,t)]$

where

c(z,	t)=	solute concentration or concentration in soil-air of a compound; (ug/mL)
Po	=	total volatilizing mass of compound; $(ug/s \cdot cm^2)$
c _{to}	=	initial (t=0) concentration in soil moisture or as indicated; (ug/mL)
z	=	direction
t	=	time; (s)
D*	=	apparent diffusion coefficient; (cm ² /s)
L	=	soil layer depth; (cm)
R _o	=	isotherm coefficients or Henry's law constant

As an example of Mayer's et al work, two solutions of equation VO-16 are presented below. This information is obtained from Thomas (1981); therefore, interested readers should refer to the original publication.

 Simplified solution of equation VO-3 for a compound that is removed rapidly from the soil surface; upward diffusion only.

$$c(z,t) = c_{to} \cdot erf[z/2(D \cdot t)^{1/2}]$$
 (VO-17)
 $P_o = c_{to}(D/\pi \cdot t)^{1/2}$

(2) Solution of previous problem; diffusion downwards has been accounted.

$$c(z,t) = 0.5c_{to}[2erf(A)-erf(A-B)-erf(A+B)]$$

$$P_{o} = D \cdot c_{to}/(\pi Dt)^{1/2}$$

$$A = z/2(D \cdot t)^{1/2}$$

$$B = L/2(Dt)^{1/2}$$
(VO-18)

Adams et al (1976) indicated that equation VO-17 gives dependable estimates when

$$t = z^2/14.4 \cdot D^*$$
 (VO-19)

From equation VO-19, it also follows that the time to reach a half concentration of a chemical, $c_0/2$, occurs when erf(x)=0.5, or when $z/2(D^{*}\cdot t)^{1/2} \simeq 0.477$, from which we obtain

$$t_{1/2} = 0.91D*/z^2$$
 (VO-20)

The Mayer et al model is not incorporated (coded) into the present version of SESOIL since its employment has not been mandated by a particular application.

3.5 Experimental Models

Experimental and application specific models are based on field and laboratory sampling programs and on statistical data analyses. Two known models are of Dow and Hamaker. (Thomas 1981.)

3.5.1 Dow

Researchers at Dow Chemical Company (1979) have performed experiments to establish the relationship between the volatilization rate of a chemical <u>applied</u> to a soil surface and the chemical properties of vapor pressure, water solubility, and adsorption coefficient. The rate constant of volatilization for chemicals applied or spilled on the soil surface was found to equal

$$k_v = 4.4 \times 10^7 \cdot (P_{vp}/K_{oc} \cdot SL)$$
 (VO-21)

and

$$c = c_0 e^{-k_v \cdot t}$$

where

kv	=	rate constant for volatilization; (day ⁻¹)
К _{ос}	=	soil adsorption coefficient (soil/water) based on organic carbon content; (ug/g)/(ug/mL)
SL	=	solubility of the compound; (ug/mL)
Pvp	=	vapor pressure of chemical; (mm Hg)
c _o ,c	=	concentration of chemical; initial, after time t; (ug/mL)
t	=	time; (day)

3.5.2 Hamaker

Hamaker (1972) proposed a relationship for estimating the volatilization rates of chemicals that are <u>strongly adsorbed</u> on soil particles. The impregnated soil layer is assumed to be semi-infinite and the expression derived is

$$P_{o} = 2c_{o} \cdot (D^{*} \cdot t/\pi)^{1/2}$$
 (VO-22)

where

- P_o = loss (total) of chemical per unit area after time t; (ug/cm²)
- co = total initial concentration of chemical in the soil; (ug/cm³)
- D* = apparent diffusion coefficient in soil-air; (cm²/sec)
- t = time; (sec)
- $\pi = 3.14$

The total initial concentration of chemical in the soil is given by equation VO-11. Other equations required together with VO-11 are the adsorption isotherm (e.g., equation AD-7, Freundlich), the porosity equation VO-7 and lienry's law equation VO-12.

.

3.6 The SESOIL Diffusion/Volatilization Models

3.6.1 Diffusion Model

The diffusion equation VO-1 (Fick's first law) is employed by the volatilization routine of SESOIL to estimate upward mass flow in the soil column and to the air. Upward diffusion is omitted since it is accounted explicitly via the volatilization model of the pollutant transport routine. Downward diffusion is omitted (as a diffusion term) since the pollutant transport routine accounts for diffusion (partitioning of phases; see Appendix PT) in the soil matrix explicitly. In that respect, downward diffusion is also accounted in a different way.

3.6.2 Volatilization Models

The selection of the appropriate volatilization model is entrusted to the user. Needs of certain simulations performed mandated the employment of two models, one of which is coded in Subroutines VOLA and VOLM. (See Appendix FC.) Only minor changes to the code are necessary to add other volatilization models. The following two sections describe models the developers of SESOIL have tested.

3.6.2.1 Farmer, Yang, Letey

The Farmer et al (1980) model, applied to HCBs (see Section 3.4.1), accounts for the Stephan equation which describes pollutant loss via volatilization of a pollutant that is <u>buried</u> under a layer of clean soil (cover). According to Farmer et al, this is an appropriate model to describe pollutant movement through soils where air diffusion is the rate controlling process of volatilization.

For convenience to SESOIL users, the model (equations)--applied over a simulation time step t--and its input parameters are summarized in a table. (Table VO-1.) The chemical-specific parameters (Henry's law and diffusion constants) are available from handbooks or may be estimated according to methods presented by Lyman et al (1981). The soil porosity values are available from the SESOIL documentation (Appendix ID), soil handbooks (USDA, USGS), or from experimental data. Temperature values for a site-specific area are obtained from NOAA reports. The compartment geometry parameters are chosen according to the needs of the investigation. Upward mass flux is accounted in SESOIL only by the existence of a concentration gradient (Fick's law).

3.6.2.2 Hamaker

The Hamaker (1972) experimental model which estimates time dependent volatilization fluxes of pollutants <u>mixed</u> in an <u>upper</u> soil layer was coded in the past in volatilization subroutine VOLATA of SESOIL. Information of this model is presented in Section 3.6.2. The diffusion coefficient of the model was obtained from handbooks. Other coefficients were estimated on-line by SESOIL. The code accompanying this version of SESOIL, however, does <u>not</u> contain the Hamaker model for various reasons.

TABLE VO-1

THE FARMER et al MODEL IN SESOIL

Governing Equation

$$P_{o} = D_{a}[(n-\theta)^{10/3}/n^{2}][c \cdot H/(R \cdot (273+T) \cdot L)] \cdot \Delta t$$
 (VO-14)

<u>Derived</u> by employing

The Stephan equation VO-13, Henry's law equation VO-12, the Millington and Quick (1961) equation VO-6, and the porosity convecting equation VO-7 which are given below.

St.eq :
$$P = -D_{sa}^{*}(c_{atm}-c_{sa})/L$$
 (VO-13)

H.law:
$$c_{sa} = c \cdot H/R(T+273)$$
 (VO-12)

MQ.eq :
$$D_{sa}^{*} = D_a (n_{air}^{10/3}/n^2)$$
 (VO-6)

Po.eq:
$$n_{air} = n - \Theta$$
 (VO-7)

where

 $P_0 =$

time Δt ; (ug/cm²)

FORTRAN Variable total pollutant flux across soil surface within PVOL

D _a	=	diffusion coefficient of chemical in air; (cm ² /s)	D
n	2	soil (total) porosity; (fraction)	N
θ	=	soil moisture content; (fraction)	THA
с	=	concentration of compound in the soil moisture; (ug/mL)	с
н	=	Henry's law constant for compound; (m ³ ·atm/mol)	н
R	=	gas constant; (8.2x10 ⁻⁵ m ³ .atm/mol ^o K)	R
Т	=	temperature; (^o C)	т
L	=	thickness of covering soil layer; (cm)	L
D [*] sa	=	apparent diffusion coefficient of chemical in soil- air; (cm ² /s)	DA

TABLE VO-1 (continued)

FORTRAN Variable

^c atm	=	0.0; concentration of compound in the atmosphere; (ug/mL)	CATM
c _{sa}	=	concentration of compound in the soil-air; (ug/mL)	CAIR
n _{air}	=	air-filled porosity (fraction)	NAIR
∆t	=	length of simulation time step; (s)	DT

Assumptions

 $c_{air} = 0.0$

Covering layer thickness (L) is equal to the length (depth) from the center of the SESOIL layer to the soil surface.

Input Parameters to SESOIL Subroutine

Parameters	Description	FORTRAN Variable
Da	Diffusion coefficient of the pollutant in air; (cm ² /s)	DA
n	Porosity (total) of the soil; (-)	N
н	Henry's law constant of pollutant; (m ³ ·atm/mol)	н
Т	Temperature; (^o C)	TA
d_u, d_L	Soil depths; (cm)	DU, DL
Z	Soil depth to groundwater; (m)	Z



3.7 Numerical Example

Assuming trichloroethylene (TCE) as a chemical compound, in a sandy loam soil and constant environmental conditions as, for example

Da	=	$0.072 \text{ cm}^2/\text{s}$
н	=	0.060 m ³ ·atm/mol
R	=	$8.2 \times 10^{-5} \text{ m}^{3} \cdot \text{atm/mol}^{\circ} \text{K}$
n	=	0.25
θ	~	0.10
т	=	14°C
L	=	2.0 m = 200 cm
∆t	=	$1 \text{ month} = 2.6 \times 10^6 \text{s}$
с	=	10 ug/mL

We estimate (equation VO-14) the pollutant mass volatilized within a month through the soil cover to the atmosphere

 $P_{o} = 0.072[(0.25-0.10)^{10/3}/0.25^{2}][10 \cdot 0.060/8.2x10^{-5}(14+273) \cdot 2.6x10^{6} = 685 \text{ ug/cm}^{2}$

4.0 DISCUSSION

As repeated consistently throughout this Appendix, SESOIL users' volatilization equations must be employed with care because of both the diffusion and the concentration definitions employed by various researchers. Since input data to SESOIL are summarized in Table VO-1 together with the corresponding model employed, special attention <u>has</u> been given by the SESOIL model developers to facilitate use of this model and to prevent incorrect input data inserts. Therefore, input data are kept to a limited number and internal (on-line) model calculations (e.g., c_{sa} versus c partitioning) assist in accomplishing this objective. Users have to make sure to input the correct number only for an input parameter. (Table VO-1.)

5.0 NOMENCLATURE

Symbol	Description	Units
A	Function, Equation VO-18	-
В	Function, Equation VO-18	-
b	Soil bulk density	g/cm ³
°C	Temperature in Celcius	°c
с	Concentration of compound in soil moisture (solute concentration)	ug/mL
C _{atm}	Concentration of compound in atmosphere	ug/mL
c _{sa}	Concentration of compound in soil-air	ug/mL
c _{to}	Initial (t=0) concentration in soil moisture	ug/mL
c _o	Concentration (total) of compound in soil matrix	ug/g soil
d[]/dx	Gradient along x	-
D _a	Diffusion coefficient of compound in air	cm ² /s
D _w	Diffusion coefficient of compound in water	cm ² /s
D _s	Diffusion coefficient of compound in soil	cm ² /s
D _{sa}	Diffusion coefficient of compound in soil-air	cm ² /s
D _{sm}	Diffusion coefficient of compound in soil moisture	cm ² /s
D*	Apparent diffusion coefficient of compound in various media	cm ² /s
e	Coefficient, Equation VO-15	mL/mL
erf[]	Error function	-

Symbol	Description	Units
F	Solute flux along a direction x, Equation VO-1	ug/cm ² ·s
g	Coefficient, Equation VO-15	ug/cm ³
н	Henry's law constant	m ³ ·atm/mol
К _{ос}	Adsorption coefficient based on organic carbon content	(ug/g)/(ug/mL)
k	Proportionality coefficient to correct for soil matrix effects	-
oK	Temperature (in Kelvin)	ο _K
L	Depth of soil cover	cm
m	Exponent, Equation VO-10	-
М	Molecular weight of compound	g/mol
n	Soil porosity (total)	fraction
n _{air}	Soil-air filled porosity	fraction
P	Pollutant flux (volatilized) across soil cover	ug/cm ² ·s
Po	Volatilized (total) chemical mass after time t	ug/cm ²
Pd	Volatilized mass after Dow	ug/cm ²
P _{vp}	Vapor pressure of compound	mm Hg
R	Gas constant = 8.2×10^{-5}	m ³ ·atm/mol· ^o K
Ro	Isotherm coefficient, Equation VO-16	-
SL	Solubility of compound	ug/mL
T	Temperature	°C
t	Time	S
x	Direction	-

Symbol	Description	Units		
Z	Vertical distance/direction	cm		
θ	Soil moisture content	fraction		
ρ	Soil particle density	g/cm ³		
ρь	Soil bulk density	g/cm ³		
π	3.14	-		

6.0 REFERENCES

Bode, L.E.; C.L. Day; M.R. Gebhardt and C.E. Goering (1973); Prediction of Trifluralin Diffusion Coefficients, Weed Science, 21(5): 485-489.

English, C.J.; J.R. Miner and J.K. Koelliker (1980); Volatile Ammonia Losses from Surface-Applied Sludge, Journal WPCF, 52(9): 2340-2349. Technical Paper 5100, Oregon Agricultural Experiment Station, Corvallis, Oregon, Projects 116 and 324.

Farmer, W.J.; K. Igue and W.F. Spencer (1973); Effects of Bulk Density on the Diffusion and Volatilization of Dieldrin from Soil, J. Environ. Qual, 2, 107-09.

Farmer, W.J.; M.S. Yang; J. Letey and W.F. Spencer (1979); Hexachlorobenzene: Its Vapor Pressure and Vapor Phase Diffusion in Soil, unpublished, undated manuscript received as personal communication from W.J. Farmer, 4 December 1979.

Farmer, W.J.; M.S. Yang; J. Letey and W.F. Spencer (1980); Land Disposal of Hexachlorobenzene Wastes: Controlling Vapor Movement in Soil, EPA-600/2-80-119, Office of Research and Development, U.S. Environmental Protection Agency, Cincinnati, Ohio.

Freeze, F.A. and J.A. Cherry (1979); Groundwater, Prentice-Hall, Inc., Englewood Cliffs, New Jersey.

Hamaker, J.W. (1972); Diffusion and Volatilization, Chapter 5 in Organic Chemicals in the Soil Environment, Vol. 3, C.A.I. Goring and J.W. Hamaker (eds.), Marcel Dekker, New York.

Jury, W.A.; R. Grover; W.F. Spencer and W.J. Farmer (1980); Modeling Vapor Losses of Soil-Incorporated Triallate, Soil Sci. Soc. Am. J., 44, 445-50.

Letey, J. and W.J. Farmer (1974); Movement of Pesticides in Soil, Chapter 4 in Pesticides in Soil and Water, W.D. Guenzi (ed.), Soil Science Society of America, Madison, Wisconsin.

Lyman, W., et al (1981) Research and Development Methods for Estimating Physicochemical Properties of Organic Compounds of Environmental Concern. Prepared by Arthur D. Little, Inc., Phase II Final Report for U.S. Army Medical Bioengineering Research and Development Laboratory, Fort Detrick, Maryland; McGraw-Hill Book Company, New York.

Mayer, R.; J. Letey and W.J. Farmer (1974); Models for Predicting Volatilization of Soil-Incorporated Pesticides, Soil Sci. Soc. Am. Proc., 38, 563-68.

Thomas, R. (1981), in Lyman, W. (1981); Chapter 16.

Weast, R.C. and M.J. Astle (eds.) (1978); CRC Handbook of Chemistry and Physics, 59th ed., CRC Press, Inc., West Palm Beach, Florida.

AD - adsorption

APPENDIX AD

ADSORPTION AND DESORPTION

		Page		
1.0	INTRODUCTION			
2.0	BACKGROUND			
3.0	MATHEMATICAL MODELING	AD-6		
	 3.1 General 3.2 The Freundlich Model 3.3 The Langmuir Model 3.4 The SESOIL Model 	AD-6 AD-7 AD-9 AD-11		
4.0	DISCUSSION	AD-13		
5.0	NOTATIONS			
6.0	REFERENCES	AD-15		

TABLE

AD-1:	REGRESSION	EQUATIONS	FOR	THE	ESTIMATION	OF	Koc	AD-10
-------	------------	-----------	-----	-----	------------	----	-----	-------
1.0 INTRODUCTION

Adsorption is the adhesion of pollutant ions or molecules to the surface or soil solids, causing an increase in the pollutant concentration on the soil surface over the concentration present in the soil moisture. Adsorption occurs as a result of a variety of processes with a variety of mechanisms and some processes may cause an increase of pollutant concentration within the soil solids--not merely on the soil surface. Processes which can contribute to increased soil concentrations include ion exchange, physical sorption, specific adsorption, partitioning, fixation, and chemisorption. The general term sorption is often used to describe these phenomena, particularly when the adsorption mechanisms are not known. Most sorption processes are reversible; the reverse processes result in desorption.

Adsorption and desorption can drastically retard the migration of pollutants in soils, therefore, knowledge of this process is of importance when dealing with contaminant transport in soil moisture to groundwater. Sorption is usually considered to occur rapidly, relative to the rate of pollutant migration in the soil matrix due to the movement of soil moisture or groundwater flow. In addition, adsorption and desorption are usually considered to be in equilibrium--in mathematical modeling studies--and are modeled as one reversible process. This assumption facilitates modeling without substantially impacting overall long-term model estimates. Nonetheless, discrete adsorption and desorption modeling is always feasible.

Various scientists have carried out numerous experiments aimed at understanding the adsorption and desorption process, for example, Rifai, et al (1956), Day and Forsythe (1958), Biggow and Nielsen (1962), Kay and Elrick (1967) and Chiou, et al (1979). Their studies, however, do not always address questions of the nature of the chemical interactions occurring; they focus, rather, on the movement of pollutants in a soil column, from which estimates of the magnitude of pollutant involved in the sorption process.

This appendix is <u>not</u> intended to thoroughly describe the sorption process of pollutants in soil; rather, it provides background information on the nature of the adsorption, mathematical modeling issues and the way adsorption is modeled in SESOIL.

2.0 BACKGROUND

Background information contained in this section was obtained from the Encyclopedia of Soil Science, Part 1 (Fairbridge and Finke, 1979), and its chapter on adsorption phenomena authored by D.R. Kenney. Additional information is presented in the literature by Lyman (1981) and Dragun (1980).

In general terms, the sorption of pollutant refers to processes that result in a higher concentration of a particular component at the surface or within a solid phase than is present in bulk solution of soils and The general term sorption is frequently used instead of sediments. adsorption because the actual sorption mechanisms are not often known. Sorption is the major general retention mechanism for many organic compounds and metals, and the sorption and desorption phenomena play an important role in controling the availability of several plant nutrients, the rate of leaching to groundwater, volatilization from soil surface, or degradation of organic compounds such as pesticides. The sorption and desorption phenomena also protect water supplies by retaining numerous potential pollutants including nutrients, heavy metals, pesticides, and pathogens. Compounds or ions adsorbed on a soil particle surface are in equilibrium with the soil solution and are capable of desorption.

In the past the sorption of inorganic ions by soils was often thought to be due largely to precipitation reactions, i.e., the formation of a sparingly soluble solid phase. However, careful studies have shown that the solubility product principle will not account for the extremely low concentrations of phosphorus and many of the metals in the bulk solution of an aerated soil (Lindsay 1972). These studies have brought about the realization that precipitation generally dominates only at relatively high concentrations of the reactants.

Cation exchange, the interchange between a cation in solution and another cation on the surface of any surface-active material, is one important adsorption mechanism for cationic plant nutrients (potassium, calcium and magnesium) in soils. A similar type of anion exchange may be involved in the retention of nitrate and chloride in acid highly weathered soils carrying a net positive charge.

Physical sorption involves the attachment of the sorbent and sorbate through weak atomic and molecular interaction forces (van der Waal forces) that operate when the electron clouds of the atoms do not overlap sufficiently to cause strong attractive forces. The activation energy for this type of attraction is characteristically low, much lower than is normally observed for ions (e.g., orthophosphate) which are bound more strongly. Most anions, as well as several heavy metals, exhibit a property termed specific adsorption. This property involves the exchange of the ion with surface ligands to form partly covalent bonds with lattice ions. The net result is that the amount of ion adsorbed is far greater than would be expected for nonspecifically adsorbed species that are adsorbed according to their relative abundance (Nott 1970).

Organic compounds, including materials such as proteins, enzymes, viruses, pesticides, and bacteria can be sorbed to soil particles (McLaren and Peterson 1965, Marshall 1971, Green 1974, Weed and Weber 1974). As with inorganic species, the mechanisms are extremely complex and are even more difficult to categorize because of differing chemical and physical characteristics of natural and synthesized organic materials.

In <u>summary</u>, sorption is the equilibrium association of pollutants by soil particles. If adsorption is the dominant process for pollutant behavior in soils, then pollutant migration to groundwater can be substantially "retarded." When no sorption occurs the pollutant can-theoretically--follow the soil moisture or the groundwater flow velocity. In <u>general terms</u> sorption processes mediate pollutant mass partitioning between the solute and adsorbed phase of a compound. This partitioning forms the concept of mathematical modeling presented in the following section.

3.0 MATHEMATICAL MODELING

3.1 General

Sorption models have been developed by various researchers by assuming the existence of a relationship between adsorbed and dissolved concentration of a compound in the soil matrix.

$$s = s(c) \tag{AD-1}$$

where

- s = adsorbed concentration of pollutant on soil particles; (ug/g soil)
- c = dissolved concentration of pollutant in soil moisture; (ug/mL)

By differentiating the above equation with respect to time, we have

$$\partial s/\partial t = (ds/dc) \cdot (dc/dt)$$
 (AD-2)

in which ds/dc represents the partitioning of the contaminant between the solution and the solids. The temporal adsorbed concentration change of a pollutant may be estimated in two ways: from (1) tabulated values of ds/dc versus c, and (2) algebraic empirical formulas giving s as a function of c, known as adsorption isotherms. The latter approach is the most common in mathematical modeling and has been also employed in SESOIL.

Tabulated values of ds/dc versus c and algebraic formulas are derived from laboratory experiments. From these experiments, the partitioning of solutes between liquid and solid phases in a porous medium is expressed in two-ordinate graphical form, where the mass adsorbed per unit mass of dry solids is plotted against the concentration of the constituent in solution. These graphical relations of s versus c and their equivalent mathematical expressions are known as isotherms because they are derived from experiments conducted at constant (isothermic) temperature. (Freeze and Cherry 1979.)

Laboratory results of adsorption experiments are commonly plotted on double logarithmic paper. For solute species at low or moderate concentrations, straightline graphical relations are commonly obtained over large range of concentrations, a fact that can be expressed by

$$\log s = b \cdot \log c + \log K$$
 (AD-3)

or by

$$s = K \cdot c^b$$
 (AD-4)

For b=1, we have the linear solution

$$ds/dc = K$$
 (AD-5)

where

s = adsorbed concentration of pollutant; (ug/g soil)

- c = dissolved concentration of pollutant; (ug/mL)
- K = adsorption (partitioning) coefficient; (ug/g
 soil)/(ug/mL)
- b = coefficient; (-)

K is a valid presentation of the partitioning between liquid and solids only if the reactions that cause the partitioning are fast and reversible and only if the isotherm is linear. Many contaminants meet this requirement.

Several mathematical descriptions of adsorption isotherms have been presented in the literature. The Freundlich and the Langmuir relations are widely used, and the literature provides numerous examples of experimental data to agree with these isotherms.

3.2 The Freundlich Model

The Freundlich sorptive model is expressed by

$$s = x/m = K \cdot c^{1/n}$$
 (AD-6)

where:

- s = adsorbed concentration of contaminant on soil particles; (ug/g soil)
- x = adsorbed pollutant mass on soil; (ug)
- m = mass of soil; (g)
- K = adsorption (partitioning) coefficient; (ug/g)/
 (ug/mL)
- c = dissolved concentration of pollutant in soil moisture; (ug/mL)
- n = Freundlich equation parameter; (-)

The Freundlich equation is frequently written as $x/m=K \cdot c^n$; therefore, care should be taken to determine the form of equation used before any value of n obtained from the literature is used. Values of 1/n in equation AD-6 are generally found to range from 0.7 to 1.1, although values as low as 0.3 and as high as 1.7 have been reported. (Hassett et al in press.) No methods are available for estimating n; therefore, in the absence of data, it is frequently assumed n=1.0. (Lyman et al 1981.)

The value of the adsorption coefficient K can be measured directly for many inorganic pollutants. For most organic pollutants, sorption (partitioning) occurs mainly on the organic portion of the soil particles. For these organic chemicals, a partitioning coefficient K_{oc} can be determined, which is the ratio of the amount of pollutant associated with the organic carbon of the soil to the pollutant remaining in solution. The adsorption coefficient K is related to K_{oc} by

$$K = K_{oc} \cdot (\% oc) / 100$$
 (AD-7)

where

- K = adsorption coefficient of compound; (ug/g)/(ug/mL)
- K_{oc} = adsorption coefficient of compound on organic carbon (oc) contained in soil; (ug/g oc)/(ug/mL)
- (%oc)= percentage of organic carbon contained in the soil or sediment; (-)

A discussion related to K_{OC} is presented by Miller (1980). Some investigators have related K_{OC} to the K of the soil on organic matter (K_{OM}) rather than on soil-organic carbon.

$$K_{\rm OC} = k \cdot K_{\rm OM} \tag{AD-8}$$

where the value of k has been found in many studies to be approximately equal to 1.724. (Lyman et al 1981.)

Values of K_{OC} may range from <1 to 10^7 . The existence of this chemicalspecific adsorption parameter has an important bearing on assessments of the fate and transport of chemicals in soils and sediments. K_{OC} is commonly used in river models, runoff models, soil and groundwater models where the transport of a specific chemical is being investigated. The degree of adsorption may not only affect a chemical's mobility but may also be an important parameter in fate processes such as volatilization, photolysis, hydrolysis and biodegradation. (Lyman et al 1981.)

Methodologies to estimate K_{oc} and n values are proposed in the literature (Lyman 1981), but it is beyond the scope of this section to outline in detail other researchers' work. Briefly, all of the available methods for K_{oc} include empirical relationships with some other property of the chemical such as water solubility, octanol/water partition coefficient and bioconcentration. The relationships are regression equations obtained from various data sets and are usually expressed in log-log form. Some factors influencing the K and K_{oc} values are temperature, pH

of soil and water, water salinity, particle size distribution and surface area. Regression equations for the estimation of K_{oc} are given in Table AD-1. The uncertainty in values of K_{oc} , K and x/m=s estimated from the equations presented, is related to a number of factors including estimation method errors, uncertainty in input data, variability in environmental factors, errors from extrapolating the linear isotherm equations and errors associated with the assumption of desorption. (Rao and Davidson 1980.)

3.3 The Langmuir Model

The Langmuir isotherm model was developed for single layer adsorption; however, it has been found to closely describe soil adsorption phenomena. (Novotny et al 1978.) It is based on the assumption that maximum adsorption corresponds to a saturated monolayer of solute molecules on the adsorbent surface, that the energy of adsorption is constant, and that there is no transmigration of adsorbate on the surface phase. The Langmuir model (Weber 1972) is described by

$$ds/dt = K_{sw} \cdot (s_e - s)$$

$$s_e = Q^0 \cdot b \cdot c/(1+c)$$
(AD-9)

where

- s = adsorbed concentration of compound on soil particles; (ug/g soil)
- K_{sw} = Langmuir equilibrium soil-water adsorption kinetic coefficient; (s⁻¹)
- se = maximum soil adsorption capacity; (ug/g soil)
- Q^O = number of moles (or mass) of solute adsorbed per unit weight of adsorbent (soil) during maximum saturation of soil; (ug/g soil)
- b = adsorption partition coefficient; (ug/mL)
- t = time; (s)
- c = concentration of pollutant in soil moisture; (ug/mL)

Laboratory studies can provide the values of Q^O and b; however, for most modeling and pollutant transport studies, these variables can be estimated only roughly from a few routinely measured soil parameters. (Krenkel and Novotny 1980.) Several authors have correlated phosphate,

TABLE AD-1

REGRESSION EQUATIONS FOR THE ESTIMATION OF Koc

Eq. No.	Equation ^a	No. ^b	۶ ² C	Chemical Classes Represented
45	log K _{oc} = -0 55 log S + 3 64 (S m mg/L)	106	071	Wide variety, mostly pesticides
46	log K _{oc} = -0 54 log S + 0 44 {S in mole fraction}	10	0 94	Mostly aromatic or polynuclear aromatics, two chlorinated
4-7 ^d	log K _{oc} = -0 557 loq S + 4 277 (S in µ moles/L)	15	0 99	Chlorinated hydrocarbons
48	log K _{oc} = 0 544 log K _{ow} + 1.377	45	0 74	Wide variety, mostly pesticides
49	log K _{oc} = 0 937 log K _{ow} - 0 006	19	0 95	Aromatics, polynuclear aromatics, triazines and diritro aniline herbicides
4 10	log K _{oc} = 1 00 log K _{ow} - 0 21	10	1 00	Mostly aromatic or polynuclear aromatics, two chlorinated
4 11	log K _{oc} = 0 94 log K _{nw} + 0 02	9	r	s Triazines and dinitroaniline herbicides
4-12	log K _{oc} = 1 029 log K _{ow} - 0 18	13	0 91	Varie ty of insecticides, herbicides and fundicides
4-13 ^d	log K _{nc} = 0 524 log K _{ow} + 0 855	30	0 84	Substituted phenylureas and alkyl N-phenylcarbamates
4 14 ^{ci,f}	109 K _{oc} = 0 0067 (P - 45N) + 0 237	29	0 69	Aromatic compounds ureas, 1,3,5 triazines, carbamates, and uracils
4-15	log K _{nc} = 0.681 log BCF(I) + 1.963	13	0 7G	Wide variety, mostly pesticides
4 16	log K _{.c} = 0.681 luq BCF(t) + 1.886	22	0 83	Wide variety, mostly pesticides

a K_{pc}^{-} soil (or sediment) adsorption coefficient S = water solubility K_{pw}^{-} - octanol water partition coefficient BCF(I) = bioconcentration factor from flowing water tests BCF(I) = bioconcentration factor from model ecosystems, P = parachor, N = number of sites in molecule which can participate in the formation of a hydrogen bond

b No = number of chemicals used to obtain regression equation

c (r² = correlation coefficient for regression equation

d Equation originally given in terms of K_{orn}. The relationship K_{orn} = K_{ec}/1.724 was used to rewrite the equation in terms of K_{orn}

e Not available

f Specific chemicals used to obtain requession equation not specified

Source: Lyman et al (1981).

phosphorus and organic chemical sorpitivity to various soil parameters, including Novotny et al (1978), Chesters (1967), and Sanks et al (1976). For example, Novotny et al (1978) proposed

 $Q^{\circ} = a_1 + e_1 \cdot 10^{-pH} + c_1((c_1ay/100)) + d_1((c_100))$ (AD-10)

$$b = a_2 + e_2 \cdot 10^{-pH} + c_2 (\% c \ln y / 100) + d_2 (\% c c / 100)$$
(AD-11)

where

a, e, c, d	=	coefficients; (-)
(%clay)	=	percent of soil clay content; (%)
(%oc)	=	percent of organic carbon content; (%)

3.4 The SESOIL Model Equation

The Langmuir isotherm is often preferred in simulation models because of the equation linearity, as contrasted to the Freundlich equation that is nonlinear and may require numerical trial and error solution algorithms. However, it has been determined that adsorption of most chemicals--and especially of organic chemicals--more nearly approximates the Freundlich isotherm. Because of this fact, more laboratory and other data are available for the Freundlich equation in the literature; therefore, the Freundlich equation is employed and coded in the version of SESOIL accompanying this documentation. Coding of the Langmuir equation into SESOIL is a minor task.

Table AD-2 presents a summary of the Freundlich model and the input parameters required for SESOIL.

TABLE AD-2

THE FREUNDLICH ADSORPTION MODEL IN SESOIL

Equation

$$P(t) = c(t) \cdot \kappa^{1/n} \cdot \rho \cdot d_{i}$$
 (AD-6)

where

$$K = K \text{ (overall adsorption)}$$
or
$$K = K_{oc} \cdot (\% \text{ oc})/100 \quad (AD-7)$$

in which

FORTRAN Variable

	P(t)	adsorbed pollutant mass at time t; ug	PADSU, PADSL
	c(t)	dissolved pollutant concentration at time t; ug/uL	CUS, CUM, etc.
*	К	overall adsorption coefficient; (ug/g)/(ug/mL)) KU. KM, KL
*	K _{oc}	adsorption coefficient on organic carbon (oc); (ug/g oc)/(ug/mL)	кос
*	% ос	organic carbon content of soil; (%)	oc
*	n	Freundlich parameter; (-)	FRN
*	di	soil layer depth i (i=1,N); (cm)	DU, DM, DL
*	ρ	soil specific weight; (g/cm ³)	RS

<u>Note</u>: The user has to input to the data file of the model, <u>either</u> K <u>or</u> the set (K_{oc},%oc), since these sets are mutually exclusive.

* = input variables to this adsorption model.

4.0 DISCUSSION

Equation AD-6 implies that desorption processes between two sequential time steps t and t+l are also explicitly accounted by SESOIL because, as long as c(t+1) is greater than c(t), adsorbed transformation will take place in the soil matrix; otherwise, [when c(t+1) less than c(t)] desorption will take place.

Adsorption and desorption processes can be also studied by writing two separate equations for each phase, namely

$$s_A = K_A \cdot c^{1/nA}$$
 for $c(T+1) \ge c(t)$; adsorption (AD-12)

and

$$s_{D} = K_{D} \cdot c^{1/nD}$$
 for $c(T+1) < c(5)$; desorption (AD-13)

We believe that these mathematical manipulations of the Freundlich isotherm will not substantially improve the output of the model though more input data will be required to make the model run. Input data for equations AD-12 and AD-13 are not always available in the literature; therefore, the sorption equation AD-6 is assumed to be satisfactory for long-term simulations. Equation AD-6 might not be satisfactory for soil-chemical systems subject to extreme fluctuations and for microscale short-term simulations (e.g., fraction of a day) because of reaction time differences (i.e., fast adsorption reaction, slow desorption reaction), though no definite conclusions can be made at this point for the accuracy of the model output.

5.0 NOTATIONS

а,	coefficients; (-)			
Ъ	adsorption partition coefficient, equation AD-4; (ug/mL)			
Ъ	coefficient; (-)			
c	dissolved concentration of pollutant in soil moisture; (ug/mL)			
d	coefficient; (-)			
e	coefficient; (-)			
К	overall sorption (partitioning) coefficient; (ug/g)/(ug/mL)			
к _А	adsorption coefficient, equation AD-12; (ug/g)/(ug/mL)			
К _В	desorption coefficient, equation AD-13; (ug/g)/(ug/mL)			
K _{oc}	adsorption coefficient of compound on organic carbon; (ug/g oc)/(ug/mL)			
Kom	adsorption coefficient of compound on organic matter; (ug/g om)/(ug/mL)			
K _{sw}	Langmuir equilibrium soil-water adsorption kinetic coefficient; (s^{-1})			
m	mass of soil; (g)			
n	Freundlich equation parameter; (-)			
s, s _A	adsorbed concentration of pollutant on soil particles; (ug/mL)			
s _D	desorbed concentration of pollutant on soil particles			
s _e	maximum soil adsorption capacity; (ug/g soil)			
t	time; (s ⁻¹)			
Qo	number of moles of solute adsorbed per unit of weight of adsorbent (soil) during maximum saturation of soil; (ug/g soil)			
x	adsorbed pollutant mass on soil; (ug)			
% clay	percent of soil clay; (%)			
% ос	percentage of organic carbon contained in the soil or sediment; (%)			

6.0 REFERENCES

Chesters, G. (1967). Terminal Report on Phase I of Insecticide Adsorption by Lake Sediments as a Factor Controlling Insecticide Accumulation in Lakes. Office of Water Resour. Res. Grant No. B-008-Wis., University of Wisconsin, Madison.

Chiou, C.T.; L.J. Peters; V.H. Freed (1979). A Physical Concept of Soil-Water Equilibria for Nonionic Organic Compounds. Science 206(16), 831-832.

Dragun, J. (1980). Sediment and Soil Adsorption Isotherm. Draft Report to U.S. Environmental Protection Agency, Office of Pesticides and Toxic Substances, Washington, D.C.

Ellis, B.G. and B.D. Knezek (1972). Adsorption Reactions of Nicronutrients in Soils, in J.J. Mortvedt, P.M. Giordano, and W.L. Lindsay (eds.). Micronutrients in Agriculture. Madison: Soil Science Society of America, 59-78.

Fairbridge, R.W. and C.W. Finkl, Jr., eds. (1979). The Encyclopedia of Soil Science, Part 1. Stroudsburg, PA: Dowden, Hutchinson & Ross, Inc. 646 pp.

Freeze, R.A. and J.A. Cherry (1979). Groundwater. Englewood Cliffs, NJ: Prentice-Hall, Inc. 604 pp.

Hassett, J.J.; J.C. Means; W.L. Banwart; S.G. Wood; S. Ali and A. Khan (in press). Sorption of Dibenzothiophene by Soils and Sediments. J. Environ. Qual.

Keeney, D.R. and R.E. Wildung (1977). Chemical Properties in Soils, in F.J. Stevenson and L.F. Elliot (eds.). Soils for Management of Organic Wastes and Waste Waters. Madison: American Society of Agronomy, 379-402.

Krenkel, P.A. and V. Novotny (1980). Water Quality Management. New York, NY: Academic Press, Inc. 672 pp.

Lyman, W., et al (1981) Research and Development Methods for Estimating Physicochemical Properties of Organic Compounds of Environmental Concern. Prepared by Arthur D. Little, Inc., Phase II Final Report for U.S. Army Medical Bioengineering Research and Development Laboratory, Fort Detrick, Maryland; McGraw-Hill Book Company, New York.

Miller, S. (1980). Adsorption on Carbon: Theoretical Considerations. Environmental Science and Technology, 14(8), 910-914. Novotny, V.; M.A. Chin; and H. Tran (1978). Description and Calibration of a Pollutant Loading Model--LANDRUN. Draft Final Report Volume 4. Chesters, G. and Diamantis, J., eds. International Joint Commission, Menomonee River, Pilot Watershed Study. Grant No. R005142. Madison, WI: University of Wisconsin-Madison.

Novotny, V.; H. Tran; G.V. Simsiman; and G. Chesters (1978). Mathematical Modeling of Land Runoff Contaminated by Phosphorus. J. Water Pollut. Control Fed. 50(1), 101-112.

Rao, P.S.C. and J.M. Davidson (in press, 1979/1980). Estimation of Pesticide Retention and Transformation Parameters Required in Non-Point Source Pollution Models, in Environmental Impact of Non-Point Source Pollution, Ann Arbor Science Publishers, Inc., Ann Arbor, Michigan.

Sanks, R.L.; J.M. LaPlante; and E.F. Gloyna (1976). Survey--Suitability of Clay Beds for Storage. Technical Report EHE-76-040CRWR-128, Center for Research in Water Resources, University of Texas, Austin.

Weber, W.J. (1972). Physicochemical Processes for Water Quality Control. Wiley (Interscience), New York.

DE - degradation

APPENDIX DE*

DEGRADATION AND DECAY

		Page
1.0	INTRODUCTION	DE-3
2.0	BACKGROUND	DE-4
3.0	MATHEMATICAL ANALYSIS	DE-5
4.0	REFERENCES	DE-7

*Contribution by K. Scow

Arthur D Little Inc

•

.

1.0 INTRODUCTION

Biodegradation is an important environmental process causing the breakdown of organic compounds. It is a significant loss mechanism in soil and aquatic systems contributing the mineralization process of organic compounds, that is their conversion to inorganic substances.

Biodegradation or decay (designated in the following paragraphs as degradation) of pollutants in the soil is a complex phenomenon involving a variety of mechanics. The quantification of these mechanics and the effects of environmental factors on the degradation rates of pollutants is an active research area.

This appendix is <u>not</u> intended to review and describe the biodegradation or decay process of chemical compounds in soil systems; rather it provides an overall background information on the nature of biodegradation and the way this chemical process is modeled. Alternative modeling approaches are also possible by SESOIL.

Detailed information regarding the biodegradation process, constant estimates and data availability has been presented in the literature by K. Scow (1981).

2.0 BACKGROUND

Several definitions of biodegradation have been proposed in the literature such as primary, ultimate and acceptable biodegradation. In this chapter biodegradation is defined as the primary degradation of organic compounds, namely any structural transformation in the parent compound that changes its identity.

Microorganisms are the most significant group of organisms involved in biodegradation. Although higher organisms, both plant and animal, are capable of metabolizing numerous compounds, microorganisms may convert to inorganic substances (H2O, CO₂, mineral salts) many organic molecules that higher organisms are unable to metabolize.

The most important natural habitats for microorganisms in relation to environmental biodegradation are soil and water. In both environments, microorganisms are essentially aquatic organisms, and certain characteristics are shared by all species (Stotzky 1979).

Soil environments have a diverse microbial population, because they offer a large variety of food sources and habitats (Hamaker 1972). The mobility of microorganisms is decreased in soil, however, because of physical barriers (such as clay aggregates) and particularly distribution of supportive microhabitats (Scow 1981).

The parameters that influence the rate of biodegradation can be grouped into two general categories:

- those that determine the availability and concentration of the compound to be degraded or that affect the microbial population size and activity (eg. population interactions) and
- (2) those that directly control the reaction rate itself (eg. population size, temperature).

Both direct and indirect variables can be classified as substraterelated, organism-related or environmental-related. Because of considerable variation in species, habitat, and chemical environment, not all variables will influence all situations in the same way. For example, low pH is likely to decrease metabolic activity in most bacteria but it favors activity in fungi (Scow 1981).

Important environmental and other parameters affecting biodegradation can be pH, temperature, moisture content in soil, adsorption, oxygen pressure (aerobic, anaerobic reactions), salinity, solute concentration of the substance in soil etc.; however, all these parameters are lumped in one quantified constant describing total loss over time. This is the biodegradation rate constant discussed in the following section.

3.0 MATHEMATICAL ANALYSIS

The process of biological degradation of pollutants is limited to microbial metabolism of the compound under aerobic conditions. Degradation is defined as any structural alteration in the parent compound at which point it disappears from the soil. Metabolic products as new chemicals would re-enter a new model run as biotic input to soil.

The equation describing biodegradation requires input of a first-order rate constant determined for the particular pollutant being modelled. It should be measured in a soil culture test under conditions similar to the site being simulated and complying with state-of-the-art technology.

Pollutant losses in soil moisture due to biological degradation is estimated by:

Although soil moisture content, temperature and other environmental parameters strongly influence biological activity, the present SESOIL routine does not describe their influence on the rate of degradation. Expansion of the equation to account for these factors is possible assuming that enough general or chemical-specific data are available to define the limits they set on degradation rates.

Equation (DE-1) has a general application in all soil zones of the soil compartment (see Appendix PT), and is employed as a first order reaction, n=1 to express loss/transformation of pollutants from the moisture content of the unsaturated soil zone.

The total pollutant mass decayed over a short period Δt in a soil compartment (layer i) can be finally expressed by

$$P_{DE} = K_{DE} \cdot \Theta \cdot c \cdot d_i \cdot \Delta t$$

where

^PDE = decayed chemical mass within Δt ; (ug/cm²) K_{DF} = biodegradation rate; (day⁻¹)

- Θ = soil moisture content; (fraction)
- c = solute concentration of pollutant in soil moisture; (ug/mL)
- d_i = soil layer depth; (cm)
- Δ_t = time step; (day)

4.0 REFERENCES

Hamaker, J.W. (1972); Decomposition: Quantitative Aspects. In: Organic Chemicals in the Soil Environment, Vol. 1, C.A.I. Goring and J.W. Hamaker (eds.), Marcel Dekker, New York.

Lyman, W., et al (1981); Research and Development Methods for Estimating Physicochemical Properties of Organic Compounds of Environmental Concern. Prepared by Arthur D. Little, Inc., Phase II. Final Report for U.S. Army Medical Bioengineering Research and Development Laboratory, Fort Detrick, Maryland; McGraw-Hill Book Company, New York.

Scow, K. (1981); see Chapter 9, in W. Lyman (1981).

Stotzky, G. (1974); Activity, Ecology, and Population Dynamics of Microorganisms in Soil. In: Microbial Ecology, A.I. Laskin and H. Lechevalier (eds.), CRC Press, Cleveland, Ohio.

HD - hydrolysis

APPENDIX HD*

HYDROLYSIS OF ORGANIC COMPOUNDS

		Page
1.0	INTRODUCTION	HI:-3
2.0	BACKGROUND	HD-4
3.0	MATHEMATICAL MODELING 3.1 Governing Equations 3.2 Numerical Example 3.3 Input Parameters	HD-8 HD-8 HD-10 HD-11
4.0	DISCUSSION	HD-13
5.0	REFERENCES	FD-14
Table	HD-1 Input Parameters to the Hydrolysis Routine	HD-12
Figur	e HD-l Typical Seasonal Variation of Soil Temperature Profile	HD-7

*Contribution by W. Lyman, Ph.D.

•

1.0 INTRODUCTION

Organic chemicals may undergo a variety of reactions with water. One family of reactions that leads to an ultimate transformation of the organic molecule is called hydrolysis. Some chemical classes (e.g., hydrocarbons) are known to be generally resistant to hydrolysis while others (e.g., alkyl halides, carbamates) are potentially susceptible. The relative importance of the hydrolysis degradation pathway is enhanced in the soil/groundwater compartment--especially at depths of approximately one meter below ground surface--since other degradation and loss pathways (e.g., photolysis, biodegradation, and volatilization) are eliminated or minimized. Other reactions involving organic molecules and water do occur, but these processes are not included in this chapter. These other reactions include reversible reactions (e.g., acid-base reactions and hydration) and additional reactions which require reaction conditions that are unlikely to occur in the environment.

The hydrolysis subroutine of SESOIL allows the user to simulate neutral, base-catalyzed and/or acid- or base-catalyzed hydrolysis reactions. The current hydrolysis subroutine is based upon the assumption that both dissolved and adsorbed organics are equally susceptible to hydrolysis. Existing experimental data regarding this assumption is contradictory, although some data show that rates of hydrolysis are not significantly affected by the presence of moderate amounts of suspended solids in aqueous systems.

The following sections provide additional background on hydrolysis reactions, the mathematical equations involved and the limitations the user should be aware of. For additional information, model users are referred to the work of Harris (1981), since much of the information presented in this appendix is a summary of a more detailed investigation presented by this author.

This appendix is <u>not</u> intended to thoroughly describe the hydrolysis process of organic compounds in soils; rather it provides background information on the nature of hydrolysis and the way this chemical process is modeled. Alternative modeling approaches are also possible by SESOIL.

2.0 BACKGROUND

Hydrolysis is a chemical transformation process in which an organic molecule, RX, reacts with water, forming a new molecule. By the normal definition of hydrolysis this involves the formation of a new carbonoxygen bond and the cleaving of the carbon-X bond in the original molecule:

$$R-X \xrightarrow{H_2O} R-OH + X^- + H^+$$
 (HD-1)

Other reactions involving water may result in the elimination of a hydrogen (H) and a leaving group (X) from neighboring carbons. This mechanism is apparently favored, for example, in the hydrolysis of Nemagon[®]:



Equations (HD-3) through (HD-8) illustrate some additional hydrolysis ractions, in which the identity of the original molecule is destroyed, although complete degradation is not implied. The initial products of hydrolysis may show wide variability in their susceptibility to subsequent degradation.

Certain types of hydrolysis reactions may be catalyzed by the presence of hydrogen ions, H^+ , and/or hydroxide ions, OH^- . The concentration of these species in water is obtained from the pH of the water. Under certain conditions the model user may desire to account for these acid-or base-catalyzed reactions by specifying second-order rate constants for the catalyzed reactions.

The rate of hydrolysis of various organic chemicals, under environmental conditions, can range over more than 14 orders of magnitude, with associated half-lives (time for one half of the material to disappear) as low as a few seconds and as high as 10^6 years. Since the time resolution of SESOIL are on the order of a month to a few years, this subroutine should <u>not</u> be used for chemicals that hydrolyze extremely fast (e.g., $t_{1/2}$ less than 1 day) and need not be considered in modeling chemicals that hydrolyze extremely slowly (e.g., $t_{1/2}$ greater than 100 yrs).

$$CH_{3}P(OCH_{3})_{2} \xrightarrow{H_{2}O} CH_{3}POCH_{3} + CH_{3}OH$$
(HD-3)

phosphonic acid diester phosphonic alcohol acid monoester

$$\begin{array}{c} O \\ \parallel \\ CH_3 OCNHC_6 H_5 \xrightarrow{H_2 O} CH_3 OH + CO_2 + NH_2 C_6 H_5 \\ carbamate \\ alcohol \\ amine \end{array}$$
(HD-4)

$$\begin{array}{c} & & \stackrel{\text{H}_2\text{O}}{\longrightarrow} & \text{HOCH}_2\text{CH}_2\text{OH} \\ \text{epoxide} & & & \text{glycol} \end{array}$$
(HD-5)



$$\begin{array}{c} CH_{3}CH_{2}CH_{2}CH_{3} \xrightarrow{H_{2}O} CH_{3}CH_{2}CH_{2}CH_{-}CH_{3} + Br^{-} + H^{+} \\ \downarrow \\ Br & OH \\ alkyl halide & alcohol & anion \end{array}$$
(HD-7)



٠

Source: Lyman et al (1981).

.

Rates of hydrolysis are usually measured under controlled laboratory conditions that include constant temperature and may include the use of buffered solutions and solvating agents. Extrapolation of such data to environmental conditions involves considerable uncertainty, particularly with regard to the influence of temperature, solution ionic strength, adsorption on soils, and the possibility of catalytic action by either dissolved material (e.g., heavy metal cations) or solid surfaces. If uncorrected laboratory data are used in SESOIL--and this may often be the only choice--then model predicted pollutant concentrations should be considered as rough approximations only. Harris (1981) provides instructions for estimating hydrolysis rate constants for certain classes of chemicals.

It may be possible for a user to employ rate constants (k) that have been corrected for the difference between the temperature used for the measured (or estimated) value, and the temperature of the soil system. For example, if the activation energy (E_A) for the reaction is known, then extrapolation from a reported value of k_1 , at temperature T_1 , to an adjusted value of k_2 at temperature T_2 is given by:

$$k_2 = k_1 \exp - \left[\frac{E_A}{R}\left(\frac{1}{T_2} - \frac{1}{T_1}\right)\right]$$
 (HD-9)

where:

- k1 = given hydrolysis rate constant of compound at T1; (sec⁻¹)
- E_A = activation energy of reaction; (cal/mol)
- R = gas constant of compound; (=1.987 cal/mol·K)
- T_1, T_2 = temperature; (^oKelvin)

If a measured value of E_A is not available, a value of 17,500 cal/mole may be assumed (Harris 1981).

Soil temperature is generally a complex function of various parameters, such as geographic location, soil nature (including water content), soil depth, air temperature, and heat flow from below. For SESOIL, the diurnal variations in soil temperature (important for approximate the top meter of soil) can be ignored because of the longer time scales used by the model. Seasonal and depth variations, however, may be important and the user should seek to correct the estimated hydrolysis rate constants to the appropriate temperature. A typical soil-temperature profile as it might vary from season to season in a frost-free region is shown in Figure HD-1.



Source: Hillel 1980.

FIGURE HD-1

TYPICAL SEASONAL VARIATION OF SOIL TEMPERATURE PROFILE

3.0 MATHEMATICAL MODELING

3.1 Governing Equations

It is generally observed that hydrolysis of organic chemicals in water follows a first-order kinetic law, that is, the rate of its disappearance is proportional to the concentration of the compound:

$$-d[RX]/dt = k_T[RX]$$
(HD-10)

or by approximating the differential to a difference, by:

$$-\Delta[RX]/\Delta t = k_T[RX]$$
(HD-11)

where:

[]	=	concentration of organic compound RX; (mol/mL)
d[] ≃	∆[] =	differential
^k т	=	hydrolysis rate constant; (day ⁻¹)
∆t	=	time, step in units compatible with $k_{T}^{}$; (days)

The rate constant in equation (HD-10) is an oversimplification for most organic hydrolysis reactions. It is often appropriate to consider ${\bf k}_{\rm T}$ as having contributions from neutral, acid-catalyzed and base-catalyzed reactions:

$$k_{\rm T} = k_{\rm O} + k_{\rm H} [{\rm H}^+] + k_{\rm OH} [{\rm OH}^-]$$
 (HD-12)

where:

k _O	=	rate constant for neutral hydrolysis; (days ⁻¹)
k _{OH}	=	rate constant for base-catalyzed hydrolysis; (days ⁻¹ mol ⁻¹ ·L)
k _H	=	rate constant for acid-catalyzed hydrolysis; $(days^{-1}mol^{-1}\cdot L)$
[H ⁺]	=	hydrogen ion concentration = 10^{-pH} ; (mol/L)
[OH-] =	hydroxyl ion concentration = $10P^{H-14}$; (mol/L)

The neutral rate constant (k_0) has units of $(time^{-1})$, while k_{OH} and k_H have units of $(time^{-1}) \cdot (concentration^{-1})$. The user of the model must determine and input the values of k_0 , k_{OH} and k_H ; if no values are given a value of zero will be assumed. If k_H and k_{OH} are non-zero, then the pH of the soil water must also be input. This may require that some average pH and temperature be considered since SESOIL does not predict the variation in these parameters in the soil column.

For the SESOIL simulations, it is assumed that dissolved and adsorbed organic species are <u>equally</u> susceptible to hydrolysis, therefore, the total pollutant mass (M) degrated by hydrolysis during a simulation time step will be:

$$M_{total} = M_{H_20} + M_{soil}$$
(HD-13)

By considering equations (HD-11) and (HD-13), we have:

$$M_{H_{2}0} = k_{T} [RX]_{H_{2}0} \cdot \Delta t \cdot V_{H_{2}0}$$
(HD-14)

$$M_{soil} = k_T [RX]_{soil} \cdot \Delta t \cdot V_{soil}$$
(HD-15)

or by employing the SESOIL designations and symbols:

$$M_{H_2O} = k_T \cdot c(t) \cdot \Delta t \cdot \theta(t) \cdot d_i \cdot AR \qquad (HD-16)$$

$$M_{soil} = k_{T} \cdot s(t) \cdot \Delta t \cdot \rho \cdot d_{i} \cdot AR \qquad (HD-17)$$

where:

M _{H2} O	-	pollutant mass in soil moisture degrated by hydrolysis; (ug)
M _{soil}	=	pollutant mass on soil particles degrated by hydrolysis; (ug)
k _T	=	hydrolysis rate constant; (day ⁻¹)
[RX] _{H20} =c(t)	=	dissolved concentration of pollutant at time t; (ug/ml)
[RX] _{soil} =s(t)	=	adsorbed concentration of pollutant at time t; (ug/g of soil)
t	=	time; (days)

∆t	= simulation time step; (days)
θ(t)	= soil moisture at time t; (fraction)
di	= depth of soil layer i; (cm)
ρ	= soil density; (g/cm ³)
v _{H2} 0	= volume of water in the compartment; (cm^3)
V _{soil}	= volume of soil in the compartment; (cm^3)
AR	= surface area of soil compartment; (cm ²)

3.2 Numerical Example

Calculate the mass of ethyl acetate hydrolyzed after one month in contact with a wet soil given: (1) an initial concentration of 100 ug/ml; (2) a soil temperature of 10° C; (3) hydrolysis rate constants at 25° C of $1.1 \times 10^{-4} \text{ L/mol} \cdot \text{s}$ for k_{H} , $1.5 \times 10^{-10} \text{ s}^{-1}$ for k_0 and $1.1 \times 10^{-4} \text{ L/mol} \cdot \text{s}$ for k_{OH} ; (4) a soil moisture pH of 8.0; (4) no pollutant adsorbed on soil; (5) a soil moisture volume of 10 mL.

(1) To correct the rate constants from $25^{\circ}C$ (298 K) to $10^{\circ}C$ (283 K), use equation (HD-13) with an assumed value of 17,500 cal/mol for E_{A} .

 $k_2/k_1 = \exp \left[-\frac{17,500}{1.987}\left(\frac{1}{283} - \frac{1}{298}\right)\right] = 0.209$

Thus, $k_{\rm H} (10^{\circ}{\rm C}) = 0.209 \cdot 1.1 \times 10^{-4} = 2.3 \times 10^{-5} \text{ L/mol} \cdot \text{s}$ $k_0 (10^{\circ}{\rm C}) = 0.209 \cdot 1.5 \times 10^{-10} = 3.1 \times 10^{-11} \text{ s}^{-1}$ $k_{\rm OH} (10^{\circ}{\rm C}) = 0.209 \cdot 1.1 \times 10^{-1} = 2.3 \times 10^{-2} \text{ L/mol} \cdot \text{s}$

These corrected values (as well as pH) would be the user's input into SESOIL which would then carry out the following calculations.

(2) From equation (HD-12):

 $k_T = 3.1 \times 10^{-11} + 2.3 \times 10^{-5} \cdot 10^{-8} + 2.3 \times 10^{-2} \cdot 10^{-(8-14)} =$ = 2.3 × 10⁻⁸ s⁻¹ = 1.99 × 10⁻³ day⁻¹

(3) From equation (HD-14) with $\Delta t = 30$ days:

$$M_{H_2O}$$
 = Mass hydrolyzed=1.99x10⁻³day⁻¹·100ug/mL·30days·10mL=59.7ug

3.3 Input Parameters

The input parameters to the hydrolysis routine are presented in Table HD-1. The chemical specific parameters (k, $k_{\rm H}$, $k_{\rm OH}$) can be obtained from handbooks (e.g., Lyman 1981). Site specific parameters must satisfy the needs of a site specific simulation and study.

TABLE HD-1

INPUT PARAMETERS TO THE HYDROLYSIS ROUTINE

	Parameter	Units	FORTRAN Variable
^k O	Neutral hydrolysis rate constant	day ⁻¹	KNH
k _H	Acid catalyzed rate constant	day ^{-l.mol-l.} L	КАН
k _{OH}	Base catalyzed rate	day ⁻¹ ·mol ⁻¹ ·L	KBH
ρ	Soil density	g/cm ³	RS
di	Compartment depths	cm	DU, DM, Z
A	Compartment area	cm ²	AR
рН	pH of soil	-	рн
4.0 DISCUSSION

It is worth emphasizing again the assumptions made for the development of the hydrolysis subroutine in SESOIL.

- First, the calculations assume that dissolved <u>and</u> adsorbed organic species are equally susceptible to hydrolysis. If subsequent experiments show this to be a serious error, then the computer code would have to be modified to account only for an "available" fraction of pollutant.
- Second, the calculations consider only simple hydrolysis reactions (neutral, acid-catalyzed and base-catalyzed) and do not follow for general catalysis by other diverse acids, bases, cations or solids. The model does not consider effects of ionic strength or the presence of other dissolved organics, nor the variation of pH in the soil compartment, or temperature along the soil column.
- Third, the model assumes that the hydrolysis rate constants that are entered have been previously corrected to the correct soil temperature for the simulation.

The above assumptions imply that the model user should <u>not</u> "blindly" use just any laboratory-measured (or estimated) rate constants, but should give some considerations to the soil system modeled and the corrections that might have to be undertaken for the input data.

5.0 REFERENCES

Harris, J. (1981), in Lyman, W. (1981); Chapter 7.

Hillel (1980) Fundamentals of Soil Physics, Academic Press, Ann Arbor, Michigan.

Lyman, W., et al (1981) Research and Development Methods for Estimating Physicochemical Properties of Organic Compounds of Environmental Concern. Prepared by Arthur D. Little, Inc., Phase II Final Report for U.S. Army Medical Bioengineering Research and Development Laboratory, Fort Detrick, Maryland; McGraw-Hill Book Company, New York. CE - cation exchange

APPENDIX CE*

CATION EXCHANGE

1.0	INTRODUCTION	CE-3
2.0	BACKGROUND	CE-4
3.0	MATHEMATICAL MODELING 3.1 Governing Equations 3.2 Input Parameters	CE-6 CE-6 CE-6
4.0	DISCUSSION	CE-8
5.0	REFERENCES	CE-9
Table	CE-1 Input Parameters to Cation Exchange Routine	CE-7

*Contribution by W. Lyman, Ph.D.

1.0 INTRODUCTION

Cation exchange is a mechanism by which cations in solution may be adsorbed on soil and thus removed from the mobile aqueous phase; charged species (cations) may exchange with various minerals and/or other soil constituents. The process of cation exchange is complex; for several ions--particularly metal cations--and under certain conditions, the cation exchange capacity of the soil is strongly correlated with the adsorption of the ion, as discussed in the next section.

The cation exchange subroutine of SESOIL is designed as an optional way of considering adsorption. Therefore, if this routine is used, the adsorption equation in SESOIL should <u>not</u> be used unless the user has selected the model inputs (cation exchange capacity and adsorption parameters) in a way to avoid any "double counting" for adsorption.

It is incumbent upon the user to insure that cation exchange is the predominant adsorption mechanism at the site being modeled. This may require considerations of the leachate characteristics (pH, ionic strength, concentration of major cations), metal speciation and soil characteristics.

This appendix is <u>not</u> intended to thoroughly describe the cation exchange chemistry of species in soils; rather it provides background information on the nature of cation exchange and the way this chemical process is modeled. Alternative modeling approaches are also possible by SESOIL.

2.0 BACKGROUND

The cation exchange capacity (CEC) of a soil is usually defined as the number of milliequivalents (m.e.) of the ion that can be exchanged (adsorbed) per 100 g (dry weight) of soil. The process is viewed as an exchange with some other cation that initially occupies the adsorption site on the solid. With clays, the exchanged ion is often calcium:

$$M^{++}$$
 + [Clay] · Ca \rightleftharpoons Ca⁺⁺ + [Clay] · M (CE-1)

Among soils, clays tend to have the highest CEC values, although materials other than clay may contribute. Brady (1974) lists, for example, the following typical CEC values (m.e./100 g) for various materials: humus, 200; monomorillonite, 100; vermiculite, 150; hydrous mica and chlorites, 30; kaolimite, 8; and hydrous oxides, 4. He also provides data showing a range of CEC values in soils of 2-60 m.e./100g. This range is somewhat larger than the 2-37 m.e./100 g range associated with 11 soils studied by Fuller (1978), which in turn is larger than the 0-4.2 m.e./100 g range given by Wang et al (1975) for 30 soils in Rhode Island. The cation exchange capacity of a soil is not an invariable property of the soil; in most soils the exchange capacity increases with pH (Brady 1974).

The actual cation exchange reation--equation (CM-1)--is probably fast and is also reversible. One cation with a high affinity for an exchange site may displace the cation previously at the site if the latter has a lower affinity. This is called the "mass action" effect. The relative strengths of soil-cation interactions are seldom available from the literature; therefore, they are not modeled in SESOIL. Furthermore, very high concentrations of certain cations (e.g., the common Ca⁺⁺, Na⁺, Fe^{++} , and K^+) may so overwhelm the exchange capacity of a soil that low concentrations of other cations--including those with higher affin-Landfill leachates and ities--will not be significantly adsorbed. aqueous industrial wastes often have very high concentrations of total dissolved solids (including Na⁺, Ca⁺⁺, etc.) and one would expect the major cations in these wastes to effectively block the adsorption (by cation exchange) of other trace cations for some significant time and distance in the migration through the soil/groundwater system.

Without some laboratory data, it is difficult to predict when cation exchange will be important and for what cations. This is not to say that some situations have not been modeled. They have, but the models usually require the use of equilibrium constants, the consideration of the "mass action" and pH effects, and also require detailed knowledge about the nature of the soil and the constitutents (e.g., other major cations and anions) of the leachate. This information will seldom be easy to assemble. Griffin and Shimp (1978) give examples of cases in which cation exchange 1s likely to be important for Pb, Cu, Zn, Cd, Na, K, Mg, and NH₄. For the heavy metals, cation exchange is unlikely to play a significant role if the leachate pH is above 7; above this pH, precipitation may be the controlling factor in mobility. It should be mentioned that the present version of SESOIL does not specifically consider the process of fixation for pollutants in the leachate. Fixation is a process whereby the pollutant (e.g., a metal cation) diffuses into the small pores or interstitial layers of the soil matrix and, following some form of chemisorption or bonding, becomes permanently bound (fixed) to the soil. Fixation can be a very important removal process for heavy metals in some situations. The fixation process might be simulated for SESOIL, however, by appropriate use of the adsorption or cation exchange routines that are available.

3.0 MATHEMATICAL MODELING

3.1 Governing Equations

The SESOIL subroutine for cation exchange is optional and when used will presumably obviate the need for any other adsorption subroutine. For modeling purposes, the process is considered to be "irreversible." The calculation of the pollutant mass immobilized by cation exchange is given by:

$$MECM = a \cdot CEC \cdot MWT/VAL \qquad (CE-2)$$

where:

MECM	=	maximum pollutant mass immobilized (cation ex- changed) by the soil; (ug/g soil)
а	=	10.0; units coefficient; (-)
CEC	=	<pre>cation exchange capacity of soil; (m.e./100g of dry wt. soil)</pre>
MWT	=	pollutant molecular (or atomic) weight; (g/mol)
VAL	=	valence of cation; (-)

Example:

Pollutant: Pb⁺⁺ (MWT = 207, VAL=2) Soil with: CEC = 3 m.e./100 g (of dry soil) MCEC = 10.3.207/2 = 3100 ug/g soil

Once the maximum capacity of the soil has been reached in a given soil element, SESOIL will assume that no further adsorption takes place unless another adsorption subroutine has also been employed (e.g., for fixation modeling). Cation exchange is assumed to be instantaneous; therefore, it is modeled as proceeding to completion before the start of all other processes. These assumptions must be made in order to model a cation exchange process that is general to many situations. When more specific data are available, modifications to this routine can be made.

3.2 Input Parameters

The input parameters to the cation exchange routine are presented in Table CE-1. Data are available from soil and chemical handbooks.

TABLE CE-1

INPUT PARAMETERS TO CATION EXCHANGE ROUTINE

Parameter	Units	FORTRAN Variable
Cation exchange capacity of the soil	(ug/100 g dry soil)	CEC
Molecular or atomic weight of pollutan	t (g/mol)	MVT
Valance of pollutant	(-)	VAL

4.0 DISCUSSION

It is incumbent upon the user to determine when it will be appropriate to use the cation exchange subroutine. This may require, as mentioned above, a consideration of the "speciation" of the pollutant, the soil pH, the presence of other cations, and the nature of the soil. For most metals the speciation can be predicted with models such as REDEQL (Ingle et al 1980). In general, no other adsorption routine should be used when the cation exchange routine is employed. The calculations in SESOIL assume a fast, irreversible removal of the cations from solution, no competition with other ions, and a higher priority than any other process in competition for the cation in solution.

5.0 REFERENCES

Brady, N.C., (1974), The Nature and Properties of Soil, 8th ed., Macmillan Publishing Co., Inc., New York.

Fuller, W.H., (1978), "Investigation of Landfill Leachate Pollutant Attenuation by Soils," Report No. EPA-600/2-78-158, U.S. Environmental Protection Agency, Cincinnati, Ohio. (NTIS Report No. PB-286 995.)

Griffin, R.A. and N. F. Shimp, (1978), "Attenuation of Pollutants in Municipal Landfill Leachate by Clay Minerals," Report No. EPA-600/2-78-157, U.S. Environmental Protection Agency, Cincinnati, Ohio.

Ingle, S.E.; J.A. Keniston; D.W. Schultz (1980); "REDEQL.EPAK: Aqueous Chemical Equilibrium Computer Program", Report No. EPA-600/S-80-049, U.S. Environmental Protection Agency, Corvallis Environmental Research Laboratory, Corvallis, Oregon.

Wang, M.C. and V.A. Nacci, (1975), "Movement of Trace Metals with Percolating Water," Report on Project No. A-052-RI to the Office of Water Research and Technology (U.S. Department of the Interior), Washington, D.C. (NTIS Report No. PB-246-104.) CM - complexation

APPENDIX CM

*

COMPLEXATION OF METALS IN SOLUTION BY ORGANIC LIGANDS

		Page
1.0	INTRODUCTION	СМ-3
2.0	BACKGROUND	CM-4
3.0	MATHEMATICAL MODELING 3.1 Mathematical Expressions 3.2 Input Parameters	СМ-7 СМ-7 СМ-9
4.0	DISCUSSION	CM-13
5.0	REFERENCES	CM-14

TABLES

CM-1	Equation Describing the Complexation Concept in SESOIL	CM-10
CM-2	Input Parameters for Complexation Algorithms	CM-12

* Contribution from W. Lyman, Ph.D.

•

1.0 INTRODUCTION

Complexation (or chelation) is the process by which metal ions and organic or other nonmetallic molecules (called ligands) can combine to form stable metal-ligand complexes.

It is well known that a number of organic materials, of both natural and anthropogenic origin, are capable of complexing with several heavy metals including (but not limited to) Cu, Pb, Fe, Zn, Cd and Ag. The complex that is formed will generally prevent the metal from undergoing other reactions or interactions that the free metal cation would.

The current level of understanding of this process is not very advanced (less so for interactions in groundwater and leachates than in surface waters) and the available information has not been shown to be particularly useful to quantitative chemical modeling (Jenne 1979; McCrady and Chapman 1979). However, complexation can have a significant effect on the behavior of metals and soils; therefore, a simplified representation of the complexation process is incorporated in SESOIL. As the process (and factors that affect it) becomes better understood and quantifiable, this routine can be improved to reflect the new knowledge.

The current complexation subroutine of SESOIL allows the user to consider only a process in which a metal ion in solution is complexed by an organic ligand resulting in the formation of a soluble complex. It is incumbent upon the model user to determine if such a process is likely in the situation being modeled and to supply the appropriate stability constant, ligand concentration and mole ratio of metal to ligand in the complex.

This appendix is <u>not</u> intended to fully describe the complexation process of metals in solution by organic ligands; rather, it provides background information on the nature of complexation and one way this chemical process is modeled. Alternative modeling approaches in SESOIL <u>are</u> possible.

2.0 BACKGROUND

It is believed that metals bind to humic and fulvic acids through three possible types of bonds as shown in equations (CM-1), (CM-2) and (CM-3). (Giesy and Alberts 1981.)

$$R - C - 0^{-} + M^{2+} + H_2 0 \longrightarrow R - C - 0 - M - 0H + H^+$$
 (CM-1)

$$c - c - o^{-} + c \cdot oH + M^{2+} = c - c - o - M - o - c + H^{+} (CM-3)$$

Similar reactions may be involved in complexation with low molecular weight carboxylic acids such as those commonly found in landfill leachate. A general equation for this type of complexation reaction can be written as:

$$M^{x^+} + bL^{y^-} \longrightarrow (ML_b)^{x-by}$$
 (CM-4)

where one mole of a metal cation (M) with a change of x+ reacts with b moles of an organic ligand (L) with a charge of y- to yield one mole of the organo-metal complex (ML_b) with charge x-by. The stability (or dissociation) constant, K, for the complex is:

$$K = \frac{[MLb^{X-by}]}{[M^{X+}][L^{Y-}]^{b}}$$
(CM-5)

where the brackets indicate molar concentration of the species. Values of K are a function of pH, ionic strength, and temperature. In some cases, the metal and ligand will form 1:1 complexes (in which case b=1) but certain ligands such as the humic and fulvic acids may have more than one effective binding site per equivalent.

A significant fraction of the humic and fulvic acids in surface waters is probably present in macromolecular or colloidal form (Gachter et al 1978; Giesy et al 1978) and in soils this material would likely be associated with the solid matrix of the soil. Low concentrations of these materials will be in solution although complexation with metals may cause some precipitation (Saar and Weber 1980) and adsorption of a soluble complex by soil is certainly possible. The complexation subroutine in SESOIL is <u>only</u> intended to model the situation where a soluble ligand reacts with a metal ion to form a soluble, nonadsorbable complex. Complexation by materials in the solid phase of the soil may be modeled (in some cases) by the cation exchange subroutine.

Considerably more information appears to be available on metal complexation in surface waters than in ground waters. Reports covering complexation in soils, sediments, groundwaters and/or leachates that may be useful to the reader include the works of Saar and Weber (1980), Kuo and Baker (1980), Davis and Leckie (1978), Knox and Jones (1979), Nriagu and Coker (1980), Oakley et al (1980), Oakley et al (1981), Pagenkopf (1978), Griffin and Shimp (1978), and Khalid et al (1977); papers by Fuller et al (1980) and O'Donnell et al (1980) report, in part, on the inverse correlation between metal mobility in soils and the total organic carbon content of the carrier fluid (leachate). The organic material in the leachate presumably contains low molecular weight carboxylic acids and higher molecular weight acids, including humic and fulvic acids, which complex the metals to form a complex that is less mobile than the free metal due to increased adsorption of the complex and/or to hindered movement of large molecules in the small pores of the soil. More general infomration on complexation may be found in the works of Geisy and Alberts (1981), Brinkman and Bellama (1978), and Sposito (1981).

In sediments, and presumably in soils, some studies (Pagenkopf 1978; Griffin and Shimp 1978) have shown that metals may be solubilized (desorbed) by the presence of complexing agents. However, just which metals can be solubilized, to what degree, and under what conditions is not predictable. If the user of SESOIL selects both the cation exchange and complexation subroutines, the model will assume that cation exchange takes precedence (i.e., happens first) and that ions involved in cation exchange are unavailable for complexation. (See Appendix CE). Other adsorption processes (e.g., via Langmuir or Freudlich equations) are modeled as being competitive with complexation. There is no clear justification for this order of operations and revisions of the subroutine are desirable as new data and better understanding are obtained.

In summary, agents or ligands can "pull" metals off the soil in certain cases, but when and how this occurs is unknown. Hence, in modeling cation exchange (see Appendix CE), we do not allow solubilization. The other sorption processes are modeled as being fully reversible, and thus complexation and adsorption will be competitive for available pollutant. However, complexation is not modeled as having an active role in the desorption of pollutant. As with cation exchange, competition from other metal ions is an important consideration in complexation. A variety of metal cations may be complexed and a particular metal ion that is complexed may be displaced by a cation (present in equal concentrations) of higher affinity or by a cation of lower affinity if the latter is present in greater concentrations. In landfill leachate, competition with iron for complexation sites may be the most important consideration (Knox and Jones 1979). The present complexation subroutine in SESOIL does not consider such competition.

The complexation reaction is relatively fast compared to the simulation time steps of SESOIL although equilibrium partitioning in some cases may not be achieved for a few days (Oakley et al 1980). Values of the stability constant appear to range from about 10^2 to 10^{16} for the complexation of some common heavy metals (Cn, Cd, Zn, Fe, Co, Ni, Pb, Hg) with humic and fulvic acids (Pagenkopf 1978). Significantly higher values (10^5 to 10^{21}) are associated with some of the manmade chelating agents such as NTA, EDTA, HEDTA and CDTA (Drake et al 1976). As mentioned previously, the values of K are a function of pH and ionic strength. In general, K increases with increasing pH, and decreases with increasing ionic strength (Khalid et al 1977).

3.0 MATHEMATICAL MODELING

The mathematical expressions employed to model the complexation process and their corresponding input parameters are discussed in the following sections.

3.1 Mathematical Expressions

The mathematical modeling of complexation in SESOIL is based upon equation (CM-5) and the constraint of conservation of mass. Thus for either of the uncomplexed species, the concentration of the free species is equal to the total input (T) of that species minus the amount complexed (c):

$$[M^{X^+}] = [M_T] - [M]_c$$
 (CM-6)

$$[L^{y^{-}}] = [L_{T}] - [L]_{C}$$
 (CM-7)

The equilibrium constant (equation CM-5) has been written so that the each mole of complex includes one mole of metal and b moles of ligand.

$$[M]_{o} = [ML]$$
(CM-8)

$$[L]_{a} = b[ML]$$
(CM-9)

By combining equations (CM-5) through (CM-9) and omitting the designation of charges we have:

$$[ML] = K ([M_{T}] - [ML]) ([L]_{T} - b[ML])^{b}$$

where:

- $[M_{\pi}]$ = total concentration of metal in solution; (mol/mL)
- [ML] = complexed concentration of metal (with organic ligand) in solution; (mol/mL)
- [L] = total concentration of organic ligand in solution; (mol/mL)
- b = number of moles of organic ligand reacting with one mole of metal cation (M); (-)

This equation is a non-linear equation (b does not necessarily = 1) and must be solved numerically. This numerically solution is performed for the monthly routines in subroutine COMP which used the same iterative procedure as used to the pollutant cycle (see appendix PT). For the annual routines, which require an analytic solution, the ligand input concentration is assumed to be large compared to the amount of ligand complexed:

$$[L]_{T} > [L]_{c}$$
 therefore
 $[L^{y}] = [L_{T}] - [L_{c}] \approx [L_{T}].$

In this case equation CM-10 reduces to

$$[ML] = \frac{K[L]^{b}[M_{T}]}{1 + K[L_{T}]^{b}}$$

The masses of free and complexed species can be obtained from the concentrations by multiplying the concentration of the complex by the volume of the subcompartment:

$$P_{ML} = a \cdot [ML] \cdot V_{H2} 0 \tag{CM-8}$$

where:

P = pollutant mass complexed during time step of simulation (mol)

$$a = 1$$
 conversion units factor (mL/cm³)

- [ML] = complexed concentration of metal in solution; (mol/ml)
- $V_{H_2O} = \Theta \cdot d \cdot A$; volume of water (moisture) in soil subcompartment; (cm³)
- θ = volumetric moisture content of soil subcompartment; (fraction)
- d = depth of soil layer; (cm)
- A = cross sectional area of soil compartment; (cm²)

In SESOIL, pollutant (i.e., metal) masses are expressed in units of micrograms (ug), in contrast to the previous expressions which are expressed in moles (mol). To convert from moles to micrograms, the molecular weight of the compound can be used as follows:

$$P(in ug) = \frac{P(in mole)}{MWT \cdot f}$$
(CM-9)

where:

P = pollutant mass
MWT = molecular weight of pollutant; (g/mol)
f = units conversion factor; (10⁶ ug/g)

Arthur D Little Inc

The expressions used in SESOIL for obtaining the mass of pollutant complexed are shown in Table CM-1. Note that the engineering designations of concentration ([M]) have been changed, so that the pollutant concentration at the step t is not designated as c(t).

3.2 Input Parameters

Input parameters to the complexation routine are presented in Table CM-2.

Compartment depths and input masses are chosen according to the needs of the simulation. Chemical/ligand specific constants are available from handbooks and laboratory studies.

Laboratory data sometime imply non-integer values for b. These noninteger values are accepted and used by SESOIL. For example, studies of the complexation of cadmium in landfill leachate by Knox and Jones (1979) indicated b values were in the range of 0.7 to 1.5 with an average of about 1.1.

For further information about input parameters and formats, see the user's manual section.

TABLE CM-1

EQUATION DESCRIBING THE COMPLEXATION CONCEPT IN SESOIL

Annual Routines (LEVEL 0, 1):

$$[L] = LIG_{i}/(\Theta \cdot d_{i})$$

$$P[ML] = \frac{K([L]/MWT_{L} \cdot 10^{6})^{b}(c(t)/MWT_{M} \cdot 10^{6})}{1 + K([L]/MWT_{L} \cdot 10^{6})} \cdot MWT \cdot \Theta(t) \cdot d_{i} \cdot 10^{6}$$
(CM-10)

Monthly Routines:

$$\begin{bmatrix} L \end{bmatrix} = LIG_{i} / (\Theta \cdot d_{i}) \\ [ML] = K \cdot ([M_{T}] - [ML]) ([L]_{T} - b[ML])^{b} \\ P[ML] = [ML] \cdot \Theta(t) \cdot MWT \cdot 10^{6} \cdot d_{i} \\ [L]_{F} = [ML] \cdot \Theta(t) \cdot b \cdot MWT_{L} \cdot 10^{6} \cdot d_{i} \end{bmatrix}$$
 (solved iteratively)

Parameter	•	Parameter Description	Units	FORTRAN Code
P[ML]	=	pollutant mass complexed during time step of simulation	(ug/cm ²)	PCOM
к	=	stability constant of the complex	(-)	SK
[L]	=	concentration of organic ligand in solution	(ug/mL)	LIGCU, LIGCL
MWTL	=	molecular weight of L	(g/mol)	MVTLIG
c(t)	2	concentration of pollutant in solution	(ug/mL)	CUM, CMM, CLM
MWT _N	=	molecular weight of pollutant	(g/mol)	MWT

TABLE CM-1 (Continued)

EQUATION DESCRIBING THE COMPLEXATION CONCEPT IN SESOIL

Ъ	=	number of moles of organic ligand reacting with one mole of the metal cation	(-)	В
θ(τ)	=	volumetric soil moisture content of soil	(-)	THA, THM
t	=	time of simulation		DT
di	=	depth of soil layer i	(cm)	DN, DM, DR
LIG _i	=	input mass of ligand	ug/cm ²	LIGU, LIGL
[L]_	=	free ligand concentration	ug/mL	LIGUF, LIGLF

TABLE CM-2

INPUT PARAMETERS FOR COMPLEXATION ALGORITHMS

Parameter			FORTRAN Variable
Name	Designation	Units	
Soil Compartment Depths	d	cm; m	DL (upper), DM (middle) DL (lower)
Complexation Stabil Constant	ity K	-	SK
Input Mass of Ligan	d LIGi	ug	LIGIN
Input Mass of Pollu	itant -	ug	POLIN
Ratio: Moles Ligan Mole of Complex	nd / b	-	В
Molecular Weight of Metal	MWT	g/mol	MWT
Molecular Weight of Ligand	MWTL	g/mol	MWTLIG

4.0 DISCUSSION

As discussed in the previous sections, the subroutine calculations for complexation in SESOIL:

- (1) are primarily for heavy metal cations in solution;
- do not consider competition with other ions or the effect of pH and ionic strength;
- (3) assume equilibrium exists at all times;
- (4) assume that complexation is fully reversible and competes with all other processes (except cation exchange); and
- (5) assume that the complex formed is soluble, does not adsorb on the soil, and does not migrate from zone to zone.
- (6) assume (for annual routines only) that the total ligand mass input is large compared to the ligand mass involved in complexation. This assumption is <u>not</u> made in the monthly routines.

It is incumbent upon the user to use this routine in the appropriate manner for environmental conditions where complexation is known to be a dominant factor in the mobility of the metal ions.

5.0 REFERENCES

Brinkman, F.E. and J.M. Bellama (eds.) (1978); Organometals and Organometalloids, Occurrence and Fate in the Environment, ACS Symposium Series 82, American Chemical Society, Washington, D.C.

Davis, J.A. and J.O. Leckie (1978); Effect of Adsorbed Complexing Ligands on Trace Metal Uptake by Hydrous Oxides, Environ. Sci. Technol. 12:1309-15.

Drake, E.; D. Shooter; W. Lyman and L. Davidson (1976); A Feasibility Study of Response Techniques for Discharges of Hazardous Chemicals that Disperse Through the Water Column, Report No. CG-D-16-77, U.S. Coast Guard, Office of Research and Development, Washington, D.C.

Fuller, W.H.; A. Amoozegar-Ford; E. Niebla and M. Boyle (1981); Behavior of Cd, Ni and Zn in Single and Mixed Combinations in Landfill Leachates, page 18-28 in Land Disposal: Hazardous Waste, Report No. EPA-600/9-81-0026, U.S. Environmental Protection Agency, Cincinnati, Ohio.

Gachter, R.; J.S. Davis and A. Mares (1978); Regulation of Copper Availability to Phytoplankton by Macromolecules in Lake Water, Environ. Sci. Technol. Vol. 12, pp. 1416-21.

Giesy, J.P., Jr.; L.A. Briese and G.J. Leversee (1978); Metal Binding Capacity of Selected Marine Surface Waters, Environ. Geol., Vol 2, pp. 257-68.

Giesy, J.P., Jr. and J.J. Alberts (1981); Trace Metal Speciation: The Interaction of Metals with Organic Constituents of Surface Waters, Chemtech (in press).

Griffin, R.A. and Shimp, N.F. (1978); Attenuation of Pollutants in Municipal Landfill Leachate by Clay Minerals, Report No. EPA-600/2-78-157, U.S. Environmental Protection Agency, Cincinniti, Ohio.

Jenne, E.A. (1979); Chemical Modeling--Goals, Problems, Approaches, and Priorities, in Chemical Modeling in Aqueous Systems, E.A. Jenne (ed.), ACS Symposium Series 93, American Chemical Society, Washington, D.C.

Khalid, R.A.; R.P. Gambrell; M.G. Verloo and W.H. Patrick, Jr. (1977); Transformations of Heavy Metals and Plant Nutrients in Dredged Sediments as Affected by Oxidation Reduction Potential and pH. Volume I: Literature Review, Contract Report D-77-4, Dredged Material Research Program, U.S. Army Engineer Waterways Experiment Station, Vicksburg, Mississippi.

Knox, K. and P.H. Jones (1979); Complexation Characteristics of Sanitary Landfill Leachates, Water Res., Vol. 13, pp. 839-46.

Kuo, S. and A.S. Baker (1980); Sorption of Copper, Zinc, and Cadmium by Some Acid Soils, Soil Sci. Soc. Am. J., Vol. 44, pp. 969-74.

McCrady, J.K. and G.A. Chapman (1979); Determination of Copper Complexing Capacity of Natural River Water, Well Water and Artificially Reconstituted Water, Water Res., Vol 13, pp. 143-50.

Nriagu, J.O. and R.K. Coker (1980); Trace Metals in Humic and Fulvic Acids from Lake Ontario Sediments, Environ. Sci. Technol., Vol. 14, pp. 443-46.

Oakley, S.M.; C.E. Delphey, K.J. Williamson and P.O Nelson (1980); Kinetics of Trace Metal Partitioning in Model Anoxic Marine Sediments, Water Res., Vol. 14, pp. 1067-72.

Oakley, S.M.; P.O. Nelson and K.J. Williamson (1981); Model of Trace-Metal Partitioning in Marine Sediments, Environ. Sci. Technol., Vol. 15, pp. 474-80.

O'Donnell, D.F.; P.A. Alesii; J. Artiola-Fortany and W.H. Fuller (1981); Predicting Cadmium Movement Through Soil as Influenced by Leachate Characteristics, U.S. Environmental Protection Agency report (in press).

Pagenkopf, G.K. (1978); Metal-Ion Transport Mediated by Humic and Fulvic Acids, in Organometals and Organometalloids--Occurrence and Fate in the Environment, F.E. Brinkman and J.M. Bellama (eds.), ACS Symposium Series 82, American Chemical Society, Washington, D.C.

Saar, R.A. and J.H. Weber (1980); Lead (II)-Fulvic Acid Complexes, Conditional Stability Constants, Solubility, and Implications for Lead (II) Mobility, Environ. Sci. Technol., Vol. 14, pp. 877-80.

Sposito, G. (1981); Trace Metals in Contaminated Waters, Environ. Sci. Technol., Vol. 15, pp. 396-403. CE - cation exchange

•

APPENDIX PH

PHOTOLYSIS

Photolysis of pollutant on soil surface layers might be another mechanism of pollutant loss. This process -- important for certain compounds -- will be incorporated in another SESOIL version.

FX - fixation

APPENDIX FX

FIXATION

Fixation is an important transformation process for certain compounds. It has been suggested by many scientists who are interested in SESOIL's modeling to incorporate into the model this process in the near future.

Arthur D Little, Inc.

81 - biologic activity

APPENDIX BI

BIOLOGIC ACTIVITIES

SESOIL is structured in a way that may allow the modeling of processes dealing with biologic activities in the soil. Although no definite plans are made by the developers regarding these issues, they believe that a SESOIL expansion in this area would lead to quite a useful tool for studying biologic soil activity due to manmade actions, as for example, POTW disposal actions on land. NV - nutrient cycle

APPENDIX NU

NUTRIENT CYCLE**

			Page
1.0	INTR	ODUCTION	NU-2
	1.1 1.2	General Literature Review	NU-2 NU-3
2.0	BACK	GROUND	NU-6
	2.1 2.2	Physical, Chemical and Biological Processes The Nitrogen Cycle 2.2.1 Forms of Soil Nitrogen 2.2.2 Nitrogen Transformations in the Soil Column	NU–6 NU–7 NU–9 NU–9
	2.3 2.4	The Phosphorus Cycle 2.3.1 Forms of Soil Phosphorus 2.3.2 Phosphorus Transformations in the Soil Column Summary of Background of Nutrient Cycles	NU-13 NU-13 NU-15 NU-16
3.0	MATH	EMATICAL FORMULATIONS	NU-18
	3.1 3.2 3.3 3.4 3.5 3.6	General Nutrient Transformations Rate Constants System of Equations Input/Output Parameters Numerical Solution Techniques of Equation Systems	NU-18 NU-18 NU-21 NU-21 NU-26 NU-26
4.0	CONC	LUSIONS	NU-30
5.0	REFE	RENCES	NU-31

^{*} This subroutine is not operational; therefore, no great emphasis has been placed in its accurate documentation. ** Contribution from Joo Hooi Ong.
1.0 INTRODUCTION

1.1 General

Certain elements are essential for the growth of plants. These elements are called nutrients and are obtained from air and from water in the soil. Carbon, oxygen, and sometimes nitrogen (e.g. legumes can use gaseous nitrogen) are obtained from air. Of the nutrients required from soil, nitrogen, phosphorus, potassium, calcium, magnesium and sulfur are needed in relatively large quantities. Essential elements used in small amounts, i.e. trace elements include iron, manganese, boron, molybdenum, copper, zinc, chlorine, and cobalt. Nitrogen and phosphorus will be considered in the model because they are the principal nutrient pollutants.

In agricultural applications, nitrogen and phosphorus are usually supplied to the soil in the form of manure and commercial fertilizers. Nitrogen is commonly applied in the form of ammonium and nitrate salts and as urea. Other major ways by which nitrogen becomes available to plants are through fixation of gaseous nitrogen by bacteria, and through nitrogen dissolved in precipitation. These latter ways are especially important in natural ecosystems. Phosphorus is mostly applied as phosphates.

Nutrients in the soil are subject to various fate processes. They are absorbed by plant roots, transformed from one chemical form to another, adsorbed onto organic matter and clays, transported from the soil surface in runoff, or leached into the groundwater zone. Environmental quality is related to nutrient fate, for example, when nitrogen or phosphorus is transported from land into waterways and lakes, eutrophication and fish kills may result. If nitrates migrate into the groundwater zone and into drinking water supplies, health effects may result from ingestion of these supplies. A form of anemia called methaemoglobanaemia which predominantly affects children is caused by ingestion of nitrates. The use of fertilizers to increase productivity in agricultural areas must therefore by managed to minimize adverse environmental effects. A tool to predict nutrient concentrations in runoff, in the soil column, and in groundwater would be an important part of this management process.

The nutrient cycle module of SESOIL simulates the transport, transformation, and storages of nitrogen and phosphorus in the soil column. The nitrogen and phosphorus cycles in the environment and the nature of the microbial and chemical reactions involved in these cycles are described in the following sections. Following this background information, the mathematical formulation of these processes and the solution technique are explained.

1.2 Literature Review

A pioneering work in modeling nutrient cycling in soils is presented in the Agricultural Runoff Management (ARM) model (Donigian, et al., 1977). The same subroutines, together with some extensions have later been incorporated into the Hydrological Simulation Program - Fortran (HSPF) model (Johanson, et al., 1979). Both the ARM and the HSPF models simulate nitrogen and phosphorus transport and transformations in the soil column, and nutrient content in sediments and runoff from small agricultural watersheds. The soil column is divided into four zones in the ARM and HSPF models: (1) surface zone, (2) upper zone, (3) lower zone, and (4) groundwater zone. The transformations of the nutrient species are described by first order kinetics with a temperature correction using the Arrhenius equation. ARM uses regression equations for estimating soil temperatures, from air temperatures. The ARM model simulates plant uptake of nitrate and phosphate by using monthly rate constants. HSPF includes uptake of ammonium by plants. It varies plant uptake on a monthly basis and distributes the uptake rate of nitrogen between nitrate-N and ammonium-N by factors, the sum of which is 1.0. The effect of moisture content on nutrient transformations is taken into account by discontinuing all transformations at very low moisture values. On the surface zone, transformations do not occur except during storm events since ARM assumes that the surface zone is dry except when runoff is occurring.

Another model which simulates nutrient transport and transformation is the CREAMS model (Knisel, 1980). This model is to be applied to field size areas and gives as output, the average concentrations of soluble nitrogen

and phosphorus in runoff, the amount of nitrate leached and its average concentration, and the amounts of nitrogen and phosphorus associated with sediments. Phosphorus compounds are not simulated in the soil column, i.e., no phosphorus is assumed to leach into the soil. The soluble nitrogen compounds leached into the soil are assumed to be nitrate or compounds that are quickly converted to nitrate and are added to the nitrate pool in the soil. The soil column in CREAMS is divided into the surface layer and the root zone. On the surface, nutrients (soluble nitrogen and phosphorus) are removed with runoff and with the sediments. In the root zone, mineralization and denitrification are simulated using first order kinetics. Plant uptake and percolation of nitrate also occur at the root zone. Two options are available for calculating plant uptake: (1) plant growth as a function of plant water use and nitrogen uptake as a function of plant nitrogen content, and (2) by assuming that nitrogen uptake follows a normal probability curve. The denitrification rate is modified by the moisture content of the soil by assuming that the rate constant is only positive when the moisture content exceeds field capacity.

The nutrient cycle module in SESOIL simulates nitrogen and phosphorus transport and transformations in the soil column, and the concentrations in sediments and in runoff. SESOIL is not limited by the size of the watershed modeled and it does not need to be calibrated.

Many of the principles in the ARM model are used in the nutrient cycle module in SESOIL. The differences between the two models are:

 SESOIL uses monthly temperature inputs for each soil zone while ARM uses regression equations for estimating soil temperatures. The advantage of having user-input temperatures is that the model does not have to be calibrated to each site. (2) SESOIL includes plant uptake of ammonium-N whereas ARM only has an uptake rate for nitrate-N. HSPF does however include plant uptake of ammonium-N. Since uptake of ammonium-N is a major nitrogen uptake path in most applications, it would be more accurate to include it.

CREAMS is useful in applications when nitrate leaching, availability of nitrogen to plants, and nutrient concentrations in runoff and sediments are the primary interests. Since it does not simulate storages of any other species except nitrate in the soil column, it cannot be used to predict nutrient concentrations in soils. The nutrient cycle module in SESOIL would, therefore, have a wider application than the CREAMS since the soil column is modeled in SESOIL.

2.0 BACKGROUND

Both the nitrogen and phosphorus cycles consist of many physical, chemical and biological processes, and very often combinations of these processes. Because of the difficulty involved in defining precisely which mechanism is involved in each phenomenon, this section provides only background information about the nitrogen and phosphorus cycles. Detailed descriptions are not given here of the process mechanisms and dependencies on environmental factors, such as temperature, organic carbon content, oxygen content, soil moisture content, and pH. Quantitative knowledge of these relationships are generally lacking. A comprehensive qualitative source for understanding the nitrogen cycle in soils is Bartholomew (1965). Similarly, Larsen (1967) provides a review of the phosphorus cycle. Other sources used in writing the following sections are Brady (1974) and Odum (1971).

2.1 Physical, Chemical and Biological Processes

Physical, chemical and biological processes all play a role in the nitrogen and phosphorus environmental cycles. A physical process is one that does not alter the nature of the chemical species involved. Chemical processes generally involve reactions in which there is a transformation of chemical species without the aid of any organisms. Biological processes are biochemical transformations of chemical species by microbial metabolism. In the natural environment it is very difficult to define in a clear-cut fashion whether a phenomenon is due to an individual, or more likely, a combination of physical, chemical, or biochemical processes. For example, adsorption is usually considered a physical process but chemisorption contributes to the concentration of the adsorbed species.

The factors which affect the growth of bacteria are important in determining the rates of transformation of the nutrient species. Some processes (e.g. mineralization) involve primarily bacteria requiring oxygen gas, or aerobic bacteria. Other processes (e.g. denitrification) involve anaerobic bacteria using mostly combined oxygen. Bacteria which can

use either of the two oxygen forms are called facultative bacteria. Soil moisture affects bacterial growth in two ways: water is needed by bacterial metabolic activities, and the moisture content in soil affects the diffusion rate of oxygen through the soil. The temperature range at which most bacterial activity is optimal is between 20°C and 40°C, with the optimum temperature at around 35°C. Under ordinary soil temperature extremes (e.g. in winter) however, bacterial activity seldom halts. Organic matter is used as the energy source for the majority of bacteria. The amount and nature of the soil organic matter determines to a certain degree the types of bacteria present in the soil and their growth rates. Under conditions of high calcium, the pH of the soil column is usually between 6 and 8 which is generally best for most bacteria, although certain bacteria species are adapted for low pH and others at high pH. The calcium content and the pH values play a role in determining the specific bacteria present.

2.2 The Nitrogen Cvcle

Figure NU-1 shows the nitrogen cycle in nature with some typical magnitudes of the transformations. In arable soils, nitrogen is acquired by soil primarily in three ways: through nitrogen fixation, additions through precipitation, and application of nitrogen in fertilizers and manures. Atmospheric nitrogen is fixed by bacteria in symbiosis with legumes. Depending on the soil conditions and the crop, the amount of nitrogen fixed can range from 50-250 lbs/acre-year (Brady, 1974). Soil conditions required are good aeration, drainage, moisture, optimal pH, and a certain amount of active calcium. The nitrogen added in this way is used by the host plant and also passed into the soil itself. Certain organisms in symbiosis with non-legumes, mostly angiosperms can also fix atmospheric nitrogen under conditions of low soil nitrogen. Free-living organisms that are not directly associated with higher plants, for example, several groups of bacteria, blue-green algae and fungi, are also capable of nonsymbiotic or free fixation of atmospheric nitrogen. The rate of fixation is assumed to be low. Direct addition of nitrogen through rain and snow is variable with season and location. In a humid temperate



btained from Brady (194).

climate, it is estimated that additions to soil average around 4.5 lbs of NH_4-N and 1.5 lbs of NO_3-N for every acre per year (Brady, 1974). In agricultural lands, the nitrogen additions in fertilizer is probably the largest component of the three contributions. The forms added are nitrate, ammonia and urea. The amount of nitrogen added depends on the kind of crop to be grown, the chemical condition of the soil, and the physical state of the soil. Application rates range from less then 100 lbs of nitrogen per acre to 300 lbs of nitrogen per acre.

Nitrogen is depleted from the soil in crop removal, drainage, erosion, and volatilization of the gaseous form. In an arable situation, crop removal is the most significant way by which nitrogen is removed. Gaseous losses are usually small but can become significant under anaerobic conditions, such as in water-logged soils.

2.2.1 Forms of Soil Nitrogen

There are basically three forms of soil nitrogen: (1) organic nitrogen associated with the soil humus, (2) ammonium fixed by certain clay minerals, and (3) soluble inorganic ammonium and nitrate compounds. Most of the nitrogen is associated with organic matter. A very small percentage (2-3%) of this organic-N is converted to inorganic forms a year (Brady, 1974). Mineral surface soils contain from 0.4 to 10% of organic matter (Brady, 1974). Typical soils contain around 4% of organic matter. Subsoils generally contain much less organic matter. Up to 8% of the total-N is fixed by clay (Brady, 1974). The soluble ammonium and nitrate compounds in soils seldom form more than 1-2% of the total-N present (Brady, 1974).

2.2.2 <u>Nitrogen Transformations in the Soil Column</u> <u>Mineralization</u>

Mineralization is a biological process whereby organic nitrogen forms are converted to inorganic forms, usually to ammonium. Soil organisms attack the organic nitrogen compounds by enzymic digestion converting the more complex proteins to ammonium. Using the example of an amino combination, the reaction is as follows:

$$R-NH_{2} + HOH \xrightarrow{\text{enzymic}}_{\text{hydrolysis}} R-OH + NH_{3} + \text{energy}$$

$$2NH_{3} + H_{2}CO_{3} \xrightarrow{(NH_{4})_{2}} CO_{3} \xrightarrow{2NH_{4}^{+}} + CO_{3}^{2-}$$
(NU-1)

In general the overall reactions are described by:

organic-N -----> ammonium-N (NU-2)

This transformation proceeds best in well-drained aerated soils but will take place to some extent under almost any conditions because of the diverse species capable of this process. Annually, only 2-3% of organic-N may be expected to be mineralized (Brady, 1974). Mineralization is a very slow process relative to the other soil reactions.

Immobilization

This phenomenon is the conversion of inorganic species to organic forms and may be regarded as the reverse of the mineralization process. Soil microbes or plants take up soluble nutrient species for growth, converting these to organic compounds which are released to the soil upon death and cell decay. Immobilization may be represented as:

$$\binom{NH_4 - N}{NO_3 - N}$$
 (NU-3)

The rate constants for these transformations are dependent on those factors affecting mineralization. Suboptimum temperature and soil mositure slow down the rate of immobilization. Anaerobic conditions may also have an important effect on slowing down the rate; however, there is a possibility of adaptation of different microbial species across a range of oxygen concentration. The sensitivity of different microbes to ranges in pH could also affect the rate constants for these reactions. Immobilization is a fast process relative to mineralization.

Nitrification

Nitrification is a process of enzymic oxidation of ammonium. It takes place in two steps due to the activity of two separate groups of bacteria. The first step is the production of nitrous acid and the second step is the oxidation of the nitrite form to nitrate. The group of bacteria primarily responsible for the first step is the <u>Nitrosomonas</u> bacteria and for the second, Nitrobacter bacteria.

$$2NH_{4} + 30_{2} \xrightarrow{\text{enzymic oxidation}} 2NO_{2}^{-} + 2H_{2}O + 4H^{+} + \text{energy}$$

$$2NO_{2}^{-} + O_{2} \xrightarrow{\text{enzymic oxidation}} 2NO_{3}^{-} + \text{energy}$$

$$(NU-4)$$

Other bacterial species are also able to oxidize and produce nitrate products but it is uncertain whether they are significant contributors of nitrate to soil. Soil conditions affecting nitrification are soil oxygen content, temperature, moisture, pH, fertilizer salts, and the nitrogen-carbon ratio. The temperature range under which nitrification occurs is between 0°C and 52°C. The most favorable temperatures are between 27-32°C (Brady, 1974). Nitrification is slow at very low or very high moisture content although it is known to proceed appreciably under very dry conditions. The most favorable soil moisture content is similar to that for growth of higher plants. Nitrification is low at low pH values but acidity itself is not significant when adequate exchangeable bases are present. Under ideal temperature, soil and moisture conditions, nitrification is a very fast process. Daily rates from 6 to 22 lbs of nitrogen per 2 million lbs of soil when 100 lbs of ammonium-N was added have been observed (Brady, 1974).

Denitrification

Denitrification is a reduction of nitrate-N to gaseous compounds. Facultative anaerobic baceria which prefer free oxygen use the combined oxygen in nitrate-N. The exact mechanisms are not known and may be chemical as well as biochemical. One way to describe it is as in reaction (b), which is basically respiration using nitrate instead of free oxygen as an oxygen source. The reactions could be adequately described in several ways:



In general, the process may be described as:

 NO_3 NO_2 NO_2

The rate of disappearance of NO_3 -N is dependent on soil oxygen content, and the presence of reducing agents and organic matter. The pH, temperature and moisture content are important factors as well. The rate of denitrification is slow under acidic conditions and high under alkaline conditions. Higher soil moisture content also increases the rate of denitrification possibly by indirectly affecting oxygen diffusion in the soil column (Bartholomew, 1965). Denitrification is most likely to occur in poorly drained soils and in acidic soils containing nitrates. Poor aeration enhances denitrification. Under conditions of poor drainage and aeration, loss of nitrogen through denitrification can be substantial.

Plant Uptake

Nitrogen is taken up by plants mostly in the forms of ammonium or nitrate. The form that is taken up depends on the conditions of the soil, the kind of plant and the stage of plant growth. Factors which influence plant uptake of nitrogen are biomass or root surface area of plants, temperature, soil moisture, soil oxygen content, season of the year, among others. When the plants are harvested, a large fraction of nitrogen is removed from the immediate location.

Adsorption-Desorption

Ammonium-N is adsorbed by clay minerals. Colloidal clay particles ordinarily carry negative charges. Positively charged ions are attracted to the colloidal crystal and are held in a non-exchangeable form. This non-exchangeable or fixed ammonium is released slowly.

NH₄+ _____ NH₄+ _____ NH₄+ (NU-6) (soil (exchangeable) (fixed) solution)

Ammonium fixation by clay minerals is greater in subsoils than in topsoil because of the higher clay content in subsoils. Organic matter or humus in soil also behaves like clay colloid particles in that ammonium-N can be fixed by these particles as well. The exact mechanism is unknown but it could be a chemical reaction by which compounds are formed between the soil organic matter and ammonium. This latter fixation is most favorable in the presence of oxygen and at high pH.

2.3 The Phosphorus Cycle

Figure NU-2 shows the phosphorus cycle in nature. Phosphorus is added to soils mostly in the form of fertilizers, especially in agricultural applications. Phosphorus is depleted from the soil in crop removal, drainage, and erosion.

2.3.1 Forms of Soil Phosphorus

In most soils, more than 50 percent of the total soil phosphorus is organic, present either as specific organic phosphorus compounds or as organic compounds linked with inorganic phosphorus groups (Larsen, 1967). Inorganic phosphorus forms present in soils depend on the pH of the soil. Under alkaline conditions, the HPO_4^{2-} ion is dominant. At low pH values, the $H_2PO_4^{-}$ ion is prevalent. Both these ions prevail at intermediate pH values.



FIGURE NU-2: MAIN PORTIONS OF THE PHOSPHORUS CYCLE

Source: Donigian, Beyerlein, Davis, and Crawford, 1977

2.3.2 Phosphorus Transformations in the Soil Column

Mineralization

Organic forms of phosphorus are converted to inorganic forms by bacteria in a process called mineralization similar to that of the mineralization of organic-N.

organic-P _____ phosphate-P (NU-7)

The same conditions which affect nitrogen mineralization regulate phosphorus mineralization.

Immobilization

Inorganic phosphorus is transformed to organic phosphorus through the process of immobilization. Soil organisms and plants take up soluble inorganic phosphorus for growth, converting these to organic forms which are released to the soil upon death.

Plant Uptake

Both the phosphate forms, HPO_4^{2-} and $H_2PO_4^{-}$ are adsorbed by higher plants. The forms available depend on the pH of the soil. Organic phosphorus cannot be used to any extent directly by higher plants but only after mineralization. Plant uptake of phosphorus is influenced by biomass or root surface area, temperature, soil moisture, soil oxygen content, season of the year, among others.

Adsorption-Desorption

The mechanism by which inorganic phosphorus ions are fixed by soil particles is not a simple adsorption mechanism. In acid soils, phosphates are precipitated by iron, aluminium, and manganese ions, and fixed by hydrous oxides and by silicate clays. At high pH values, phosphates are precipitated by calcium compounds. Therefore, it is more accurate to assume the process as a combination of precipitation and adsorption-desorption.

2.4 Summary of Background of Nutrient Cycles

Table NU-1 shows typical dominant pathways in a soil column and transformations that may be expected in each soil zone. It must be emphasized that the relative rates vary with climate, soil type, vegetation, among other factors.

TABLE NU-1: DOMINANT NUTRIENT PAINWAYS IN EACH SOLL ZONE

	Ammonium-N in Solution	Ammonium-N Adsorbed	Nitrate-N in Solution	Organic-N In Solution	Inorganic-P Adsorbed	Inorganic-P Adsorbed	Organic-P in Solution
Surface	Runoff Leaching Volatilization	Loss in Sediments	Runoff Leaching	Runoff Leaching	Runoff Leaching	Loss in Sediments	Runoff Leaching
	Plant Uptake Adsorption-Desorption Immobilization Mineralization Nitrification	Adsorption- Desorption	Plant Uptake Immobilization Nitrification	Immobilization Mineralization	Plant Uptake Adsorption- Desorption Immobilization Nineralization	Adsorption- Desorption	Immobilization Mineralization
Upper Unsaturated	Leaching		Leaching	Leaching	Leaching		Leaching
	Plant Uptake Adsorption- Desorption Immobilization Mineralization Nitrification	Adsorption- Desorption	Plant Uptake Immobilization Nitrification Denitrification	Immobilization Mineralization	Plant Uptake Adsorption- Desorption Immobilization Mineralization	Adsorption- Desorption	Immobilization Mineralization
Lower Unsaturated	leaching		Leaching	leaching	Leaching		Leaching
	Plant Uptake Adsorption-Desorption Immobilization	Adsorption- Desorption	Plant Uptake Immobilization Dentrification	Lmanobilization	Plant Uptake Adsorption- Desorption Immobilization	Adsortpion- Desorption	Lmmobilization

3.0 MATHEMATICAL FORMULATION

3.1 General

The important nutrient species included in this modeling effort are: organic nitrogen in soil solution (org-N), ammonium in soil solution $(NH_4-N(s))$, ammonium adsorbed on soil particles $(NH_4-N(a))$, nitrate in soil solution (NO_3-N) , organic phosphorus in soil solution (org-P), orthophosphate in soil solution $(PO_4-P(s))$, and orthophosphate adsorbed on soil particles $(PO_4-P(a))$. These are the dominant species in a soil column. Each of the organic nutrient species is considered as a group because they are insufficiently characterized and because data are usually available for the organic nitrogen and organic phosphorus as individual groups. Nitrite (NO_2-N) is not considered as a separate species by itself since it is unstable and usually exists at concentrations that are orders of magnitude less than nitrate-N. The fluxes in and out of the NO₂-N pool are expected to be high; the storage capacity is very small.

Schematic diagrams of the nitrogen and phosphorus cycles modeled by the nutrient module of SESOIL are shown in Figures NU-3 and NU-4 respectively. These models are simplified representations of the cycles in nature.

Table NU-1 gives the processes in the soil column that affect transport and transformation of nutrient species in each soil zone. The physical processes of losses in runoff, sediments, leaching, and volatilization are modeled by the pollutant transport routine in SESOIL and will not be repeated here. The pollutant transport routine is called by the executive program of the model.

3.2 Nutrient Transformations

Each of the nutrient transformations is modeled as a first order reaction. Assuming that the soil being modeled is microbially active, at the low nutrient concentrations normally present in soils, first order kinetics best estimates what happens in the soil column. Since the Freundlich, the Langmuir and an overall adsorption equation are modeled by the





LEGEND:



7

storages

transformation pathway

FIGURE NU-4: SCHEMATIC DIAGRAM OF PHOSPHORUS CYCLE MODELED

adsorption-desorption routine of SESOIL, a choice is given to the user depending on the data availability and other factors.

3.3 Rate Constants

Table NU-2 shows the rate constants used in the model to describe the transformations in the soil column. These are first order rate constants to be specified by the user for each reaction and for each soil zone being modeled. The dependency of the rate constant on temperature is described by the Arrhenius equation.

$$\kappa_{\rm T} = \kappa_{35^{\circ}\rm C} \gamma^{(\rm T-35)}$$

where:

 K_T = rate constant at T°C (day⁻¹) $K_{35°C}$ = optimum K at 35°C (day⁻¹) γ = temperature coefficient (constant) T = temperature (°C)

The temperature coefficients for the nutrient transformations are shown in Table NU-2. The user has to input the values of rate constants at 35°C and the temperature coefficient for each of the rate constants. Some typical values of rate constants and temperature coefficients are given in Table NU-3.

3.4 System of Equations

The system of equations governing the nitrogen and phosphorus transformations in soil is given in Table NU-4. The transformations are based on nutrient mass per mass of soil or soil water in each zone, i.e., massbased concentrations. The ammonium desorption rate KD, the nitrogen mineralization rate KM, the phosphate desorption rate KDP, and the phosphorus mineralization rate KMP are based on per soil mass. All the other rates are based on per soil water mass in each zone.

To calculate soil and water masses in each zone, soil bulk densities have to be specified for each zone, and soil moisture content has to be

Rate	Temperature	
Constant	Coefficient	Transformation Process in Soil Column
(day ⁻¹)		
Nitrogen		
ΚΛ	ТКА	Adsorption of ammonium from solution to adsorbed phase
KD	TKD	Desorption of ammonium from adsorbed phase to solution
KN	TKN	Nitrification of ammonium in solution to nitrate
KDN	TKDN	Denitrification of nitrate to gaseous nitrogen
KJN	TKIN	Immobilization of nitrate to organic nitrogen
KIA	ТКІЛ	Immobilization of ammonium in solution to organic nitrogen
КМ	ткм	Mineralization of organic nitrogen to ammoniun in solution
КРЛ	ТКРЛ	Plant uptake of ammonium in solution
KPN	TKPN	Plant uptake of nitrate

Phosphorus

клр	ТКАР	Adsorption of phosphate from solution to adsorbed phase
KDP	TKDP	Desorption of phosphate from adsorbed phase to solution
KIP	TKIP	Immobilization of phosphate in solution to organic phosphorus
КМР	TKMP	Mineralization of organic phosphorus to phosphate in solution
КРР	ткрр	Plant uptake of phosphate in solution

.

TABLE NU-3: TYPICAL RATE CONSTANTS AND TEMPERATURE COEFFICIENT VALUES

PROCESS	RATE CONSTANT (day ⁻¹)	TEMPERATURE COEFFICIENT
Nitrogen:		
Mineralization	,	2
org-NNH ₄ -N	$0.001 - 0.0078^{\perp}$ (11 soils with wide	1.072
Immobilization	range of properties)	
NH4-N	0.15 ¹ (Ontario loam)	1.07 ²
NO3-Norg-N	0.15 ¹ (Ontario loam)	1.07 ²
Nitrification		
NH ₄ -NNO ₂ -N	0.22 ¹ (Salinas clay)	1.072
NO2-NNO3-N	9.0 ¹ (Salinas clay)	1.07 ²
NH ₄ -N>NO ₂ -N	0.143 ¹ (Milville loam)	1.07 ²
$NO_2 - N \longrightarrow NO_3 - N$	9.0 ¹ (Milville loam)	
NH ₄ −N→NO ₃ −N	0.0033 - 1.11 ¹ (various loams)	1.07 ²
Denitrification		2
$NO_3 - N - (N_2 + N_2 O)$	0.004 - 0.192 ¹ (various loams)	1.07-
Adsorption	2	2
$M_4 - N(s) \rightarrow N_4 - N(a)$	1.0000	1.050
Desorption	2	
$NH_4 - N(a) \longrightarrow NH_4 - N(s)$	1.0000 ³	1.0503
Plant Uptake ⁴	_	
NO3-N>Plant-N	$0-0.0975^{3}$	1.070 ³
NH ₄ -N>Plant-N	0-0.0975 ⁵	1.070 ⁵
Phosphorus:		
Mineralization		
$org-P \longrightarrow PO_4 - P(s)$	$0.002 - 0.02^3$	1.070 ³

•

TABLE NU-3: (Continued)

PROCESS	RATE CONSTANT (day ⁻¹)	TEMPERATURE COEFFICIENT
Immobilization PO ₄ -P(s)org-P	0 ³	1.070 ³
Adsorption $PO_4 - P(s) \longrightarrow PO_4 - P(a)$	1.0000 ³	1.050 ³
Desorption PO ₄ -P(a)PO ₄ -P(s)	0.0015 - 0.0150 ³	1.050 ³
$\frac{\text{Plant Uptake}^{4}}{\text{PO}_{4}-P(s) \longrightarrow \text{Plant-P}}$	0 - 0.0525 ³	1.070 ³

- 3. Donigian, et al., 1977. P-2 Watershed, Watkinsville, PA.
- 4. Range due to seasonal dependence.
- 5. Obtained by assuming equal uptake of NH_4 -P and NO_3 -N.

^{1.} Adapted from Knisel, Davidson, <u>et al.</u>, 1978. Laboratory values under various conditions.

^{2.} Brady, 1974 gives a typical doubling of a rate constant for every 10°C increase in temperature, which would indicate a temperature coefficient of 1.07 for biochemical reactions.

TABLE NU-4: DIFFERENTIAL EQUATIONS DESCRIBING NITROGEN AND PHOSPHORUS TRANSFORMATIONS

Organic Nitrogen

 $\frac{d}{dt} \{ org-N \} = KIA \{ NH_4-N(s) \} + KIN \{ NO_3-N \} - KM \{ org-N \}$

Ammonium in Solution

 $\frac{d}{dt} \left\{ NH_4 - N(s) \right\} = KM \left\{ org - N \right\} + KD \left\{ NH_4 - N(a) \right\} - (KPA + KIA + KA + KN) \left\{ NH_4 - N(s) \right\}$

Adsorbed Ammonium

 $\frac{d}{dt} \left\{ NH_4 - N(a) \right\} = KA \left\{ NH_4 - N(s) \right\} - KD \left\{ NH_4 - N(a) \right\}$

Nitrate

$$\frac{d}{dt} \left\{ NO_3 - N \right\} = KN \left\{ NH_4 - N(s) \right\} - (KIN + KDN + KPN) \left\{ NO_3 - N \right\}$$

Organic Phosphorus

$$\frac{d}{dt} \left\{ org-P \right\} = KIP \left\{ PO_4 - P(s) \right\} - KMP \left\{ org-P \right\}$$

Phosphate in Solution

 $\frac{d}{dt} \left\{ PO_4 - P(s) \right\} = KMP \left\{ org - P \right\} + KDP \left\{ PO_4 - P(a) \right\} - (KIP + KAP + KPP) \left\{ PO_4 - P(s) \right\}$

Adsorbed Phosphate

$$\frac{d}{dt} \left\{ PO_4 - P(a) \right\} = KAP \left\{ PO_4 - P(s) \right\} - KDP \left\{ PO_4 - P(a) \right\}$$

obtained from the hydrologic cycle routine of SESOIL. Soil mass in each zone is the product of the bulk density and volume of the zone. Knowing soil mass and soil water mass, the nutrient concentrations may be calculated appropriately to be used in the transformation equations. $NH_4-N(a)$, $PO_4-P(a)$, org-N and org-P are therefore in the units of mass per soil mass and the other nutrient species are in the units per soil water mass.

3.5 Input/Output Parameters

The input and output parameters used in the nutrient cycle module of SESOIL are summarized in Table NU-5.

3.6 Numerical Solution Techniques of Equation Systems

Table NU-4 gives the nitrogen and phosphorus equations modeled by the nutrient subroutine of SESOIL. An analytical solution of first order differential equation systems might be possible (theoretically), but for practical reasons SESOIL employs numerical algorithms.

Various solution algorithms are available in the literature, such as the simple Euler integration technique employed by the Agricultural Runoff Management (ARM) model [Donigian, et al., 1977] and the Runge-Kutta techniques [Abramowitz and Segun, 1968]. SESOIL employs a second order and fourth order Runge-Kutta solution method for the annual and the monthly simulations respectively.

For a given system of two first order differential equations as an illustration:

The "second order" Runge-Kutta discretized solution is obtained from:

$$y_{n+1} = y_n + \frac{1}{2}(k_1 + k_2) + 0(h^3)$$

$$z_{n+1} = z_n + \frac{1}{3}(1_1 + 1_2) + 0(h^3)$$
(NU-10)

Arthur D Little, Inc.

	PARAMETER	UNITS	
INP	UT		
1.	Loadings of:	kg/m ²	
	NH ₄ -N		
	NO ₃ -N		
	org-N		
	PO ₄ -P		
· 2.	Times of applications of:	historical	dates
	NH ₄ -N		
	NO ₃ -N		
	org-N		
	PO4-P		
3.	Method of incorporation of loadings		
4.	Monthly rate constants (35°C) for nitrogen and phosphorus uptake by plants for each soil zone. KPA; KPN; KPP	day ⁻¹	
5.	Rate constants (35°C) for each transformation other than plant uptake for each soil zone. KA; KD; KN; KDN; KIN; KIA; KM; KAP; KDP; KIP, KMP	day ⁻¹	
6.	Temperature coefficients for each rate constant. TKA; TKD; TKN; TKDN; TKIN; TKIA; TKM; TKPA; TKPN; TKAP; TKDP; TKIP; TKMP; TKPP	(-)	
7.	Monthly temperatures in each soil zone.	°C	
8.	Volume of soil in each soil zone.	m ³	
9.	Bulk density of each soil zone.	g/cm ³	
10.	Soil moisture content (from hydrologic simulations	;). (-)	
11.	Time step of simulations.	month	

•

TABLE NU-5: INPUT/OUTPUT PARAMETERS (Continued)

PARAMETER

UNITS

OUTPUT

1. Monthly concentrations of nutrient species in each soil zone. (A) $NH_4-N(a)$ org-N $PO_4-P(a)$ org-P(B) $NH_4-N(s)$ NO_3-N $PO_4-P(s)$ $\frac{mg}{kg \text{ of soil water}}$ where:

$$k_{1}=h \cdot f(x_{n}, y_{n}, z_{n})$$

$$k_{2}=h \cdot f(x_{n}+h, y_{n}+k_{1}, z_{n}+k_{1})$$

$$l_{1}=h \cdot g(x_{n}, y_{n}, z_{n})$$

$$l_{2}=h \cdot g(x_{n}+h, y_{n}+k_{1}, z_{n}+l_{1})$$

The "fourth order" Runge-Kutta discretized solution is obtained from:

$$y_{n+1} = y_{n+\frac{1}{6}}(k_1 + 2k_2 + 2k_3 + k_4) + 0(h^5),$$

$$z_{n+1} = z_n + \frac{1}{6}(1_1 + 21_2 + 21_3 + 1_4) + 0(h^5)$$

where:

$$k_{1}=hf(x_{n}, y_{n}, z_{n})$$

$$k_{2}=hf\left(x_{n}+\frac{1}{2}h, y_{n}+\frac{1}{2}k_{1}, z_{n}+\frac{1}{2}l_{1}\right)$$

$$k_{3}=hf\left(x_{n}+\frac{1}{2}h_{1}y_{n}+\frac{1}{2}k_{2}, z_{n}+\frac{1}{2}l_{2}\right)$$

$$k_{4}=hf(x_{n}+h, y_{n}+k_{3}, z_{n}+l_{3})$$

$$l_{1}=hg(x_{n}, y_{n}, z_{n})$$

$$l_{2}=hg\left(x_{n}+\frac{h}{2}, y_{n}+\frac{k_{1}}{2}, z_{n}+\frac{1}{2}\right)$$

$$l_{3}=hg\left(x_{n}+\frac{h}{2}, y_{n}+\frac{k_{2}}{2}, z_{n}+\frac{l_{2}}{2}\right)$$

$$l_{4}=hg(x_{n}+h, y_{n}+k_{3}, z_{n}+l_{3})$$

The FORTRAN code of the Runge-Kutta solution techniques is presented in subroutine RUNGEK, Appendix FC.

.

.

4.0 CONCLUSIONS

The nutrient cycle of SESOIL simulates the transport, transformations and storages of nutrient species in soil and gives as an output the storages of each species in each zone of the soil column.

The accuracy of the nutrient cycle output depends on the accuracies of the hydrologic and pollutant transport simulations because it uses both these routines in its simulations. This module can be used in the management of agricultural runoff, nutrient residues in the soil column, and the contamination of groundwater by nutrients via leaching.

5.0 REFERENCES

Abramowitz, Milton and Irene A. Segun, eds. <u>Handbook of Mathematical</u> Functions, Dover Publications, Inc., New York; 1968.

Bartholomew, W.V. and F.E. Clark, eds. <u>Soil Nitrogen</u>, Agronomy Monograph No. 10, American Society of Agronomy, Madison, Wisconsin; 1965.

Brady, Nyle C. <u>The Nature and Properties of Soils</u>, 8th edition, MacMillan Publishing Company, Inc., New York; 1974.

Davidson, James M., D.A. Graetz, P.S.C. Rao, and H.M. Selim. <u>Simulation</u> of Nitrogen Movement, Transformation, and uptake in Plant Root Zone, Ecological Research Series, Environmental Research Laboratory, Athens, GA; 1978. EPA-600/3-78-029.

Donigian, A.A., Jr., D.C. Beyerlein, H.H. Davis, Jr., and N.H. Crawford. Agricultural Runoff Management (ARM) Model Version II Refinement and Testing, Environmental Research Laboratory, Athens, GA, U.S. EPA, Washington, D.C.; 1977. PB-273 105.

Johanson, Robert C., J.C. Imhoff, H.H. Davis, Jr., <u>User's Manual for the</u> <u>Hvdrological Simulation Program - Fortran</u>, Hydrocomp, Inc., Preliminary Draft; September 1979.

Knisel, Walter G., ed. <u>CREAMS: A Field Scale Model for Chemicals, Run-off, and Erosion from Agricultural Management Systems</u>, U.S. Department of Agriculture, Conservation Research Report No. 26; 1980.

Larsen, S. "Soil Phosphorus," in A.G. Norman, ed., <u>Advances in Agronomy</u>, American Society of Agronomy, Vol. 19, Academic Press, New York; 1967, p. 151-210.

Odum, Eugene P. <u>Fundamentals of Ecology</u>, 3rd edition, W.B. Saunders Company, Philadelphia, PA; 1974.

PT - pollutant cycle

APPENDIX PT

POLLUTANT CYCLE

		Page	
1.0	INTRODUCTION	PT-3	
2.0	MODELING BACKGROUND	P T-4	
	 2.1 Literature Review 2.2 The SESOIL Pollutant Transport Routine Concept 2.3 Chemical Partitioning 2.4 Pollutant Mass Balance 2.5 Compartments 	PT-4 PT-5 PT-7 PT-9 PT-12	
3.0	POLLUTANT CYCLE ROUTINES	PT-15	
	3.1 General 3.2 Annual Pollutant Cycle Routine (LEVELO, LEVELI) 3.2.1 General 3.2.2 Governing Equations 3.2.3 Solution Procedure	PT-15 PT-15 PT-15 PT-16 PT-16	
	 3.3 Monthly Pollutant Cycle Routine (LEVEL2, LEVEL3) 3.3.1 General 3.3.2 Governing Equations LEVEL2, LEVEL3 3.3.3 Numerical Solution Procedures 3.3.3.1 General 3.3.2 Constraint Criteria 3.3.3 Simulation Time Step 3.3.4 Moisture Molecule Penetration Constraint 	PT-28 PT-28 PT-28 PT-40 PT-40 PT-55 PT-55 PT-57	
	3.4 Storm-by-Storm Pollutant Cycle (LEVELN)	PT-59	
4.0	DISCUSSION	PT-59	
5.0	REFERENCES P		

.

FIGURES

PT-1:	CONCEPTUAL PRESENTATION OF THE SOIL-LAYER ACTING AS A "POLLUTANT MASS CARRIER" OVER TIME	PT-6
PT-2:	SCHEMATIC OF PHASES DESCRIBING THE SOIL MATRIX, AND INTERRELATION OF SOIL LAYERS	PT-8
PT-3:	PARTITIONING AND OTHER TIME OR NON-TIME DEPENDENT PROCESSES WITHIN Δt .	PT-11
PT-4:	SCHEMATIC OF MATHEMATICAL CONVERGENCE CRITERIA OF EQUATION SYSTEMS	PT- 56
TABLES		
PT-1:	SUMMARY OF LEVEL FEATURES	PT-14
PT-2:	ANNUAL POLLUTANT CYCLE EQUATIONS LEVELS 0, 1	PT-17
PT-3:	SOLUTION TO POLLUTANT CYCLE EQUATIONS LEVELS 0, 1	PT-26
PT-4:	MONTHLY POLLUTANT CYCLE EQUATIONS LEVEL2	PT-29
PT-5:	MONTHLY POLLUTANT CYCLE EQUATIONS LEVEL3	PT-41

.

1.0 INTRODUCTION

The ultimate fate or distribution of a pollutant in the SESOIL compartment is governed by the hydrologic cycle processes, the sediment cycle processes and the interaction of the various chemical fate processes related to the pollutant involved. The actual quantity or mass of a pollutant in any one process will depend at a particular time on the "competition" among all the processes for the available pollutant mass. This competition constitutes the basic philosophy of the pollutanttransport routine of SESOIL that will be described in the following sections.

The hydrologic cycle processes have been described in appendix HY. The sediment cycle processes have been outlined in appendices SW and SR. The individual chemical fate processes (diffusion, volatilization, sorption, etc) have been described separately in previous appendices. In SESOIL all these processes are considered interrelated and are combined under different equation systems -- depending on the level of SESOIL operation. The solutions of these systems determine the spatial and temporal distribution of a pollutant in the various subcompartments (soil-air, soil-moisture, soil, etc) of the SESOIL "environment." The mathematical routine/equation that combines all the previously described processes (from appendix HY to appendix NU) is designated as the "pollutant cycle" routine of SESOIL.

The processes of sedimentation, soil resuspension, photolysis, fixation, biologic activity and nutrient transportation are <u>not</u> modeled in this version (1981) of SESOIL. Such a modeling effort is anticipated by the model developers in the near future (maybe 1982) in connection with the watershed features of SESOIL.

This appendix presents the pollutant cycle routine of the model including: (a) a short literature review of previous modeling efforts; (b) the rationale governing the pollutant cycle equation structure and the actual equations employed; and (c) the numerical techniques developed for each level of SESOIL operation. Information regarding input data requirements to the pollutant cycle routine is presented in section 3.0, User's Manual, of this report. One of the model objectives has been to provide the literature with a "user friendly" package; therefore input parameters to this routine have been kept to a minimum by using many values that are estimated on-line by the model. Thus, although SESUIL models a complex system of many processes and interactions, it is intended to be relatively easy to use.

2.0 MODELING BACKGROUND

2.1 Literature Review

The watershed modeling features of SESOIL are not presented in this documentation; therefore the following literature review focuses only in the unsaturated soil zone modeling aspects of the soil environment.

Previous sophisticated pollutant transport modeling efforts have mainly employed the one- or two-dimensional time dependent diffusive, convective mass transport differential equation in homogeneous isotropic soils, which in a one-dimensional domain is written

$$\frac{\partial(\Theta c)}{\partial t} = \frac{\partial}{\partial t} (\Theta \cdot K_D^* \frac{\partial c}{\partial z}) - \frac{\partial(Vc)}{\partial z} - \rho \frac{\partial s}{\partial t} + \Sigma P \qquad (PT-1)$$

where Θ = soil moisture content, c = dissolved pollutant concentration in soil moisture, K_D^* = apparent diffusion coefficient of compound in soil-air, V = Darcy velocity of soil moisture, ρ = soil density, s = adsorbed concentration of compound on soil particles ΣP = sum of sources or sinks of the pollutant within the soil volume and z = depth.

The above equation has been solved principally: (a) numerically over a temporal and spatial discretized domain, via finite difference or finite element mathematical techniques (eg. Bonazountas et al 1979); and (b) analytically, by seeking exact solutions for simplified environmental conditions (eg. Enfield et al 1980). It must be pointed out that the theoretical derivation of the mass transport equation PT-1 is based upon a mass balance consideration of the pollutant in a representative soil element of volume dV=dxdydz.

The principal scientific deficiencies when modeling pollutant transport via equation PT-1 are: (1) only diffusion, convection, adsorption and possibly decay can be modeled, whereas other processes such as fixation or cation-exchange have to be neglected; (2) this equation is applicable mainly to pollutant transport of organics, whereas transport of metals which can be strongly affected by other processes cannot be directly modeled; (3) this equation can predict volatilization only implicitly via boundary diffusion constraints, however, experimental studies have frequently demonstrated an overestimation or underestimation of the theoretical volatilization rate; (4) no experimental or well accepted equation for a process (eg. volatilization) can be incorporated in PT-1, since the model has its own predictive mechanism; (5) pollutant concentrations are estimated only in the soil-moisture and on soilparticles, whereas pollutant concentrations in the soil-air are omitted; (6) the discretized version of the equation has a pre-set temporal and spatial discretization grid that results in high operational costs (professional time, computer time) of the model, since input data have

to be entered into each node of the grid; and (7) it is difficult and costly to run above models for typical canonical environmental scenarios (soils, climates, pollutants) for reasons discussed in section 3.16 canonical environments, of the User's Manual.

In order to bypass the above and other deficiencies of the traditional mathematical modeling, the SESOIL model developers have designed -- based on their experience with the traditional modeling (Bonazountas et al 1979) -- a new pollutant transport routine, which is presented conceptually in the following section.

2.2 The SESOIL Pollutant Transport Routine Concept

In SESOIL, any soil subcompartment of irregular cross area A_i and of depth d_i is considered a "pollutant mass carrier," that can receive pollutant mass from other subcompartments, store pollutant mass and export pollutant mass to other subcompartments. Assuming, for example, that a compartment (Figure PT-1) contains originally a pollutant mass M_{orig} -- of a particular pollutant -- and receives during an infinitesimal time step $\Delta t=(t)-(t-1)$ an impulse of pollutant mass M_{input} , then for this subcompartment we can write the equation

$$M_{\text{orig}}(t) + M_{\text{input}}(t) = M_{\text{trans}}(t) + M_{\text{rem}}(t) + M_{\text{out}}(t) \quad (PT-2)$$

where

Morig(t) = initially available pollutant mass in the soil compartment, at time t
Minput(t) = input pollutant mass to the soil compartment in (At)
Mout(t) = time dependent exported pollutant mass by the pollutant carrier
Mtrans(t) = time dependent pollutant transformation (or loss) within the compartment
Mrem(t) = remained pollutant mass in the compartment due to various reasons.

Some of the processes of individual terms of equation PT-2 are time dependent over the infinitesimal time step Δt (eg. volatilization), others are assumed time independent (eg. adsorption). These issues are discussed in later sections.

The mass balance concept (above) has been applied in SESOIL to a soil matrix consisting of three phases: (a) solids (soil); (b) liquid (soil-moisture); and (c) gas (soil-air). Phases and subcompartments in SESOIL


- trans(t) = Pollutant mass transformation within the soil layer during the time step \Deltate(t)-(t-1).
- out(t) = Pollutant mass exported by soil compartment at the end of time (t).
- rem(t) = Remaining pollutant mass in the compartment at time (t).

FIGURE PT-1: CONCEPTUAL PRESENTATION OF THE SOIL-LAYER ACTING AS A "POLLUTANT MASS CARRIER" OVER TIME are interrelated (Figure PT-2). It is the pollutant cycle that simulates these interrelations by modeling both the chemical partitioning among the three phases, and transport between subcompartments i.

2.3 Chemical Partitioning

The soil environment consists in SESOIL of three media: (1) soil-air (gaseous phase); (2) soil-moisture (liquid phase); and (3) soil (solid phase). The fate (transport/transformation) of pollutants in the soil column -- and consequently their ultimate fate -- depends on the pollutant partitioning among these three phases. This partitioning is a function of various parameters such as the chemical-specific partition coefficients (eg. air/soil moisture) and rate constants.

In SESOIL, the three phases are assumed to be in equilibrium at all times. Thus once the concentration in one phase is known, the concentrations in the other phases can be calculated. In SESOIL, the pollutant cycle is based on the pollutant concentration in the soil-moisture. The concentration in the soil-air is then calculated by Henry's law, and the concentration in the soil is calculated from the adsorption, cation exchange, and other sorption processes included in the model. Henry's law and the sorption processes are briefly described below.

The solute (dissolved) concentration of a compound is related to its soil-air concentration via Henry's law (see equation VO-12).

$$c_{sa} = c \cdot H/R(T+273)$$
 (PT-3)

where

c_{sa} = pollutant concentration in soil-air; (ug/mL) c = dissolved pollutant concentration; (ug/mL) H = Henry's law constant; (m³·atm/mol) R = gas constant; (8.2x10⁻⁵ m³·atm/mol·°K) T = temperature; (°C) °K = °C + 273

In SESOIL, the pollutant concentration on the <u>soil</u> is determined from the <u>sum</u> of the concentrations of the pollutant adsorbed, cation-exchanged, and/or otherwise associated with the soil particles, eg. via adsorption isotherms as discussed in appendix AD (adsorption) and in appendix CE (cation exchange). One commonly used adsorption isotherm equation is the Freundich equation

$$s = K \cdot c^{1/n}$$
 (PT-4)



FIGURE PT-2: SCHEMATIC OF PHASES DESCRIBING THE SOIL MATRIX, AND INTERRELATION OF SOIL LAYERS

.

where

- s = absorbed concentration of compound; (ug/g soil)
- K = partitioning coefficient; (ug/g soil)/(ug/mL)
- c = dissolved concentration of compound; (ug/mL)
- n = Freundich constant; (-)

All these sorption processes are -- and are expressed in SESOIL as -- a function of the dissolved concentration of the pollutant in the soil moisture.

The total concentration of a chemical in a soil matrix can be calculated from the concentration of the pollutant in each phase and the related volume of each phase (see also appendix VO) by

$$c_0 = (n-0)c_{sa} + (0)c + (\rho_b)s$$
 (PT-5)

where

2.4 Pollutant Mass Balance

The pollutant concentration in the SESOIL compartment changes with both time and space due to the change of pollutant mass within the subcompartment of the soil column. According to the law of mass conservation for a representative element, the change of pollutant mass and over a small time step Δt in that element will equal

 $\Delta M = M_{in} - M_{out} - M_{trans}$ (PT-6)

where: ΔM = change in pollutant mass in the element; M_{in} = mass entered the element; M_{out} = mass left the element; and M_{trans} = mass transformed (ie. degraded) within the element.

Assuming that pollutant mass entered rapidly (instantaneously) at the beginning of the time step, and by discretizing equation PT-6 over the finite time step from, for example, t=0 to t=t (ie. $\Delta t=t$), we have

$$\begin{bmatrix} \Delta M \end{bmatrix}_{o}^{t} = M_{t} - M_{o} = M_{in,o} - M_{out,t} - M_{trans,t}$$
(PT-7)

where: M_t = pollutant mass in the element at time t; M_o = original pollutant mass in the element; $M_{in,o}$ = input pollutant mass at t=0; $M_{out,t}$ = pollutant mass lost in time t; and $M_{trans,t}$ = pollutant mass transformed in time t.

When the original (initial) mass in the element, and the input to the element are known, the fate of the pollutant at time t can then be determined by rearranging the terms of equation PT-7 and by expressing the right hand side of the equation as a function of unknown variables such as the dissolved, the adsorbed, and the vapor concentration of the pollutant, or

$$M_{in,o} + M_o = M_t(c,s,c_{sa},t) + M_{trans,t} (c,s,c_{sa},t) + (PT-8) + M_{out} t(c,s,c_{sa},t)$$

The above equation, which forms the basic philosophy of SESOIL, is consistent with a discretized version of equation PT-1 for a constant soil moisture content Θ :

$$\Theta(c_{t} - c_{t-1})/\Delta t = f(c,s,D^{*},V,t)$$
(PT-9)

however, in PT-8 pollutant mass balance (and not transport) is performed for all three pollutant phases (vapor, liquid, solid) as contrasted to PT-9 whose pollutant transport is studied only in the liquid and solid phase. Pollutant transport will result from the mass balance equation (PT-8) where the soil matrix is considered as a "pollutant carrier" as previously described and discussed below.

In addition, each of the terms of PT-8 is expressed in SESOIL as the weighted sum of the contributions (see equation PT-5) of various individual processes that cannot be accounted for by PT-9. For example, the term $M_{trans,t}$, the mass transformed during an infinitesimal time period Δt , is equal to the sum of the masses involved in hydrolysis from water (soil moisture), hydrolysis from soil solids and other degradation processes. This is a feature that cannot be studied via equation PT-9.

Finally, the SESOIL pollutant cycle equation is formulated for individual processes and for an infinitesimal time step Δt (Figure PT-3), and not





FIGURE PT-3: PARTITIONING AND OTHER TIME OR NON-TIME DEPENDENT PROCESSES WITHIN Δt .

for a differential time step f(t)-(t-1). This formulation allows SESOIL to easily prioritize strength of individual processes when necessary (eg. cation exchange takes precedence over all other processes, only that mass not able to be exchanged is available for other sorption processes) a modeling approach <u>not</u> followed before. This infinitesimal approach for interrelated soil subcompartment on soil layers, in <u>each</u> of which all pollutant phases are further interrelated, allows model developers to be able to employ more than one well accepted equation of the literature for one and the <u>same</u> process; for example, volatilization via Hamaker (1979), Farmer et al (1980), others -- see appendix VO.

The individual components of the SESOIL basic equation PT-8 are different depending on the number of soil layers considered/modeled, the nature of the process (eg. time-, non-time dependent), and the temporal resolution of the simulation. This results in the employment of different numerical procedures for solving the equation systems and in other scientific issues discussed in the following sections. Model users, however, need not be concerned with these issues which are handled internally (on-line) by the model.

2.5 Compartments

The mass balance equation PT-8 presented in the previous section was formulated for any representative soil element that was assumed to be homogenous and isotropic. The model calculated pollutant concentration for this element reflects an average over the entire compartment, which means that spatial variation within the selected compartment is ignored. However, the user can select a fine cross-sectional (areal) resolution of his compartment in order to increase areal-accuracy of his predictions as discussed at the end of this section.

To increase the depth dependent spatial resolution of the model, without incurring the numerical and computational difficulties of formulating and solving discretized partial differential equations, the SESOIL compartment has been treated as a series of interconnected layers. Each layer then has its own mass balance equation, and can both receive and release pollutant to and from other layers (above or below).

Presently, SESOIL can handle simulations with two or three soil zones (Figure PT-2). A top layer exposed to the atmosphere, a middle layer and a lower layer are of <u>user</u> specified depth. There is no optimal advice for the physical boundaries of these soil layers. In many cases, these layers can be used to simulate a shallow root zone of 5-25 cm; in other cases a wider depth may be used. However, the minimum depth is 1 cm, for mathematical reasons only.

Multiple soil layers (eg. N instead of 3) can be simulated even with this version of SESOIL; however, potential users <u>should</u> contact model developers (Bonazountas, Wagner) for this issue.

In SESOIL, a soil column can be specified to cover any area from 1 cm^2 (finite approach) to several km² (watershed approach). To simplify the mathematics and to avoid computer round-off errors, the hydrologic cycle, the sediment cycle and the pollutant cycle equations of SESOIL are normalized to an area of 1 cm^2 . Pollutant fluxes across and within an element boundary have been simulated via equation PT-8 which has been formulated as

$$P_{in,o} + P_o = P_t + P_{out,t} + P_{transf,t}$$
(PT-10)

where

^P in,o	normalized (per cm ²) pollutant input flux; (ug/cm ²)
Po	normalized original pollutant mass in compartment; (ug/cm ²)
P _t	normalized pollutant mass in compartment at time ty (ug/cm ²)
Pout,t	normalized pollutant output flux; (ug/cm ²)
trans,t	normalized pollutant mass transformed within time t (ug/cm^2)

Equation PT-10 has been expanded to include various processes (hydrologic cycle, chemistry) taking place in the soil compartment. Time has become part of each individual processes within the time step of each simulation or level of operation. (Table PT-1).

TABLE PT-1

SUMMARY OF LEVEL FEATURES

LEVEL (#)	TIME Resolution	SPATIAL ³⁾ Resolution	OTHER
0	Annual	2 soil layers	Hydrocycle is user input
1	Annual	2 soil layers	Hydrocycle is estimated ²⁾
2	Monthly ¹⁾	2 soil layers	Hydrocycle is estimated ²⁾
3	Monthly ¹⁾	3 soil layers	Hydrocycle is estimated ²⁾

1) Provides annual averages of monthly estimates.

2) Hydrocycle estimate infiltration and groundwater recharge participation for mass balance purposes.

3) User should employ n+1 layers, where n is the user's needs. The +1 layer (next to groundwater) is for mass balance purposes.

3.0 POLLUTANT CYCLE ROUTINES

3.1 General

Currently SESOIL can be operated at four different levels -- LEVELO, to LEVEL3 (Table PT-1) -- each level serving a different purpose and providing a different temporal and spatial resolution of processes simulated (eg. soil quality, pollutant mass to groundwater). LEVELO and LEVEL1 are associated with annual simulations and two soil layers, whereas LEVEL2 and LEVEL3 are associated with monthly simulations and either two soil layers (LEVEL2) or three soil layers (LEVEL3).

The SESOIL theory is structured around the temporal resolution of a simulation. The spatial resolution is separately studied within each temporal subset (sections 3.2 and 3.3 below). For information as to the appropriate level of model operation for a particular application, consult section 3.0, user's manual, of this document.

3.2 Annual Pollutant Cycle Routine (LEVELO, LEVEL1)

3.2.1 General

Rough annual pollutant cycle simulations can be performed via LEVELO and LEVEL1 of SESOIL. These two levels differ only in the hydrologic cycle.

In LEVELO, the hydrologic cycle components (soil moisture, infiltration, groundwater, runoff, etc) are <u>user</u> input to the model. In LEVEL1, these components are estimated by the model via its annual hydrologic cycle routine (HYDROA) and the site specific climatological (ie. NOAA) and soil data. The hydrologic cycle drives the pollutant cycle.

The same pollutant cycle routine (TRANSA) is used for both levels, LEVELO and LEVEL1. The routine design is based upon:

- a discretization of the soil compartment into two zones, upper unsaturated zone (watershed) exposed to the atmosphere and lower unsaturated zone which extends to the groundwater table; and
- (2) the formulation of annual mass-balance equations PT-10 for each of the two soil layers.

The principal assumptions made for the formulation of the annual pollutant cycle equations are:

- the total pollution entering the subcompartment over the year is a "one-hit-event" which takes place at the start of the simulation;
- (2) the total pollutant mass entering the subcompartment is instantaneously distributed throughout the subcompartment;

- (3) the soil column is uncontaminated at the start of the simulation;
- (4) the simulation is to be performed for only one year;
- (5) the soil moisture content is represented by its long-term annual average value and does not vary with time within the year or spatially along the soil column.

LEVELO and LEVELI have specialized applications, for example: screening of a large number of chemicals that have to be compared for their environmental effects when released into a soil compartment. These levels should be employed with <u>care</u> (see also section 1.4 in the user's manual); therefore, if any of the previously given assumptions are not relevant to a particular application, the user has to select a higher level for his simulations, such as LEVEL2 or LEVEL3.

3.2.2 Governing Equations

Each of the two soil layers is considered a pollutant carrier with its own individual processes. The pollutant cycle equation system is solved analytically for reasons described in the following section and is coded in FORTRAN in subroutine TRANSA and in its related programmed functions -eg. VOLA, ADSA. The equation system, formulated for an upper soil layer and a lower soil layer, is presented in Table PT-2.

3.2.3 Solution Procedure

The annual pollutant routine is designed for approximate studies, where the emphasis in both LEVELO and LEVEL1 is placed on the differences among predicted concentrations -- for various environments and chemicals -and not on "accurate" values of the concentrations. Therefore, all processes in the annual routines are modeled as being of first-order, thus allowing a simple analytical solution of the systems to be obtained. A moisture penetration constraint is incorporated in both LEVELO and LEVEL1 simulations. This is discussed in detail in section 3.3.4 for all levels of SESOIL operations.

Because all of the individual fate processes/terms of the mass balance equation are linear functions of the pollutant concentration in the soilmoisture of each zone, the equations have been rearranged to give the pollutant concentrations in the soil-moisture directly as a function of the input parameters to the model (Table PT-3). The pollutant concentrations in the media of soil-air and soil-solids are calculated from the dissolved pollutant concentration, either via the partitioning equations presented in Section 2.3, or by dividing the pollutant masses previously calculated as being in each phase by the volume of that phase.

TAB, · rT-2

ANNUAL POLLUTANT CYCLE EQUATIONS - LEVELS 0,1

All terms in paragraphs 1, 2, and 3 below are in units of $\mu g/cm^2$. General Mass Balance Equation $P_{IN,i} + P_{O,i} = P_{REM,i} + P_{OUT,i} + P_{TRANS,i}$ 1. 2. Individual Processes - Upper Zone $P_{IN,U} = P_{INP,U} + P_{INF,U}$ Input mass: $P_{0,U} = 0.0$ (assumption 3, section 3.2.1) Mass available at beginning of year: $P_{\text{REM},U} = P_{\text{MOI},U} + P_{\text{ADS},U} + P_{\text{CEC}-F,U} + P_{\text{COM},U} + P_{\text{VAP},U}$ Mass remaining at end of year: Mass out in year: $P_{OUT,U} = P_{RS} + P_{INF,L} + P_{VOL,U}$ Mass transformed within year: $P_{TRANS,U} = P_{HYD-H_2O,U} + P_{HYD-S,U} + P_{DEG,U}$ 3. Individual Processes - Lower Zone $P_{IN-L} = P_{INP-L} + P_{INF-L}$ Input mass: $P_{0,L} = 0.0$ (assumption 3, section 3.2.1) Mass available at beginning of year: $P_{\text{REM},L} = P_{\text{MOI},L} + P_{\text{ADS},L} + P_{\text{CEC}-F,L} + P_{\text{COM},L} + P_{\text{VAP},L}$ Mass remaining at end of year: $P_{OUT,L} = P_{RG}$ Mass out in year: P TRANS, L = P HYD-H₃O, L + P HYD-S, L + P DEG, L Mass transformed within year:

TABLE rT-2 (Continued)

ANNUAL POLLUTANT CYCLE EQUATIONS - LEVELS 0,1

Where:

1		subscript for soil layer i
U		subscript for the upper soil layer
L		subscript for the lower soil layer
P _{IN}	-	pollutant mass input to soil compartment (total)
Р _О	-	pollutant mass originally in soil compartment
PREM	2	pollutant mass remaining in soil compartment
^р оит	=	pollutant mass output from soil compartment
P _{INP}	=	pollutant mass directly input to soil compartment
PINF	2	pollutant mass infiltrating to a zone
P _{MOI}	=	pollutant mass dissolved in soil moisture
PADS	=	pollutant mass adsorbed on soil particles
PCEC-F	=	final pollutant mass cation exchanged
^р сом	=	pollutant mass complexed
P _{VAP}	**	pollutant mass in vapor phase (soil air)

TABI 'T-2 (Continued)

P _{RS}	*	pollutant mass in surface runoff
P _{VOL}	=	pollutant mass volatilized
P TRANS	=	pollutant mass transformed in the soil compartment
^P iiyd-h ₂ o	=	pollutant mass hydrolyzed from soil moisture
^P HYD-S	=	pollutant mass hydrolyzed from soil solids
P DEG	=	pollutant mass degraded (other than by hydrolysis)
P _{RG}	=	pollutant mass in groundwater recharge

TABLL rT-2 (Continued)

ANNUAL POLLUTANT CYCLE EQUATIONS - LEVELS 0,1

Individual Fate Process Equations

4.

$$\begin{split} P_{INP,1} &= POLIN_{1} \\ P_{INP,U} &= I_{A} \cdot a_{SL} \cdot SL \\ P_{INF,L} &= c_{U}(A) \cdot I_{A,dU} \\ P_{MOI,1} &= c_{1}(A) \cdot 0 \cdot d_{1} \\ P_{COM,1} &= \left(\frac{(c_{1}(A)/(MWT \cdot 10^{6})) \cdot k_{eq} \cdot ([L]_{1}/(MWT_{LIG} \cdot 10^{6}))^{b}}{1 + (k_{eq} \cdot [L]_{1}/(MWT_{LIG} \cdot 10^{6}))^{b}}\right) \cdot MWT \cdot 10^{6} \cdot 0 \cdot d_{1} \\ P_{VAP,1} &= c_{1}(A) \cdot (U/(R \cdot (T_{a} + 273))) \cdot (n - \theta) \cdot d_{1} \\ P_{ADS,1} &= c_{1}(A) \cdot K_{1} \cdot \rho \cdot d_{1} \\ P_{CEC,1} &= Calculated by comparing the total ion exchange capacity of the soil with the input mass \\ T_{CEC,1} &= (Calculated by comparing the total is exchanged P_{CEC,1} = P_{IN,1} \\ Case 1: All available pollutant is exchanged P_{CEC,1} = T_{CEC,1} \\ P_{RS} &= c_{U}(A) \cdot R_{S,A} \cdot I_{RS,A} \\ P_{VOL,U} &= \left[c_{u}(A) \cdot (H/(R \cdot (T_{a} + 273) \cdot (d_{u}/2)))\right] \left[D_{a} \cdot ((n-\theta)^{10/3}/n^{2})\right] \cdot (365 \cdot 86400) \end{split}$$

TABL. ^JT-2 (Continued)

ANNUAL POLLUTANT CYCLE EQUATIONS - LEVELS 0,1

$$P_{HYD-H_2O,i} = c_i(A) \cdot K_{T,i} \cdot \theta \cdot d_i \cdot 365$$

$$P_{HYD-S,i} = (c_i(A) \cdot K_i \cdot K_{T,i} \cdot \rho \cdot d_i \cdot 365) + (P_{CEC,i} \cdot K_{T,i} \cdot 365)$$

$$P_{DEG,i} = c_i(A) \cdot K_{DE,i} \cdot \theta \cdot d_i \cdot 365$$

$$P_{RG} = c_L(A) \cdot R_{G,A}$$

5. Supporting Equations

$$I_{A,d_{U}} = R_{C,A} + ((I_{A} - R_{C,A}) \cdot (d_{L}/Z))$$

MWT = MWT + b . MWT_{LIG}

$$K_i = K_{oc} \cdot (\% oc_i)/100$$

(for organics)

oc_L = oc_U ·
$$a_{oc}$$

CEC_L = CEC_U · a_{CEC}
K_{T,1} = K_N + K_H · $10^{(-pH_1)}$ + K_{OH} · $10^{-(14-pH_1)}$
pH_L = pH_U · a_{pH}
P_{CEC-F,1} = P_{CEC,1} - P_{CEC,1} · K_{T,1} · 365
[L]₁ = LIG₁/(θ · d_1)
K_{DE,L} = K_{DE,U} · $a_{K_{DE}}$

TABLE PT-2 (Continued)

ANNUAL POLLUTANT CYCLE EQUATIONS - LEVELS 0,1

DEPTH =
$$(I_{\Lambda} + R_{G,\Lambda})/(2 \cdot \theta \cdot n)$$

Where:

U		subscript for upper soil layer	
L		subscript for lower soil layer	
1		subscript for any soil layer	
POLIN	=	annual direct pollutant input; (µg/cm ²)	POLINU, POLINL
IA	=	annual rainfall infiltration; (cm)	IA
^a sl	=	concentration of pollutant in the rainfall infiltration as fraction of solubility; (-)	ASL
s _L	=	pollutant aqueous solubility; (µg/mL)	SL
°1(A)	=	annual concentration of pollutant in soil moisture (layer 1); (µg/cm ²)	CUA, JUA
I _{A,d} U	=	annual infiltration at depth d _U ; (cm)	IADU
d 1	-	depth of soil layer; (cm)	DU,DL
θ		average annual soil moisture content; (fractional)	THA

Fortran Variables

TAB/ `T-2 (Continued)

			Fortran Variables
ρ	=	soil bulk density; (g/cm ³)	RSA
MWT	=	molecular weight of pollutant; (g/mol)	MWT
MWTLIC	; =	molecular weight of ligand; (g/mol)	MWTLIG
MWTML	-	molecular weight of pollutant-ligand complex; (g/mol)	MWTML
k eq	=	stability constant of pollutant-ligand complex; (-)	SK
[L] ₁	=	concentration of ligand in soil moisture; (µg/mL)	LIGCU,LICCL
Ъ	=	number of moles of ligand per mole of pollutant complexed; (-)	В
н	=	llenry's Law constant; (m ³ . atm/mol)	н
R	=	gas constant; $(8.2 \times 10^{-5} \text{ m}^3 \text{ . atm/mol} - ^{\circ}\text{K})$	R
Ta	=	temperature; (°C)	ТА
n	=	soil effective porosity; (fractional)	N
К	=	adsorption coefficient on soil; [(µg/g)/(µg/mL)]	к
T _{CEC,}	í ⁼	total compartment (i) cation exchange capacity; (µg/cm ²)	TCECU, TCECL
CEC,1	=	soil layer i cation exchange capacity; (milli equivalents/100g soil)	CECU, CECL
VAL	=	pollutant valence; (-)	VAL

TABLL ~T-2 (Continued)

			Fortran Variables
^R s,∧	=	annual surface runoff; (cm)	RSA
¹ rs,A	=	index (=0,1) of surface runoff participation in pollutant distribution; (~)	IRSA
DA	-	diffusion coefficient of pollutant in air; (cm ² /S)	D
^K T,1	=	total hydrolysis rate constant; (day ⁻¹)	KTN
K _{DE,1}	=	degradation rate constant; (day^{-1})	KDE
^R g,a	8	annual groundwater recharge (cm)	RGA
Z	n	depth to groundwater; (cm - data input in m)	2
Koc	-	adsorption coefficient on organic carbon; [(μg/g OC)/(μg/mL)]	кос
ос	=	soil organic carbon content; (%)	OC
a oc	=	ratio soil organic carbon content lower: upper; (-)	AOC
^a CEC	=	ratio soil cation exchange capacity lower: upper; (-)	ACEC
к _. N	-	neutral hydrolysis rate constant; (day ⁻¹)	KN
ĸ _{II}	=	acid-catalyzed hydrolysis rate constant; (day ⁻¹ . mol ⁻¹ /L ⁻¹)	клн
к _{ОН}	=	base catalyzed hydrolysis rate constant; (day ⁻¹ . mol ⁻¹ /L ⁻¹)	Квн

TABLE 2 (Continued)

			Fortran Variables
рН _і	=	pll in layer 1; (-)	РН
a pil	=	ratio soil pli lower: upper; (-)	APII
^a K _{DE}	=	ratio degradation rate constant lower: upper; (-)	AKDE
P _{CEC-F,i}	-	final pollutant mass remaining cation exchanged; $(\mu g/cm^2)$	PCECU, PCECL
DEPTH	-	depth rain penetration since start of simulation; (cm)	DEPTH
LIG,	-	ligand input mass	LICU, LIGL

TABLE PT-3

SOLUTION TO POLLUTANT CYCLE EQUATIONS FOR LEVELS 0,1

Upper Zone: $C_{U}(A) = [P_{INP,U} + P_{CEC,U}]/[0 \cdot d_{U} + k_{U} \cdot \rho \cdot d_{U} + H \cdot (n-0) \cdot d_{U}/R \cdot (T_{a} + 273) + R_{S,A} \cdot I_{RSA} + I_{A,d_{U}} + K_{T,U} \cdot 0 \cdot d_{U} \cdot 365 + K_{T,U} \cdot k_{U} \cdot \rho \cdot d_{U} \cdot 365 + k_{DE} \cdot 0 \cdot d_{U} \cdot 365 + H/(R \cdot (T_{a} + 273) \cdot (d_{U}/2)) \cdot D_{A} \cdot (n-0)^{10/3}n^{2} \cdot 86400 \cdot 365 + (1/(MWT \cdot 10^{6})) \cdot k_{eq} \cdot ([L]_{U}/(MWT_{LIG} \cdot 10^{6}))^{b}/(1 + (k_{eq} \cdot [L]_{U}/(MWT_{LIG} \cdot 10^{6}))^{b}) \cdot MWT \cdot 10^{6} \cdot \theta \cdot \frac{1}{2}$ $S_{U}(A) = (c_{U}(A) \cdot k_{U} \cdot \rho \cdot d_{u} + P_{CEC-F,u})/(\rho \cdot d_{u})$

PT-26

Lower Zone:

$$C_{L}(A) = [P_{INP,L} + P_{CEC,L}]/[n \cdot d_{L} + k_{L} \cdot \rho \cdot d_{U} + H \cdot (n-\theta) \cdot d_{L}/R \cdot (T_{a} + 273) + R_{GA} + K_{T,L} \cdot \theta \cdot d_{L} \cdot 365 + k_{DE,L} \cdot \theta \cdot d_{L} \cdot 365 + (1/(MWT \cdot 10^{6})) \cdot k_{eq} \cdot ([L]_{L}/(MWT_{LIG} \cdot 10^{6}))^{b}/(1 + (k_{eq} \cdot [L]_{L}/(MWT_{LIG} \cdot 10^{6}))^{b}) \cdot MWT \cdot 10^{6} \cdot \theta \cdot d_{L}]$$

$$S_{L}(A) = (C_{L}(A) \cdot k_{L} \cdot \rho \cdot d_{L} + P_{CEC-F,L})/(\rho \cdot d_{L})$$

$$C_{SA,L}(A) = c_{L}(A) \cdot H/R \cdot (T_{a} + 273)$$

TABLE PT-3

SOLUTION TO POLLUTANT CYCLE EQUATIONS FOR LEVELS 0,1

Where:

$$C_{i}(\Lambda)$$
 = pollutant concentration in moisture of zone i

 $S_i(A) = pollutant concentration on soil of zone i$

$$C_{SA,i}(\Lambda) = \text{pollutant concentration in soil-air of zone i}$$

For other symbols and supporting equations, see table PT-2.

Because such a direct solution can be obtained only when all equation terms are linear with respect to the dissolved pollutant concentration, only the linear Freundlich adsorption isotherm (n=1) has been incorporated into these levels. In addition, the equations of the time dependent chemical processes are solved with an annual time step, which in fact causes numerical computational inaccuracies in the predicted pollutant concentrations. This is one of the reasons why these levels of operation should not be employed for site specific simulations.

3.3 Monthly Pollutant Cycle Routine (LEVEL2, LEVEL3)

3.3.1 General

Both LEVEL2 and LEVEL3 routines simulate monthly cycles. They differ only by the number of user specified layers that form the soil column. In LEVEL2, the soil column has been divided into two soil zones; the upper unsaturated soil zone and the lower unsaturated soil zone. LEVEL3 has three soil zones; the upper, the middle and the lower unsaturated zones. A monthly pollutant cycle subroutine has been formulated for each of the above layers.

The principal assumptions made for the formulation of the pollutant transport equations are:

- the total pollutant mass enters a subcompartment sequentially (at a user specified rate) during the simulation period;
- (2) all physical phases of the subcompartment (soil, air, soil-moisture, soil-particles) are in equilibrium within a time step;
- (3) the soil-moisture content is represented by its monthly long-term averaged value and does not vary over the course of the month or along the vertical of the soil column (an improved version of SESOIL is underway).

It is believed that these levels of SESOIL will cover a wide range of applications with predictive accuracies within expected limits and simultaneously offer great savings in user input data effort and computer time. For situations where the above assumptions are not considered satisfactory development (and use) of a fully discretized (over time and space) numerical version of SESOIL is necessary. Model developers (Bonazountas & Wagner) have planned for this level of operation --LEVELN.

3.3.2 Governing Equations LEVEL2, LEVEL3

For each level, the individual pollutant carrying layers account for different sets of individual processes. Table PT-4 presents the derivation of the pollutant cycle equations for LEVEL2, Table PT-5 presents

TAB PT-4

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

All terms in paragraphs 1, 2, and 3 below are in units of $\mu g/cm^2$.

July.

PT-29

- 2. Individual Processes Upper Zone Input mass: Mass available at beginning of time step: Mass remaining at end of time step: Mass out in time step: Mass transformed within time step: P(t)_{1N,U} = P(t)_{1NP,U} + P(t)_{1NF,L} + P(t)_{COM,U} + P(t-1)_{VAP,U} P(t)_{REM,U} = P(t)_{NOI,U} + P(t)_{ADS,U} + P(t)_{CEC-F,U} + P(t)_{COM,U} + P(t)_{VAP,U} P(t)_{OUT,U} = P(t)_{RS} + P(t)_{INF,L} + P(t)_{VOL,U} + P(t)_{SINK,U} P(t)_{OUT,U} = P(t)_{RS} + P(t)_{INF,L} + P(t)_{VOL,U} + P(t)_{SINK,U} P(t)_{TRAN,U} = P(t)_{HYD-H2}O,U + P(t)_{DEG,U} + P(t)_{TRANS,U} + P(t)_{HYD-C,U}
- 3. <u>Individual Processes Lower Zone</u> Input mass: $P(t)_{IN,L} = P(t)_{INP,L} + P(t)_{INF,L}$ Mass available at beginning of time step: $P(t)_{0,L} = P(t-1)_{MOI,L} + P(t-1)_{SOIL,L} + P(t-1)_{COM,L} + P(t-1)_{VAP,L}$ Mass remaining at end of time step: $P(t)_{REM,L} = P(t)_{MOI,L} + P(t)_{ADS,L} + P(t)_{CEC-F,L} + P(t)_{COM,L} + \cdots + P(t)_{VAP,L}$ Mass out in time step: $P(t)_{0UT,L} = P(t)_{RG} + P(t)_{VOL,L} + P(t)_{SINK,U}$ Mass transformed within time step: $P(t)_{TRAN,L} = P(t)_{HYD-H_20,L} + P(t)_{DEG,L} + P(t)_{TRANS,L} + P(t)_{HYD-C,L}$

TABLE FT-4 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

Where:

i		subscript for soil layer i
t		indicates the current time step
t-1		indicates the previous time step
U		subscript for the upper soil layer
L		subscript for the lower soil layer
PIN	=	pollutant mass input to soil compartment (total)
Р _О	÷	pollutant mass originally in soil compartment
P _{REM}	=	pollutant mass remaining in soil compartment
POUT	=	pollutant mass output from soil compartment
P TRAN	=	pollutant mass transformed in the soil compartment
P _{INP}	=	pollutant mass directly input to soil compartment
P _{INF}	=	pollutant mass infiltrating to a zone
P _{MOI}	.=	pollutant mass dissolved in soil moisture
PSOIL	=	pollutant mass associated (adsorbed, exchanged) with soil
^Р сом	=	pollutant mass complexed

TABL1-4 (Continued)MONTHLY POLLUTANT CYCLE EQUATIONS- LEVEL 2

P _{VAP}	=	pollutant mass in vapor phase (soil air)
PADS	-	pollutant mass adsorbed on soil particles
P _{CEC-F}	=	final pollutant mass cation exchanged
P _{RS}	=	pollutant mass in surface runoff
PVOL	=	pollutant mass volatilized
P SINK	=	pollutant mass in other sinks (e.g., sediment transport)
^Р нчd-н ₂ 0	=	pollutant mass hydrolyzed from soil moisture
P _{HYD-S}	=	pollutant mass hydrolyzed from adsorbed pollutant
P HYD-C	=	pollutant mass hydrolyzed from exchanged pollutant
P DEG	8	pollutant mass degraded (other than by hydrolysis)
P TRANS	8	pollutant mass in other transformations (e.g., fixation)
P _{RG}	=	pollutant mass in groundwater recharge

TABLE PT-4 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

PT-32

4.

Individual Fate Process Equations

P(t) INP,1 = POLIN₁/NI P(t) INF, U = I_M · a_{SL} · SL/NI $P(t)_{INF,L} = c_U(t) \cdot I_{M,dU}/NI$ $P(t)_{MOI,1} = c_1(t) \cdot \theta(t) \cdot d_1$ P(t)SOIL, i = $s_i(t) \cdot \rho \cdot d_i$ $P(t)_{VAP,i} = c_i(t) \cdot \left(\frac{H}{R} \cdot (T_a + 273)\right) \cdot (n - \theta(t)) \cdot d_i$ P(t)ADS,1 = $c_1(t)^{1/FRN} \cdot K_1 \cdot \rho \cdot d_1$ P(t) CEC,i = Calculated by comparing the total ion exchange capacity of the soil with the input mass $T_{CEC,i} = (CEC_i \cdot MWT/VAL) \cdot 10 \cdot \rho \cdot d_i$ Case 1: All available pollutant is exchanged $P(t)_{CEC,i} = P(t)_{IN,i}$ Exchange capacity of soil is exceeded P(t)_{CEC,1} = T_{CEC,1} Case 2: = c_U(t) · R_{S,M} · i_{RS,M}/NI P(t)_{RS} $P(t)_{VOL,U} = \left[c_{U}(t) \cdot \left(\frac{H}{R} \cdot (T_{a} + 273) \cdot (\frac{d_{U}}{2}) \right) \right] \left[D_{a} \cdot \left((n - \theta(t))^{10/3} / n^{2} \right) \right] \cdot \left(\frac{30 \cdot 86400}{NI} \right)$

TABLE I (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

P(t)_{VOL.1} = Calculated according to concentration gradient

Case 1: $c_{II}(t) < c_{L}(t)$ (Gradient is upward.)

 $P(t)_{VOL,L} = \left[c_{L}(t) \cdot \left(\frac{1}{(R \cdot (T_{a} + 273) \cdot (d_{U} + d_{L}))}{2}\right)\right] \left[D_{\Lambda} \cdot \left(\frac{(n-\theta(t)^{10/3})}{n^{2}}\right)\right] \left(\frac{30 \cdot 86400}{NI}\right)$ Case 2: $c_{U}(t) \ge c_{L}(t)$ (Gradient is downward or zero.)

$$P(t)_{VOL,L} = 0.0$$

$$P(t)_{SINK, 1} = P_{SINK, 1}/NI$$

$$P(t)_{HYD-H_{2}O, 1} = c_{1}(t) \cdot K_{T, 1} \cdot O(t) \cdot d_{1} \cdot (30/NI)$$

$$P(t)_{HYD-S, 1} = (c_{1}(t)^{1/FRN} \cdot K_{t} \cdot K_{T, 1} \cdot \rho \cdot d_{1} \cdot (30/NI))$$

$$P(t)_{DEG, 1} = c_{1}(t) \cdot K_{DE, 1} \cdot O(t) \cdot d_{1} \cdot (30/NI)$$

$$P(t)_{TRANS, 1} = P_{TRANS, 1}/NI$$

$$P(t)_{RG} = c_{L}(t) \cdot R_{G, M}/NI$$

$$P(t)_{HYD-C, 1} = (P(t)_{CEC, 1} \cdot K_{T, 1} \cdot (30/NI))$$

$$P(t)_{COM, 1} = [ML]_{1} \cdot MWT \cdot 10^{6} \cdot O(t) \cdot d_{1} \quad SK = [ML]_{1}/\left[\left(\frac{c_{1}(t)}{MWT \cdot 10^{6}} - [ML]_{1}\right)\left(\frac{[L]_{1}}{MWT_{LIC} \cdot 10^{6}} - b[ML]_{1}\right)^{b}\right]^{*}$$

* Solved numerically in subroutine COMP

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

Supporting Equations $= \left(R_{G,M} + (I_M - R_{G,M}) \cdot (d_L/Z) \right) \cdot \frac{k(1)_u}{k(1)_z}$ ^IM,d_U k(1)_z $= \frac{Z}{\frac{d_U}{k(1)_{U}} + \frac{d_L}{k(1)_{U}}}$ = MWT + b . MWT MWT ML. = $K_{OC} \cdot (\% OC_{i})/100$ K, (for organics) = oc_u . a_{oc} oc I $CEC_{L} = CEC_{U} \cdot a_{CEC}$ $= K_{N} + K_{H} \cdot 10^{(-pH_{i})} + K_{OH} \cdot 10^{-(14-pH_{i})}$ K_{T,1} $^{pH}L = {}^{pH}U \cdot {}^{a}pH$ $K_{DE,L} = K_{DE,U} \cdot n_{KDE}$ $P(t)_{CEC-F,i} = P(t)_{CEC,i} - P(t)_{CEC,i} \cdot K_{T,i} \cdot (30/NI)$ $S_{i}(t) = (p(t)_{ADS, i} + P(t)_{CEC-F, i}) / (p \cdot d_{i})$ DEPTH(t) = $(I_M + R_{C,M})/(2 . 0 . n . NI)$

5.

TABL Y-4 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

 $\begin{bmatrix} L \end{bmatrix}_{i} = LIG_{i} / (\theta \cdot d_{i})$ $\begin{bmatrix} L \end{bmatrix}_{i, FREE} = \left(\begin{bmatrix} L \end{bmatrix}_{i} \cdot d_{i} \cdot \theta - B \cdot P_{COM} \cdot \left(\frac{MWTLIG}{MWT} \right) \right) / (d_{i} \cdot \theta)$ $c_{SA,i}(t) = c_{i}(t) \cdot H / (R \cdot (T_{a} + 273))$

TABLE $PT-\overline{4}$ (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

Where:

t	indicates current time step	
t-1	indicates previous time step	
U	subscript for upper soil layer	
L	subscript for upper soil layer	
i	subscript for any soil layer	
POLIN	= monthly direct pollutant input; (µg/cm ²)	POLINU, POLINL
NI	<pre>= number of numerical iterations per month; (-1)</pre>	NI
I _M	<pre>= monthly rainfall infiltration; (cm)</pre>	IM
^a SL	= concentration of pollutant in the infiltration as fraction of solubility; (-)	ASL
s _L	= pollutant aqueous solubility; (μg/mL)	SL
c ₁ (t)	<pre>= concentration of pollutant in soil moisture (layer i); (µg/cm²)</pre>	CUM, CLN
ι _{M,dU}	= monthly infiltration at depth d _U ; (cm)	IMOU
ďi	<pre>- depth of soil layer; (cm)</pre>	DU,DL
θ	<pre>= average soll moisture content; (fractional)</pre>	ТНА
s _i (t)	= concentration of pollutant on soil solids (layer i); (μg/g soil)	SUM,SLM

TAB! 'T-4 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

Fortran Variables = soil bulk density; (g/cm^3) ρ RS MWT molecular weight of pollutant; (g/mol) MWT = MWTLIG = molecular weight of ligand; (g/mol) MWTLIG MWTML = molecular weight of pollutant-ligand complex; (g/mol) MWTML = stability constant of pollutant-ligand complex; (-) k eq SK [L] concentration of ligand in soil moisture; (ug/mL) LIGCU, LIGCL = number of moles of ligand per mole of pollutant Ь = complexed; (-) B = Henry's Law constant; (m³. atm/mol) H H = gas constant; $(8.2 \times 10^{-5} \text{ m}^3 \cdot \text{atm/mol} - ^\circ\text{K})$ R R Ta temperature; (°C) Т = soil effective porosity; (fractional) n = N adsorption coefficient on soil; $[(\mu g/g)/(\mu g/mL)]$ K K = Freundich exponant; (-) FRN FRN = total compartment cation exchange capacity; $(\mu g/cm^2)$ T_{CEC,1} TCECU, ICECI. = CEC,1 soil cation exchange capacity; (milli equivalents/100g soil) CECU, CECL = VAL pollutant valence; (-) = VAL

TABLE PT-4 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

		Fortran Variables
^R s,m	= monthly surface runoff; (cm)	RSM
¹ RS,M	index (=0,1) of surface runoff participation in pollutant distribution; (-)	IRSM
DA	= diffusion coefficient of pollutant in air; (cm ² /S)	D
^P out,1	= other pollutant sinks per month; (µg/cm ²)	POUTU, POUTL
К _{т,1}	= total hydrolysis rate constant; (day ⁻¹)	KTN
к _{DE}	<pre>= total degradation rate constant; (day⁻¹)</pre>	KDE
^P TRANS, N	other pollutant transformations per month; (ug/cm ²)	PTRANU, PTRANL
R _G	<pre>= monthly groundwater recharge</pre>	RGM
Z	= depth to groundwater; (cm)	Z
кос	<pre>= adsorption coefficient on organic carbon; [(µg/g OC)/(µg/mL)]</pre>	КОС
OC	= soil organic carbon content; (o/o)	oc
^a 0C	= ratio soil organic carbon content lower: upper; (-)	ΛΟΟ
^a CEC	<pre>= ratio soil cation exchange capacity lower: upper; (-)</pre>	ACEC
к _N	= neutral hydrolysis rate constant; (day ⁻¹)	KN

-4 (Continued) TABLE

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 2

Fortran Variables acid-catalyzed hydrolysis rate constant; ĸ_{II} $(day^{-1}, mo1^{-1}/L^{-1})$ KAH base catalyzed hydrolysis rate constant; KOII $(day^{-1} . mo1^{-1}/L^{-1})$ KBII soil pll; (-) pll f PH = = ratio soil pH lower: upper; (-) **V**bh a_{pH} ratio degradation rate constant lower: upper; (-) AKDE aKDE final pollutant mass remaining cation exchanged; $(\mu g/cm^2)$ P(t) CEC-F,i PCECU, PCECL = average intrinsic soil permeability; (cm²) k(1) K1 intrinsic permeability of soil zone i; (cm²) k(1); K1U,K1L = DEPTH _ ~ depth rain penetration since start of simulation; (cm) DEPTH 2 pore disconnectedness index; (-) С = saturated hydraulic conductlvity; (cm/day) K(1) BK1 = ligand input mass; (µg/cm²) LIGU, LIGL LIG, [L] i, FREE free ligand concentration; (µg/mL) LIGCUF, LIGCLF = = pollutant concentration in soil air; $(\mu g/cm^3)$ c_{Sa,i}(t) CUSA, CLSA = concentration of ligand-pollutant complex (mol/mL) [ML] MLC

July, 1982

С

Arthur D Little, Inc

the derivation of these equations for LEVEL3. The LEVEL2 equations are coded in FORTRAN in subroutine TRANSM, those of LEVEL3 are coded in subroutine TRANS3. Both subroutines call several functions containing individual fate equations (see appendix FC, FORTRAN code).

3.3.3 Numerical Solution Procedures

3.3.3.1 General

All of the individual fate processes which compose the SESOIL mass balance equation are -- and are expressed as -- functions of

- a variety of rate, partitioning and other constants; and
- (2) the pollutant concentration in the moisture of each zone.

Some of the concentration terms are non-linear; therefore, these equations can not be solved directly as in the case of the annual cycle routines. An iterative solution procedure has been developed to solve this system efficiently.

The solution procedure involves the following steps for each layer -- starting at the surface of the soil column:

(1) an initial value c (t) = 0.0 is assumed;

(2) the mass balance equation PT-10 is solved iteratively

P_i = P_{in,i} +P_{o,i} - P_{a,i} - P_{out,i} - P_{transf,i}

by incrementally increasing the value of c(t) by Δc (Δc is large at the beginning) until P_i meets one of the five constraint criteria described in the next section;

- (3) the incremental interval Δc is decreased to a new value $\Delta c_1 < \Delta c$ and the system is resolved within a narrower numerical range by following again the above rationale.
- (4) Above procedure is repeated until:
 - (a) $P_n \leq \epsilon$, where ϵ is a confidence limit presently specified as 0.01, or
 - (b) the accuracy limits of the computer (machine) is reached (i.e. accumulation of round-off becomes significant).

TABL Γ-5

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

All terms in paragraphs 1-4 below are in units of $\mu g/cm^2$.

 $\frac{1}{2}$. <u>General Mass Balance Equation</u> $P(t)_{IN,i} + P(t)_{REM,i} = P(t)_{A,i} + P(t)_{OUT,i} + P(t)_{TRAN,i}$

2. Individual Processes - Upper Zone

1...1...

Input mass:

$$P(t)_{IN,U} = P(t)_{INP,U} + P(t)_{INF,U}$$
Mass available at beginning of time step:
Mass remaining at end of time step:
Mass out in time step:

$$P(t)_{O,U} = P(t-1)_{MOI,U} + P(t-1)_{SOIL,U} + P(t-1)_{COM,U} + P(t-1)_{VAP,U}$$

$$P(t)_{REM,U} = P(t)_{MOI,U} + P(t)_{ADS,U} + P(t)_{CEC-F,U} + P(t)_{COM,U} + P(t)_{VAP,U}$$

$$P(t)_{OUT,U} = P(t)_{RS} + P(t)_{INF,M} + P(t)_{VOL,U} + P(t)_{SINK,U}$$

$$P(t)_{OUT,U} = P(t)_{RS} + P(t)_{INF,M} + P(t)_{VOL,U} + P(t)_{SINK,U}$$

$$P(t)_{TRAN,U} = P(t)_{HYD-H_2O,U} + P(t)_{DEG,U} + P(t)_{TRANS,U} + P(t)_{HYD-C,U}$$

3. Individual Processes - Middle Zone

Input mass: $P(t)_{IN,M} = P(t)_{INP,M} + P(t)_{INF,M}$ Mass available at beginning of time step: Mass remaining at end of time step: Mass out in time step: Mass transformed within time step: $P(t)_{REM,M} = P(t)_{MOI,M} + P(t)_{ADS,M} + P(t)_{CEC-F,M} + P(t)_{COM,M} + P(t)_{VAP,M}$ $P(t)_{OUT,M} = \dot{P}(t)_{INF,L} + P(t)_{VOL,M} + P(t)_{SINK,M}$ $P(t)_{OUT,M} = \dot{P}(t)_{INF,L} + P(t)_{VOL,M} + P(t)_{SINK,M}$ $P(t)_{TRAN,M} = P(t)_{HYD-H_2O,M} + P(t)_{DEG,M} + P(t)_{TRANS,M} + P(t)_{HYD-C,M}$
TABLE PT-5 (Continued) MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

4. Individual Processes - Lower Zone

Input mass:
$$P(t)_{IN,L} = P(t)_{INP,L} + P(t)_{INF,L}$$
Mass available at beginning of time step: $P(t)_{0,L} = P(t-1)_{MOI,L} + P(t-1)_{SOIL,L} + P(t-1)_{COM,L} + P(t-1)_{VAP,L}$ Mass remaining at end of time step: $P(t)_{REM,L} = P(t)_{MOI,L} + P(t)_{ADS,L} + P(t)_{CEC,L} + P(t)_{COM,L} + P(t)_{VAP,L}$ Mass out in time step: $P(t)_{0UT,L} = P(t)_{RG} + P(t)_{VOL,L} + P(t)_{SINK,U}$ Mass transformed within time step: $P(t)_{TRAN,L} = P(t)_{IIYD-H_20,L} + P(t)_{HYD-S,L} + P(t)_{DEG,L} + P(t)_{TRANS,L} + P(t)_{HYD-C,L}$

TABLE -5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

Where:

1		subscript for soil layer i
t		indicates the current time step
t-l		indicates the previous time step
U		subscript for the upper soil layer
M		subscript for the middle soil layer
L		subscript for the lower soil layer
P _{IN}	=	pollutant mass input to soil compartment (total)
Р _О	=	pollutant mass originally in soil compartment
PREM	-	pollutant mass remaining in soil compartment
P _{OUT}	-	pollutant mass output from soil compartment
P TRAN		pollulant mass transformed in the soil compartment
PINP	=	pollutant mass directly input to soil compartment
P _{INF}		pollutant mass infiltrating to a zone
P _{MOI}	=	pollutant mass dissolved in soil moisture
^p soil		pollutant mass associated (adsorbed, exchanged) with soil
PCOM	3	pollutant mass complexed

TABLE PT-5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

^P VAP	=	pollutant mass in vapor phase (soil air)
PADS	8	pollutant mass adsorbed on soil particles
P _{CEC} -F	-	final pollutant mass cation exchanged
P _{RS}	=	pollutant mass in surface runoff
PVOL		pollutant mass volatilized
PSINK	-	pollutant mass in other sinks (e.g., sediment transport)
^P hyd-II ₂ 0	=	pollutant mass hydrolyzed from soil moisture
P _{HYD-S}	R	pollutant mass hydrolyzed from adsorbed pollutant
P HYD-C	12	pollutant mass hydrolyzed from exchanged pollutant
P _{DEG}	=	pollutant mass degraded (other than by hydrolysis)
P TRANS	=	pollutant mass in other transformations (e.g., fixation)
P _{RG}	=	pollutant mass in groundwater recharge

TABLF -5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

5.

P(t) INP,1 = POLIN₁/NI P(t) INF, U = I_M · a_{SL} · SL/NI $P(t)_{INF,M} = C_{IJ}(t) \cdot I_{M,dU}/NI$ P(t) INF,L = $C_{M}(t) \cdot I_{M,dM}/NI$ $P(t)_{MOI,i} = c_i(L) \cdot \theta(L) \cdot d_i$ P(t) SOIL, i = $s_1(t) \cdot \rho \cdot d_1$ $P(t)_{COM,i} = [ML]_{i} \cdot MWT \cdot 10^{6} \cdot \theta(t) \cdot d_{i} SK = [ML]_{i} / \left[\frac{C_{i}(t)}{MWT \cdot 10^{6}} - [ML]_{i} \right] \left(\frac{[L]_{i}}{MWT_{LIC} \cdot 10^{6}} - b[ML]_{i} \right)^{b} \right]^{*}$ $P(t)_{VAP,i} = c_i(t) \cdot (H/(R \cdot (T_a + 273))) \cdot (n - \theta(t)) \cdot d_i$ P(t)ADS, i = $c_1(t)^{1/FRN} \cdot K_1 \cdot \rho \cdot d_1$ P(L) CEC,1 Calculated by comparing the total ion exchange capacity of the soil with the input mass = (CEC, \cdot MWT/VAL) \cdot 10 $\cdot \rho \cdot d_{1}$ TCEC,1 All available pollutant is exchanged $P(t)_{CEC,1} = P(t)_{IN,1}$ Case 1: Exchange capacity of soil is exceeded $P(t)_{CEC,i} = T_{CEC,i}$ Case 2: $P(t)_{RS} = c_U(t) \cdot R_{S,M} \cdot i_{RS,M}/NI$

Individual Fate Process Equations

Arthur D Little. Inc

TABLE PT-5 (Continued)

.

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

$$P(t)_{VOL, U} = \left[c_{U}(t) \cdot \left(h/(R \cdot (T_{a} + 273) \cdot (d_{U}/2)\right)\right] \left[D_{a} \cdot \left((n-\theta(t))^{10/3}/n^{2}\right)\right] \cdot \left(\frac{30 \cdot 86400}{NL}\right)$$

$$P(t)_{VOL, H} = Calculated according to concentration gradient
Case 1: $c_{U}(t) \leq c_{H}(t)$ (Gradient is upwards.)
$$P(t)_{VOL, M} = \left[C_{H}(t) \cdot \left(u/(R \cdot (T_{a} + 273) \cdot (d_{U} + \frac{d_{H}}{2})\right)\right)\right] \left[D_{A} \cdot \left(\left(n-\theta(t)^{10/3}\right)/n^{2}\right)\right] \left(\frac{30 \cdot 86400}{NL}\right)$$
Case 2: $C_{U}(t) \geq C_{H}(t)$ (Gradient is downward or zero.)
$$P(t)_{VOL, M} = 0.0$$

$$P(t)_{VOL, L} = Calculated according to concentration gradient
Case 1: $C_{U}(t)$ and $C_{H}(t) \leq C_{L}(t)$ (Gradient is upward.)
$$P(t)_{VOL, L} = \left[C_{L}(t) \cdot \left(u/(R \cdot (T_{a} + 273))\right) \cdot \left(d_{U} + d_{H} + \frac{d_{L}}{2}\right)\right] \left[D_{A} \cdot \left(\left(n-\theta(t)^{10/3}\right)/n^{2}\right)\right] \left(\frac{30 \cdot 86400}{NL}\right)$$
Case 2: $C_{U}(t)$ or $C_{H}(t) \geq C_{L}(t)$ (Gradient is downward or zero.)
$$P(t)_{VOL, L} = 0.0$$

$$P(t)_{SINK, 1} = P_{SINK, 1}/NI$$

$$P(t)_{HVD-H_{2}O, 1} = c_{1}(t) \cdot K_{T, 1} \cdot \theta(t) \cdot d_{1} \cdot (30/NI)$$$$$$

PT-46

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

$$P(t)_{HYD-S,i} = (c_{i}(t)^{1/FRN} \cdot K_{i} \cdot K_{T,i} \cdot \rho \cdot d_{i} \cdot (30/NI))$$

$$P(t)_{DEG,i} = c_{i}(t) \cdot K_{DE,i} \cdot \theta(t) \cdot d_{i} \cdot (30/NI)$$

$$P(t)_{TRANS,i} = P_{TRANS,i}/NI$$

$$P(t)_{RG} = c_{L}(t) \cdot R_{G,M}/NI$$

$$P(t)_{HYD-C,i} = P(t)_{CEC,i} \cdot K_{T,i} \cdot (30/NJ)$$
Supporting Equations

^I_{M,d}_U =
$$(R_{G,M} + (I_M - R_{G,M}) \cdot (d_M + d_L/Z)) \cdot \frac{k(1)_u}{k(1)_z}$$

$$I_{M,d_{L}} = (R_{G,M} + (I_{M} - R_{G,M}) \cdot (d_{L} / Z)) \cdot \frac{k_{2}}{k_{z}}$$

$$k_{z} = \frac{Z}{\frac{d_{U}}{k(1)_{U}} + \frac{d_{M}}{k(1)_{M}} + \frac{d_{L}}{k(1)_{L}}}$$

= MWT + b . MWT_{LIG}

^{MWT}ML

 $K_i = K_{OC} \cdot (\% OC_i)/100$ (for organics)

 $OC_{M} = OC_{U} \cdot a_{OC}^{2}$

6.

TABLE PT-5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

= oc_u · a_{oc} OC_L = CEC_U · a²CEC CECM CECL $= CEC_{U} \cdot a_{CEC}$ = $K_{\rm N} + (K_{\rm H} \cdot 10^{(-\rm pH_{i})}) + (K_{\rm OH} \cdot 10^{-(14-\rm pH_{i})})$ K_{T,i} $pH_{M} = pH_{U} \cdot a2_{pH}$ $pH_L = pH_U \cdot a_{pH}$ $K_{DE,M} = K_{DE,U} \cdot a_{KDE}^2$ = K_{DE,U} . a_{KDE} ^KDE,L $P(t)_{CEC-F,i} = P(t)_{CEC,i} - P(t)_{CEC,i} \cdot K_{T,i} \cdot (30/NI)$ $S_{i}(t) = \left(p(t)_{ADS,i} + P(t)_{CEC-F,i}\right) / (\rho \cdot d_{i})$ DEPTH(t) = $(I_M + R_{G,M})/(2 \cdot \Theta \cdot n \cdot NI)$

TABLE F (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

$$\begin{bmatrix} L \end{bmatrix}_{i} = LIC_{i}/(\theta \cdot d_{i})$$

$$\begin{bmatrix} L \end{bmatrix}_{i, FREE} = \left(\begin{bmatrix} L \end{bmatrix}_{i} \cdot d_{i} \cdot \theta - B \cdot P_{COM} \cdot \left(\frac{MWTLIC}{MWT} \right) \right) / (d_{i} \cdot \theta)$$

$$c_{SA,i}(t) = (c_{i}(t) \cdot H / R \cdot (T_{i} + 273))$$

$$k_{2} = \frac{d_{U} + d_{M}}{\left(\frac{d_{U}}{k(1)_{U}} + \frac{d_{M}}{k(1)_{M}}\right)}$$

TABLE PT-5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

Where:

Fortran Variables

t	indicates current time step	
t-1	indicates previous time step	
U	subscript for upper soil layer	
м	subscript for middle soil layer	
L	subscript for lower soil layer	
1	subscript for any soil layer	
POLIN	= monthly direct polluLant input; (μg/cm ²)	POLINU, POLINL
NI	<pre>= number of numerical iterations per month; (-1)</pre>	NI
I _M	<pre>= monthly rainfall infiltration; (cm)</pre>	IM
^a SL	<pre>= concentration of pollutant in the infiltration as fraction of solubility; (-)</pre>	ASL
s _L	= pollutant aqueous solubility; (µg/mL)	SL
c _i (t)	<pre>= concentration of pollutant in soil moisture (layer i); (µg/cm²)</pre>	CUM, CLM
^I M, d _i	= monthly infiltration at depth d _i ; (cm)	IMOU
d ₁	= depth of soil layer; (cm)	DU, DL
θ	= average soil moisture content; (fractional)	ТНА
8 ₁ (t)	= concentration of pollutant on soil solids (layer i); (ug/g f ⁽¹))	SU7′M

T/ 'FT-5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

		Fortre " lables
ρ	= soil bulk density; (g/cm ³)	RS
MWT	<pre>= molecular weight of pollutant; (g/mol)</pre>	MWT
^{MWT} LIG	<pre>= molecular weight of ligand; (g/mol)</pre>	MWTLIG
MWTML	<pre>= molecular weight of pollutant-ligand complex; (g/mol)</pre>	MWTML
k eq	<pre>= stability constant of pollutant-ligand complex; (-)</pre>	SK
[L] ₁	= concentration of ligand in soil moisture; (ug/mL)	LIGCU, LIGCL
b	= number of moles of ligand per mole of pollutant complexed; (-)	В
н	= Henry's Law constant; (m ³ . atm/mol)	н
R	= gas constant; $(8.2 \times 10^{-5} \text{ m}^3 \cdot \text{atm/mol} - ^{\circ}\text{K})$	R
т _а	= temperature; (°C)	Т
n	<pre>= soil effective porosity; (fractional)</pre>	N
К	<pre>= adsorption coefficient on soil; [(µg/g)/(µg/mL)]</pre>	к
FRN	= Freundich exponant; (-)	FRN
^T cec, 1	= total compartment cation exchange capacity; (µg/cm ²)	TCECU, TCECL
CEC,1	<pre>= soil cation exchange capacity; (milli equivalents/100g soil</pre>) CECU, CECL
VAL	= pollutant valence; (-)	VAL

TABLE PT-5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

		Fortran Variables
R _{s,m}	= monthly surface runoff; (cm)	RSM
ⁱ rs,M	index (=0,1) of surface runoff participation in pollutant distribution; (-)	IRSM
D _A	<pre>= diffusion coefficient of pollutant in air; (cm²/S)</pre>	D
^P out, i	<pre>= other pollutant sinks per month; (µg/cm²)</pre>	POUTU, POUTL
^K T,1	= total hydrolysis rate constant; (day ⁻¹)	KTN
к _{DE}	<pre>= total degradation rate constant; (day⁻¹)</pre>	KDE
^P TRANS, N	<pre>= other pollutant transformations per month; (µg/cm²)</pre>	PTRANU, PTRANL
R _G	monthly groundwater recharge	RGM
Z	= depth to groundwater; (cm)	z
кос	<pre>= adsorption coefficient on organic carbon; [(μg/g OC)/(μg/mL)]</pre>	кос
OC	= soil organic carbon content; (o/o)	oc
^a oc	= ratio soil organic carbon content lower: upper; (-)	AOC
^a CEC	= ratio soil cation exchange capacity lower: upper; (-)	ACEC
ĸ _N	= neutral hydrolysis rate constant; (day ⁻¹)	KN

TABL' ~-5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

Fortran Variables

<pre>= acid-catalyzed hydrolysis rate constant; (day⁻¹ . mol⁻¹/L⁻¹)</pre>	клн
<pre>= base catalyzed hydrolysis rate constant; (day⁻¹ . mol⁻¹/L⁻¹)</pre>	КВН
= soil pH; (-)	PH
= ratio soil pH lower: upper; (-)	Арн
= ratio degradation rate constant lower: upper; (-)	AKDE
= pollutant mass remaining cation exchanged; $(\mu g/cm^2)$	PCECU, PCECL
= average intrinsic soil permeability; (cm ²)	К1
= intrinsic permeability of soil zone 1;(cm ²)	K1U,K1L
= depth rain penetration since start of simulation; (cm)	DEPTH
= pore disconnectedness index; (-)	С
<pre>= saturated hydraulic conductivity; (cm/day)</pre>	BK1
<pre>- = ligand input mass; (µg/cm²)</pre>	LIGU,LIGL
= free ligand concentration; (μg/mL)	LICCUF, LICCLF
= pollutant concentration in soil air; $(\mu g/cm^3)$	CUSA, CLSA
	 actu-catatyzed hydrolysis fate constant; (day⁻¹. mol⁻¹/L⁻¹) base catalyzed hydrolysis rate constant; (day⁻¹. mol⁻¹/L⁻¹) soil pll; (-) ratio soil pll lower: upper; (-) ratio degradation rate constant lower: upper; (-) pollutant mass remaining cation exchanged; (µg/cm²) average intrinsic soil permeability; (cm²) intrinsic permeability of soil zone i; (cm²) depth rain penetration since start of simulation; (cm) pore disconnectedness index; (-) saturated hydraulic conductivity; (cm/day) ligand input mass; (µg/cm²) free ligand concentration in soil air; (µg/cm³)

TABLE PT-5 (Continued)

MONTHLY POLLUTANT CYCLE EQUATIONS - LEVEL 3

Fortran Variables

k ₂	-	average permeability of upper and middle zone; (cm ²)	К2
a2 _{OC}		ratio soil organic carbon content middle: upper; (-)	A2OC
a2 CEC	=	ratio soil cation exchange capacity middle: upper; (-)	A2CEC
a2 pli	=	ratio soil ph middle: upper; (-)	А2РН
a 2 KDE	=	ratio soil degradation rate middle: upper; (-)	A2KDE
{NL}	=	concentration of ligand-pollutant complex; (mol/mL)	MLC

3.3.3.2 Constraint Criteria

Five convergence/constraint criteria have been designed to assure a solution within limits (ϵ). These are described below and are graphically presented in Figure PT-4.

(1) $abs(P_{i}[c(t+1)]) \leq abs(P_{i}[c(t+1)])$ (PT-11)

This criterion assures movement of the initial P_i value towards the zero-axis.

(2)
$$\operatorname{abs}(P_i) \leq \varepsilon = 0.01$$
 (PT-12)

This criterion implies convergence to a zero value has occurred.

(3) $(P_i[c(t+1)]) \cdot (P_i[c(t+1)]) < 0$ (PT-13)

This criterion assures the convergence to zero, without overpassing the zero value. In the latter case the value would jump the zero-axis (from the one or the other direction) and would continue indefinitely.

(4)
$$abs(P_{i}[c(t+1)]) = 0$$
 (PT-14)

This constraint reflects the case where no pollution is in the column any more.

(5) c(t+1) has been calculated to 6 significant digits. For extreme concentrations (either very small/very large pollutant load), the equations may not balance to within 1% (criterion 2) due to accumulation of round-off error. In such a case, this criteria forces convergence.

3.3.3.3 Simulation Time Step

For LEVEL2 and LEVEL3 the pollutant cycle equations are formulated on a monthly basis, and the results are reported for each month simulated. However, since all terms that are time dependent are written with an explicit time step (Δ t), the model has to be run for smaller time steps (eg. week, day) in order to account for the non-linear processes (eg. volatilization). In this case the obtained monthly reports (output) represent the iterations and summations of results of many smaller iterations.

The number of iterations per month (NI) is preset in the FORTRAN code. The actual simulation time step (in days) is equal to 30 (davs/month) divided by the number of iterations per month NI, or $\Delta t=30/NI$. A large



FIGURE PT-4: SCHEMATIC OF MATHEMATICAL CONVERGENCE CRITERIA OF EQUATION SYSTEMS

number of iterations per month will increase the accuracy of the model output, and will also result in increased computational time. Presently, the model performs four iterations per month (30/4=7.5 days per simulation). It is felt that this number of iterations is a reasonable compromise between accuracy and computational cost. However, any number of iterations per month can be reprogrammed when necessary.

3.3.4 Moisture Molecule Penetration Constraint

A pollutant originating from any unsaturated soil zone layer will theoretically reach an underlying soil layer (or the groundwater) as soon as a moisture drop originating from the first layer reaches the latter layer (or the groundwater). In practice, however, a "retardation" of the pollutant front will take place with respect to the bulk mass of moisture movement. This retardation is mathematically described by Freeze and Cherry (1979 p. 404) via a retardation factor related to the adsorption capability of the pollutant on the soil particles. If no adsorption is assumed, the pollutant front will follow the seepage soil moisture velocity.

It is not necessary to separately account for such a pollutant in SESOIL, since the pollutant transport routine of the model will retard pollutant mass traveling vertically via the adsorption and other processes modeled. However, it is important to know whether a polluted soil moisture molecule (carrying a dissolved pollutant) originating in an upper layer has penetrated -- even at a negligible concentration -- through the entire layer to the underlying soil layer or has reached the groundwater. If so, the pollutant transport routine of the underlying layer has to be activated; otherwise, no pollutant mass input to the underlying layer will take place. The moisture molecule penetration constraint is incorporated in all levels of SESOIL operations.

Two methods of approximating estimation of the penetration depth of a soil moisture molecule into the underlying soil compartment are relevant to the SESOIL analysis: (1) via Darcy's law, (2) via soil dynamics, and these are combined below into one equation for any soil layer.

The average linear soil flow velocity in a saturated soil is given by Freeze & Cherry (1979, p.71).

$$\overline{\mathbf{v}} = \mathbf{Q}/\mathbf{n} \cdot \mathbf{A} = \mathbf{Q}/\mathbf{n} \tag{PT-15}$$

where

 \overline{v} = average linear velocity

Q = volumetric flux (or specific discharge, or Darcy's
 velocity)

. . .

n = soil porosity

A = cross section of soil matrix (assumed 1)

The average linear soil moisture velocity -- known also as interstitial pore water velocity (Enfield et al 1980) -- is given by

$$\overline{\mathbf{v}}(\Theta) = \overline{\mathbf{v}}/\Theta = Q/\mathbf{n}\cdot\Theta$$
 (PT-16)

where

 $\overline{v}(\Theta)$ = average linear soil moisture velocity

 Θ = soil moisture content

SESOIL employs the theoretical hydrologic routine of Eagleson (1978) as adapted for time dependent moisture storage and transfer in the course of the months. Therefore, Eagleson's principal equations have been also employed here to describe volumetric fluxes in a soil layer.

The monthly average linear volumetric flux or specific discharge in a month for a soil layer is given in SESOIL (eg. HY-28.1) by

$$Q = (I(M) + (Rg(M))/2$$
 (PT-17)

Therefore, by combining above equations, we have the average moisture molecule penetration depth after t months

$$d = t \cdot (I(M) + Rg(M)) / (2\Theta_n)$$
 (PT-18)

where

d = average soil moisture penetration depth (cm)

t = 1,2,3, ... months elapsed; (integer #)

I(M) = monthly infiltration depth to soil layer (cm)

Rg(M) = monthly percolation(depth)from soil laver (cm)

 Θ = soil moisture content (fraction)

n = soil porosity (fraction)

Numerical example (data from Eagleson 1977):

- Assume: t=12 (ie. months), Θ =s_:n=(0.67):(0.35)=0.235, I(M)=63.6/12=5.3 cm

(ie. 5.3 cm/month and for the 12 months), Rg(M)=19.8/12=1.65 cm, n=0.35

- then d = 12(5.3+1.65)/(2*0.235*0.35) = 507 cm (ie. cm/yr)

3.4 Storm-by-Storm Po.lutant Cycle (LEVELN)

The LEVELN of SESOIL has not been developed; however, designers have conceptualized the development of this model feature. For additional information call 617/864-5770.

4.0 DISCUSSION

SESOIL is a "user frierdly" model and as such the pollutant cycle subroutines are designed to be easily expandable. Processes not currently included in the simulation can be incorporated simply by adding another term to one of the mass balance terms. For example, an expression for the degradation of pollutant by soil bacteria could be added to the model by including the bacterial degradation term with the other pollutant transformation equations.

The pollutant mass added to the groundwater is estimated by all levels of operation of the model. The behavior of pollutant within the groundwater is not currently described by SESOIL, although the simulation of a groundwater "layer" can be developed and model developers have some long-term plans. However, SESOIL is also adaptable to provide information for (and/or to be interfaced with) other groundwater models of the literature.

5.0 REFERENCES

Bonazountas, M.; W. Hawes; W. Tucker (1979). Heuristic Unsaturated Flow/Quality Soil Zone Model. AGU, Spring Meeting, Washington, DC.

Eagleson, P.S. (1978). Climate, Soil, and Vegetation (1-7). Water Resources Research, Vol. 14, No. 5, pp. 705-776.

Enfield, C.G.; R.F. Carsel; S.Z. Cohen, T. Phan & D.M. Walters (1980). Method of Approximating Transport of Organic Pollutants to Groundwater EPA - Internal draft. EPA - Robert S. Kerr, Environmental Research Laboratory, Ada, Oklahoma.

Farmer, W.J.; M.S. Yang; J. Letey; W.F. Spencer (1980). Land Disposal of Hexachlorobenzene Wastes: Controlling Vapor Movement in Soil, EPA-600/2-80-119, Office of Research and Development, U.S. Environmental Protection Agency, Cincinnati, Ohio.

Fiksel, J.; M. Bonazountas; H. Ojha; K. Scow (1981). An Integrated Geographic Approach to Developing Toxic Substance Control Strategies. Arthur D. Little Report to U.S. EPA, Office of Policy and Resource Management, Washington, DC. EPA Contract No. 68-01-6160.

Freeze, F.A.; J.A. Cherry (1979). Groundwater. Englewood Cliffs, NJ: Prentice-Hall, Inc.

Hamaker, J.W. (1972. Diffusion and Volatilization, Chapter 5 in Organic Chemicals in the Soil Environment, Vol. 3, G.A.I. Goring and J.W. Hamaker (eds.), Marcel Dekker, NY.

ID - input data

SESOIL updites Sin not church

TABLE OF CONTENTS

			Page
LIST	OF TA	ABLES	ID-3
LIST	OF FI	LGURES	ID-4
1.0	INTRO	DDUCTION	ID-6
2.0	CLIMA	ATOLOGICAL INPUT DATA	ID-6
	2.1	General	ID-6
	2.2	Manual Estimations of Climatologic Data	ID-7
		2.2.1 Monthly Time-Specific Simulations	ID-12
		2.2.2 Monthly Long-Term Simulations	ID-16
		2.2.3 Annual Time-Specific Simulations	ID-16
		2.2.4 Annual Long-Term Simulations	ID-19
	2.3	Automated Estimation of Input Parameters	ID-19
		2.3.1 Monthly Time-Specific Simulations	ID-19
		2.3.2 Monthly Long-Term Simulations	ID-23
		2.3.3 Annual Time-Specific Simulations	ID-23
		2.3.4 Annual Long-Term Simulations	ID-23
		2.3.5 Program Notes	ID-23
3.0	SOIL	ÍNPUT DATA	ID-27
	3.1	Summary of Default Values	ID-27
	3.2	Soil Intrinsic Permeability k(1)	ID-29
		3.2.1 Definitions	ID-29
		3.2.2 Hydrologic Characteristics of Soil Types	ID-31
		3.2.3 Soil Hydraulic Conductivity K(1)	ID-33
		3.2.4 Factors Affecting K(1)	ID-35
		3.2.5 Guides to Estimating K(1)	ID-37
		3.2.6 Mean Soil Particle Size Estimation	ID-37
		3.2.7 K(1) vs Particle Size	ID-37
		3.2.8 Section of Working Values/Curves	ID-42
		3.2.9 $K(1)$ vs $k(1)$	ID-44
		3.2.10 k(1) SESOIL Default Values	ID-44
	3.3	Soil Effective Porosity n	ID-46
		3.3.1 Definitions	ID-46
		3.3.2 Soil Hydraulic Properties	ID-49
		3.3.3 n SESOIL Default Values	ID-51
	3.4	Soil Disconnectedness Index c	ID-51
		3.4.1 Definition	ID-51
		3.4.2 The c Index Sensitivity	ID-54
	3.5	Soil Parameter Calibration	ID-59
		3.5.1 General	ID-59
		3 5.2 Calibration of k(l), c via s	ID-59
		5.3 Calibration of s via k(l), n_{2} , c	ID-62
		5.4 Automated Calibration	ID-62

TABLE OF CONTENTS (continued)

		Page
4.0	CHEMISTRY INPUT DATA	ID-63
5.0	CANONICAL CLIMATIC-SOIL COMPARTMENTS	ID-63
6.0	REFERENCES	ID-64

•

LIST OF TABLES

Table No.		Page
ID-1	SESOIL Default Soil Input Parameters	ID-30
ID-2	Range of Values of Intrinsic Permeability and Hydraulic Conductivity; Unit Conversion Factors	ID-32
ID-3	Hydraulic Conductivity Classes According to the USDA-SCS	ID-34
ID-4	General Relationship Between Soil Texture and Saturated Hydraulic Conductivity	ID-40
ID-5	Soil Texture, Representative Particle Size Contents, and Mean Diameters	ID-41
ID-6	Soil Texture vs Particle Diameter ^(d) , Soil(k) Conductivity and Intrinsic Permeability ^(k)	ID-45
ID-7	k(1) Default Values for SESOIL	ID-47
ID-8	Range of Values of Porosity	ID-48
ID-9	Representative and SESCIL n -Default Values	ID-52
ID-10	Default c Values for SESOIL	ID-58

LIST OF FIGURES

Figure No.		Page
ID-1	Survey of TD-1440 For: Montana	ID-8
ID-2	Sample NOAA Monthly Data Sheet	ID-9
ID-3	Sample NOAA Annual Summary Data Sheet	ID-10
ID-4	Sample Data Matrix of Input Parameters for a Two-Year Monthly Simulation	ID-13
ID-5	First Sample Data Block from Monthly NOAA Data Sheets	ID-i4
ID-6	Middle Sample Data Block from Monthly NOAA Data Sheets	ID-15
ID-7	Sample Hourly Precipitation Data Block from Monthly NOAA Data Sheets	ID-17
ID-8	Sample of Annual NOAA Data	ID-13
ID-9	User Input Data File (IP DATA) sample Values and Formats	ID-20
ID-10	Sample Program Output Monthly Parameter Values for the Water Years 1945-1950	ID-21
ID-11	Sample Program Output Monthly Parameter Averages from 1945-1950	ID-24
ID-12	Sample Program Output Annual Parameter Averages by Year from 1945-1950	ID-25
ID-13	Sample Program Output Parameter Average for All Observations from 1945-1950	ID-26
ID-14	Generalized Correlation of Independent Soil Parameters; k(l), n , c e	ID-28
ID-15	Guide for USDA-SCS Soil Textural Classification Showing Points for Which Mean Particle Diameters Have Been Calculated	ID-38
ID-16	Comparison of Various Soil Particle Siz Ranges Used by Various Agencies	ID-39

LIST OF FIGURES (continued)

Figure No.		Page
ID-17	Soil Texture Permeability Curves	ID-43
ID-18	Relationships Between Porosities and Soil Grain Size	ID-50
ID-19	Hydraulic Conductivity vs Soil Moisture	ID-53
ID-20	 Functional Relationship, c-f(S); Saturated Intrinsic Permeability 	ID-56
ID-21	Saturated Permeability vs Pore Size Distribution Index (from Eagleson, Personal Communication)	ID-47
ID-22	Water Balance Solutions Using Soil Properties from Equation ID-21	ID-61

1.0 INTRODUCTION

Four major input data categories are required by SESOIL

- (1) Climatological Data
- (2) Soil/Vegetational Data
- (3) Chemical Data
- (4) Application Specific Data

Data for these categories are handled and input to the model in different ways, depending upon the simulation level. Currently four levels of operation are available, varying from general "Exposure Assessment" simulations (annual, monthly) to "site-specific" simulations. The four levels of operation are

- LEVELO Annual General Exposure Assessment Simulations (2 soil layers)
- (2) LEVEL1 Annual Site-Specific Exposure Assessment Simulations (2 soil layers)
- (3) LEVEL2 Monthly Site-Specific and Exposure Assessment Simulations (2 soil layers)
- (4) LEVEL3 Monthly Site-Specific and Exposure Assessment Simulations (3 soil layers)

This appendix will provide in the future a thorough background to effectively compile all input data for all above categories, in order to <u>prove</u> to potential users how <u>easy</u> it is to run SESOIL contrasted to other sophisticated models of the literature. The following sections, however, give detailed description only for the climatologic data compilation, since time and budget constraints prevented the developers (Bonazountas, Wagner 1981) to invest more time in this aspect of the model use.

Data for all data categories are stored in permanent SESOIL data files for easy retrieval at any time.

2.0 CLIMATOLOGICAL INPUT DATA

2.1 General

Site-specific simulations require monthly or annual and time and sitespecific climatological input data. Non-site-specific (hypothetical) simulations require long averaged (monthly or annual) climatological input data. Data compilation for these for types of simulations can be performed

- Manually from the National Oceanographic and Atmospheric Administration (NOAA) climatological data records (sheets); and
- (2) Computerized via a subroutine that compiles on-line data from NOAA climatological data tapes.

The LEVELO simulation does <u>not</u> require climatologic input data because the hydrologic cycle components are input to the model by the user (see Section 3.0, User's Manual). However, this level of operation requires soil/vegetational, chemical and application specific data, which can be compiled and stored permanently in the SESOIL data files as described in Section 3.0 through 5.0 of this appendix.

The climatological input data for levels 1, 2 and 3 can be obtained from NOAA records in the form of data sheets or data tapes. Information can be requested -- by indicating the weather station number and location (Figure ID-1) -- from the

U.S. Department of Commerce, NOAA National Climatic Center, Federal Building Asheville, North Carolina 28801 704/258-2850 (ext. 208, Digitized Data Dept.)

The following paragraphs describe the two methods of analyzing NOAA climatological data for input to SESOIL: (1) hand calculation based on the NOAA Monthly Data Sheets, and (2) a computer subroutine (IPDATA) which uses NOAA data tapes. The hand calculation method will be described first to provide some background on the ten parameters and how they are delivered.

2.2 Manual Estimations of Climatologic Data

NOAA reports provide daily, monthly (Figure ID-2) and annual (Figure ID-3) summaries of climatologic data for designated sites throughout the United States.

The ten SESOIL climatologic input parameters are

- L latitude; (N°)
- TA temperature; (°C)
- NN fractional cloud cover -- 24 hrs average; (-)
 - S humidity (fractional)
- A albedo; (-)
- REP rate of evapotranspiration; (cm/day)
- MPA mean (annual or monthly) precipitation; (cm/yr or cm/month)

SURVEY OF TD-1440

FOR: MONTANA

STATION NO.	SERVICE	STATION		PERIOD		
						•
24033	W	Billings/Logan Fld.			01/48-	
24132	F	Bozeman/Belgrade/Gallatin			01/48-12/54	
24135	F	Butte/Silver Bow Fld.			01/48-12/60	
24040	F	Custer			01/48-05/50	
24137	F	Cut Bank/ Mun			01/48-	
24138	F	Dillon/Beaverhead Cnty.			01/48-06/73	
24139	F	Drummond			01/48-12/54	
24034	W	Glasgow/WBO		×	01/48-10/55	
94008	W	Glasgow/Int.		×	10/55-	
94010	Α	Glasgow			10/58-06/68	
24143	W	Great Falls/Int.			01/48-	
24112	A	Great Falls/Malstrom			01/49-12/70	
24035	W	Havre/WBO		*	01/48-12/48;	
				*	05/50-05/61	
94012	W	Havre/City-Cnty.		*	02/61	
24144	W	Helena			01/48-	
24146	W	Kalispell/Glacier Prk.	ŧ	*	01/48-	
24036	F	Lewistown/Mun.			01/48-	
24150	F	Livingston/Mission			01/48-12/54	
24037	F	Miles City/Mun.			01/48-	
24153	W	Missoula/Johnson-Bell Fld.			01/48-	
24159	F	Superior			01/48-11/53	
24161	F	Whitehall			01/48-12/54	

.

SAMPLE NOAA MONTHLY DATA SHEET

JANUART 1978 TOPERA, KANSAS AT MEATHER SERVICE FOST OFC

Local Climatological Data



ID-6

SAMPLE NOAA ANNUAL SUMMARY DATA SHEET

Local Climatological Data Annual Summary With Comparative Data 0 1979



Page 1 of form

TOPEKA, KANSAS

Narrative Climatological Summary r deriv (frem 20 postaria of alla a mod protocity)(pation n – Mar (all a of all a port for a union post post Ang danga filly Jarlag (for highest pation a basily mini 1 0 :: 0 to also a set of the s

NOAS

Ś С Meteorological Data For The Current Year - _ _ 5 March Long Sol 1111 , ÷ 1 -: . 7 . I E ···:. ., II-: -7

Normals Means And Extremes

· · · · · · · · · · · · · · · · · · ·			
and a second sec			
the second secon		· · · · · · · · · · · · · · · · · · ·	. –
		· · · · · · · · · · · · · · · · · · ·	
· · · · · · · · · · · · · · · · · · ·	1 11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		-
		1 1 1 1 1 2 1 2 1 2 2 2 2 2 2 2 2 2 2 2	1
	'		•
		1	
			1 21
			1 2.
and a start of the land and a start of the	and all fames '		
and the sector of and and and and and a sector of	with many the term of the standard	مسيب واسلوه ليواجد مناها مستخلصا	
the second	where provide the second se		
A to H to see to the same set of same a long and	and the second se		

	-	

ţ

FIGURE ID-3 (Continued)





Page 3



ο

.

- MTR mean (annual or monthly) storm duration; (days)
- MN mean number of storm events during (annual or monthly) simulation period; (-)
- MT mean length of rain period (in a year or in a month);
 (days)

Aside from L all other data for the above parameters are handled in four different ways when compiling SESOIL input, depending upon the simulation level; namely depending upon the time interval (months or years) and the time period (specific or general/long-term) analyzed.

2.2.1 Monthly Time-Specific Simulations

Values for all ten parameters must be provided for each month of year(s) under study in a 10 x 12 matrix, as shown in Figure ID-4. Matrix elements for each month are obtained or calculated from the NOAA climato-logical forms/sheets as follows

- L The latitude is given as two digits, minutes only, and is found in the top left corner of the NOAA forms (Figure ID-5). Latitude can be also expressed (input) in decimal degrees; eg. 39° 04' = 39+(4/60) = 39.067°.
- TA The average temperature for the month is given in $^{\circ}$ F at the bottom of Column 4 (Figure ID-5) and must be converted to $^{\circ}$ C via C $^{\circ}$ = 5(F $^{\circ}$ -32)/9.
- NN Cloud cover is taken as the average monthly sky cover (in tenths) from column 22 (Figure ID-5) and is divided by 10 to convert to fractional cloud cover.
- S The average monthly humidity is estimated by averaging the relative humidity given for the 8 observations during the day in the data block shown in Figure ID-6, and is divided by 100 to convert to fractional humidity.
- A Albedo is taken from the corresponding table of Appendix HY (Table HY-1).
- REP Because evapotranspiration is estimated by SESOIL from the five climatic parameters previously described, this data entry can be input as 0.0. In case site-specific values are available (e.g. Table HY-1), users may input REP and disregard compilation of the previous five parameters.
- MPM The mean monthly precipitation is given at the bottom of column 10 (Figure ID-5), in inches and must be converted to centimeters (1 inch = 2.54 cm).

SAMPLE DATA MATRIX OF INPUT PARAMETERS FOR A TWO-YEAR MONTHLY SIMULATION

	7(1-1+	4. KA1.9	• • • •	C 10100 0	1.'	- 51 8 181		376	•			
	16/70	11/75	127.2	1272	211.	1/7 .	4/73	4/73	1/12	7/7.	1.	.11.
ł	34.0	37.0	1 *	1.1	1	27.1	1)	11.0	11.0	4.2.		
T :	12.50	* . 11	.J .	C .	1.1	4.11	1	14.00	2.2	1 1 1 1	1.11	
• • •	• 32	. 69		. ()	. ()	. 67		- 60	54	. ()	4.4	
<u> </u>	. (513	. 745	0		7 . 6	. 7500	7 . 7	5 . 71 4	7.1	7	/ . 7	. 7.0.0
7	0.1	0.1	0.1	0.1	0.1	0.1	0.1	0.1	0.1	1).1	0.1	0.1
110	0.0	5.0	0.0	6.0	0.0	0.0	e . 9	0.0	0.0	2.0		0.0
115 11	0.91	· · 1 ·	1. '	4.14	1.10	10.01	()		14. 20	1	10.	1
·· T I	1.	17		. 2 4	. 17	. 14	-1.3	. 1 3	. '0			
	2.0	10.0	· ·	· . (·	1 a 11	11.0	1(.)	10.0		1	11.	4
•• T	30.5	11.5	5).	1 C	10.1	30.	24.	36.10	30.	· · · ·	••••	10.1
	TLUE				1070			000				
		на кала: 11/76	· · · · · ·		· 17/7 ·	こうだい ししゃ	м(н) на та 1			7 4 7		
	70770	11/7		1//9		1717	4///		6//	////	1.1.1	- / / .
L.	39.0	37.0		··· ()	14.6	34.0	· · · · ·	34.0	39.0	17.0	. · · · · ·	
10	1.5 • 35	4.44	1 • • • "	· •	ť•	4 • · · ·	1	17.11	5 4 . 72	· · •		د ۱ - ۱
• •	• 4 0	• 4 4	• • •	• (4	• * *	• ^ 4	• 17	• * • 1	• 14	• 1		• 4 7
۰.	• E H 7 5	• • • • • •	• • • • •	• 7013	• 10 2	· ', ', 7 ',	• 1 M > 0	• • • •	• ^ ! · 1 - !	• ', * ''	• •	. / 1 .
A .	C • 1	J.1	J • 1	L.1	6.1	(· • 1	(•1	(· • 1	$0 \cdot 1$	0.1	· • 1	• • 1
116	0.0	0.0	9.0	C •0	0.0	0.0	0.0	0.0	0.0	9.0	4.0).(
1.1.1.	10.4	4.57	0.13	2.4	2.31	10.54	2.52	12.32	12.17	2. 11	14.24	3.02
51 T 4	.27	.24	4	. 34	• 27	. 30	-14	.05	. 09	. 1 15	. 1 4	.00
1 * * s.	7.0	6.0	1.4	•••	7 • C	10.0	(.)	10.0	3.0	5.0	12.9	5 . C
	36.45	3.3. 5	10.1	31.41	1 *	5.1.5	4	111.1	311.1.	11		

FIRST SAMPLE DATA BLOCK FROM MONTHLY NOAA DATA SHEETS



ID-11

Arthur D Little, Inc

MIDDLE	SAMPLE	DATA	B	LO	CK	FR	OM	MC	ONT	HLY	1	IOAA	DATA	SHEETS
			SUMMARY BY HOURS											
				OVEROGES PESULIANT										
			100 HOUP	SAT COVER	B FAF LON PAESSURE 1 N	15 FT	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		151.01 14E	(140 SFEED 7 7.4	I RECTION	3PLE0 7 P.L		
			00 03 05 12 15 15	5 6 6 7 7 7 6	29.33 29.32 29.32 29.35 29.35 29.35 29.29 29.30	15 13 15 21 25 22	14 13 12 13 19 22	0 7 10 12	78 79 78 74 81 89	7.8 6.8 7.3 8.8 9.8 10.6 7.5	34 34 33 34 33 33 33 35	3 9 3.3 4 1 4.3 3.8 4.3 3.8		
		L	ال ا S	<u>6</u>]	29_32 Σ i=	1	تد_ا hun	nid B	it	y _i)	. /	100	•	

FIGURE ID-6

- MTR Mean storm duration is obtained from the Hourly Precipitation chart shown in Figure ID-7. Three conventions are followed in demarcating storm events
- (1) Hours showing the symbol "T" for "trace" precipitation are not included in storm duration if they appear at the beginning or the end of an event, or if they occur alone.
- (2) "Trace" hours are counted in the middle of a rain event.
- (3) Rain events that continue into the next month are included in the month which had the most hours of that event.

The mean storm duration (MTR) is obtained by: counting the number of distinct storm eve ts (MN); counting the total number of hours with quantifiable amounts of precipitation and those hours with trace amounts according to convention (2) above; and dividing the total hours by the number of events. This result must be divided by 24 (hours) to convert the units into days.

- MN The mean number of storms is reported as counted above for determing the MTR.
- MT Presently the model does not distinguish between months with 30 or 31 days, therefore, for a month full of rain the length of the monthly time interval MT is input as 30.5 (365÷12). In the near future, the simulation will handle the exact number of rainy days during a month (as if they were the rainy season of a year). MT in this case will be obtained by bracketing the rainy days as shown in Figure ID-7.

2.2.2 Monthly Long-Term Simulations

If the simulation is to provide a monthly analysis of an unspecified time period, data for at least 5 specific years can be averaged to "damp out" annual variations. The parameters which do not require averaging are: L, A, REP and MT. The remaining parameters are first handled as described above for each month of the 5 years, (with the exceptions of TA and MPM because the mean temperature (TA) and the mean rainfall (MPM) for the period of record are provided by NOAA with the annual summary, as shown in Figure ID-8) and then averaged.

2.2.3 Annual Time-Specific Simulations

To perform an analysis which covers a user specified period of time in annual time-steps (LEVEL 1), data analysis is carried out by averaging


<u>Notes</u>: 1. This figure shows 6 rain events according to SESOIL conventions. The sixth may carry over into the next month and requires verification prior to assigning it to either. For demonstration purposes, however, it is assumed to be one complete event. Thus MN=6.

1

- There are 14 hours of rainfall contained within the events bracketed above. Thus MTR=0.43.
- 3. The value of MT is given as 30.5 even though the rainy period covers 25 days of the month.

Arthur D Little Inc.

SAMPLE OF ANNUAL NOAA DATA

Average Temperature

Precipitation

۰

The part is an iso and part iso						Max	lung	in the second	Aug	Sent	Oct	Nov	Dec	Annual		Vear	1.00	Feb	Mar	Anz	May	hune	luly	A.110	Sent	Oct	Nov	Dec	Annual				
The monthly averages; for no specific time. TA monthly for a specific year. TA monthly for a specific year.	Year	Jan	reo	Mar	Apr	may	June	JUIY	AUK 1	Sehr 1	UCI 1	100	Uec			1041	2011			~ ~ ~	may	June	30.7										
The monthly averages; for no specific time. TA monthly, for a specific year. TA monthly, for a specific year.	1939	20.1	37.2	51.0 •7.0 •7.2	34.3 34.3 34.3	64.6 70.6 64.9	79.2		77. e	77.0	62.2 63.2	•0.1	39.7	54.4 54.4		1939	1.14 0.70 1.32	1.92	1.27	3.50	1.33	2.3	1.20	3.91	1.29	2.04	;;;	0.74	28.33				
<pre> A to be a set of the se</pre>	1941 1942 17 1 1944 1945	31.0 31.0 20.1 30.2 31.5	32.3 32.4 37.2 36.4 3•.8	40,4 44,6 38,8 39,4 31,0	58.6 60.2 37.7 30.2 51.6	70.4 63.7 62.7 69.1 61.0	74.8 74.2 74.7 77.0	81.2 ⁴ 87.1 81.8 79.0 79.0	70.8 77.2 82.6 77.6	71.3 67.9 66.7 70.5	59.5 59.3 56.4 50.8 57.6	45.7 44.2 41.4 40.4	34.0 30.4 70.2 24.0 25.4	97.0 95.7 93.2 99.9 59.9		1941 1942 1943 1944 1944).70 0.11 0.15 0.54	0.31 1.85 0.64 0.97	1.41 1.D1 0.60 4.93	3.32 4.16 1.40 8.65	3 78 5.29 4.93 2.81 6.68	3.19	1-38 1-92 4-29 3-43 2-75	7.74 7.72 2.10 6.91 0.87	3.81 4.86 4.00 3.01 6.20	10.61	0.01	2.01 2.74 1.76 5.04	44,45 40.97 34,30 44,32 38,35				
<pre> TA monthly averages; for no specific time. TA monthly for a specific year. TA monthly for a specific year. TA monthly for a specific year. TA</pre>	1748 1747 1748 1747	33.4 31.4 23.8 23.4	43.7 27.4 18.6 30.0 34.9	55.4 36.0 37.3 61.2 39.1	59.8 59.0 50.5 57.8	62.0 60.4 63.6 67.6	76.5 71.6 73.0 73.8 73.8	81.0 76.1 78.1 79.2 71.9	77.4 84.4 77.9 75.6 71.7	67.2 72.6 70.8 63.4 68.7	59.2. 65.4 34.8 38.0 62.6	24.6 53.4 67.6 27.6	37,4 34,4 33,4 35,6 29,9	50.1 54.3 53.8 54.1 52.9	•	19+6 1947 1948 1949 1950	1.98 0.43 0.40 5.24	0.52	2.83 3.74 4.23 2.23 0.63	3.1 0.01 1.7 9.41 2.01	3.3 2.68 4.85 5.16 4.12	\$.5 \$.5 \$.5	0.84 3.57 5.50 4.35 12.08	6.30 3.37 2.71 7.01 5.01	3.33 1.13 1.32 1.77 2.39	1.42	1.02	1.87 3.94 6.78 1.93 0.01	23.03 36.98 30.72 45.73 38.91				
<pre></pre>	1751 1752 1755 1755	20.0 32.0 33.6 27.6 31.1	34.1 38.6 37.6 44.4	37.8 39.4 45.0 41.0 42.0	30.5 33.0 49.7 60.5 61.4	64.7 69.0 63.7 60.0	69.1 87.1 81.7 77.1 70.9	74,9 80,1 79,3 65,3 64,3	17.3 78.0 77.5 80.* 79.8	64.0 70.0 71.0 79.2 72.4	36.2 53.0 61.3 59.1 39.1	30.2 41.6 43.5 47.5 34.4	29.5 31.4 34.9 34.8 30.1	92.3 93.3 57.0 57.4 58.3		1931 1932 1933 1934 1934	1.00	1.53 0.46 1.57 8.04 2.52	1.80 2.76 3.10 0.67	3.00 3.00 1.32 2.22 2.51	4.12 3.70 4.53 4.82 9.62	10.54	11.92 0.09 4.13 3.25 2.25	9.01 9.11 9.11 9.75	6.94 0.68 2 71 0.70 7.51	3.07	1 31 2.20 2.31 7	0.41	48.40 21.33 24.31 30.03 28.44				
<pre>its its its its its its its its its its</pre>	1956 1957 1958 1959 1959	29.7 23.1 30.9, 24.0, 24.0,	32.9 36.4 26.2 31.7 20.4	•3.4 •2.0 36.4 ••.7 27.4	51.6 52.3 53.1 53.0 30.1	67.3 67.8 67.8 65 8 63.3	76.8 72.2 71.0 74.3 78.2	0.0 0.5 75.1 74.4 77.0	79.6 78.4 76.2 80.2 78.2	70.9 63.0 67.5 67.6	82.8 33.4 38.4 32.6 38.4	42.3 41.4 46.9 96.4	34.4 37.6 29.9 37.8 31.0	55,4 53,6 53,5 53,5 53,6 54,6		1936 1937 1938 1938 1939	0.79 0.39 1.32 0.77	0.70 0.36 1.13 0.58 2.19	2 - 50 2 - 50 2 - 50 2 - 50 2 - 50	3.94 5.01 1.76 1.76 2.21	2.30 9.18 2.29 4.70 1.70	5.14 5.10 7.53 2.41	4.44 5.21 6.30 2.92 3.10	2.35 1.46 3.71 2.65 3.70	1.02 5.15 3.76 2.68	1.33	1.90 1.91 3.21 0.11	0.70	25.35 38.14 38.73 27.10 25.64				
<pre>iss st. st. st. st. st. st. st. st. st. s</pre>	1761 1762 1763 • 1764	20.8 21.2 10.0 34.2 31.0	35.6 37.8 37.5 34.3 30.6	43.3 38.4 47.3 40.0 31.0	50.2 52.2 57.4 54.4 57.2	5°.8 72.7 63.3 68.9	12.0 12.5 17.5 18.1 18.1	76.7 76.2 80.8 81.2 77.0	74.4 78.7 78.5 74.5	69.0 69.4 71.9 68.1	57.0 60.0 67.6 56.6 37.3	41.0 44.0 47.4 46.2 49.7	25.0 91.9 23.4 29.9	53.4 53.7 57.6 57.0 54.6		1963 1963 1964 1964	0.01	1.34 1.28 0.16 0.29 1.27	6.32 1.43 2.53 1.71 1.36	2.61 1.21 0.81 4.70 2.31	9.24 4.23 4.9L 2.09 9.41	2.74	7.25 3-68 3-13 1-81	1.92 6.64 1.30 6.24 3.93	5.64 4.50 1.11 1.11 7.22	4,88 1,36 0,94 0,14	2.30 1.04 0.74 0.27	1.01 0.43 0.23 0.94 2.44	41.00 32.26 19.07 33.90 37.97				
TA monthly averages; for no specific time. TA monthly for a specific year. TA	1708 1787 1968 1968 1968	26.0	31.0 33.4 31.6 33.9	47.0 47.0 56.4 37.1	30.6 37.4 53.5 51.5 51.5	63.8 60.8 39.1 63.3 69.1	13.7 17.1 74.1 44.7 71.4	83.1 79.0 76.4 79.4 77.8	74.8 72.1 79.9 79.5 80.6	67.8 07.0 60.0 67.0	51.4 24.7 57.3 52.7 54.2	44.1 41.7 40.0 42.3 41.2	30.7 33.0 29.0 29.9	53.8 53.5 53.0 52.9 54.1		1966 1967 1968 1969 1970	0.10	0.34 0.2L 0.34 0.34	0.10 1.91 0.46 1.17 1.02	1.94 0.12 0.20 7.14 3.49	0.41 5.07 3.37 3.77 3.44	8.87 13.20 2.14 8.49 5.65	0.75 3.05 10.17 3.20 1.30	3.62 1.84 7.40 0.87 0.83	1.46 9.64 2.50 2.03 7.70	0.42 4.10 3.90 2.49	0.24	0.74 3.11 2.13 1.29 1.65	19.50 30.44 40.56 33.46 31.62				
TA monthly averages; for no = TA specific time. TA monthly for a specific year. TA monthly. for a specific year. TA monthly. for a specific year. TA	1971 1972 1973	24.1 25.6 27.5	26.0 .31.0 .33.9 	41.1 66.7 67.3	56.9 54.7 57.6	61.4 63.5 61.3	77.9	74.9	79.4	71.6	01.0 34.0 00.4 39.3	43.4 37.4 44.8	34.8	54.2 93.0 54.5	1_	1971 1972 1973 1974	15.1 74.0 74.5 74.5	3.40 0.50 3.71 1.20	0.64	1.08 3.91 4.03 3.74 8.20	4.83 2.90 4.37 3.80 3.88	3.10	4.81 4.81 10.16	0.20 3.20 2.84 1.04	1.35	3.87 2.11 4.57 5.14	20.6 29.6 21.5 21.5	1.81 1.76 4.30	20.73 31.21 60.89				
TA monthly averages; for no = TA specific time. TA monthly for a specific year. TA	1976	27.8	\$2.7 37.5		\$7.0 60.2	60.4 70.1	72.7 73.2	78.0 79.9	76.9 76.4	69.8 71.6	50.3 58.7	19.4 42.7	24.9 30.1	19.7 59.4	Γ,	1976 1977	8. 1	0.51	1.38	4.85	4.63 7.83	1.69	2.0 1.37	11.10	1.12	3.01 9.72	0.04 3.39	0.21	20.75	I			
= TA monthly averages; for no specific time. TA monthly for a specific year. = MPM, monthly; for a specific year. = MPM, monthly; for a specific year. = MPM annual general = TA = MPM, monthly; for a specific year. = TA	HEAM MAX MIN	31.0	- 12-1		41.0 66.3 53.7	44-3 75.3 51.7	79-2 63-9	- 78 - 9 - 8 - 6 - 7 - 9	77.4 88.5 88.3	69.2 66.6 97.7	57.8 69.6	43.5 34.0 32.9	32.6 41.9 23.3			RECORT	0.17	1.27	2.04	1.01	4.30	4.7	_1.92	4.04	3.5	2,33	1.64	1.	33,41				
specific time. TA monthly for a specific year. TA monthly for a specific year. annual = MPM, monthly; for a specific year. annual = MPM ann general = TA general = TA	4	= 1	CA n	iont	h1y	av	erag	ges;	fo	or n	0		=	TA			= M	РΜ,	moi	nth:	ly;	for	no	sp	eci	fic	tin	ne.	= MPN	1			
	TA 1	s non t	pec hly	ifi ,fo	c t r a	ime sp	ecif	ic	yea	r.		1	ant gene		TA	= MPI	м, п	ion t	hly	; f	or a	a sp	peci	fic	ye	ar.		an ge	nual nera	= 1	MPM spe	ann ecif	u i

ID-15

Arthur D Little, Inc

the twelve monthly averages for the year for the following five parameters: TA, NN, S, MTR and MN.

Parameters L, A, and MT do not change with time. The value for total annual mean precipitation (MPA) is found either by adding the average monthly precipitation (MPM) for all twelve months or from the totals provided in an annual summary table (Figure ID-8) and converting from centimeters to inches (if not previously converted). While MT varies seasonally, the variation is considered only for annual simulations.

2.2.4 Annual Long-Term Simulations

The data for a general annual simulation (i.e., an unspecified time period) are averaged over annual data for five or more specific consecutive years. The specific years' data are determined as described above, excepting the values for MPM and TA, which may be taken directly from NOAA data, by referring to the 40 year mean for the period of record.

2.3 Automated Estimation of Input Parameters

The ten parameters described in Section 2.2 above are determined automatically by the IPDATA subprogram, using tapes for data input. Two of the parameters (latitude and albedo) require user input to the subprogram.

The subprogram reads the data taken at a weather station from two NOAA tapes (TD-1440 and TD-9924), obtainable from the National Climatic Center by calling their Asheville offices. This Office also supplies an index of station numbers for the U.S. and the world where observations have been made called "SURVEY of TD-1440, DATA FORMAT," and user manuals for the tapes.

The data stored on TD-1440 tapes are used to calculate the monthly values for the following six parameters: TA, NN, S, MPM, MTR and MN. In addition, the IPDATA subprogram stores the total number of hours precipitation (TPPT) counted in each month. The value for REP is supplied as 0.0 by the subprogram, while MT is supplied as the number of days in the current month. The user may override the value for REP if desired by editing the input parameters (see Section 3.0, User's Manual). This would occur only in the case where certain of the input parameters were lacking or poorly supplied with data. The user might determine this either before or after scanning the data on the tapes. The second tape, TD-9924 is required to obtain values for mean monthly precipitation.

2.3.1 Monthly Time-Specific Simulations

To run the IPDATA subprograms for a simulation which will produce 10x12 matrices (10 parameters, 12 months) for all the years within a specific time period, the user first edits the FILE IP DATA. This file, along with the appropriate formats are shown in Figure ID-9. All entries but the value for REP must be changed for each site. Typical Annual Output is shown in Figure ID-10.

USER INPUT DATA FILE (IP DATA) SAMPLE VALUES AND FORMATS

STATION NAME ((STANAK)	GUANT	ANAMC. C	AY
HINST YEAR THY OUTPUT FILE O ALGEDO, LATITU	(NFILE), DIVISION(NDIV) JJC (_AT), REP	45 99 1.u	50 59 30.0	C. 0
Line Number	Format			
1	40x-, 10A4			
2	40X, I2, 5X, I2			
3	40X, I2, 5X, I2			
4	38X, 3(F7.2)			

	040300 9) (J 040400	********* 04/17/2/14/14 *******	• + + + + + + + + + + + + + + + + + + +	¢ • • • • • • • • • • • • • • • • • • •	00)0v0)9 0000c0)0	30000060 1170 20200306	000000000 6 79 4000000	***********	********** *********	**********		\$\$\$\$\$\$\$\$\$	*********
e) e	****	0101100000 Paka4211	400000000 ER VALUE	10000100 5 FOR TH	60000000000000000000000000000000000000	00003064 YEAR 194	60000665 5 TU 194	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	*******	********		0000(006	******
		UCT	NON	DEC	JAN	F=B	MAR	APRI	L MAY	JUNE	JULY	AUG	SEPT
	L TNN S A T E M M T M N T P T T A A A A A A A A A A A A A A A A	3 C • 0 2 o • 5 0 • 7 0 • 7 1 • 0 0 • 2 1 • 0 3 1 • 0 4 o • 0	30.C 23.0 9999.0 0.8 1.0 0.5 2.5 0.1 17.0 30.0 43.0	30.0 23.7 9999.0 0.3 1.0 0.0 5.1 0.1 10.7 31.0 22.0	30.0 24.2 9799.0 3.3 1.0 0.3 7.5 0.3 1.3 31.3 1.0 1.0	30.0 23.9 9,99.0 0.7 1.0 0.C 10.2 0.1 4.0 28.0 7.3	30.0 24.4 9999.0 0.7 1.0 0.0 12.7 0.0 10.0 31.0 15.3	30.0 25.0 999.0 0.7 1.0 0.3 15.2 0.3 14.0 30.0 90.0	30.0 26.1 2999.0 0.8 1.0 C.0 17.8 0.1 15.0 31.0 39.0	30.0 26.9 9999.0 0.3 1.0 0.0 20.3 0.0 20.3 5.0 30.0 5.0	30.0 27.5 9599.0 U.7 1.0 0.0 22.9 0.1 5.0 31.0 8.0	J0.0 27.7 9959.0 0.7 1.0 C.C 25.4 0.1 5.0 31.0 8.0	3C.0 27.1 9999.0 C.8 1.0 0.0 27.9 C.1 5.0 30.3 1.2.0
Ą		PARA ET	ER VALUE	S FUR TH	E NATER	YEAR 174	6 TO 194	7					
-18		001	V 011	DEC	N + L 	F28	MAR	APRI	L ***	JUNE		AUG	SEPT
	LA NN SA REM MPM MTR MT MT TPP T	0.0 27.1 27.1 5.7 0.0 0.0 0.0 0.0 1.7 1.0 20.0 20.0	30.0 25.2 9973.0 0.8 1.0 0.0 2.5 0.1 11.0 30.0 24.0	30.0 24.5 5959.0 0.8 1.0 0.0 5.1 0.1 6.0 31.0 7.0	JU.J 24.5 J99J.J U.J 1.J 0.J 7.5 0.1 5.J 31.0 3.D	30.9 23.4 999.0 0.8 1.0 0.0 1C.2 C.1 14.C 28.0 35.0	30.0 24.9 9993.0 0.0 12.7 0.1 4.0 31.0 10.0	30.0 26.0 9999.0 0.7 1.0 0.0 15.2 0.0 5.0 30.0 6.0	30.0 26.4 9999.0 0.8 1.0 0.0 17.8 0.1 7.0 31.0 16.0	J0.0 27.2 9995.0 0.8 1.0 0.0 20.3 0.1 20.0 30.0 41.0	30.0 28.2 9999.0 0.8 1.0 0.0 22.9 0.1 8.0 31.0 13.0	30.0 28.5 9999.0 0.8 1.0 25.4 0.1 9.0 31.0 17.0	30.0 27.5 0.8 1.0 0.0 27.9 0.1 21.0 3C.0 50.0

SAMPLE PROGRAM OUTPUT

MONTHLY PARAMETER VALUES FOR THE WATER YEARS 1945-1950

Arthur D Little, Inc.

0 40 000	4949696464 1134 ARA 4	ER VALUES	F34 THE	#ATER	YEAR 194	7 TU 1948	344644644	+****	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			
	JCT	V 0 V	DEC	JAN	FEB	маң	APRIL	V A Y	JUNE	JULY	AUG	SEPT
L TNN S A R E M M N M N M N T P P T	30.0 27.4 0.4 5.8 1.0 0.0 0.0 0.0 0.1 1.3.0 31.0 52.0	30.0 26.8 5979.0 0.8 1.0 0.0 2.5 0.1 4.0 30.0 6.0	30.0 25.0 3.2 3.3 1.3 0.0 5.1 0.1 5.0 31.0 9.0	J0.0 24.7 J99J.0 U.3 1.3 U.3 7.6 0.J 3.0 J1.0 J.9	J0.9 25.1 9999.0 0.7 1.0 0.0 10.2 J.1 8.9 23.0 12.0	$ \begin{array}{r} 30.0\\ 20.0\\ 999.5\\ 3.7\\ 1.0\\ 0.0\\ 12.7\\ 9.77.0\\ 0.0\\ 31.0\\ 3.0\\ 9.0\\ \end{array} $	30.0 26.3 9999.0 0.7 1.0 0.0 15.2 0.1 11.0 30.0 19.0	30.0 27.5 999.0 0.8 1.0 0.0 17.8 0.1 24.0 31.0 73.0	30.0 27.9 5999.0 C.5 1.0 0.0 20.3 C.2 12.0 30.0 47.0	30.0 23.7 9959.0 0.7 1.0 0.0 22.9 0.1 6.0 31.0 12.0	J0.0 29.1 9,59.6 0.7 1.0 0.0 25.4 0.0 7.0 31.0 7.0	3 C • O 2 E • O 9 9 9 5 • O 0 • 8 1 • O C • O 2 7 • 9 C • 2 1 4 • O 3 C • C 7 C • O
*****	0000000000 PA (1.4ET8	COCOCCER VALUES	SCH THE	********* ******	******** YEAR 194	\$*\$***** 5 TU 1949	********* }	******	*****	******	******	*******
	001	V CM	DEC	JAN	FEB	MAD	APRIL	MAY	JUNE	JULY	AUG	SEPT
L TA NN S A REP MPM MTR MN MT TPPT	JU.J 27.2 9.J9.0 J.B 1.9 J.0 U.1 U.1 LB.9 J1.9 J1.9	30.0 26.5 0.3 0.8 1.0 0.0 2.5 0.1 13.0 30.0 17.0	30.0 24.9 0.2 0.9 1.0 0.3 5.1 0.0 2.0 31.0 2.9	30 · 7 23 · 1 0 · 2 0 · 7 1 · 3 3 · 3 7 · 5 3 · 2 4 · 0 3 1 · 9 2 3 · 3	$ \begin{array}{r} 30 \cdot 0 \\ 23 \cdot 7 \\ 0 \cdot 1 \\ 0 \cdot 7 \\ 1 \cdot 0 \\ 0 \cdot 0 \\ 10 \cdot 2 \\ 0 \cdot 0 \\ 2 \cdot 0 \end{array} $	30.0 24.2 0.2 0.7 1.0 0.0 12.7 0.0 6.0 31.0 6.0	30.0 25.7 0.3 0.8 1.0 0.0 15.2 0.1 5.0 30.0 7.0	30.0 25.7 0.6 C.8 1.0 0.0 17.8 C.1 23.0 31.0 55.0	30.0 26.7 C.6 C.9 1.0 C.0 20.3 C.1 13.0 3C.3 27.0	30.0 27.5 0.5 0.7 1.0 0.0 22.9 0.0 22.9 0.0 31.0 2.0	30.0 27.6 9.5 0.8 1.0 0.0 25.4 0.1 12.C J1.0 24.0	30.0 27.3 0.5 0.8 1.0 0.0 27.9 0.1 11.0 30.0 25.0
****	¢ ¢¢ ¢ ¢¢¢¢ Parame ti	¢¢¢¢¢¢¢¢¢¢ ER VALUES	00000000000000000000000000000000000000	¢¢¢≎¢¢¢¢ ∀AT.2 ₹	\$*\$\$*\$ YEAH 194	******** 7 TO 1950	********)	******	********	*******	*******	* * * * * * * * * * * *
	100	NÚV	UEC	JA'I	FLB	MAR	APRIL	MAY	JUNE	JUL Y	AUG	SEPT
L TA NN 5 A PPM MTR MN MT TPP T	50.0 20.1 0.5 0.8 1.0 0.0 0.0 0.1 20.0 31.0 49.0	30.0 25.3 0.5 0.8 1.0 0.0 2.5 0.1 13.0 30.0 17.0	$ \begin{array}{r} 30.0 \\ 24.0 \\ 0.4 \\ 0.5 \\ 1.0 \\ 0.0 \\ 5.1 \\ 0.1 \\ 12.0 \\ 31.0 \\ 15.0 \\ \end{array} $	30.3 23.3 0.2 0.8 1.3 0.0 7.6 0.0 3.3 31.0 3.0	30.3 23.0 0.3 0.8 1.0 0.0 10.2 0.0 10.2 0.0 28.0 9.0	30.0 24.1 3.2 0.8 1.0 0.0 12.7 0.0 2.0 31.0 2.0	30.0 24.3 0.4 0.7 1.0 0.0 15.2 0.1 5.0 30.0 9.0	30.0 25.8 0.4 0.8 1.0 C.0 17.8 0.1 9.0 31.0 14.0	3C.9 27.1 0.8 1.0 0.0 2C.3 0.1 3.0 30.0 d.0	30.0 27.3 G.4 0.7 1.0 C.0 22.7 0.1 9.0 31.0 14.0	30.0 27.5 0.5 0.8 1.0 0.0 25.4 0.1 20.0 31.0 35.0	3 C • 0 2 6 • 8 0 • 6 C • 8 1 • 0 C • 0 2 7 • 9 C • 1 2 0 • 0 3 0 • 0 4 5 • 0
0000000	00000000	*****	000000000000	*****	******	* * * * * * * * * * * * * * * * * * * *	*****	*****	*****	*****	******	******

ID-19

The necessary tapes for the station under study must be loaded and labeled according to the protocols of the user's computation center. Presently the subprogram requires labels for ITAPE1 and ITAPE2, defined in the execution file for IPDATA. This latter file also determines the location to which the output is sent, i.e., disc storage, line printer, or terminal.

Output is printed as matrices (Figure ID-10), one matrix for each year in the range between the input values for IYR1 and IYR2.

2.3.2 Monthly Long-Term Simulations

The subroutine will automatically derive average values for each parameter, by month, if more than five year's data are analyzed. The results are presented as a table labeled "Average Monthly Parameter Values over the Period 19-- through 19--" where the last digits of the period are IYRL and IYR2. Typical output shown in Figure ID-11.

2.3.3 Annual Time-Specific Simulations

Average values for each parameter over the space of a year, for each year within the specified period are also output by IPDATA. These data are output in a table titled "Annual Parameter Averages for All Months, By Year from 19-- to 19--", where the last digits of the period are IYR1 and IYR2. Typical output is displayed in Figure ID-12. Note that MT=365 and MPA is the sum of the monthly MPM values.

2.3.4 Annual Long-Term Simulations

The averages stored in the ANAV matrix described above, may again by averaged to obtain an array consisting of one average value for each parameter over, say, a 20-year period. IPDATA automatically calculates these values for any period of record specified as greater than 1 year. Typical output is shown in Figure ID-13.

2.3.5 Program Notes

The IPDATA subprogram was written to retrieve data stored by NOAA on tape. Some of the quirks used in this form of data storage necessitated the approaches used in the subprogram. This includes, checking each entry for missing, invalid or mistyped (i.e., out of range) data entries. When these circumstances are encountered, the data will be skipped. If an entire month's data were missing or invalid, as might occur in the case of monitoring equipment malfunction, the appropriate output value will be 9999. This holds true for monthly, annual and period-ofanalysis averages as well. In such cases, the user will have to use best judgment as to how to proceed.

It is foreseen that averages will eventually be taken for many stations within a region and thus all IPDATA output are designed to be stored for this purpose.

The formats for all IPDATA output correspond to SESOIL-81's read format. The annual matrices must be transferred to the GE file for use as SESOIL input.

Arthur D Little, Inc.

	οιτ	110 V	DEC	JAN	FEB	448	APRIL	MAY	JUNE	JULY	AUG	SEPT
L	30.0	30.0	30.0	30.0	30.0	30.0	30.0	30.0	30.0	30.0	30.0	30.0
T 4	20.1	23.8	24.4	24 •)	23.5	24.7	25.5	20.3	27.2	27.7	25.1	27.4
NN	J.5	0.4	0.2	0.2	0.2	0.2	0.3	0.5	0.0	0.4	0.5	0.6
5	0.3	0.8	5.8	3.3	0.7	0.7	0.7	0 • d	C • 8	ƕ7	C.7	C.8
Ă	1.0	1.0	1.0	1.0	1.0	1.0	1.0	1.0	1.0	1.0	1.0	1.0
NEP	3.0	0.0	0.5	ა. ა	0.0	0.0	Ú.O	0.0	C.U	0.0	0.0	0.0
NPM	0.0	2.5	5.1	7.5	10.2	12.7	15.2	17.8	20.3	22.9	25.4	27.9
MTR	0.1	9.1	2.1	2.1	0.1	0.0	0.1	0.1	0.1	0.1	0.1	0-1
MN	17.4	11.6	7.0	3.4	7.2	4.4	8.0	15.6	10.6	0.2	10.6	15.0
NT	33.5	30.5	30.5	33.5	30.5	30.5	30.5	30.5	30.5	30.5	30.5	30.5
TPPT	35.0	21.8	11.4	7.5	13.2	0.8	26.2	39.2	25.6	9.8	18.2	41.6
*****	* * * * * ** ** **	*****	*******	****	*********	******	*****	******	******	*****	****	****

ID-21

Arthur D Little, Inc.

FIGURE ID-11

SAMPLE PROGRAM OUTPUT

MONTHLY PARAMETER AVERAGES FROM 1945-1950

YLAR L . MPA .	1 • 1 45 TA • NH MTR • MN	• 5 • MT	• A • REP • TPPT	30.0 167.6	25.7 0.1	0.7 9.0	U.P 365.0	1 • 0 2 4 • 7	0.C
YEAH L . MPA .	2 • 1940 TA • NY MTR • MN	• S • MT	• A • REP • TPPT	30.C 107.C	25.1 0.1	0.5 10.8	0.8 365.0	1.0 20.9	c.o
YсА√ L • МРА •	3 • 1947 TA • NH MT-2 • MN	• 5 • MT	• A • PEP • TPPT	JC.0 107.0	26.9	0.3 9.3	Ú.9 J65.0	1.0 24.7	0.0
YEAR L . MPA .	4 • 1948 TA • NN MTR • MN	• 5 • MT	• A • REP • TPPT	30.0 157.6	25.) 0.1	0•4 9•3	U • 9 365 • 0	1.0 18.4	0.0
YEA .	5. 1949 TA NY MTR MN	• MT	а терт	- 30°•0 167•0	25•4 0•1	10.3		1.0	0.C
***		*****	600000000000000000000000000000000000000	*****	******	*****	*******	*****	*****

FIGURE ID-12

SAMPLE PROGRAM OUTPUT

ANNUAL PARAMETER AVERAGES BY YEAR FROM 1945-1950

*****	***************************************
PARAMETER AVERAG	IS FUR ALL OBSERVATIONS OVER THE PERIOD 1945-1950
L	30.00
1 A	
NN	0.47
S	0.76
Ā	1.00
REP	G • 0
NPA	167.64
NTI	0-08
MN	9,75
MT	125-00
TODT	21.42
*****	, , , , , , , , , , , , , , , , , , ,

SAMPLE PROGRAM OUTPUT

PARAMETER AVERAGE FOR ALL OBSERVATIONS FROM 1945-1950

- 3.0 SOIL INPUT DATA
- 3.1 General
- 3.2 Soil Intrinsic Permeability (k)
- 3.3 Soil Porosity (n)
- 3.4 Soil Pore Disconnectedness Index (c)

The soil pore disconnectedness index c in SESOIL, is the exponent relating the "wetting" and "drying" intrinsic permeability k(s) of the soil to its saturated intrinsic permeability k(l), and is given by Eagleson (WRR p. 723 e.g. 7). This relation is given in the literature by Brook and Corey (1966) and its validity for both a cohesive and a cohesionless soil is graphically presented by Eagleson (WRR p. 724 Figure 3).

To obtain the c parameter for various soils a user has to follow the work of Talma (1974) and Moore (1939). Eagleson (1978) has presented in this work typical c parameters (Table ID-1), however, these values have to be employed with caution. The authors (Bonazountas and Wagner) of SESOIL intend to author a section for this model providing: (a) c values for the various soils of the USDA soil classification triangle, and (b) a discussion for estimating the effective "real (of the field)" hydrologic/soil properties of soils from observations of vegetation density via the model itself. This task has not been performed yet for budget reasons only.

However, it has to be emphasized that any unsaturated soil zone of the literature (e.g. Bonazountas et al 1979) requires as a user input instead of a curve relating "permeability-k vs. capillarity head ψ ." This curve is obtained from experimental data. It is almost impossible to obtain this curve off-the-shelf for any type of soil, but it also is extremely difficult to obtain this curve in the laboratory. Therefore, the "one-variable" approach of Eagleson (1978) employed in SESOIL greatly simplifies data gathering. Authors of this report advise the user to employ values of Table ID-1, and to interpolate for different types of soils using also the work of Freese and Cheery (1979 p. 29).

4.0 CHEMISTRY INPUT DATA

Chemical parameters, coefficients, etc. might be compiled from the handbook "Research and Development Methods for Estimating Physicochemical Properties of Organic Compounds of Environmental Concern" (Lyman et al. 1981), or any other handbook.

TABLE ID-1

INDEPENDENT SOIL PROPERTIES/PARAMETERS¹⁾

Properties ⁴⁾	SOIL TYPE										
(SESOIL Variable)	Clay	Clay-Loam	Silty-Loam	Silty-Loam	Sand ²⁾						
k(1)	1×10^{-10}	2.8×10^{-10}	1.2×10^{-9}	2.5×10^{-9}	1×10^{-7}						
n	0.45	0.35	0.35	0.25	0.25						
c	12	10	5 ³⁾	4	3.5						

ID-25

- 1) See Table HY-3
- 2) Compiled from various sources
- 3) Personal conversation with Eagleson
- 4) A single relationship between k(1), n and c does not exist.

Main Source: Eagleson (1978)

5.0 REFERENCES

Bonazountas, M., W. Hawes and W. Tucker (1979) "Heutistic Unsaturated Flow/Quality Soil Zone Model," American Geophysical Union, Spring Meeting, Washington, D.C.

Brooks, R.H., and A.T. Corey (1966) "Properties of Porous Media Affecting Fluid Flow," J. Irrig. Drain. Div. Amer. Soc. Civil Eng., IR2, 61-88.

Eagelson, P.S. (1978) "Climate, Soil, and Vegetation," Water Resources Res., 14.

Lyman, W. <u>et al.</u> (1981) "Research and Development Methods for Estimating Physicochemcial Properties of Organic Compounds of Environmental Concern." Prepared by Arthur D. Little, Inc., Phase II Final Report for U.S. Army Medical Bioengineering Research and Development Laboratory, Fort Detrick, Maryland; McGraw-Hill Book Company, New York.

Moore, R.E. (1939) "Water Conduction from Sahllow Water Tables," Hilgardia 12(6), 383-426.

Talsma, T. (1974) "The Effect of Initial Moisture Content and Infiltration Quantity on Redistribution of Soil Water," Aust. J. Soil Res., 12, 15-26.



3.0 SOIL INPUT DATA

3.1 Summary of Default Values

The hydrologic routine of SESOIL employs the one-dimensional water balance model of Eagleson [1978a-g] which has been written in terms of five surface vegetation and soil parameters: the vegetation canopy M, the species-dependent plant water use coefficient k, the soil <u>effective</u> porosity n, the saturated intrinsic permeability k(1), and the soil pore disconnectedness index c. A state variable of the problem has been the average long-term soil moisture concentration s The analytic structure of this model is summarized in appendix HY, however, tailored to the needs of SESOIL, roughly for bare soils and monthly simulations. The bare soil requirement reduces the five surface parameters to the following three independent soil input parameters:

- k(1) = saturated soil intrinsic permeability; (cm²)
- n = effective soil porosity; (fractional)
 - $= \Delta[\ln(k)]/\Delta[\ln(s_{o})];$ soil pore disconnectedness index; (-)

There is no unique association of the particular c and n regional values of a soil type with the value of a k(1). Therefore, model users should be very careful when employing the default independent soil property values presented at the end of this section. The soil properties are critical to the moisture fluxes and vary tremendously spatially. Use of point measured soil properties can yield results of only local (and hence not areally averaged) character. It would be also appropriate to use some observed water balance element such as average basin yield (surface runoff and groundwater runoff) to evaluate the effective soil properties, as discussed in section 3.5; soil parameter calibration.

A schematic variation of soil hydraulic properties with textural class is presented in figure ID-14. Of course, soils evolve having a continuous spectrum of textures from clay through silt to sand and gravel, and the critical hydraulic properties of soils vary widely even within the same textural class, but over the variety of classes their range is enormous. Figure ID-14 is only a gross generalization of the overall soil characteristics [Eagleson 1982, p.326]. More detailed information is provided in the subsequent sections.

In a natural ecological system, Eagleson suggested [Eagleson & Tellers 1982, p.341] that there may be ecological pressures for change in natural soil-vegetation systems, which drive a synergistic development toward a water- or energy-limited equilibrium state in a given climate. Identification of the conditions for this equilibrium should allow an a priori specification of one or more of the physical parameters of the soil and vegetation, a fact that may lead to elimination of k(1) from the water balance equation in terms of the long-term average soil moisture concentration in the root zone s_{2} , a state variable of the



Variation of soil hydraulic properties with textural class. (a) After Chilingar [1964] (b) After Bear [1979].





- k(l) = intrinsic soil permeability; (cm²) n = total soil porosity; (-) n = effective soil porosity; (-)
 c = soil pore disconnectedness indes; (-) Mo = vegetation canopy density at maximum soil moisture = potential transpiration efficiency of k v
 - plant species

Figure ID-14 - Generalized Correlation of Independent Soil Parameters; k(1), n, c

ID-28

budget theory. In a bare-soil system, however, all three input variables --k(1), n_{a} , and c -- have to be known and input to the model.

Ideally, one should test input values by having direct observations of k(1), n and c -- in addition to the various climate parameters -- from an array of spatially homogeneous natural systems covering a wide range of the dimensionless climate-soil parameter E (see appendix HY, equation HY-23). In practice, however, at least the soil parameters of natural systems are highly variable spatially [Nielsen et al 1973; Libardi et al 1982], so even if dense observations of them were available (which is rare), the problem of how to average them areally would arise [Eagleson 1982, p.342]. Because of the large spatial variability of the properties of natural soils, and because of the high degree of non linearity of the fluxes, spatial averaging over the large area elements of either climate or water resource models become a non-trivial problem [Eagleson 1982, p.325]. Therefore, given an initial set of input parameters k(1), n, c, it is recommended to calibrate the input data set of the model by varying the intrinsic permeability k, the pore disconnectedness index c, and the effective porosity n of the soil, towards obtained field records of soil moisture content or basic yield (see section 3.5; Soil Parameter Calibration).

In case of total absence of site specific input data and when non-site specific long term pollutant fate modeling efforts have to be performed for canonical climatic environments, SESOIL users may employ the information compiled in Table ID-1, namely the:

- USDA (U.S. Department of Agriculture) soil textural classification system as a guide for soil type selection; and
- * Default values of k(1), n, c compiled for the various USDA soil-types of the soil triangle.

The figure and the table of table ID-1 are self-explanatory for easy use and for canonical environments only. The rationale behind the default values generated and the uncertainty when deriving them are outlined in the following sections for each individual soil parameter. Only fundamental concepts are discussed as they relate to the SESOIL use, therefore, model users are advised to consult original publications of this appendix.

3.2 Soil Intrinsic Permeability -- k(1)

3.2.1 Definitions

When various fluids of density ρ and dynamic viscosity μ are run through a porous medium consisting of uniform glass beads of diameter d, and under a constant hydraulic gradient dh/dl, the following proportionality relationships are observed [Freeze & Cherry 1979, p.27]:

$$v = v(d^2), \quad v = v(\rho,g), \quad v = v(1/\mu)$$
 (ID-1)

SESOIL Default Soil Input Parameters



	Textural Soil Class		SESOIL Defay	it Valge	,11)	Suppor	ting_Infor	mation to De:	fault_Val		
() 	(USDA triangle) Designation	ر ي هر) (<u>د م</u> ر)	k(]) ³⁾	n	e ⁵⁾	2 ⁵⁾	4 ⁷⁾	£10)	\$ base \$)	sil2 ⁸⁾	ciay ³⁾
1	Clay very fine; 12)	1.33	0.75x10-10	0.20	12	0.70	0.0328	7.5x10	10	13	-7
(2)**	Clay (medium fine) ¹³⁾	1.35	2-5x10-10	0.20	12	0.08	0.050	2.5=10-3	18	19	53
3	Clay (fine) ¹²⁾	1.35	5.0 <u>=10</u>	0.22	12	0.64	0.0785	6.0x10 ⁻³	25	25	50
-	Silty clay	1.35	0.5x10-10	0.25	12	0.59	0.0236	5.0=10	6	4 7	47
ذ	Silty clay loss	1.35	0.85x10-10	0.27	10	0.59	0.0362	8.5x10-4	10	55	35
(6)	Clav loam	1.35	6.5x10 ⁻¹⁰	0.30	7.5	0.58	0.103	6.5x10 ⁻³	33	33	34
7	Loan	1.35	8.0m10 ⁻¹⁰	0.30	6.5	0.52	0.124	8.0x10 ⁻³	40	÷Q	20
(8)	Silt loam ¹²⁾	1.35	3.5×10 ⁻¹⁰	0.35	5.5	0.54	0.0726	3.5×10 ⁻³	22	65	13
9	Silt	1.35	0.5010-10	0.27	12	0.49	0.0240	5.0x10 ⁻⁴	5	90	5
10	Sandy CLAV	1.35	15x10 ⁻¹⁰	0.24	ġ	0.44	0.157	1.5×10 ⁻²	52	6	-2
11	Sandy clay loam	1.35	25x10-10	0.26	4	0.45	0.176	2.5×10-2	58	15	27
(12)	Sandy Loam	1.35	20=10-10	0.25	0.33	0.44	0.167	2.0%10-2	55	25	10
13	Loanv sand	1.35	500x10 ⁻¹⁰	0.28	3.9	0.33	0.250	5.0x10 ⁻¹	33	10	7
14	Sand	1.35	100=10-10	0.30	2.7	0.35	0.285	1.0=10 ⁻¹	95	3	2

¹ See USDA soil classification triangle of figure ID-15 (pictured above also)

Set Cars our corresponds to an overall proposed soil scenario related to the USDA soil triangle; in figure with y

- 3 k(1) = intrinsic soil permeability; see table ID-7
- n_{e} = effective soil porosity; see table ID-9 and figure 13-18 5
- c = soil pore disconnectedness index; see table ID-10
- n = soil porosity; see table ID-9 and figure ID-18
- d soil particle median dismeter: see table ID-6
- 8 see table 12-5
- fee table to 5
 for soil bulk density an overall constant value of 1.35 g/cm³ is assumed;
 see Suchman 5 Bracy (1969). Esglason (1970, p.237). Holton 6 Whelan (1967)
- 10 K soil conducting; see table 1D-6
- il only for non-with specific modeling efforts
- 12 averaged value of table 12-3
- 13 average values of soils #1 and #3.

Source: Bonazountas & Wagner [1981].

Together with Darcy's original observation that v=v(-dh/dl), the above three relationships lead to a new formulation of Darcy's law

$$v = -(C d^2 \rho. g/\mu)(dh/d1)$$
 (ID-2)

The parameter C is a constant of proportionality affected by the soil grain size, sphericity and other factors. Comparison of ID-2 with the Darcy equation

$$v = -K \, dh/dl \tag{ID-3}$$

leads to

$$K = Cd^2 \rho \cdot g/\mu \qquad (ID-4)$$

In this equation, ρ and μ are functions of the fluid alone and Cd^2 is a function of the medium. If we define

$$k = Cd^2$$
; (cm²) (ID-5)

then $K = k\rho g/\mu$; (cm/sec) (ID-6)

The parameter k is known as the specific or intrinsic permeability (or permeability) and K is known as hydraulic conductivity (or sometimes the coefficient of permeability). Table ID-2 provides values of k and K for a variety of geological materials, and values of unit conversion factors.

Beyond the above definition, Eagleson [Eagleson 1978, p.23] defined the effective intrinsic permeability (cm^2) of a soil as

$$k(s) = (\mu/\gamma_{..}) K(s)$$
 (ID-7)

where γ the specific weight of pore water in dynes per cm³ and based on the work of Brooks and Corey [1966] he indicated for his water balance that:

$$k(s) = k(1) s^{C}$$
 (ID-8)

For saturated conditions --i.e. s=1 (see app. HY)-- k(1)=k and K(1)=K as indicated in expressions ID-5 and ID-6. For unsaturated soil conditions we have the definition $k(\theta)$ and $K(\theta)$, or k(s) and K(s) to be consistent with Eagleson.

3.2.2 Hydrologic Characteristics of Soil Types

Over 4000 soils of the United States have been classified into four hydrologic groups [Chow 1964]; designated as A, B, C and D by the Soil Conservation Service (SCS). The majority of the assignments are based on the judgment of soil scientists and correlators [Chiang 1971]. A short description [Novotny 1976] of the four soil types follows:

TABLE ID-2

Range of Values of Intrinsic Permeability and Hydraulic Conductivity; Unit Conversion Factors

Unconsolidated k . ۲ ٦ ~ Rocks deposits (darcy) (cm²· ·cm/s) (m/s) (ga /day/ft²) r 105 r 106 - 10⁻⁵ - 10 clean sand ----- Gravel ---104 - 10-1 Fractured igneous and I ractured igneous and I rinestone and I rinestone and Jandsterr 105 1C3 10-2 104 - :C-3 1C2 103 10-4 -10-7 -10-2 10 1C² 10-8 - 10-3 10-5 10 - 10-9 -10-4 10-6 10-1 -10 - 10⁻⁵ - 10-7 10-2 :C-1 10-3 - 10-5 - 1C - :C-e :0⁻² i 10-4 10⁻¹² 10 10[.] l 10-3 1C-13 10-5 1C-1C 10⁻⁸ Unfractured metamorphic and ---10-4 10-6 igneous rocks 10-9 10-14 !O"" Shale --10-5 10-7 10-10 0-15 10-12 10-6 10-11 L 10-13 0-16 L L 10-7

Range of Values of Hydraulic Conductivity and Permeability

Conversion Factors for Permeability and Hydraulic Conductivity Units

		Permeability k*		Hydraulic conductivity, K						
	cm²	ft²	darcy	m/s	ft/s	gal/day/fi:				
cm ²	1	1.08 • 10-1	1 01 108	9 80 102	3 22 103	1 85 10*				
ft²	9 29 102	1	9 42 * 1010	9 11 103	2 99 100	1 71 1012				
darcy	9.87 10-9	1 06 × 10 ⁻¹¹	1	966 10-0	3 17 10-5	1 82 101				
m/s	102 • 10-1	1 10 \ 10~0	1 04 > 103	1	3 28	212 10.				
ft/s	3.11 × 10**	3,35 🔨 10-7	3.15 × 104	3 05 - 10-1	1	5 74 101				
gal/day/ft ²	5.42 × 10 ⁻¹⁰	5.83 × 10 ⁻¹³	5.49 × 10-2	4 72 10-7	1 74 · 10-•	1				

•To obtain k in ft², multiply k in cm² by 1.08×10^{-3}

Source: Freeze & Cherry [1979, p.29]

- * Group A: Soils of low total runoff potential have high infiltration rates even when thoroughly wetted; these consist chiefly of deep, well to excessively-drained sand or gravel and therefore possess a high rate of water transmission.
- * Group B: Soils of low-moderate total runoff potential have moderate infiltration rates when thoroughly wetted, and range from moderately-deep, moderately-well to well-drained soils having moderately-fine to moderately-coarse texture. Consequently, these soils have a moderate rate of water transmission.
- * Group C: Soils of high-moderate total runoff potential have slow infiltration rates when thoroughly wetted and consist chiefly of soils with a layer that impedes downward movement of water, or soils with moderately-fine to fine texture. These soils have a resultant slow rate of water transmission.
- * Group D: Soils of high total runoff potential have very slow infiltration rates when thoroughly wetted, and consist chiefly of clay soils with a high swelling potential, soils with a permanently high water table, soils with a clay pan or clay layer at or near the surface, and shallow soils over nearly impervious material. These have the expected very slow rate of water transmission.

Potential storage of soil moisture can be partitioned into two moisture classes: 1) gravitational water, i.e. that held between saturation and 0.33 bar tension, and 2) plant-available water, i.e. that held between 0.33 and 15 bar tension. Moisture content at 0.33 bars is assumed to represent field moisture capacity or the lower limit of gravitational water, and 15 bars the permanent wetting percentage in medium textures soils. Gravitational water, G, is derived by subtracting 0.33 bar volume percentages from total porosity. Available water capacity (AWC) is the difference between the moisture contents (volumes) at 0.33 and 15 bar tensions. [Novotny 1976.]

3.2.3 Soil Hydraulic Conductivity -- K(1)

Hydraulic conductivity of saturated soils, vary greatly and natural soils and soil materials, therefore, behave quite differently. Based on these differences in performance, soils can be classified according to their K(1) rate. For example, for earth dams the U.S. Bureau of Reclamation generally classifies soils with K-values above 10^{-4} cm/sec as pervious and soils with K-values below 10^{-6} as impervious.

In tile drainage, K-rates may be used in selecting depth and spacing of tile drains. Using relative permeabilities, guides for average depth and spacing for tile drains have been established by various agencies or investigators such as that given in table ID-3.

TABLE ID-3

Hydraulic Conductivity Classes According to the USDA-SCS

K Values								
inch/hr	<u>cm/sec</u>	^m /day						
Less than 0.5	Less than 0.035 x 10^{-3}	Less than 0.3						
0.05 to 0.20	0.035×10^{-3} to 0.14×10^{-3}	0.3 to 0.12						
0.20 to 0.80	0.14×10^{-3} to 0.56×10^{-3}	0.12 to 0.48						
0.80 to 2.50	0.56×10^{-3} to 1.75×10^{-3}	0.48 to1.52						
2.50 LO 5.00	1.75×10^{-3} to 3.5×10^{-3}	1.52 to 3.0						
5.00 to 10.00	3.5×10^{-3} to 7.0×10^{-3}	3.0 to 6.0						
More than 10.00	More than 7.0 x 10^{-3}	More than 6.0						
	<u>inch/hr</u> Less than 0.5 0.05 to 0.20 0.20 to 0.80 0.80 to 2.50 2.50 to 5.00 5.00 to 10.00 More than 10.00	K Valuesinch/hrcm/secLess than 0.5Less than 0.035×10^{-3} 0.05 to 0.20 0.035×10^{-3} to 0.14×10^{-3} 0.20 to 0.80 0.14×10^{-3} to 0.56×10^{-3} 0.80 to 2.50 0.56×10^{-3} to 1.75×10^{-3} 2.50 Lo 5.00 1.75×10^{-3} to 3.5×10^{-3} 5.00 to 10.00 3.5×10^{-3} to 7.0×10^{-3} More than 10.00More than 7.0 $\times 10^{-3}$						

Source: Horn [1971]

3.2.4 Factors Affecting K(1)

A summary of the work of Horn [1971] is presented in this paragraph.

Particle size, gradation (particle size distribution analysis), arrangement of soil particles (fabric, structure, micromorphology), organic matter content, iron oxide content, clay mineral composition, exchangeable sodium percentage, and total concentration of salts are among the more important basic factors affecting pore size distribution and continuity, and hence, K.

Particle size and shape (it is assumed for purposes of particle size analysis that soil particles are roughly spherical in shape although many particles are platy or lath-shaped) determines the size of normal packing voids. If a soil is well graded, i.e. has a good distribution of particles throughout all size ranges, the smaller particles fit into the interstices associated with the larger particles and reduce the total porosity. Smaller clay particles may also be washed into the subsoil and deposited as films that isolate or block off larger voids that conduct water is low. Slowly permeable soils of this type are often referred to as claypan soils. Other special cases of particle arrangement occurring in natural soils are the fragipan horizons of silty soils, which also have severely restricted K.

Organic matter tends to increase the K of clayey soils by promoting aggregation. However, this is only significant when calcium ions are dominant on the soil exchange and in the soil solution. When this condition exists calcium humate compounds are formed that link clay particles into large water stable aggregates.

Iron oxides also cement finer particles together, or form coatings on aggregates that prevent their dispersion. The high K of many red, tropical soils is a result of high iron oxide contents. However, under certain conditions, iron compounds may be translocated downward in the soil and reprecipitated to eventually form dense ironpans which are very slowly permeable or sometimes impervious to water movement.

Similarly other compounds, notably those of calcium, may be precipitated in the subsoils of arid or subarid regions. Once formed, these caliche pans may drastically inhibit movement of water. Where such pans are near the soil surface waterlogging may develop under irrigation.

The kind and amount of clay in soils is extremely important in determining rate of water flow through soils. Considering the size of clay particles alone it is clear that the theoretical K rate of soil masses composed of particles with equivalent spherical diameters of less than 0.002 mm is extremely low. If it were not for the aggregation of clays into larger units (peds) most clayey soils would be virtually impervious. Some clays (those of the 2:1 layer montmorin group) swell tremendously when they are wetted and reduce permeability greatly. Others such as kaolinite do not swell. Thus, kaolinitic soils are generally quite permeable while montmorillonitic soils have very low permeability or are impervious.

Exchangeable sodium (Na) content if in excess of 10% to 15% of the soil cation exchange complex will cause dispersion of the soil particles. A highly dispersed soil is very massive and dense and lacks the high proportion of void spaces associated with strongly structured, flocculated and aggregated soils. The K may be reduced by as much as 90% due to Na-induced dispersion in some soils (Bonazountas et al 1981). Such soils include the alkali soils and have pH values of the order of 8.5 to 10.0, although NA dispersion may also be associated with soils of acid pH. Conversely, soils with a high percentage of exchangeable calcium (Ca) are usually well aggregated soils and generally have moderate to rapid permeability.

Soils containing montmorillonite, even in amounts as low as 5% to 10%, are particularly sensitive to changes in the status of the exchangeable cations and the soil solution. A well aggregated Ca-dominated soil containing montmorillonite may be affected very quickly and adversely by the addition of sodium-rich alkaline water. Therefore, the quality of irrigation water should be carefully scrutinized before applied to these soils.

Compaction of soils, by the passage of cattle, or heavy equipment (which produce traffic pans) over a field surface often causes unwanted reduction in soil infiltration and permeability rates by reducing the size and continuity of pores. Intentional compaction of canal bottoms and sideslopes causes desirable reduction in seepage losses. Frequently, however, soils will rebound significantly from initial compaction particularly where water levels fluctuate markedly.

Cultivation of soils, particularly clay soils of low organic matter content, when saturated or nearly saturated often causes destruction of natural soil structure which may then result in substantial reduction of permeability. Wet cultivation (puddling) may be done intentionally as in paddy rice cultivation to cause destruction of the large natural soil aggregates (peds) which favor rapid permeability. Puddled soils produced in this manner lose less water to deep percolation when irrigated because of the destruction of their continuous macropore system.

For most crops, however, a compacted or puddled soil is deleterious because it inhibits air, water, and root penetration. After puddling, it may take several months for the soil to fully recover its natural structure through alternate wetting and drying (thawing and freezing, or swelling and shrinking), and the influence of root activity and organisms in the soil.

Temporarily increased K-rates can be achieved by mechanical action such as subsoiling. Other features affecting permeability, but not considered inherent soil properties, include relatively short-lived macropores such as wormholes and various other animal burrows, root channels, etc. Biotic features such as these are most common to noncultivated soils and less so in cultivated soils.

3.2.5 Guides to Estimating K(1)

Most guides developed as aids in making estimates of soil K-values are based on the relationship between soil texture and various ranges in rates or K classes. The USDA has schematically presented soil textural classification with a triangle (figure ID-15). In addition the USDA-SCS rates soils into seven relative K classes as shown in table ID-3. Examples of the empirical relationships that have been established by various researchers are given in table ID-4.

The chief weaknesses of these guides are first, the inconsistency that arises in the definition and identification of various soil textural classes, and secondly, the fact that the differences in permeability due to factors other than texture (as described previously) are not accounted for [Horn 1971]. To avoid the difficulty associated with using textural class names, i.e. sandy loam, clay loam, etc., a more quantitative approach has been undertaken using mean particle size, as described subsequently.

3.2.6 Mean Soil Particle Size Estimation

This value may be calculated from particle size distribution analyses (mechanical analyses) which are often made routinely in laboratories as a part of project investigations and with much more ease than permeability determinations. This calculation is based on selecting a mean particle size (equivalent spherical diameter) of 0.3 mm for the sand fraction, 0.01 mm for the silt fraction, and 0.002 mm for the clay fraction. Values for sand and silt represent the midpoint in the size range representing each of these two fractions from a log scale (figure ID-16). For the clay fraction, its upper limit was selected. The particle size limits utilized are those of the USDA but can be equated quite readily to size limits used by other agencies by means of the scale in figure ID-16.

Using the textural triangle (figure ID-15) central points or values of sand, silt, and clay percentages representing each of the soil textural classes were selected. The clay textural class was subdivided [Horn 1971] into very fine clay (more than 60% clay) and fine clay. These values and the mean particle diameters (weighted averages) calculated from them are summarized in table ID-5 and in table ID-1 (Summary, section 3.1).

3.2.7 K(1) vs Particle Size

The mean particle diameters in millimeters as given in table ID-5 for each textural class are plotted on the ordinate axis of a log-log graph



GUIDE FOR USDA-SCS SOIL TEXTURAL CLASSIFICATION SHOWING POINTS FOR WHICH MEAN PARTICLE DIAMETERS HAVE BEEN CALCULATED

.



COMPARISON OF VARIOUS SOIL PARTICLE SIZE RANGES USED BY VARIOUS AGENCIES

TABLE ID-4

GENERAL RELATIONSHIP BETWEEN SOIL TEXTURE AND SATURATED HYDRAULIC CONDUCTIVITY

	Soil Type	Hydraulic	K (1)	
_		<u>m</u> /day	inch/hr	cm/sec
**	Jonnan et al. [1947]			
	Coarse sand	120	196.8	140.0×10^{-3}
	Sand	12	19.7	14.0×10^{-3}
	Fine sand	4.8	7.9	5.5×10^{-3}
	Very fine sand	2.4	3.9	2.8×10^{-3}
	Loamy sand	1.2	2.0	1.4×10^{-3}
	Sandy loam	0.24	0.4	0.28×10^{-3}
	Very fine sandy loam	0.12	0.2	0.14×10^{-3}
	Loam	0.048	0.08	0.05×10^{-3}
	Silt loam	0.024	0.04	0.028×10^{-3}
	Silty clay loam	0.012	0.02	0.014×10^{-3}
	Silty clay	0.0024	0.004	0.0028×10^{-3}
	Clay	0.0012	0.002	0.0014×10^{-3}
**	Israelsen and Hansen [1967]		
	Sandy	2 (1 - 10)	1.4×10^{-3}	1.2
	Sandy loam	1 (0.5 - 3)	0.7×10^{-3}	0.6
	Loam	0.5 (0.3 - 0.8)	0.35×10^{-3}	0.3
	Clay loam	0.3 (0.1 - 0.6)	0.21×10^{-3}	0.18
	Silty clay	0.1 (0.1 - 0.2)	0.07×10^{-3}	0.06
	Clay	0.2 (0.05 - 0.4)	0.14×10^{-3}	0.12

Source: Horn [1971]

ID-40

TABLE ID-5

SOIL TEXTURE, REPRESENTATIVE PARTICLE SIZE CONTENTS, AND MEAN DIAMETERS

	Textural Class (USDA) (1)	Sand, as a percentage (2)	Silt, as a percentage (3)	Clay, as a percentage (4)	Mean diameter, in millimeters (5)
1.	Sand	95	3	2	0.285
2.	Loamy sand	83	10	7	0.250
3.	Sandy clay loam	58	15	27	0.176
4.	Sandy loam	55	25	10	0.167
5.	Sandy clay	52	6	42	0.157
6.	Loam	40	40	20	0.124
7.	Clay loam	33	33	34	0.103
8.	Clay (fine)	(a)40	10	50	0.122
		(Ъ)25	25	50	0.0785
		(c)10	40	50	0.035
9.	Silt loam	(a)34	53	13	0.107
		(Ъ)22	65	13	0.0726
		(c) 7	80	13	0.029
10.	Silty clay loam	10	55	35	0.0362
	Clay(very fine)	(a)22	1	77	0.067
		(b)10	13	77	0.0328
		(c) l	22	77	0.007
12.	Silt	5	90	5	0.0240
13.	Silty clay	6	47	47	0.0236

Source : Horn [1971]

Note: (b) describes central point; (a) and (c) describe the range (see Fig. 2).

against K-rate on the abscissa (figure ID-17). It should be noted that these mean diameters represent only one of many possible values within each given textural range [Horn 1971].

The theoretical K's relative to various particle sizes, as calculated by Zunkar [1930], represent a straight line relationship (Figure ID-17). These K's are calculated for spherical particles of the same diameter with each occurring as a discrete particle. In natural soils, of course, particles are only roughly spherical or often platy, and rarely, if ever, possess uniformity of diameter. Also, cementation of particles into aggregates commonly occurs. Thus, natural soils would be expected to <u>differ</u> in K-rate from the theoretical. Nevertheless, the theoretical K serves as a good base from which K-curves for natural soils may be anticipated. With coarse, clean, well sorted sands the actual and the theoretical values approach each other but such sandy soils are quite uncommon.

3.2.8 Selection of Working Values/Curves

For a particular application and a particular soil region, a working curve representing the soil texture-k relationship can be selected using best judgment of how existing local conditions and soil properties tend to affect soil conductivity. If some K measurements are available for known particle distributions these can be used in selecting a working curve for extrapolation to other soils. Ideally, a large number of reasonably accurate K measurements of soils along with corresponding particle size distribution data would be on hand for the locality in order to improve the accuracy of extrapolated values; however, this is rarely the case.

Without actual correlations, the working curve must be based on judgment. Best judgments are made if a thorough understanding is had of fundamental physical, mineralogical, and chemical properties of soils. Field observations, and soils data that may be available will further guide judgment. For small local areas it can often be safely assumed that nonparticle size related factors affecting soil permeability rate such as organic matter content, biotic activity, climatic conditions, clay mineralogy, and chemistry of the soil solution are more or less the same throughout. Realistic values and valid comparisons of soil K for the local area on then be obtained from a single working curve.

Hydraulic conductivities of the very sandy soils and, at the other extreme the very clayey soils, are not affected by nonparticle size related factors as much as the in-between soils. The latter comprise the majority of natural soils and vary considerably in response to these factors.

Some very general groupings of soils according to their K are separated in figure ID-17 by the horizontal and vertical dashed lines. For example, soils with mean particle diameters greater than 0.2 mm have K rates usually in excess of about 10^{-3} cm/sec and soils with diameters





less than 0.015 mm have rates of 10^{-4} cm/sec and are essentially impervious. Soils between these ranges are greatly affected by the nonparticle size factors controlling K; these factors vary considerably from place to place and, in addition, are subject to in situ changes in relatively short periods of time. As a consequence the intermediate soils can range from impervious to permeable [Horn 1971].

Rigorous application of the K curves in figure ID-17 presupposes the availability of data from particle size distribution analyses for all major horizons of the soils in question. Depending on intended application, K in a vertical direction of an entire soil section may be rated according to the least permeable layer, recognizing that for some soils, the presence of very thin continuous clay laminae or other abruptly contrasting layers may exert a profound influence on the permeability behavior of the soil. Lateral K of soil sections may be rated according to the layer with greatest K. Alternatively, average K rates may be taken for the entire section.

The methods for obtaining mechanical analysis data are much easier and the results more dependable than those associated with measuring K-rates. Of course, collection of representative soil samples and laboratory facilities capable of conducting particle distribution measurements are necessary in order to quantify the soil texture variable in the estimation of K-rates. In cases where both mechanical analyses and soil K determinations are to be made, it is recommended that soil samples are taken from the same location for laboratory analyses. This not only results in reducing sampling costs and field time but also provides valuable data for correlating permeability rates with the particle size data and other measured soil properties.

When only field estimations of soil texture are available, the K estimate is less quantitative. Nevertheless, using the limits indicated in figure ID-17, a guided judgment method is provided from which tenable soil K values can be obtained for a variety of field applications.

3.2.9 K(1) vs k(1)

The hydraulic conductivity (K) is related to the intrinsic permeability (k) via equation ID-6. Since only fluid (ie. water, soil moisture) is assumed to flow in a soil system, the conversion factor from K-to-k is a constant number, to be obtained from table ID-2. By employing: the values of table ID-2; the soil classification system of table ID-15; and the soil texture graph of figure ID-17; the intrinsic permeabilities (k) for the USDA soil texture triangle of figure ID-15 have been derived (table ID-6).

3.2.10 k(1) SESOIL Default Values

As discussed in section 3.2.6, the theoretical k rates derived would be expected to differ from the field k rates of natural soils. The last column in table ID-6 (Eagleson's estimates) indicates a consistent

TABLE ID-6

SOIL TEXTURE VS. PARTICLE DIAMETER^(d) SOIL CONDUCTIVITY^(K) AND INTRINSIC PERMEABILITY(k)

Te	xtural Soil				Eagelson
C1	ass (USDA)	d	K	k	[1978,1982]
<u>#</u>	Designation ¹	(mm)	(cm/sec)	(cm^2)	(cm ²)
			-4	-9	
1	Clay (very fine)	0.0328	7.5×10^{-4}	7.5×10^{-9}	_10
$(2)^2$	Clay (medium fine)	0.050	2.5×10^{-3}	2.5×10^{-6}	1×10^{-10}
3	Clay (fime)	0.0785	6.0×10^{-3}	6.0×10^{-0}	
4	Silty clay	0.0236	5.0×10^{-4}	5.0×10^{-9}	Г _10
5	Silty clay loam	0.0362	8.5×10^{-4}	8.5×10^{-9}	2.8 x 10
(6)	Clay loam	0.103	6.5×10^{-3}	6.5×10^{-6}	$\{6.95 \times 10^{-11}\}$
7	Loam	0.124	8.0×10^{-3}	8.0×10^{-8}	L 10
(8)	Silt loam	0.0726	3.5×10^{-3}	3.5×10^{-0}	$\{1.41 \times 10^{-10}\}$
9	Silt	0.0240	5.0×10^{-4}	5.0×10^{-9}	1.20×10^{-9}
10	Sandy clay	0.157	1.5×10^{-2}	1.5×10^{-7}	L1.20 x 10
11	Sandy clay loam	0.176	2.5×10^{-2}	2.5×10^{-7}	0
(12)	Sandy loam	0.167	2.0×10^{-2}	2.0×10^{-7}	2.5×10^{-9}
13	Loamy sand	0.250	5.0 x 10^{-1}_{-1}	5.0×10^{-0}	
14	Sand	0.785	1.0×10^{-1}	1.0×10^{-0}	

•

¹ See Figure ID-15.

Class in () corresponds to an overall proposed soil scenario related to the USDA soil triangle (Figure ID-15 and Table ID-1).

natural k-rate decrease of 10^{-1} to 10^{-2} contrasted to the theoretically derived values. Since Eagleson's theory deals with the entire seasonal water budget of the soil compartment, it is more or less expected that his field k(1) values obtained via model calibration [Eagleson & Tellers 1982, p.353] will be lower because of the number of water budget processes accounted (eg. infiltration, exfiltration). It may be appropriate, therefore, to reduce the k values derived and presented in table ID-6 to two orders of magnitude (i.e. 10^{-2}) in order to obtain some consistency with the calibrated k(1) values of Eagleson. There exist no theoretical justification for doing so (aside from engineering judgment), however, for scenarios and canonical environmental simulations or modeling efforts, the proposed action is not anticipated to drastically alter SESOIL output results. No sensitivity model analysis is performed at this stage to justify above action.

The finally derived saturated intrinsic permeabilities to be input as default values to SESOIL when dealing with fictitious or canonical environments are given in table ID-7, and in table ID-1 (summary; section 3.1).

3.3 Soil Effective Porosity -- n

3.3.1 Definitions

If the total unit volume V_T of a soil or rock is divided into the volume of the solid portion V_s and the volume of the voids V_v , the volumetric total porosity (or porosity) n is defined as

$$n = V_v / V_T$$
 (ID-9)

and is usually reported as a decimal fraction or a percent. Table ID-8 [Freeze & Cherry 1979, p.37] lists representative porosity ranges for various geologic materials. In general rocks have lower porosities than soils; gravels, sands, and silts, which are made up of angular and rounded particles, have lower porosities than soils rich in platy clay minerals; and poorly sorted deposits have lower porosities than well sorted deposits [Freeze & Cherry 1979].

The porosity n is an important controlling influence on hydraulic conductivity K. In sampling programs carried out within deposits of well sorted sand or in fractured rock formations, samples with higher n have in general also higher K, but the relationship does not hold on a regional basis across the spectrum of possible rock and soil types. Clay rich soils, for example, usually have higher porosities than sandy or gravelly soils but lower hydraulic conductivities. In practice, it is difficult to saturate a soil sample and then dry it and measure its porosity. It is usual, however, to estimate the total porosity (or porosity) from the relationship [Eagleson 1970, p.286]

$$n = 1 - (\rho_b / \rho_s)$$
 (ID-10)

Table ID-7

k(1) Default Values for SESOIL

Textural Soil Class (USDA)

k(1)

#	Designation	(cm ²)
1	Clay (very fine)	$7.5 \times 10^{11} = 0.75 \times 10^{11}$
$(2)^2$	Clay (medium fine)	$2.5 \times 10^{12} = 2.5 \times 10^{-1}$
3	Clay (fine)	$6.0 \times 10^{-10} = 6.0 \times 10^{-10}$
4	Silty clay	$5.0 \times 10^{-11} = 0.5 \times 10^{-11}$
5	Silty clay loam	$8.5 \times 10^{-11} = 0.85 \times 10^{-11}$
6)	Clay loam	$6.5 \times 10^{-10} = 6.5 \times 10^{-10}$
7	Loam	$8.0 \times 10^{-10} = 8.0 \times 10^{-10}$
8)	Silt loam	$3.5 \times 10^{-10} = 3.5 \times 10^{-10}$
9	Silt	$5.0 \times 10^{-11} = 0.5 \times 10^{-11}$
0	Sandy clay	$1.5 \times 10^{-9} = 15 \times 10^{-10}$
1	Sandy clay loam	$2.5 \times 10^{-9} = 25 \times 10^{-10}$
12)	Sandy loam	$2.0 \times 10^{-9} = 20 \times 10^{-10}$
3	Loamy sand	$5.0 \times 10^{-8} = 500 \times 10^{-7}$
4	Sand	$1.0 \times 10^8 = 100 \times 10^7$

1
2 See figure ID-15.
2 Class in () corresponds to an overall
proposed soil scenario related to the
USDA soil triangle (figure ID-15, table ID-1).

Table ID-8

Range of Values of Porosity

Material	Porosity r	
	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	
Unconsolidated deposits		
Gravel	25-50	
Sand	35-50	
Clay	40-70	
Rocks		
Fractured basalt	5-50	
Karst limestone	5-50	
Sandstone	5-30	
Limestone, dolomite	0-20	
Shale	0-10	
Fractured crystalline rock	0-10	
Dense crystalline rock	0-5	

Source: Freeze & Cherry [1979, p.37].

.

where  $\rho_{\rm b}$  is the bulk mass density of the sample, and  $\rho_{\rm s}$  is the particle mass density. The bulk density is the oven-dried mass of the sample divided by its field volume. The particle density is the oven-dried mass divided by the volume of the solid particles, as determined by a water displacement test. [Freeze & Cherry 1979, p.337.] In case no great accuracy is required, it can be assumed for most mineral soils  $\rho_{\rm c}$ =2.65 g/cm³.

One should be careful with the use of porosity values, when employing an unsaturated soil zone model such as SESOIL. The problem is similar to the specific yield and the porosity issue of a phreatic aquifer [Bear 1979, p.88]. As water is being drained from the interstices of the soil, the drainage is never a complete one. A certain amount of water is retained in the soil against gravity by capillary forces. After drainage has stopped, the volume of water retained is an aquifer per unit (horizontal) area and unit drop of the water table is called "specific retention"  $S_r$ , and is related to the specific aquifer yield  $S_r$  y

$$n = S_v + S_r$$
 (ID-11)

For this reason S (less than n) is also called "effective porosity" n. The correlation between porosity n, effective porosity n and median soil particle (grain) size is schematically presented in Figure ID-18 [Davis & DeWiest 1966]. SESOIL requires the effective porosity n as an input parameters.

#### 3.3.2 Soil Hydraulic Properties

An ecological and soil genesis pedologic discussion is presented by Eagleson [1982a, p.326], according to whom the overall soil properties behavior may be generalized as presented in figures ID-14 a,b (section 3.1). As discussed by Eagleson, soils evolve having a continuous spectrum of textures from clay through silt to sand and gravel, and the critical hydraulic properties of them vary widely even within the same textural class, but over the variety of classes their range is enormous. Brooks and Corey [1966] show that k(1) is related to both the shape (i.e. tortuosity) of the pores and to their total size (i.e. porosity) of a soil. Davis [1969] proposed the overall relationship [Eagleson 1982a, p.327]:

$$k(1) = Ae^{Bn}$$
 (ID-12)

where the coefficients A and B vary with textural class as sketched in figure ID-14 (section 3.1).

As discussed by Eagleson, for the fine-particled clayey soils the total particle surface area is enormous. The total pore volume which can be occupied by water that is bound to these surfaces through molecular forces is correspondingly large and the inactive porosity n dominates. With increasing sand content this volume decreases and the effective


Source: Bear [1979, p.88].

# Figure ID-18 Relationships Between Porosities and Soil Grain Size

porosity rises. As the sandy soil becomes gravelly, the falling total porosity takes over from the decreasing surface area and the effective porosity becomes smaller. This is sketched in figure ID-14b. We see from this diagram that over the clay to sand textural range encompassing most soil, permeability and effective porosity are directly related.

# 3.3.3 n SESOIL Default Values

The range of values of the porosity n or the effective porosity n is known to be quite small in field studies -- from about 0.25 to about 0.45 -- and does not have a large effect [Tellers & Eagleson 1980, Chan & Eagleson 1980] on solutions of the water balance equation HY-32 (appendix HY) as other parameters may have. The effective porosity affects directly the sigma function (equation HY-34) and consequently the gamma function of the model and the down-the-road remaining calculations, but the sensitivity of sigma vs n is not large.

Eagleson reports in 1978 [WRR, p.769, table 2] effective porosity n (i.e. n) values for various soils corresponding to the "porosity" curve of figure ID-14b. In his latest publication in 1982, however, his n values are slightly decreased [WRR, 1982, p.329, table 1]. The newly introduced 1982 porosity vs. permeability graph [WRR 1982, p.326, figure 1] indicates potential need for employing the "effective porosity" curve of figure ID-14b (instead of the n-curve). We will follow the latter logic in the following pages when developing default values for SESOIL.

Table ID-9 lists representative porosity n values (columns (1) & (2)). They are obtained from various sources and are reported for the soil classification scheme presented on table ID-7 (derived from the USDA soil triangle). The average or the best (using best judgment) value of the various sources is designed with n. This value has been "adjusted" to an  $\bar{n}$  value (column (5)) with the aid of figure ID-18, and the latter has been reported on table ID-1 (section 3.1) as a SESOIL default value. Adjustment has been performed by correlating the  $\bar{n}$ ,  $\bar{n}$  and keeping the  $\bar{n}$  estimates for clay (#2), clay loam (#6), silt loam (#8) and sandy loam (#12) to their fixed values of 0.2, 0.30, 0.30 and 0.25 correspondingly.

#### 3.4 Soil Disconnectedness Index -- c

3.4.1 Definition

In Eagleson's model, the soil pore disconnectedness index c is defined as [Eagleson 1982, p.328]:

$$c = \Delta[\ln(k)]/\Delta[\ln(s_{a})]$$
(ID-13)

where k  $[cm^2]$  the intrinsic soil permeability and s the long-term average soil moisture concentration in the root-zone. In other words, c represents the slope of a k-s, or K-s curve as graphically shown in figure ID-19.

#### lable 10-9

# Representative and SESOLL $n_e$ -Default Values

	Textural Soil Class		Porosity (n)		Effective Porosi	.ty (n _e )
	USDA Designation	Perrier & Gibson[1980] (1)	(fractional) Freeze & <u>Cherry[1979]</u> (2)	_ 7) n <u>estimate</u> (3)		ne eetimate (5)
1	Clay (very fine)			0.70	م) ع) 0.45	0.20 ⁵⁾
(2) ⁸⁾	Clay (medium fine)	0.680	0.4-0.7	0.68		0.20
3	Clay (fine)			0.64		0.22
4	Silty clay	0.592		0.59		0.25
5	Silly clay loam	0.588		0.59		0.27
(6)	Clay Joan	0.582		0.58	0.35, ¹⁾ ,0.30 ²⁾	0.30
7	Loam	0.521		0.52		0.30
(8)	Silt loam	0.535		0.54	0.35 ¹⁾³⁾ ,0.3 ²⁾ .4 ⁴⁾	0.35
9	SIIt		0.35-0.50	0.49		0.27
10	Sandy clay	0.442		0.44	0.350-0.458 ⁴⁾	0.24
11 .	Saudy clay loam	0.453		0.45		0.26
(12)	Sandy Joam	0.442		0.44	0.251)	0.25
13	Loamy sand	0.330		0.33		0.28
14	Sand	0.389	0.25-0.40	0.35	0.379-04224)	0.30

1 Eagleson [WRR, 1978, p.769] 2 El-Hemry & Engleson [1980]

³ Averaged value from clay-to-sand [Chan & Engleson 1980]

⁴ Eagleson [1970, p.287]

5 Free estimate based on Bear [1980]

⁶ Interpolated values between fixed  $\bar{n}_e$  values for solls # 2,6,8,12, reported on table ID-1 ⁷ n,  $\bar{n}_e$  final estimates as SESOII, default values

8 class in () corresponds to an overall proposed soil scenario related to the USDA soil triangle (see table ID-1)



Source: Eagleson [1979].



Brooks and Corey [1966] show that during wetting or drying soil cycles the functional relationship

$$k(s) = k(1).s^{C}$$
 (ID-14)

allows integration of the simplified Burdine [1958] equations governing the relationship between effective permeability and capillary pressure in irregular pore structures to obtain the relationship

$$c = (2+3m)/m$$
 (ID-15)

employed in Eagleson's work [Eagleson 1978, p.723]. In that respect the pore disconnectedness index c in SESOIL can also be defined as the exponent relating the "wetting" or "drying time dependent intrinsic permeability k(s) of the soil to its saturated intrinsic permeability k(l). Relation to ID-14 has been defined by Eagleson. Theoretically k is a soil property, therefore, it should be independent from the water or moisture soil content (see equation ID-5). In practice, however, because of the effective porosity issue discussed in section 3.3.2 (figure ID-14b), k becomes moisture dependent, a fact that may have resulted to the definition of equation ID-14 via the c exponent.

At this point it is worth emphasizing that any unsaturated soil zone of the literature [eg. Bonazountas et al 1979] requires as a <u>user</u> input instead of a curve relating "conductivity k vs. capillarity head  $\psi$ ." This curve is obtained from experimental data. It is almost impossible to obtain this curve off-the-shelf for any type of soil, but it is also extremely difficult to obtain this curve in the laboratory. Therefore, the "one-variable" approach of Eagleson [1978] employed in SESOIL greatly simplifies data gathering and data input to a modél.

#### 3.4.2 The c Index Sensitivity

A hydrologic balance sensitivity based on the two independent soil properties k(1) and c is presented in figure HY-38, section HY-3.5. Roughly speaking, the compartment surface runoff is insensitive to c but very sensitive to k(1).

In relation to the various soil types, and when dealing with vegetated areas, Eagleson & Tellers [1982, p.347] proves that the range of climax values of c (that is values of c at maximum vegetation canopy, given a specified soil moisture content and an ecological optimality) has been only

$$4.74 < c < 5.50$$
; c > 3.0 (ID-16)

for six catchments studied. In addition the lower limit of c is reported [Brooks & Corey 1966] to be 3.0. Typical values of c lie around 4 to 5 and the reason for this small range has been recently analyzed by Eagleson & Tellers [1982, p.348].

The correlation of c and s is evaluated via the f(s) function [Eagleson & Tellers 1982, p.349].

$$(c-3).g_{i}(c,s_{0}) = f(s_{0})$$
 (ID-17)

which by its expected value results to the very useful empirical relationship

$$(c-3).g_{i}(c,s_{o}) = 0.75$$
 (ID-18)

that is presented for 6 catchments in the lower half part of figure ID-20. This relationship allows elimination of c from the water balance equation HY-32 in terms of the state variable s and when dealing with ecologically balanced systems. However, because variation of c is within certain limits, we may employ the equations derived by Eagleson & Tellers [1982, p.349] from ID-18, i.e.

$$s_{o}[k(1)]^{1/(c+5)} = f(c)$$
 (ID-19)

and

$$k(1) = (0.058/s_0)^{c+5}$$
 (ID-20)

to make some rough estimates for c values based on k(1) values as following: for a set of soils (sand, sandy loam, silty loam, loam and clay) Tellers & Eagleson [1980] report from their experience with equations ID-19, ID-20, the expression

$$k(1) = (m/512.7)^{2.75}$$
(ID-21)

where m = 2/(c-3) as given by parameter set in HY-29 (app. HY) and graphically reported in figure ID-21. Therefore:

* In general (regression line; figure ID-21):

$$c = k(1)^{1/2.75} (512.7/2)+3$$
 (ID-22)

* For sandy soils (m=3, figure ID-21):

$$c = 3.7$$
 (ID-23)

* For sandy loam soils (m=0.5, figure ID-21), and

$$c = 6.33$$
 (ID-24)

Table ID-10 summarizes obtained and compiled c values for the USDA soil texture classification triangle. Values of the last column are reported in table ID-1 also. Based on information obtained from the literature



(b)

Source: Eagleson [1982], Eagleson & Tellers [1982].

Figure ID-20 Functional Relationship, c-f(S₀); Saturated Intrinsic Permeability



Figure ID-21 Saturated Permeability vs Pore Size Distribution Index (from Eagleson, Personal Communication)

# Table ID-10

Default	С	Values	for	SESOIL
---------	---	--------	-----	--------

Te	_K tural Soil Class USDA	c-Soil	ties	SESOIL Default c-values ³⁾				
#	Designation	k(1)	c1)	د ₂ 2)				
		(cm ² )	(-)	(-)	-			
1	Clay (very fine)	7.5x10 ⁻¹¹	25	12	12.0			
(2)5	) Clay (medium fine)	$2.5 \times 10^{-10}$	16	12	12.0			
3	Clay (fine)	$6.0 \times 10^{-10}$	16	8.54 ⁴⁾	12			
4	Silty clay	$5.0 \times 10^{-11}$	23		12			
5	Silty clay loam	$8.5 \times 10^{-11}$	10.6	3.8 ⁴⁾	10			
(6)	Clay loam	$6.5 \times 10^{-10}$	13	10,5.05	7.5			
7	loam	$8.0 \times 10^{-10}$	11.0	8.5 ⁴⁾	6.5			
(8)	Silt loam	$3.5 \times 10^{-10}$	16.0	6,4.95	5.5			
9	Silt	$5.0 \times 10^{-11}$	23.0		12			
10	Sandy loam	$1.5 \times 10^{-9}$	6.33		6			
11	Sandy clay loam	$2.5 \times 10^{-9}$	8.70		4.0			
(12)	Sandy loam	$2.0 \times 10^{-9}$	8.70	4	4.0			
13	Loamy sand	$5.0 \times 10^{-8}$	5.00		3.9			
14	Sand	$1.0 \times 10^{-8}$	3.7		3.7			

 $^{\rm l}$  approximate values by employing equation ID-22 and the k(l) values of table ID-7

² Eagleson [1977, 1978, 1981, 1982]

. -

- 3  Employ primarily values of soils # 2,6,8 and 12
- ⁴ Talsma [1974], Moore [1939], Brook & Corey [1966] from Eagleson [1978a-g].

⁵ Soil class in () corresponds to an overall proposed soil scenario related to the USDA soil traingle; in figure with

the c value range has been confined between c=12 [Eagleson 1978] and c=3.7 [see figure ID-21 for clay]. The c₁ theoretical estimates are kept as a guidance. The value of soil #6 (clay loam, c=7.5) equals the average of the two Eagleson studies [1982, 1978]. The same applies to soil #8 (silt loam). For the remaining soil types rough linear interpolation is performed, because the changes of c do not affect water balance estimates strongly. As information becomes available, these default values will be improved. For site specific application, model calibration to estimating the independent soil parameters c, k(1), n is recommended.

### 3.5 Soil Parameter Calibration

# 3.5.1 General

A single general relationship for k(1), n and c does not exist, because the vegetation canopy density and evapotranspiration -- both related to n, k(1) and c -- affect the interrelation of the three parameters in an area. Tellers & Eagleson [1980] and Eagleson & Tellers (1982] have employed the water balance theory of Eagleson [1978a-g] to estimate the effective hydrologic properties of soils from observation on vegetation density. Bonazountas et al [1981] have employed the monthly water balance of SESOIL to estimate soil moisture contents of soil, where field data were available.

Three types of input parameters are associated with Eagleson's hydrobalance routine: climatic, soil and vegetation. The climatic and vegetal properties are easily obtained from observations; this leaves the soil parameters to be determined from relationship between climate, soil and vegetation [Eagleson 1978a-g]. Four soil parameters are associated with Eagleson's theory; three independent soil properties (k(1), n, c) and the soil moisture state variable s. In calibrating the model, two methods are of major importance: (1) calibration of the soil parameters k(1), c via s and n, or (2) calibration of s via k(1), c and n. The vegetation part of Eagleson's theory has been eliminated from SESOIL, however, calibration procedures remain the same.

# 3.5.2 Calibration of k(1), c via s

The range of values of the porosity, n, is known to be quite small, from 0.25 to about 0.45 (see section 3.3), and does not have a large effect on solutions of the water balance equation. Assuming a known value for n, this fact leaves the soil moisture, intrinsic permeability, and pore disconnectedness index as unknowns. To solve for these variables, two equations or relationships are needed which incorporate the soil and climate as well. The first relationship is the water balance, Equation HY-32, which expresses the soil moisture, s, as an implicit function of the climate and soil. The second expression used is a rather weakly correlated regression between k(1) and m (figure ID-21, equation ID-21). With the information produced by Eagleson & Tellers [1982] for vegetated

sites, the following calibration procedure can be designed, to estimate effective soil parameters, given a set of climatic parameters.

- (1) A value of n is assumed, in order to estimate E (equation HY-37).
- (2) The lowest possible value for c, approximately 3.5, is selected as an initial value.
- (3) k(1) is calculated from equation ID-21.
- (4) With these values for the three soil parameters,  $n_e$ , k(1), and c, it can be seen from equation HY-37 that s remains as the only variable needed for determining E. With E known from step (1), s is calculated.
- (5) Annual precipitation is calculated via equations HY-32 through HY-43.
- (6) If the annual precipitation from the above step is not equal to the actual mean rainfall, c is incremented upward from its initially low value and steps (3)-(5) are repeated.
- (7) Due to the approximation introduced by using equation ID-21, the precipitation,  $P_A$ , calculated in step (5) may never exactly equal the actual mean value,  $m_{PA}$  for any value of c. P_A will approach  $m_{PA}$  as c is increased, coming to within  $\Delta P_A$  of equality at intermediate c before diverging again for large c. For low values of c, the calculated k(1) is large, representing a soil with high permeability and well connected pores. With evapotranspiration specified at the optimum (i.e. minimum) value, a large precipitation is therefore calculated in order to produce the inevitably large groundwater yield of the highly porous soil. For large c and small k(1), the soil is extremely impervious and the surface yield will be high. With minimum evapotranspiration, a large value for precipitation is again needed. Somewhere between these two extremes, a set of suitable soil parameters is obtained which gives an annual precipitation,  $P_A$ , which is closest to the actual mean,  $m_{P_A}$ . This relationship is illustrated in figure ID-22. m_{PA}. Holding c constant at the value which gives the minimum  $\Delta P_{i}$ , k(1) is then deviated from regression equation ID-21 until another minimum in calculated precipitation is reached. If this value is above the mean precipitation, c is decreased, if it is below the mean, c is increased. Another search is done on k(1) until the minimum precipitation is found. This step is repeated until the minimum calculated precipitation is equal to the mean.
- (8) If the values obtained for k(1) and c are not consistent with the assumed porosity, n is adjusted to a more appropriate



Source: Tellers & Eagleson [1980].



value corresponding to a more pervious or impervious soil type depending on the values of k(1) and c. Steps (1) through (7) are repeated.

The soil parameters obtained from steps (1-8) are used to construct the CDF of annual yield in the same manner as Eagleson [1978g]. The above procedure can be repeated for 3 input data categories; i.e. climate, soil and vegetation in case equation HY-32 were not simplified to the bare soil SESOIL needs (see section ID-3.1).

# 3.5.3 Calibration of s via k(1), n, c

The concept of procedure previously described can be repeated to calibrate SESOIL either via existing s field measurements, or via USGS data records for basin yields (surface runoff and groundwater runoff) [Eagleson 1978a-g]. In this particular case

- (1) a n, c set is obtained from table ID-1.
- (2) the model is run, and model yield output (equation HY-42) is compared to USGS basin yields.
- (3) Depending upon the distribution of the surface vs groundwater runoff yield predictions, parameters n, c, and k(l) are adjusted to new values, primarily by adjusting k(l) (see section ID-3.2) and by following the sensitivity consensus of figures HY-7 and HY-8.
- (4) Above steps are repeated until basin yields and averaged soil moisture s predictions have reached their field values.

### 3.5.4 Automated Calibration

It is feasible to write a program automatically calibrating SESOIL and all its input parameters given USGS data records for basin yields and some basic precompiled information such as the default data of table ID-1. However, development of such a computer code has been beyond the scope of the developers involvement.

### 4.0 CHEMISTRY INPUT DATA

Chemical parameters, coefficients, etc. may be compiled from the handbook "Research and Development Methods for Estimating Physicochemical Properties of Organic Compounds of Environmental Concern" [Lyman et al 1981], or any other handbook. In the future, authors of SESOIL intend to compile a data base for chemical properties relevant to the model use and a number of pollutants.

### 5.0 CANONICAL CLIMATIC-SOIL COMPARTMENTS

Many environmental studies, such as human exposure assessments related to hazardous waste sites, may require the design of typical or canonical soil-compartments; and SESOIL is well suited for such simulations [Bonazountas et al 1981]. The design, however, of a canonical compartment is a function of both the climate and the soil-type of the environment; therefore, only climatic or only soil canonical (default) data sets would not suffice for a model user if he has to describe a canonical soil-compartment. There may be a way to design for the entire U.S. canonical soil environments accounting for both climatologic and soil type default values, but since model developers have not yet finalized their thinking regarding this issue, and they are not confident regarding their current technical approach, they have decided not to present this information in this section. Some information may follow in the future.

#### 6.0 REFERENCES

Bear, J. (1979). "Hydraulics of Groundwater," New York: McGraw Hill International Book Company.

Bonazountas, M.; W. Hawes and W. Tucker (1979). "Heutistic Unsaturated Flow/Quality Soil Zone Model," American Geophysical Union, Spring Meeting, Washington, D.C.

Bonazountas, M.; J. Wagner and B. Goodwin (1981). "Evaluation of Seasonal Soil/Groundwater Pollutant Pathways." EPA Contract No. 68-01-5949, Task 9. Washington, D.C.: Monitoring and Data Support Division, U.S. EPA.

Bonazountas, M.; J. Wagner; J. Segna; and M. Slimak. (1982). "The 'SESOIL' Model and Exposure Studies." J. Environ. Monit. & Assessment (submitted for publication).

Brooks, R.H. and A.T. Corey (1966). "Properties of Porous Media Affecting Fluid Flow," J. Irrig. Drain. Div. Amer. Soc. Civil Eng., IR2, 61-68.

Buckman, H.O. and N.C. Brady (1969). "The Nature and Properties of Soils." London: The Macmillan Company, 653 pp.

Burdine, N.T. (1952). "Relative Permeability Calculations from Pore-size Distribution Data." Petrol. Trans., Am. Inst. Mining, Met. Petrol. Engrs., Vol. 198, 71-77.

Chan, S. and P.S. Eagleson (1980). "Water Balance Studies of the Bahr El Ghazal Swamp," Report No. 261, Ralph M. Parsons Laboratory for Water Resources and Hydrodynamics, Department of Civil Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts.

Chiang, S.L. (1971). "A Runoff Potential Rating Table for Soils." J. Hydrology Vol. 13, 54-62.

Chow, V.T. (1964). "Runoff." In: Handbook of Applied Hydrology. New York: McGraw Hill Book Company, Inc.

Corey, A.T. (1977). "Mechanics of Heterogeneous Fluids in Porous Media." Water Resources Publication, Fort Collins, Colorado.

Eagleson, P.S. (1970). "Dynamic Hydrology," New York: McGraw-Hill Book Company.

Eagleson, P.S. (1977). "Climate, Soil and the Water Balance: A Framework for Their Analytical Coupling," Department of Civil Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts. Eagleson, P.S. (1978a-g). "Climate, Soil, and Vegetation," Water Resources Research, 14(5), 705-776. Eagleson, P.S. (1982). "Ecological Optimality in Water-Limited Natural Soil-Vegetation Systems. 1. Theory and Hypothesis," Water Resources Research, 18(2), April 1982; 325-340.

Eagleson, P.S. and T.E. Tellers (1982). "Ecological Optimality in Water-Limited Natural Soil-Vegetation Systems. 2. Tests and Applications," Water Resources Research 18(2), April 1982; 341-354.

El-Hemry, I.I. and P.S. Eagleson (1980). "Water Balance Estimates of the Machar Marshes," Report No. 260, Ralph M. Parsons Laboratory for Water Resources and Hydrodynamics, Department of Civil Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts.

Freeze, R.A. and J.A. Cherry (1979). "Groundwater," Englewood Cliffs, New Jersey: Prentice-Hall, Inc.

Grayman, W.M. and P.S. Eagleson (1969). "Streamflow Record Length for Modelling Catchment Dynamics," Report No. 114, Hydrodynamics Laboratory, School of Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts.

Holton, H.N.; C.B. England and D.E. Whelan (1967). "Hydrologic Characteristics of Soil Types. J. Irrig. Drainage Div. IR3, 33-39.

Horn, M.E. (1971). "Estimating Soil Permeability Rate," J. Irrigation and Drainage Div., 97(IR2), p.263.

Libardi, P.L.; K. Reichardt; C. Jose; M. Bazza and D.R. Nielsen (1982). "An Approximate Method of Estimating Soil Water Diffusivity for Different Soil Bulk Densities," Water Resources Research 18(1), February 1982; 177-181.

Lyman, W. et al. (1981). "Research and Development Methods for Estimating Physicochemical Properties of Organic Compounds of Environmental Concern." Prepared by Arthur D. Little, Inc., Phase II, Final Report for U.S. Army Medical Bioengineering Research and Development Laboratory, Fort Detrick, Maryland. New York: McGraw-Hill Book Company.

McWhorter, D.B. and D.K. Sunada (1977). "Ground-water Hydrology and Hydraulics," Water Resources Publication, Fort Collins, Colorado.

Metzger, B.H. and P.S. Eagleson (1980). "The Effects of Annual Storage and Random Potential Evapotranspiration on the One-Dimensional Annual Water Balance," Report No. 251, Ralph M. Parsons Laboratory for Water Resources and Hydrodynamics, Department of Civil Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts.

Moore, R.E. (1939). "Water Conduction from Shallow Water Tables," Hilgardia 12(6), 382-426. Nielsen, D.R.; J.W. Biggar and K.T. Erh (1973). "Spatial Variability of Field Measured Soil-Water Properties." Hilgardia Vol. 42(7), 215-260.

Novotny, V. (1976). "Hydrological and Hydraulical Conceptual Models Applicable to Overland and River Transport Modeling," Literature Review No. 4, Water Resources Center, University of Wisconsin, Madison, Wisconsin.

Perrier, E.R. and A.C. Gibson (1980). "Hydrologic Simulation on Solid Waste Disposal Sites (HSSWDS), Contract No. EPA-IAG-D7-01097, Cincinnati, Ohio: Municipal Environmental Research Laboratory, ORD, U.S. Environmental Protection Agency.

Talsma, T. (1974). "The Effect of Initial Moisture Content and Infiltration Quantity on Redistribution of Soil Water," Australia. J. Soil Res., 12, 15-26.

Tellers, T.E. and P.S. Eagleson (1980). "Estimation of Effective Hydrologic Properties of Soils from Observations of Vegetation Density," Report No. 254, Prepared by Ralph M. Parsons Laboratory for Water Resources and Hydrodynamics, Department of Civil Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts.

Zunker, F. (1930). "Das Verhalten des Bodens zum Wasser." Handbuch der Bodenlehre, Vol. VI, 66-200.

DF - data files

### APPENDIX DF

# DATA FILES (INPUT/OUTPUT)

1.0	INTRODUCTION	DF-3
2.0	INPUT DATA ARRAYS (IDA)	DF-5
	<ul> <li>2.1 General</li> <li>2.2 Climatologic IDA Arrays</li> <li>2.3 Soil IDA Arrays</li> <li>2.4 Chemistry IDA Arrays</li> <li>2.5 Geometric IDA Arrays</li> <li>2.6 Application Specific IDA Arrays</li> </ul>	DF-5 DF-5 DF-8 DF-10 DF-11 DF-12
3.0	OPERATIONAL/RETRIEVAL ARRAYS (ORA)	DF-16

# FIGURES

DF-1 SCHEMATIC OF "SESOIL" DATA FILE/ARRAY. OVERALL STRUCTURE	DF-4
DF-2 CLIMATOLOGICAL INPUT DATA ARRAYS (IDA)	DF-6
DF-3 SOIL, CHEMISTRY, GEOMETRIC INPUT DATA ARRAYS (IDA)	DF-9
DF-4 APPLICATION SPECIFIC INPUT DATA ARRAYS (IDA)	DF-13
DF-5a OPERATIONAL/RETRIEVAL ARRAYS (ORA)	DF-17
DF-5b OPERATIONAL/RETRIEVAL ARRAYS (ORA)	DF-18

#### 1.0 INTRODUCTION

This appendix is intended to guide users through details of the data management and the computational procedures of the FORTRAN code of SESOIL. It will also help model developers (Bonazountas and Wagner) in the future, since model design and development has not been completed.

SESOIL is operational at various levels as discussed in Section 3.0, User's Manual. To organize data management and provide a quality assurance control during operations, the input, output and intermediate data are stored in data files (arrays).

SESOIL is structured around

- (1) An input data file system (IDFS), and
- (2) A data management array system (DMAS)

The input data file system (IDFS) contains user input or permanently stored input data (climatic, soil, chemistry; see appendix ID) required to operate the model. This input data file system consists of 5 input data files, the GE DATA, LO DATA, L1 DATA, L2 DATA and EXEC DATA. Detailed information regarding these files is given in section 3.4 (Input Data Files), of section 3.0 (User's Manual) of this documentation; therefore, it is not discussed in this appendix.

The data management array (DMAS) system contains two major types of arrays:

- (2.1) Input data arrays (IDA) and
- (2.2) Operational/Retrieval arrays (ORA)

The input data arrays (IDA) contain the input data relevant to a <u>parti-</u> <u>cular</u> simulation. These data are read from the input data file system (IDFS) and are stored in the input data arrays (IDA) by subroutine RFILE (<u>Read FILE</u>). RFILE can access any of the 5 data files: GE DATA, EXEC DATA, LO DATA, L1 DATA, L2 DATA.

Operational/Retrieval arrays (ORA) contains data or results of intermediate calculations, and/or data for transfer to other subroutines. These arrays can be accessed (from the FORTRAN code) by a programmer or user to check on the correctness of intermediate computational steps of the model. As such, ORA-arrays serve as "control nodes" of the model code.

A schematic presentation of the SESOIL operations via its data file system is shown on the next page (Figure DF-1). Most of the DMAS arrays used in SESOIL and their contents are discussed in the following sections.



TIGURE DF-1 -- SCHEMATIC OF "SESOIL" DATA FILE/ARRAY OVERALL STRUCTURE 2.0 INPUT DATA ARRAYS (IDA)

2.1 General

The input data arrays contain the input data required for a <u>particular</u> simulation. Four major imput data arrays are available: climatologic soil, chemistry, geometric and application specific arrays. The array name, and the parameters (in FORTRAN code and their units) in each space of the arrays are given in the following sections.

2.2 Climatologic IDA Arrays

The climatological input data arrays are schematically presented in the next page (Figure DF-2) and are descirbed below:

CLIMA1(6) -- annual climatic parameters of the region

Each line of the array contains one year of data for the following 6 parameters:

- Latitude; (L;°N)
- 2. Average annual soil surface temperature; (T;°C)
- 3. Average annual fraction of sky covered by clouds; (NN;-)
- 4. Average annual relative humidity (fractional); (S;-)
- 5. Average annual shortwave albedo of the surface; (A;-)
- 6. Average daily evapotranspiration; (REP; cm/day)

CLIMA2(6) -- annual storm parameters of the region

Each line contains one year's data of six storm-related parameters of the region:

- 1. Mean annual precipitation; (MPA;cm)
- 2. Mean annual storm duration; (MTR; days)
- 3. Mean number of storms per year; (MN;-)
- 4. Mean length of rainy season within a year (MT;days)
- 5. Empty space in array (no data)
- 6. Empty space in array (no data)

L	TA	NN	ຣ	Å	REP
°N	ື ເ	-	-	-	٣٧







CLIMAZ (6)

CLIMA1 (6)

CLIMM1(6,12,10)







FIGURE DF-2 -- CLIMATOLOGICAL INPUT DATA ARRAYS (IDA)

CLIMA3(12) -- monthly storm depths (Oct-Sep)

This array contains one year's data of the monthly distribution of storm depths (MPM:cm) in the area, starting with the month of October (hydrologic year).

Note: Information contained in this array is not currently used by the program (see discussion of CLIMM1 and CLIMM2). The array was built into the code for a previous model version and is maintained for potential future use.

CLIMM1(6,12,10) -- monthly climatic parameters

Each block 6x12 of the 6x12x10 array contains one year of climatic parameters of a region. Ten years of data may be stored in total (IYR=1,10).

Each line of the 6x10 block contains 12 values of a particular parameter for the 12 months of the hydrologic year (Oct-Sep;IMO=1,12). These parameters are:

CLIMM1(1,1 ,IYR): (L; °N)

CLIMM1(2,IMO,IYR): surface temperature; (T;°C)

CLIMM1(3, IMO, IYR): fraction of sky covered by clouds; (NN;-)

CLIMM1(4,IMO,IYR): relative humidity (fractional); (S;-)

CLIMM1(5, IMO, IYR): shortwave albedo; (A;-)

CLIMM1(6, IMO, IYR): daily evapotranspiration rate; (REP; cm/day)

CLIMM2(6,12,10) -- monthly storm parameters of a region

Each 6x12 block of the 6x12x10 array contains one year of storm-related parameters of the region. Ten years of data may be stored in total (TYR=1,10).

Each line contains the values of a particular parameter for the twelve months of the hydrologic year (Oct-Sep;IMO=1,12). The parameters are:

CLIMM2(1, IMO, IYR): mean monthly precipitation; (MPM; cm)

CLIMM2(2, IMO, IYR): mean storm duration; (MTR; days)

CLIMM2(3, IMO, IYR): number of storms per month; (MN;-)

CLIMM2(4, IMO, IYR): mean length of rainy period within the month; (MT; days)

CLIMM2(5, IMO, IYR): empty space in array

CLIMM2(6,IMO,IYR): empty space in array CLIMM3(12,10) -- currently an empty array

2.3 Soil IDA Arrays

The soil input data arrays are schematically presented on the next page (Figure DF-3). They are:

SOIL1(6) -- basic regional soil parameters

This array contains the following six basic soil parameters:

- Soil density; (RS;g/cm³)
- 2. Intrinsic average (depth) soil permeability; (K1;cm²)
- 3. Soil pore disconnectedness index; (c;-)
- 4. Effective soil porosity; (N;cm³/cm³)
- 5. Organic carbon content of the soil; (oc;%), and
- 6. Carbon content of the soil; (cc;%).

SOIL2(6) -- other soil related parameters

This array contains additional soil parameters of site specific simulations.

- 1. Soil cation exchange capacity; (CEC;m.e./100g dry w.t. soil)
- 2. Intrinsic soil permeability of upper soil layer; (KIU;cm²)
- 3. Intrinsic soil permeability of middle soil layer; (KIM;cm²)
- 4. Intrinsic soil permeability of lower soil layer; (KIL:cm²)
- 5. Empty space for future use
- 6. Empty space for future use

S011(6)	R5 1 9/3	2 2 CW	<u>C</u> 3	N 	5 %	<u>()</u> () ()	I													
SDILS (6)	BIA LCEC	N12 2 2	N C KAH	E. E KIL	•	•														
	]st	koc	A	KDE	ŧ	X	Hwt	VAL	KNH	K B H	KAH		5K	ß	MWTLIG				Ŧ	
CHEM1 (18)	< 7m/8n	48/84 N	cw ² /s w	day -1 E	marate G	45/9 ml	3/mol ~	8	day-1 b	L/wel-day 5	- Vaul. day	•	13	<del>بر</del> ۱	g/md 5	•	•	•		
NUT1(6)	en	mpty		<b>TN 0</b>	સ															
	AR	N	PO	Ma	שר	A2LDE	A20C	A2 CC	A KDE	AOC	Αςς	1580	ASL	FRN	PH	A2PH	APH	A2CEC	A CEC	
GED <b>M</b> (2 <b>0</b> )	1	2	3	4	5	6	7	8	٩	10	11	12	13	14	15	16	17	18	19	·
	N N N	£	CM	C M	л З	١	١	١	١	١	١	١	١	١	١	١	١	١	١	•

FIGURE DF-3 -- SOIL, CHENISTRY, GEOMETRIC INPUT DATA ARRAYS (IDA)

### 2.4 Chemistry IDA Arrays

The chemistry arrays (Figure DF-3) are:

CHEM1(18) -- chemical specific parameters

This array contains 14 basic parameters of a compound, and has 4 empty spaces. These parameters are:

- 1. Solubility of the compound; (SL;ug/mL or mg/L)
- Adsorption coefficient of the compound based on organic carbon; (KOC; (ug/g oc)/(ug/mL))
- 3. Diffusion coefficient of pollutant in air;  $(DA; cm^2/s)$
- 4. Degradation rate in the upper unsaturated soil zone; (KDE;day⁻¹)
- 5. Henry's Law Constant of the compound; (H;m^{3.atm/mol})
- Overall adsorption coefficient of compound on soil; (K;(ug/g soil)/(ug/mL))
- 7. Molecular weight of compound; (MWT;g/mol)
- 8. Valence of compound; (VAL;-)
- 9. Neutral hydrolysis constant (KNH;day⁻¹)
- 10. Base hydrolysis constant (KBH;L/(mol·day))
- 11. Acid hydrolysis constant (KAH;L/(mol·day))
- 12. Empty space
- 13. Stability constant of compound-ligand complex; (SK;-)
- 14. Number of moles of ligand per mole of compound complexed (B;-)
- 15. Molecular weight of ligand; (MWTLIG;g/mol)
- 16. Empty space
- 17. Empty space
- 18. Empty space

NUT1(6) -- nutrient cycle parameters.

The array can contain up to 6 parameters relating to the nutrient cycle. This array is presently empty, since the nutrient cycle routine is not operational

2.5 Geometric IDA Arrays

The geometric array (Figure DF-3) is:

GEOM(20) -- application related geometric parameters

The array contains 17 parameters of the application geometry and 3 empty spaces. The number of parameters input depends upon the level of operation. The 17 parameters are:

- 1. Area of the compartment for all levels; (AR;cm²)
- 2. Depth to groundwater for all levels; (Z;m)
- 3. Depth of the upper soil zone for all levels; (DU;cm)
- 4. Depth of the middle soil zone for level 3 (DM;cm)
- 5. Depth of the lower soil zone for all levels; (DL;cm)
- Ratio of biodegradation, middle/upper soil zone; (A2KDE;-) for level 3
- Ratio of organic carbon content, middle/upper soil zone (A20C;-) for level 3
- Ratio of clay content, middle/upper soil zone (A2CC;-) for level 3
- Ratio of biodegradation, lower/upper soil zone for all levels (AKDE;-)
- Ratio of organic carbon content, lower/upper soil zone for all levels; (AOC;-)
- Ratio of clay content, lower/upper soil zone for all levels; (ACC;-)
- 12. Index of pollutant participation in surface runoff for levels 0 and 1; (ISRA:-)
- Ratio of concentration of pollutant in rainfall to maximum solubility for levels 0 and 1; (ASL;-)

- 14. Freundich coefficient for levels 2 and 3; (FRN;-)
- 15. pH in upper soil zone for all levels; (PH;-)
- 16. Ratio of pH middle/upper for level 3; (A2PH;-)
- 17. Ratio of pH, lower/upper for all levels; (APH;-)
- 18. Ratio of CEC, middle/upper for level 3; (A2CEC;-)
- 19. Ratio of CEC, lower/upper for all levels; (ACEC;m-)
- 20. Empty space

2.6 Application Specific IDA Arrays

The application specific arrays (Figure DF-4) are:

RUNLO(6) -- LEVELO parameters

The array contains values of 4 parameters required to run LEVELO. It also has 2 empty spaces. The 4 parameters are:

- 1. Soil moisture; (THA;-)
- 2. Infiltration; (IA; cm)
- 3. Groundwater runoff; (RGA; cm)
- 4. Surface runoff; (RSA;cm)
- 5. Empty space
- 6. Empty space

LOAD(6) -- LEVELO and LEVEL1 pollutant loadings to compartment The array contains 4 loading parameters and 2 empty spaces:

- 1. The total loading in the upper zone; (POLINU; $ug/cm^2$ )
- 2. The total loading in the lower zone (POLINL; $ug/cm^2$ )
- The total ligand mass input to the upper zone; (LIGU;ug/cm²)
- The total ligand mass input to the lower zone; (LIGL;ug/cm²)
- 5. Empty space
- 6. Empty space

2)  

$$n \frac{1}{12} \frac{1}$$

RUNLO (6)

RUN M1 (10,12)

RUNM2 (10,12)

	OC Nr. De	Jon ten 4	2. 42 403 Jun Juns 640 6
4	CPM	ug/mL	
2	TRANSU	ug/cm²	>
3	TRANSM	ug/cu ²	>
u	TRANS L	ug/m²	$\rightarrow$
5	SINKU	¥g/m²	->
G	SINKM	ug/cm²	
7	SINKL	ug/cyz	
8	cigu	9/ cm ²	<b>→</b>
q	LIGM	g/ cw ²	<b></b>
lo	LIGL	8/cw²	

TITLES (5,12)

REGIONAL TITLE
 SOIL TITLE
 COMPOUND TITLE
 NUTRIENT TITLE
 APPLICATION AREA TITLE

FIGURE DF-4 -- APPLICATION SPECIFIC INPUT DATA ARRAYS (IDA)

RUNM1(10,12) -- LEVEL2 and LEVEL3 loading parameters

The array contains values of 7 monthly parameters for above levels of operation. The columns represent months from October to September. The content of the lines is:

- 1. Concentration of pollutant in the upper zone soil moisture; (CUM;ug/mL)
- 2. Concentration of pollutant in the middle zone soil moisture; (CMM;ug/mL)
- Concentration of the pollutant in the lower zone soil moisture; (CLM;ug/mL)
- 4. Monthly loading in the upper soil zone; (POLINU;ug/cm²)
- 5. Monthly loading in the middle soil zone; (POLINM; $ug/cm^2$ )
- 6. Monthly loading in the lower soil zone (POLINL;ug/cm²)
- 7. Multiplier for pollutant in surface runoff by month; (ISRM;-)
- 8. Empty spaces
- 9. Empty spaces
- 10. Empty spaces

RUNM2(10,12) -- LEVEL2 and LEVEL3 pollutant parameters

The array contains 7 monthly pollutant parameters for above levels. The columns represent the months, from October to September. The lines contain:

- Concentration of pollutant as fraction of solubility in the rainfall; (ASL;ug/mL)
- 2. Rate of pollutant transformation in the upper zone;  $(TRANSU; ug/cm^2)$
- 3. Rate of pollutant transformation in the middle zone;  $(TRANSM;ug/cm^2)$
- 4. Rate of pollutant transformation in the lower zone;  $(TRANSL;ug/cm^2)$
- 5. Pollutant loss (by processes of source/sink) in the upper zone (SINKU;ug/cm²)
- Pollutant loss (by processes of source/sink) in the middle zone (SINKM;ug/cm²)
- 7. Pollutant loss (by processes of source/sink) in the lower zone (SINKL;ug/cm²)
- 8. Ligand mass input to the upper zone; (LIGU;ug/cm²)

9. Ligand mass input to middle zone; (LIGN, ug/cm²)

10. Ligand mass input to lower zone; (LIGL;ug/cm²) TITLES(5,12A4) -- titles of the particular simulation This alphanumeric array contains all titles of a SESOIL.

- 1. Line 1 contains the regional title
- 2. Line 2 contains the soil title
- 3. Line 3 contains the compound title
- 4. Line 4 contains the nutrient cycle title, and
- 5. Line 5 contains the application area title

3.0 OPERATIONAL/RETRIEVAL ARRAYS (ORA)

The overall use of the operational/retrieval (ORA) arrays has been discussed in section 1.0 (Figure DF-1). Important operational arrays are present in the figures on the next page (Figure DF-5a,b). The parameters, the symbols in FORTRAN and the units of each array are presented below.

HYDBAL(13,10) -- hydrologic cycle array

This array is used to store and transfer results of the hydrologic cycle routine. The first 12 lines contain data for the months of the hydrologic year (Oct-Sep). The last line contains the total or the average over the year. The columns contain:

- 1. Moisture content; (THA; fractional)
- 2. Monthly precipitation; (MPM;cm)
- Monthly infiltration; (IM;cm)
- 4. Monthly evapotranspiration; (REP;cm/day)
- 5. Monthly surface runoff; (RSM; cm)
- 6. Monthly groundwater recharge; (RGM;cm)
- 7. Convergence function; (GZ;-)
- 8. Empty spaces
- 9. Empty spaces
- 10. Empty spaces

PINP(13,6) -- pollutant input parameters array

This array is used to store the pollutant input masses calculated for each month of the monthly simulations (levels 2 and 3). The first 12 lines contain data for the months of the hydrologic year (Oct-Sep). The last line contains the total inputs for the year. The columns contain:

- 1. Pollutant input mass via rainfall; (PINFU;ug/cm²)
- 2. Pollutant mass input directly to the upper zone; (POLINU;ug/cm²)
- 3. Pollutant mass input directly to the lower zone; (POLINL;ug/cm²)
- 4. Pollutant mass input directly to the middle zone; (POLINM;ug/cm²)
- 5. Empty (spaces)
- 6. Total pollutant input mass for soil column; (PIN;ug/cm²)



FIGURE DF-50 -- OPERATIONAL/RETRIEVAL ARRAYS (ORA)


PCONC (13,15) -- pollutant concentrations array

This array is used to store the pollutant concentrations in the various media (air, soil, soil moisture) for each month of the monthly simulation (levels 2 and 3). The first 12 lines contain data for the months of the hydrologic year (Oct-Sep). The last line contains the total inputs for the year. The columns contain:

- 1. Pollutant concentration in the upper zone soil moisture; (CUM;ug/mL)
- 2. Pollutant concentration in the middle zone soil moisture; (CMM;ug/mL)
- 3. Pollutant concentration in the lower zone soil moisture; (CLM;ug/mL)
- 4. Pollutant concentration on the soil in the upper zone; (SUM;ug/g)
- 5. Pollutant concentration on the soil in the middle zone; (SMM;ug/g)
- 6. Pollutant concentration on the soil in the lower zone (SLM;ug/g)
- Pollutant concentration in the soil air of the upper soil zone; (CUSA;ug/mL)
- Pollutant concentration in the soil air of the middle soil zone; (CMSA;ug/mL)
- Pollutant concentration in the soil air of the lower soil zone; (CLSA;ug/mL)
- Free Ligand concentration in the soil moisture of the upper zone; (LIGCUF;ug/mL)
- 11. Free Ligand concentration in the soil moisture of the middle zone; (LIGCMF;ug/mL)
- Free Ligand concentration in the soil moisture of the lower zone; (LIGCLF;ug/mL)
- 13. Depth of rainfall "front"; (DPTH;cm)
- 14. Empty column
- 15. Empty column

POLBAL(13,45) -- pollutant mass array

This array is used to store the pollutant masses involved in the individual fate processes for each month of the monthly simulation for levels 2 and 3. The first 12 lines contain data for the months of the hydrologic year (Oct-Sep). The last line contains the total or the remaining mass at the end of the year. The columns contain:

- 1. Pollutant mass in surface runoff; (PRSM;ug)
- 2. Pollutant mass volatilized from upper zone (PVOLU;ug)
- 3. Pollutant mass in other sinks from upper zone; (PSINKU;ug)
- 4. Pollutant mass adsorbed in upper zone (PADSU;ug)
- 5. Pollutant mass degraded in upper zone (PDEGU;ug)
- 6. Pollutant mass transformed in upper zone; (PTRANU;ug)
- 7. Pollutant mass released to groundwater; (PRGM;ug)
- 8. Pollutant mass in other sinks from lower zone; (PSINKL;ug)
- 9. Pollutant mass adsorbed in lower zone; (PADSL;ug)
- 10. Pollutant mass degraded in lower zone; (PDEGL;ug)
- 11. Pollutant mass transformed in lower zone; (PTRANSL;ug)
- 12. Pollutant mass dissolved in soil moisture in upper zone; (PMOIU;ug)
- Pollutant mass dissolved in soil moisture in lower zone; (PMOIL;ug)
- 14. Empty space

15. Empty space

- 16. Pollutant mass cation exchanged in upper zone; (PCECU;ug)
- 17. Pollutant mass cation exchanged in lower zone; (PCECL;ug)
- 18. Pollutant mass hydrolyzed from moisture in upper zone; (PIIYDMU;ug)
- 19. Pollutant mass hydrolyzed from moisture in lower zone; (PHYDML;ug)
- 20. Pollutant mass complexed in upper zone; (PCOMU;ug)

Pollutant mass complexed in middle zone; (PCOML;ug) 21. 22. Pollutant mass in other sinks in middle zone; (PSINKM;ug) 23. Pollutant mass adsorbed in middle zone; (PADSM;ug) 24. Pollutant mass degraded in middle zone; (PDEGM;ug) 25. Pollutant mass transformed in middle zone; (PTRANM;ug) 26. Pollutant mass in moisture of middle zone; (PMOIN;ug) 27. Empty space Pollutant mass cation exchanged in middle zone; (PCECM;ug) 28. Pollutant mass hydrolyzed from moisture in middle zone; (PHYDNM;ug) 29. 30. Pollutant mass complexed in middle zone; (PCOMM;ug) 31. Pollutant mass volatilized from middle zone; (PVOLM;ug) Pollutant mass volatilized from lower zone; (PVOLL;ug) 32. Pollutant mass hydrolyzed from upper soil layer; (PHYDSU;ug) 33. Pollutant mass hydrolyzed from middle soil layer; (PHYDSM;ug) 34. Pollutant mass hydrolyzed from lower soil layer; (PHYDSL;ug) 35. Pollutant mass hydrolyzed (upper) from cation exchanged 36. pollutant; (PHYDCU;ug) 37. Pollutant mass hydrolyzed (middle) from cation exchanged pollutant; (PHYDCM;ug) Pollutant mass hydrolyzed (lower) from cation exchanged 38. pollutant; (PHYDCL;ug) Pollutant mass in soil air of upper layer; (PSAU;ug) 39. Pollutant mass in soil air of middle layer; (PSAM;ug) 40. Pollutant mass in soil air of lower layer; (PSAL;ug) 41. 42. Empty space 43 Empty space Empty space 44 45. Empty space

FC - Fortran code

# **AP** - applications

## APPENDIX AP

### APPLICATION SAMPLES

This section will contain abstracts and executive summaries of applications discussed in sections 2.0 and 3.0 of this documentation. Eg. see references Bonazountas et al (1981), Wagner & Bonazountas (1982) in section 3.0.

A typical data base and typical input/output model results are presented in the following pages for all levels 0, 1, 2 and 3.

```
NUN : 1
                         ###### LEVELD SESULL MODEL UPERATION ######
NUN-SITE SPECIFIC FATE ASSESTENT
        1
       0 041
                                                                                                600u0
V4004 SEJJIL-91:
                         SEASUNAL CYCLES OF WATER. SEDIMENT. AND POLLUTANTS IN SOIL ENVIRONMENTS
                                                                                                *****
                                                                                                00000
        00700
       000033 JEVELOPERS: M. BUHAZJUNTAS.ARTHUR D. LITTLE INC. .(617)864-5770.X5871
000000 J. #AGNER .APTHUR D. LITTLE INC. .(617)864-5770.X2585
                                                                                                *****
                                                                                                ****
                                                                                                ****
        63033
£4843
        00:00 VLRJION:
                         DECEMBER 1901
                                                                                                61000
        04139
                                                                                                00000
        14443
        1-1
11
                         INU_X
                                   CLINTON.
            REGION
                      : (
                            1)
                                              MA. (SUB-HUMID):
            SULL TYPE : (
CUMPOUND : (
ŀ-:
                            5)
                                   CLAY-LOAA GENERIC
                                   TEST CHEN
                           15)
APPL.APLA : (
                                   CLENTON.
   AP-
                                              MA. (503-HUMID):
                            1)
   Ň
            SIMULATION INPUT PARAMETERS
            •
1.1.1.1
         -- APPLICATION PARAMLTERS --
    A +LA(SU.LH.): 1.0
1
    DEPTH TJ GR#(4): +10E+03
هد
بر
    UPPER SUIL ZIFE DEPTHICH): 10.
    PH UPPER ZONE(-): 7.0
-"İ
1.1
    PH RATIO LUNDRIUPPEN ZCHE (-): 1.0
TEMPERATURE (JES. C): 0.4
...
    JEGRADATION RATIO LOW RIUPPLR ZUNE(-): .50
    DIGANIC CARUDA CONTENT RATIO LUNER: UPPER ZONE(-): .50
44
    CLAY CUNTENT RATIO LOWER: UPPER ZONE( -): .50
<u>^</u>
    POLLUTANT HUNDEF MULTIPLICE: 1.0
    LLACHING (INPUT) RATE (FRAC. UF SOLUBILITY): .10E-02
CATIO: ECCHANGE CAPACITY RATIO LDMLR:UPPER ZONE(-): .0
4
...
    PULLUTANT 1 PUT MASS-UPPER ZONE(UG/S).CM): .202+03
    PULLUTANT I PUT MASS-LUWER LUNC(UU/SJ.C4): .10E+04
LIGAND INPUT MASS-UPPEN ZJNE(UG/SJ.C4): .10E+04
51
    LIGAND LIPUT MASS-LUVER ZUNL(UG/SU.C4): .201+04
اړ,
```

1

```
-
 51
 6
 ŀ
              -- HYDROLOGIC PARAMETERS --
 5
       SUIL MOISTURE(%): 26.
       INFILTRATION(CH): 01.
 ¦.,
       GRA RECHARGE(CM): 20.
 50
       SURFACE RUNUFF(C4): J4.
 11
 3
               -- CHEMICAL PARAMETERS --
 14
       SULUSILITY(US/ML): .11E+0+
AUSURP: CULF-(NUC): .36E+33
11.1
       UIF. CUEF. IN AIR(SU.CM/SEC): .83E-05
 !....
       DLG RADATION RATC(/DAY): .20
HENRYS CON.(CJ.M-AIM/MOLE): .94E-02
        ALSJAM. CJEF. UN SUIL(K): .0
 ₽₽
       MUL_CULA( AT.(J/MUL): 5.0
VALENCE(-): 2.0
       NIUTRAL IYDAJLYSIS CONSTANT (/DAY): .708-35
JASE HYDROLYSIS CONSTANT(L/MUL-DAY): 8.0
 ய்
       ACID HYD (JLYSIS CJNJIANT(L/MUL-DAY): ).0
LIJANJ-PJLLUTANT STAUILIIY CUNST.(-): .10E+00
NU. MULEJ LIJANJ/MJLE PULLUTANT(-): 2.0
 LIJAND YJULUJUAR WEIGHT (G/MULE): 13.
 ","
 1:
               -- JJIL PARAMETERS --
 11
        SULE DENSITY(S/CU.CM): 1.3
       DISCOUNSECTEDNESS INDEX(-): 10.
 11
 1341
 irl
        PU-J3117(-): .J5
        DRGANIC CARDON CONTENT(4): 1.3
 e i
        CLAY LANJUN CUNTENT(X): .20E-01
       CATION CALLANGE CLL'. (MILLI EJ./100G DRY SOIL): .0
INT-INJL PERALAJILITY-UPPER ZUNE(SJ.CM) .0
INT-INJL PERALAJILITY-MUDDLE ZUNE(SJ.CM) .0
INT-RINJL PERALAJILITY-LJ-LR ZUNE(SJ.CM) .0
 -11
 14
  ...
```

----

.

SINULATION DUTPUT **************** -- PULLUTANT INPUT TO SOIL COLUMN --PULLUTANT INPUT #/ RAIN 56.8 UG DIRECT PULLUTANT INPUT:U 200. ŪĞ DIRECT PULLUTANT INPUT:L . 100E+04 UG TUTAL PULLUTANT INPUT .1276+34 UG -- POLLUTANT FATE ASSESSMENT IN SOIL COLUMN--UPPER SULL ZONE DISTRIBUTION (MASS): X OF TOTAL INPUT VUL ATILIZED £ U 0.11 % 1.44 IN JJIL MJISTURE 2.33 US 3.1d X ADSURIED ON SUIL IMMUNICIZED-CATION EA. 50.4 υj 4.35 % • 0 ŪĴ 0.0 % J.00 % CJAPLEKED ·137E-JJ UG IN SUIT AIR IN SUIFACL RUNJEF HYDRULYZED-FRUM MOI. . 344 UG 0.03 % 1.07 % 20.5 U J .7312-32 UG 3.00 % HYDRULY L-D-FROM SUIL .175 UG 3.01 % HIJRJEYZED-FRJM CEC .0 UG 0.3 % 170. DECRADED UG 13.20 % LUACE SOIL ZUNE DISTRIBUTION (MAJJ): 14 SOLE MUISTURE Augurged on Sule Immujilized-Cation ex. 20.1 υů 1.02 X 251. UG 12.47 % • J ĴĜ 0.0 % .127: -07 UG CU APLE YEU 0.00 ~ UG IN SULL ALK 5.24 X 3.01 HYDAJLYZED-FRUM MUI. HYDRJLYZED-FRUM 501L • 202E-JI υS 0.01 % .110 UG U.JU X HYDRJLYZED-FROM CEC • ) υū ን-ሀ አ DEGHAJLU 701. 05 57.13 % LI CUTANCE CLA UG 0.0 X • J

11

11

3

Ą

÷

			•	
		GARCH SOLL ZONE DISTRIBU	TIUN (CONC	ENTHATIUNS):
		IN SOLL MUISTURE Adsumbed un soll In soll aim	.0033 2.820 .2463	UG/ML UG/G UG/ML
		LUWEN SOIL ZONE DISTAIDU	TION (CONC	ENTRATIONS):
		IN SULL MUISTURE Ausured un sull In sull air	.81276-02 .19026-01 .3J222-02	UG/ML UG/G UJ/ML
		MAX. DEPTH OF POLLUTANT	+H]•	Сч
.	***	****	******	##EXECUTION COMPLETED###################################
	AP-5			
;	•			
-				
1				
1				
1				

. , . . RUN : 2 AVOGGO LEVELI ŜISJIL MOUEL OPERATION 400000 10 SITE SPECIFIC ANNUAL FATE ASSESMENT **** ., 00007 SCASUNAL CYCLES OF WATER. SEDIMENT. AND POLLUTANTS IN SULL ENVIRONMENTS ***** COOMA SCOULTSI: **** 20001 M. HONAZUUNTAS.ANTHUN D. LITTLE INC. .(017)864-5770.X5871 J. WAGNER .ARTHUN D. LITTLE INC. .(017)864-5770.X2585 60101 13 00000 DEVELOPERS: 46043 00000 **** ..... 15 04030 VERSION: DECEMBER 1741 00000 .7 66044 20070 **** • 000.00 1 . --| INUCK REGION CLINTON. MA. (SUB-HUMID): : - ( 1) SUIL TYPE : C ų, 5) CLAY-LOAM GENERIC CJYPOUND : ( TEST CHEM 3 15) ₽ CLI ITUN. MA. (SUB-HUNID): APPL. AREA: ( 1) •• 1 SINULATION INFUT PARAMETERS 4 12 .... 1.1 -- APPLICATION PAHAMLTERS --13 14 15 AHLA(SU.14.): 1.0 DEPTH TJ GR#(4): .10E+03 41 UPPER SULL ZUIE DEPTHICMI: 15. PH UPPER ZUNE(-): 7.0 PH HATIU LUALA: UPPER ZONE(-): 1.0 14 DEGRAJATION RATIO LORER: UPPER ZUNE(-): .50 UNSANIC CARUDA CUNTENT RAILO LUGERIUDPEN ZONE(-): .50 CLAY CUNTENT RATIO LUMERIUPPER ZONE(-): .50 CATION EXCHANCE CAPACITY RATIO LUMEN: UPPER ZONE(-): .0 .1 1 3 POLLUTANT LAPUT 4435-UPPLR ZUNL(UG/SJ. CM): .202403 POLLUTANT INPUT MASS-LUALR ZUNL(UG/SJ. C4): . 10E+04 LIJANJ I IPUT MASS-UPPER ZUNE (UG/SJ. CM): . LOE+04 LIGAND INPUT MASS-LDALH ZJNE(UG/SU. C4).20E+04 ų,

-

. . . . .

٠,

0

- - - - - - -

UPPER SUIL ZUNE DISTRIBUTION (CONCENTRATIONS):

- ----

IN SOLL MUISTURE	•6039	UG/ML
APPORTED ON POIL	2.020	UG/G
IN JJIL AIR	-2-59	UJZML

- -

### LOWER SOIL ZONE DISTRIBUTION (CUNCENTRATIONS):

.

IN SULL MUISTURE	SC-29516.	UG/ML
AUSORALD JH SJIL	·1/025-J1	US/G
IN SJIL AIR	• 7 2 5 7 5 - 2 5	UG/ML

MAX. DEPTH OF POLLUTANT 453. CM

AP-7

#### 

```
LATITUDE(JEC. DEG): 43.

TEHPERATURE(JEG C): 8.4

CLOUD CUVER(FRAC.): .35

REL. HURIDITY(FRAC.): .30

SHURTAAVL ALHEDJ(-): .30

EVAPJIRANSPIRATIDI(CM/LAY): .0

ANHUAL PRECIPITATUI(CM): V4.

MEANTIM_ UF RATE(JAY2): .32
```

-- CLIMATIC PARAMLTERS --

MEAN TIME OF RAIN(JAYS): .32 Alan Hundle of Sturm Events(-): .11E+03 Mean Length of Rainf Seasin(days): .37E+03 .

151

1:

1:

1.1

1.

i

#### -- CHEMICAL PARAMETERS --

SUL JULL ITY (UG/AL): .11E+0. AJSUAP. CUEF.(KUC): .36E+03 DIF. CJEF. I AIK(SO.CM./JEC): .83E-05 ULJ (AUATIDA RAT(/UAY): .20 MEX.YS CJA.(CJ.M-AT4/MOL): .94E-02 AJJ (A. CJLF. UN SIL(K): .0 MOLLCULA( AF.(G/AUL): .0 MOLLCUL

-- JUIL PARAMETERS --

SDIL DENJITY(J/CJ.C4): 1.3 141. PERALAJLITY(J/CJ.C4): 1.3 DISCONNECTNESS INDEX(-): 100 PÚNJÍTY(-): 1J3 UNGANIC CANDY, CUNTENT(X): 1.3 CLAT CANDUN CJNT, IT(X): .2JE-01 CATION EXCHANGE CJEF. (MILLI ÉG./1005 DRY SOIL): .0 INT (INSIC PERALAHILITY-DPER ZUNÉ(SUNCH).0 INT (INSIC PERALAHILITY-DPER ZUNÉ(SUNCH).0 INT (INSIC PERALAHILITY-DER ZUNÉ(SUNCH).0 INT (INSIC PERALAHILITY-DER ZUNÉ(SUNCH).0

INGTUC ALIJA JUMI C		
ANNUAL HYDRCLOG	IC CYCLE P	ARAMETERS -
SOIL MOISTURE CONTENT(THA) PRECIPATION(PA) NET IFIETRATION(IA) EVAPJERANSPIRATION(ETA) SURFACE RUNDEF(RSA) VILLO (YA)	25.7 94.5 00.7 34.5 35.2 20.2	× × × × × × × × × × × × × ×
RATIU PAZAPA (-)	1.00	
POLLUTANT INPUT	'TO 301∟ C	DLUMN
POLLUTANT INPUT #7 RAIN Direct pjllutant input: Direct pjllutant input:	66•7 U 200•	UG UJ 4 UG

Į١.

UPPER SULL ZUNE DISTRIBUTION (MASS): & UF TUTAL INPUT VULATILIZED IN SUIL MUISTURE U, 0.11 % 1.45 2.33 ŪĞ 0.10 X ADSURBED UN SUIL IMAUSILIZED-CATION EX. 4.J5 X 56.0 UG UĞ 0.0 % • • CUIPLEXED .157C-)3 UG J.OU X IN JUL AIR .345 υG 0.63 X 14 SJ (FACL RUNUFF 20.5 UG 1.39 % HIJ HALIZED- FRJM VOI. .7396-02 UG 0.00 % HYU YJLYLED-FRUM SUIL .173 UĠ 0.01 % HIU IJLY LO-FRJA CEC • 0 υj J.U X 170. DEN RAJED 13.20 % Ju LOWER SUIL ZUNE DISTRIBUTION (MASS): IN SULE MULSTURE Ausurged un sule Imnustelized-cation ex. 20.7 UG 1.62 % 231 . ŨĞ 19.48 X 0.0 % • • Uŭ CUAPLE KLD IN SULL AL C .1276-37 UG 0.00 % 3.09 ŪG 0.24 % HYJRILYZED-FAUM MOI. .602c-31 UJ 0.01 % HYDRILYZEU-FRJM SOIL •770 UG 0.36 % HTURILTLED-FRUM CEC • 0 υĠ 0.0 % 52.13 X うとしょううしつ 701. υj TU GRUUNDAATER ŬĞ J.U % • •

i

ł

-- PULLUTANT FATE ASSESSMENT IN SOIL COLUMN--

ľ

1.12.8.7.8

1:

h.

<u>،</u> ،

۰,

; -

11

,

.

.....

| ' |-

```
RUN : J
                     ANCANA LEVELS SESDIE MODEL OPERATION AARAAA
                          PUNTHLY SITE SPECIFIC SIMULATION
     20031
                                                                                   01000
      00031 SENJIL-31:
                     SLASUNAL CYCLES OF HATER. SEDIMENT. AND POLLUTANTS IN SOIL ENVIRONMENTS
                                                                                   61363
      44042
                                                                                   ****
      4++++ 0EVELOPERS:
                     M. BUNAZUUNTAS.ARTHUR D. LITTLE INC. . (017)864-5770.85871
                                                                                   0.000
      66963
                     J. #AGNER
                                 • ARTHUR D. LITTLE INC. • (017)864-5770• X2585
                                                                                   41443
      0303-
                                                                                   46603
      ALCON VERSION:
                     DECEMBER 1931
                                                                                   01031
      64663
                                                                                   010:3
      04012
                                                                                   0000r
      1 ND こ X
          REGION
                  : (
                        4)
                             CLINTON+MASS+(40 YR AVERAGED DATA FOR TEST UNLY)
          SULL TYPE : (
COMPOUND : (
                       5)
                             CLAY-LUAM GENERIC
                       15)
                              TEST CHEN
AP-
          APPL. AREA: (
                        د ب
                              TEST RUN (CLINTON. MA. 40 YEARS)
÷
          GENERAL INPUT PARAMETERS
          -----
      -- APPLICATION PARAMETERS --
  AREA(50.04.): 1.0
  DEPTH TJ JR4(4): +10E+03
  UPPER ZOIL DEPTH(CM): 15.
  PH UPPER ZUHL(-): 0.0
  PH HATIO LUNCHIUPPER ZONL(-): .88
  DEGRADATION RAFID EDRER: UPPER ZONE(-): 1.1
  JAGANIC LARD) + CONTENT RATIO LDARROUPPER ZONE(-): 1.2
CLAY CONTINUESTING ATTO LOACHODER ZONE(-): 1.3
  LATI 14 _ (CHANGE CAPACITY 4ATIO LOWER: UPPER ZONE(-): .10E-02
  TREUNDELLA EXPUNENT(-): 1.4
```

. 2

į,

1.

1.

110

. .

ι,

1,

IC IS

1.54

ŀ

1

1 1

1 -

#### -- CHEMICAL PARAMETERS --

. 1

1

2

٦,

1

.

<u>`</u>1

4

3

:

.'

SOLUBIL [TY(UG/4L): .11E40+ AJGOAN- CJEF.(KOC): .360+0J DIF. CG2+ .14 AI*(GA:C*+/SEC): .3JE-J5 DLGMAJATION WAT_(/DAY): .20 MENNYS CJ.4.(CJ.44-4TM/MOLC): .94E=02 AJGJAN- CJEF. U+ SJIL(A): .0 MJL_CULAY #1.(S/ MDL): .0 MJL_CULAY #1.(S/ MDL): .0 MJL_CULAY #1.(S/ MDL): .0 MJL_CULAY #1.(S/ MDL): .0 MJL-CULAY #1.(S/ MDL): .0 MJL-CULAY #1.(S/ MDL): .0 MJL-CULAY #1.(S/ MDL): .0 ACLO HYDRJLYJIS CONSTANT(L/MDL-DAY): .10E+06 HJGANJ MULCULAR #LIGHT(G/MDL): IJ.

-- UJIC PARANCTERS --

SUL JEN, ITY(J/CJ.CH): 1-3 141. PE44EAULITY(SJ.CM.): .28E-JV UISCOMMULTEUNES INDEX(-): 10. PJ4JSITY(-): .JD J4GANIC CAAJJ+ CONTENT(X): 1-J CLAY CAAJJ+ CONTENT(X): 1-J IATALAJL PEALABLLITY-UPELA 2012(SJ-CA) -J INT (1-SIL PLALABLLITY-UPELA 2012(SJ-CA) -J INT (1-SIL PLALABLLITY-UPELA 2012(SJ-CA) -J

1

1

ì٠

1

1.

*************															
		YL A	1 MON 1	HLY INPUT	PARAMETE	K5									
							·								
	UCT	HOV	DEC	JAN	FEU	MAR	APR	MAY	JUN	JUL	AUG	SEP			
CLIMATIC PAYAMETERS															
LATIIUUE(JEG) 42.5 42.5 42.5 42.5 42.5 42.5 42.5 42.5															
CLUUD L / R(FRAC.) REL. HUH. (FRAC.)	• J20 • 700	• 320 • 700	· 320 • 7 0 3	. J20 . 700	· 320 • 700	.320	.320	• 320 • 700	• 320 • 700	• 320 • 700	• 329 • 700	.329			
ALSLUJ(-) EVAPUT.(EM/DAY):	. J 00 . V	.300 .0	• 300 • 0	. 300	- 300	.300	.300	.300	.303	. 300	.300	.355			
PR:CIP.(CA):	7.94	7.54	7.54	7.84	7.04	7.34	7.84	7.34	7.84	7.04	7.64	7.44			
M. JURA (U.(-): M. JURA (U.(-):	9.0d	9.00 10.00	9.03 10.4	- 520 J - 08 30 - 4	9.00 V.0	• 320 9•98	• 52 0 9 • 0 9	9.00	• 32 J 9• 08	- 320 2-08	• 320 9•08 30-6	• • • • • • • • • • • • • • • • • • •			
		5004	5014	50.4	5014	50.4	5014		3314	5014	50.4	50.4			
RUN DATA-5	L[ ]														
NUIS. CUIC-UP.(UG/	ML) .0	• 0	•0	• •	•0	• 0	•0	• 0	•0	• 0	• 0	• 2			
POL - INP-U(U3/53 C POL - INP-U(U3/53 C	M) 200		•0		•0	•0	•0	•••	••	.0	.0	•0			
SUN . PUL. MULTIPLI	ER 1.00	1.00	1.00	1.00	1.00	• • •	.0	.0	.0	. 0	.0	.0			
KUN DATA-J	er 2														
POL . IN PALATE RAC-	3L) .100	1-02 .0	• •	• 0	• 0	•0	•0	. •Ŭ	•0	• •	• 0	•0			
「パッション・ション・コート(リッノン・コート)	CM) .200	) •U	• • •	••	•0	•0 •9	•0	•0	•0		• • • • • • • •	••			
5[1k5-L(JU/J) CM)	• 5 J (	.0	•0	:0	•0	•0	•0	•0	•0	. 0	•0	•0			
LIG. 14201-0(06/37		)c+04 .0	• 9	• 0	• •	•0	• 0	•0	• 0	• 0	• 2	.0			
LIG. INPUT-LUG/51	. C 4) . 200	E+04 .0	•0	• •	• 0	.0	•0	• 0	.0	.0	.0	•0			

		YEAR-	יינא ו בב בב	THLY RESU		JT)						·		
HYDROLUGIC CYCLE COMPONENTS														
	061	NOV	DEC	JAN	FLB	MAR	APR	MAY	JUN	JUL	AUG	SEP		
SULL MUISTURE( %) PRECIPATIUN(CM) NET INFILTR.(CM) EVASDIRAISP.(CA) SURFACE RUISF(CM GRA RUISF(CM) YIELD (CA)	25.8 7.50 5.02 2.72 2.34 2.30 5.14	25.8 1.60 2.02 2.02 2.04 2.30 2.30 5.14	25.8 7.86 5.02 2.72 2.84 2.30 5.14	25.5 7.80 5.02 2.72 2.84 2.30 5.14	25.8 7.60 5.02 2.72 2.54 2.50 2.50 5.14	25.8 7.46 5.02 2.72 2.54 2.50 5.14	25.8 7.96 5.02 2.72 2.84 2.30 5.14	25.8 7.86 2.02 2.72 2.34 2.30 5.14	25.0 7.00 5.02 2.72 2.54 2.33 5.14	25 • 8 7 • 80 5 • 02 2 • 72 2 • 34 2 • 30 5 • 14	25.8 7.56 5.02 2.72 2.84 2.30 5.14	25.8 7.96 5.92 2.72 2.34 2.30 5.14		
RATIO PA/MPA(G2)	1.00	1.30	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00		
POLLUTANT	MASS INPU	, ד דט כם	LUMN(UG) ·	•••·										
	0CT	NOV	DEC	JAN	FEB	MAR	APR	MAY	JUN	JUL	AUG	SEP		
P (ÉC[PATION UTHER(UPPER) UTHER(LUJÉR)	5.52 200. .10(E+04	•0 •0 •0	• 0 • J • J	• 0 • 0	•0 •0 •0	• 0	• • • •	•0 •0	•0 •0	•0 •0 •0	•0	• C • O • O		
FJTHL LIPUT	•121E+04	• 0	•0	• 0	• 0	• 0	• 0	• 0	• 0	- 0	•0	.0		

----

	PULLUTANT	MASS DIST	NUTION I	N COLUMN	(UG)								
	UPPER SULL ZONE	:											
	SURFACE RUNCFF VJEATIEIZED UTMER SINKS ADS. UN SUIE IMMUUIEIDECE US RADED MYUNCEYZEDEMDI	3.58 .230 .400 101. .0 29.3 .2212-32	4.70 .310 .0 118. .0 J9.5 .291E-02	3.14 .207 .0 8.4.7 .0 25.7 .1.4E-02	2.18 .144 .0 09.7 .3 17.9 .135E-02	1.50 .103 .0 55.4 .0 12.8 .965E=03	.0 .771E-01 .0 45.7 .0 9.57 .722E-03	.0 .595E-01 .U J8.2 .0 7.39 .530E-03	.0 .457E-01 .0 J2.3 .0 .0 .0 .0 .435E-J3	•0 •372E-01 •0 27.5 •0 ••62 •349E-03	•0 • 30 1E -01 •0 23• 7 •0 3• 73 •25 2E -03	•0 •245E-01 •0 20•0 •0 3•05 •230E-03	•0 •202E-01 •3 10.C •0 2.51 •1092-03
	HYJROLY223-531 HYDROLY223-626 CJAPLEY25 JTHLR TKANS. IN JOLE AIST. IN SOLE AIST.	- 47JE-U1 - U - 143E-U3 - 200 8 - 30 1 - 21	• 791 • 0 • 5555 • 0 • • 45 • 791	- 4512-01 -0 - 3052-03 -0 3-70 -537	.347:-01 .0 .345E-03 .0 2.60 .370	.2732-01 .3 .3072-03 .3 1.84 .274	.222E-01 .0 .337E-03 .0 1.44 .209	-105E-01 -U -194E-03 -U 1-12 -103	- 1 302 - 31 - 387L - 03 - 0 - 884 - 128	-132E-01 -0 -355E-03 -0 -703 -103	• 1142-01 • C • J & 72-03 • U • 575 • 8 J 4 E - 01	•952E-02 •0 •357E-03 •9 •471 •633E-01	.8552-02 .0 .3372-03 .0 .337 .5552-01
	LUMER JUIL ZONE	:											
	INTU GRWAFER VULATILIZED UTMER SINKS	• 0 • 0	• 0 • 0	• 0	• 9 • 0	• 0 • 0	• 0	• U • U	• Ú • O	• 0	•0	• 0 • 0	• C • O
,	AUS. JN JUIL 14NJUL120-CEC 0208AJ50 HY08ULY20-MJI	972. .0 21.8 .8025-C3	936. •0 33.2 •151£=32	-0 -0 -36-2 -14-5E-02	908. •U 34.4 •136E-U2	830. 32.0 .127E-02	605. .0 J1.0 .122E-02	770. .0 29.4 .110E-02	749. .0 27.9 .110E-02	722. •0 20.5 •105E-02	697. .0 25.3 .100E-02	673. •0 24.1 •652E-03	051. 22.9 .907E-03
	HYD ()LY220-CCC HYD ()LY220-CCC CJ (PLC72) UI (27 144 45. I ( 501L MJ15T.	• 181 • 181 • 300 5 • 98	•2•5 •0 •130 •5 5•65	-0 -223 -0 5-33	-205 -205 -0 5-11	• 202 • 0 • 6 4 • 84	• 0 • 215 • 0 • 0 • 0	• 2 4 3 • 0 • 0 • 0 4 • 3 7	•196 •0 •196 •0 •0 4•15	•191 •3 •211 •0 3•95	• 18 • • 0 • 1 4 9 • 0 J. 7 0	• 175 • 0 • 227 • 6 3• 55	•172 •0 •188 •0 J•41
	IN SCIL AIR	• 468	. 32 4	• 761	.741	./03	• 668	• ( ) 4	.03	• 5 7 3	• 14 0	•520	• * *3
	POLLUTANT	CJHCENTRA	110N5 (UG)	ME) OR (	16/6)								
	M015TURE-UPPER 5016-UPPER A15-UPPER FREE E1644D-UPPER	2 • 16 8 • 11 • 85 5 2 5 7 •	1 • 4 1 5 • 7 7 • 575 2 5d •	•955 4.53 •370 256.	・671 3・52 ・275 253・	•486 2.80 •199 208•	• 372 2• 31 • 152 258•	•289 1•43 •118 258•	•228 1•63 •9332-01 258•	•183 1•39 •747E-01 258•	•148 1•26 •6065 =01 258 •	.121 1.34 .497E-01 208.	•100 •607 •4112-01 259•
	MƏLİTURË-LOMER Səle-Ləmlə AIR-Ləmlə Free LIVAND-LUMER	.232E-02 .73/E-01 .948E-03 .7/5	•220E-02 •710E-01 •402E-03 •775	•2095-02 •0542-01 •3532-03 •77.0	・1ッ8さー02 ・05 3さー01 ・HU9ミー03 ・775	.1002-02 .6345-01 .7602-03 .775	-179E-02 -611E-01 -729E-03 -775	•169L-02 •589E-01 •693E-03 •775	•151E-02 •568E-01 •653E-03 •775	•153E-02 •544E-01 •626E-03 •7/5	•1468-02 •5298-01 •5968-03 •775	-137E-02 -511E-01 -568E-03 -775	.132E-02 .494E-01 .541E-03 .775
	MAX. PUL.DEPTH(CM)	45.9	15.9	1.08.	139.	170.	201.	232.	263.	294.	324.	355.	385.

----

TJTAL INPUTS (UG) JPD-2 SJL 2018 LDALA SUL ZONE AYDAJLUGIC CYCLE CUMP	235. .1J0E+04 DNENTS	
TJTAL INPUTS (UG) JPPLR SUIL ZUHE LOMEA SUIL ZUHE HYDHJLUGIC CYCLE CUMP	235. .1305+04 DNENTS	1
UPPER SUIL ZURE LUMEN SUIL ZURE AYURDLUGIC CYCLE CUMP	235. .130E+04 DNENTS	i
AYUAJLUGIC CYCLE CUMP	DNENTS	
TUTAL PALLIPATICA(CA) TUTAL INFILTAATION (CA) TUTAL INFILTAATION (CA) TUTAL SURFACE RUNDFF(CA) TUTAL SURFACE RUNDFF(CA) TUTAL SURFACE RUNDFF(CA) TUTAL YILLO (CA)	25.0 74.3 00.2 32.6 34.1 27.4 01.7	
PULLUTANT MASS DISTRI UPPER SOIL ZONE:	BUTION (UG)	    }
TJIAL SURFACE RUNDEF TJIAL VILATILIZCO TJIAL VILATILIZCO FIAL AJS. UN SIL FIAL AJS. UN SIL FIAL AJS. UN SIL FIAL AJS. UN SIL TJIAL HYJRJEYZLJ-MJI TJIAL HYJRJEYZLJ-MJI TJIAL HYJRJEYZLJ-UI TJIAL CHUSALYZLJ-CEC TJIAL CHUSALYZLJ-CEC TJIAL CHUSALYZLJ- FIAL IN SJIL MJIST. FINAL IN SJIL AIR	15.2 1.30 .433 13.0 .6 151. .121z - 31 .314 .3 .437E - 02 .233 .339 .339	
LUNER SUIL ZONE: Tutal Intu Stantla	• 0	
TUTAL UTALALIZAD TUTAL UTALALIZAD FILAL ADJ. UH JULL FILAL LADJ. UH JULL FILAL LADJ. ULZZD-CEC TJTAL MAJRJLAZD-JUL TJTAL MAJRJLAZD-JUL TJTAL COMPLEXCO TJTAL UTMER TAANS.		
	PULLUTANT MASS DISTRICULUPLE SOIL ZONE: TJIAL SURFACE RUNOFF TJIAL VILATILIZED TJIAL UTA JIAJILIZED FI AL ADJ. UN SJIL FI AL ADJ. UN SJIL TJIAL HYDRUTZED-CEC TJIAL UN SULYZED-UI TJIAL ADJ. UN ST. FIAL IN SJIL ATA LUAZA SUIL ZGNC: TUTAL UTALATALIZED TJIAL VILATALIZED TJIAL VILATALIZED TJIAL UN SULZED-CEC TJIAL UN SULZED-CEC TJIAL UN SULZED-SUI TJIAL MISHIZZED-SUI TJIAL MISHIZZED-SUI TJIAL MISHIZZED-SUI TJIAL MISHIZZED-SUI TJIAL MISHIZZED-SUI TJIAL UTALATALIZED TJIAL MISHIZZED-SUI TJIAL UTALATALIZED TJIAL TALATALIZED TJIAL	POLLUTANT MASS DISTRIBUTION (UG) UPPER SOIL ZONE: TJIAL JARA JARA JARA JARA JARA JARA JARA JA

-- 1VER NUISTURE-UPPER SUIL-UPPER AIR-UPPER FRLE LIGAND-MUISTURE-LOW SUIL-LURER AIR-LURER AIR-LURER MAX. PUL. DE APP 1

-- AVERAGE POLLUTANT CONCENTRATIONS-(UG/ML)OR (UG/G) --

siines).

.

MULSTURE=UPPER SULL=UPPER AIR=UPPER FRL=11.0500=00000	+ 2 7 3 2 + 7 4 + 2 + 3 2 1 1 -
MUTSTURE-LUNER	-1772-0
SUIL-LUMER	•0J0E=0
AIRTLUALR	•724E-J
FREL LIGAND-LOWER	.715
MAX. PUL. DEPTH(M)	3.30

		YLAR-	2 MO:41	(HLY [NPU]	PARAMETI	.KS							
	0C 1	NOV	DEC	JAN	FE3	MAR	PGA	MAY	PUL	JUL	AUG	SEP	
CLIMATIC P	ARAMETER S												
LATITUJE(UĽG) TEMP+(DEG C) CLUUD CV?(FAZC) ALL- 104+(FRAZ) ALD-JJ(-) LVAPUF+(CM/DAY):	42.5 8.40 .320 .700 .300 .3	42.5 8.40 .320 .700 .300 .0	42.5 3.40 .323 .730 .303 .0	42.5 5.40 .320 .700 .300 .0	42.5 8.40 .320 .700 .300 .0	42.5 8.40 .320 .700 .300 .0	42.5 8.40 .320 .703 .309 .0	42.5 8.40 .320 .700 .300 .300	42.5 8.49 .J20 .709 .J0J .J0J	42.5 8.40 .320 .700 .300 .0	42.5 3.40 .320 .700 .300 .0	42.5 8.40 .320 .700 .300 .0	
PRECIP.(C4): M.TIR. RAIN(DAYS): M. STORINU.(-): M. SEASUN(DAYS):	7.34 .J20 7.U8 .J8	7.84 .320 7.08 Ju.4	7.84 .32J 7.03 3J.4	7.84 .320 9.03 30.4	7。34 。 うごり ダ - ひき ろ J - 4	7.04 .320 5.08 30.4	7.84 .320 3.08 30.4	7.84 .320 9.08 30.4	7.04 .320 9.03 30.4	7.84 .320 9.08 JU.4	7.34 .320 9.08 30.4	7.34 -329 9.08 30.4	
•													
RJN DATA	ET 1												
MUIS. CUNC-UP.(UG/ Muis. CUNC-UP.(UG/ Put. INP-U(UU/Su C Put. IMP-L(UG/SU C SUM. PUT. MULTIPLI	41) -0 41) -0 41 -0 41 -0 41 -0	- 0 - 0 - 0 - 0 - 0	- 0 - 0 - 0 - 0 - 0	.0 .0 .0 .0	0 • 0 • 0 • 0	• 0 • 0 • 0	.0 .0 .0 .0	.0 .0 .0 .0	- 0 - 0 - 0 - 0	- 0 - 0 - 0 - 0 - 0	• 0 • 0 • 0	•0 •0 •0	
-+ RUN DATA-5	ET 2												
PUL - IN イムIN(FRAC=) IR43FUイイビン・U(JG/JJ IA45FUイイビン・U(JG/JJ IA45FUイイビン・L(JG/SJ JI455FU(JG/SJ C4) SI455FU(JG/SJ C4)	SL) .J CM) .J CM) .D .J	• 0 • 0 • 0 • 0 • 0	. U . O . U . O	• 0 • 0 • 0 • 0 • 0	•0 •0 •0 •0	• 0 • 0 • 0 • 0	• 0 • 0 • 0 • 0	• 0 • 0 • 0 • 0	• 0 • 0 • 0 • 0	• 0 • 0 • 0 • 0	• • • • • • • • •	•0 •0 •0 •0	
14. 1907-0(06/30 14. 1907-1(06/30	.См) .0 .см) .0	• Ŭ • U	• 0 • 0	• Ŭ • Ŭ	• U • O	• 0 • 0	• 0 • 0	• 0	•0	• 0 • 0	• 0 • 0	• • •	

3 4 3 4			¥£&R-	2 MU 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	IHLY RESU	JL 15(001P(	JT ) ===						
1 1 1 1	HYDROLOGIC	CYCLE (	LUMPONENTS	,									
		001	NUV	∂ć C	JAN	FLB	MAR	APR	MAY	NUL	JUL	AUG	SEP
	SOLL MOLSTURE( %) PRECIPATION(CM) NLT INFILTRA(CM) EMAROTRANSPA(CA) SJMFACL RUNDFF(CA) GRA HUNJFF(CA) YLLD (CA)	25.8 7.50 5.02 2.72 2.84 2.30 5.14	25.8 7.86 9.02 2.72 2.34 2.33 3.14	25.5 7.50 5.02 2.72 2.64 2.50 5.14	25-8 7-80 5-02 2-72 2-84 2-30 5-14	25.8 1.86 5.02 2.72 2.84 2.30 5.14	25.8 7.36 5.)2 2.72 2.84 2.30 5.14	25.8 7.65 5.02 2.72 2.54 2.30 5.14	25 • 8 7 • 30 5 • 02 2 • 72 2 • 3 • 2 • 3 • 5 • 1 4	25.8 7.65 5.02 2.72 2.54 2.30 5.14	25.8 7.86 5.02 2.72 2.44 7.30 5.14	25.8 7.86 5.02 2.72 2.54 2.54 2.54 2.54 2.54	25.8 75.02 2.72 2.32 2.30 5.14
AP-19	RATIJ PAZAPA(JZ)	1.00	1.00	1.03	1.03	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00
31 37 31	PJLLUTANT	MASS IN	PUT TU COL	LU4N(UG)	··								
		uct	NUV	DEC	JAN	FEB	MAR	APR	MAY	JUN	JUL	AUG	SEP
	PRECIPATION Other(Upp_r) Uther(Lumer)	• 0 • 0 • 0	• D • C • O	•0 •0 •0	• 0 • 0 • 0	0.0 0.0	• 0 • 0 • 0	• • • •	•0	• 0 • 0	0 • 0 • 0 •	• 0 • 0 • 0	• 0 • C • 0
	TJTAL INPUT	- 0	• 0	• 0	• 0	• U	.0	. 0	•0	• G	•0	• 0	•0

¦¦ 1 30 -- POLLUTANT HASS DIST (IBUTION IN COLUMN (UG) --

	JPPER SOLL ZONES	:		•									
3 4 3 0 1 2 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3 4 4 3 4 4 4 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	SURFACE AUAUFF VJEATIEIZEU UTMERSIANS AUSEUN SILE 1MAUMIEIZU-CEC DEGRAUEU HYURSEYZEU-MOI HYURSEYZEU-MOI HYURSEYZEU-SUI HYURSEYZEU-SUI HYURSEYZEU-SUI DIMERTIKAISE 14 MUE AUISTE IN SOLE XIN	• 0 • 108=-01 • J • 5 • 8 • 0 • 1552-03 • 750E-02 • J • J • J • J • 5 • 6 • 0 • 108 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0	.0 .141E-J1 .c .0 .132E-J3 .002E-J2 .0 .350E-J3 .274 .373E-01	• 0 • 120E - 01 • 0 12 - 4 • 0 1 - 4 ) • 112E - 33 • 303E - 02 • 3 • 307E - 03 • 335E - 01	.0 .102E-01 .0 1.1.1 .0 1.27 .955E-04 .52502 .0 .3A7L-03 .0 .149 .149 .289L-01	•0 •d78E-02 •7 •0 •0 •0 •0 •0 •0 •0 •0 •0 •0 •0 •0 •0	.0 .753E-02 .0 .941 .713E-04 .42%E-02 .0 .397E-03 .0 .148 .215E-01	-U -OJBE-O2 -V B-IO -O -OI7 -OI62-O4 -JU42-O2 -JU7E-O3 -U -IU7E-O1	•0 •573E=02 •0 7-41 •537E=04 •537E=04 •542E=02 •0 •357E=03 •9 •113 •104E=01	• 0 • 5 0 3E - 0 2 • 0 • 7 5 • 0 • 6 2 5 • 4 7 1 E - 0 4 • 3 1 7 E - 0 2 • 3 6 7 E - 0 3 • 9 7 2 E - 3 1 • 1 4 4 E - C 1	•0 •4432-02 •0 •18 •550 •4152-04 •372-03 •372-03 •6 •8762-01	.0 .3v2F-02 .0 .436 .3v7E-04 .20JE-02 .174E-03 .9 .775E-01 .112E-01	• 0 • 34 dE - 02 • 21 • 3 • 4 3? • 4 3? • 32 dE - 34 • 24 3! - 02 • 3 3 7E - 03 • 6 3 7E - 01 • 6 3 7E - 01
1	COMER 5010 2006	:											
AD_00	1413 GRAATER VULATILIZED OTHLA SINKS AUSS GN SOLL IMADJILIZO-CEC DEGRADED HTD GLTZLO-MOI HTD GLTZLO-MOI HTD GLTZLO-SOL HTD GLTZLO-CC CIMPLEX (445. I.4 SOLL AIA	.0 .0 .0 21.9 .004E-03 .160 .0 .15 .0 J.25 .472	.0 .0 .0 20.6 .824E-J3 .151 .0 .130 .0 .131 .0 .130 .0	• 0 • 3 • 5 3 9 • 4 • 7 8 7 2 - 0 3 • 1 5 5 • 0 • 2 5 4 • 0 2 • 9 6 • 4 3 3	-0 -0 507- -0 -7512-03 -150 -3 -250 -2 -0 -2 -0 -2 -0 -2 -0 -2 -0 -2 -0 -2 -0 -2 -0 -2 -0 -0 -0 -0 -0 -0 -0 -0 -0 -0	.0 .0 551. .3 .18.2 .146 .0 .146 .0 .253 .3 2.70 .392	• 0 • 0 • 0 • 34 • • 0 • 17 • 4 • 0 • 0 • 0 = 0 3 • 14 1 • 0 • 1 3 • • 0 • 1 3 • • 0 • 1 5 • 0 • 1 5 • 0 • 1 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0	• 0 • 0 • 0 • 17• • 0 • 050E = 03 • 157 • 0 • 153 • 3 • 2•47 • 30 7	• 0 • 0 • 0 • 501 • • 0 • 15 • J • 02 dE - 03 • 13 2 • 0 • 17 6 • 0 • 0 • 2 • 3 7 • 34 4	.0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .2 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0	.6 .0 .471. .0 14.6 .576E-03 .124 .2 .2 .2 .2 .0 2.17 .315	.0 .0 .457. .0 14.0 .5522-03 .121 .0 .149 .0 2.08 .352	.0 .3 .6 .3 .3 .3 .5 30E-03 .117 .173 .0 .2 .00 .2 .00
25) 77  78	PULLUTANT	CUNCENTRA	11085 (US	ML) OR (	UG/G)								
	MOISTURE-UPPER SJIC-UPPER AIR-UPPER FREE EIGARS-UPPER	. 840E-01 . 7/8 . 343E-01 205.	.708€-01 .705 .207€-01 258.	・601E-01 ・623 ・245E-01 250・	.514E -01 .562 .210E -01 258.	.443E-01 .505 .151E-01 258.	.393E-01 .455 .157E-01 253.	• 333E - 01 • 412 • 130E - 01 258•	•291E-01 •374 •119E-01 298•	•256E-01 •341 •105E-01 258•	•226E-01 •312 •924E-02 253•	.200E-C1 .256 .818E-02 258.	•173E-01 •263 •728E-02 255•
5 F E F E	MJISTJRE-LUMER SJIL-LUM AIR-LUMEA FRLL LIGAND-LUMER	•1260-02 •4776-01 •9162-03 •775	•120E-02 •401E-01 •472E-03 •775	•113E-02 •4495-01 •4736-03 •773	•110E-02 •432E-01 •4432-03 •775	•105E-02 •415E-01 •429E-03 •775	.100E-02 .405E-01 .4101-03 .775	•959E=03 •392E=01 •392E=03 •775	•918E-03 •390C-01 •375E-03 •775	•879L-03 •369E-01 •359E-03 •77)	.842[03 .355f-01 .344E-03 .775	.8082-03 .3472-01 .3302-03 .77%	•775E-03 •337E-01 •317E-03 •775
2 2 2	MAA. POL.DEPTH(CM)	417.	448.	477.	510.	541.	572.	503.	634.	005-	645.	727.	758.

----

1			
	Y =	/EAR - 2 ANNUAL SUNMARY REPORT	J
ſ,`			i
19	FJTAL INPUTS (UG)		1
5	UPPER SULE ZONE	• 0	1
;	LJAER SOIL ZJHE	• 0	i
	WARE DECOMP		
6	AVERAGE JULL HUISTURE( %)	25.3	
121	TUTAL PRICIPATION(CM) TUTAL DREALTAATION (CM)		i
<b>1</b> ,1	TUTAL EVAPUTRANSP. (CM)	32.6	1
11	TUTAL SURFACE RUNDEF(CM)	34.1	
1.1		01.7	•
`·i			
۰.	- PULLORANT MASS DISTRI		
:	UPPER SJIL ZONE:		
ĭ,	1111 M. C. 1115 A.C.S. 11110016		
- -	TOTAL SURFACE RUNDEF TOTAL VULATILIZED	• J • J J T E = 01	1
	TUTAL JTHLA SINKS		•
5	FITAL AUST ON SULL FITAL INAUSILIELU-CEC		
i si	TITAL DEITAUED	12.3	
10	TJTNE HYJRJEYZED-901 T9TNE HYJRJEYZED-901	• /24 - 703 • 52 7F = 01	:
1.	TJTAL HYD BULYZED-CLC	• )	
<u>i</u> 1	ていてみに ていりいじんたい	• • • • 55 = - 0.2	1
5-1	FINAL LA SOLL MOIST.	. 6 JOE - 01	
27	FIRAL IN SOIL AIN	-1006-01	
11			
5	LJAER SUIL ZUNC:		1
11	TUTAL INTU GRAATER	• 4	
17	TUTAL VILATALIZED		
11	FINAL AUS. OF SUIL	4 4 4 4 4	1
	TITAL DESCAJUD	237.	
•	TJTAL HYJ (JL/2EU- ACI	. 5135-02	
1	1313L HYDRULY2CU#331 1313L HYDRULY2CU#CCC	1 • C J	
1	TJT+L CJ4+LLGU	2 • 50	
"1	TUTAL UTHER TKA'S. Flime I e Jule Muljt.	ل ه د ( ه د	
	FLIAL LA IJIL ALM		

- -- ----

AVEN	AGE PULLUTANT	CUNCENTRATIONS- (UG/ML)OR	(UG/G)	••
------	---------------	---------------------------	--------	----

MUISIURE-UPPER SUIL-UPPER AIR-UPPER FHEE LIGAND-UPPER	・タンスニーフト ・アトレ ・メノンミーフト ングフィーフト
MU131Uれビーレジョビネ 501レービジョース AIポービジョース デネビー LIGANJーレジョビス	•114[-02 •4 ,2[-0] •4 ,7[-0] •4 ,7[-0] •5 +3
MAX. PUL. DEPTH(M)	7.38

#### 

....

1-

C11111 ECCHANNE LAUDELTA - VIID LUALA SONE(-): 1.1 ליונחי דעראשזיר כאיעכווא יינוח שוחתרבירסארא לחמר(-): ויז C-11 :(-)PLC7 + 1040:81+07 CI198 1.91860 4870 (*1 : (-) ? + C7 + 2 ddn: 2 7001 + C117+ [+ -1+02 / 477 1(-) TALE 1 2000 112 HOT CLIVE IN TALE 2 100 2 2000 12 1000 1.2 1.1 1.1 C CARAIN CINTERT 10 MIDDLETURDER 2006 (-): 1.5 101 :(-)202 HAUAN:810 FOT 01108 KONEC - 236 202 - 9202 HAUAN:810 FOT 01108 KONEC - 2026 202 - 220210 + 22110 HONEC + 2026 99 - 2020 HAUAN:820 - 2020 - 2020 99 - 2020 HAUAN:820 - 2020 - 2020 - 2020 - 2020 0*1 :(-) THD7 HTCCOIN OTIA: Hd 0 ** : (-)_+07 Haden Hd 10-301 : (HO)HIATC DHEZ DIES TOCOTH 066-4 2014 2314 01614(CW): 10* DC014 10 754(4): *10E+03 0 *1 :( N ) * 0 * 1 * 4 -- SHALAMANA DARAMETERS ---------SPATAWANAG TURUT JAFANDU NUP 1231 ) :V3AV 'lddv 11 WTHD ISTI ( 51 01:004402 ) : CLAT-LUAN GENERIC SJIL TYPE ( 9 ) : CLINION HOT STATES (40 YR AVERGED DATA FOR TEST GNLY) ( + : NCIONY ) X TCNI 01202 10630 00110 10630 00030 1061 830N3330 INDISHIA COOLO 00010 10200 -28251+0772-+08(110). .ONI 31111 .0 501184. NOA+ .L 10130 13020 600-4 TL65X*0229-996(219)* *DPI 37111* *O 2008188*58180028809 *W ***** 05AFF 0bE42: 000 20 **** **** SEASONAL CYCLES OF ARTER, SEDIMENT, AND POLLUTANTS IN SOIL ENVIRONMENTS 116-710575 40000 00070 60000 MOHIHER SILE SPECIFIC SIMULATION(3 LAVERS)

1

..

• • •

.

..

1

.

÷

~

... *:

57

۴. . '

...

-

1

,

• 1

1

---

...i

2

1

•

1 POP

-- CHLMICAL PARAMETERS --

- --

:

SJUJJULITY(UJ/4U): .116+J4 AJJJ22. CJCF.(KJC): .302+03 JI. CJCF. 14 AIM(SJ.CM/SEC): .832-05 ULJKAJATIUN RATU(ZDAY): .20 ML4(Y) CJU.(UJ 4-AT"/4UU): .44E-02 AJJJK1. CJLE. 04 SJUL(K): .0 MLCULA2 MI.(G/4JL): 5-J VALUCE(-): 2-U NLJT24L 4YDRJLYSIS CUNSTANT(ZDAY): .70E+J5 JAJL HYDRJLYSIS CUNSTANT(ZMGL-DAY): 9-0 MCID HYDRJLYSIS CUNSTANT(L/MOL-DAY): 9-0 MCID HYDRJLYSIS CINSTANT(L/MOL-DAY): 9-0 HUAHJ-HJULUTAAT STABLUTY CUNST.(-): .10E+J6 40. HJULS LISAHDYAJLE POLLUTANT(-): 2-0 LIGNUM MJLCJLAR MLIGHT(J/4UL): 10. .

.

Ë

::

.

.

1

....

÷

-- JUIL PARAMETERS --

UE+>114( >/SU.C(): 1.3 [4]. PERALAULITY(SU.C(): -20E-09 >ISLUATECTEDTUSS INDEX(-): 10. PU+25IT4(-): -35 U(JANIC CARD)+ CUNTENT(X): 1.3 CLAT CARDUA CUNTENT(X): 1.3 CLAT CARDUA CUNTENT(X): -0 INFALASIC PERALECTY - 0PER 20NE(SU.CM): -0 INFALASIC PERALAULITY-0PER 20NE(SU.CM): -0 INFALASIC PERALAULITY-UNER 20NE(SU.CM): -0

____

----

----

... ....

		0. c. c. c. c. c. c. c. c. c.	0 0 0 0 0 0	0. 0. 0. 0. 0. 0.	C C C C	0 • 0 • 0 • 0 • 0 • 0 •	GCC00G00000000000000000000000000000000	0.000.000 0.000 0.000			Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant     Constant	<pre></pre>
· · · · · · · · · · · · · · · · · · ·	0•	<b>0</b> •	0.	0.•	0*	0•	0•	0 •	0 •	ñ•	0 - 20-3001	רחויכי ואזוינ חפיאר) יינאן יינאיי רחויכי ואזוינ חפיאר)
•											•	- 6 141-7140 MDz
- - - - - - - - - - - - - - - - - - -	0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 • 0 • 0 • 0 • 0 •	0* 0* 0* 0* 0*	0 • 0 • 0 • 0 • 0 •	0 • 0 • 0 • 0 • 0 •	0 • 0 • 0 • 0 •	00*1 0* 0* 0* 0*	00 • 1 0 • 0 • 0 • 0 • 0 •	0.00      	• - 0 1 5 0 E + 0 4 • - 2 - 0 • - 0 0 • - 0 0 • 0 0 • 0 0 • 0	2044 40012+(1=4+0=1) 1 b0f 14b-f(0?2)*(2) 50f 14b-f(0?2)*(4) b0f 14b-0(0?20*(4) W12* C04(-10*(0?44f) W12* C04(-02*(0?44f) W12* C04(-02*(0?44f)
2									· -		、 <u> </u> -	- 1 132-414C NUS
2]	48.7 42.4 40.4 40.4	46.7 60.4 46.4 7	++0E P0-6 V2E+ V2-1	64 - 7 60 - 9 60 - 9 60 - 9 60 - 9 7 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	30 ° ¢ 5 ° 0 9 6 ° 4 7 ° 9 7 ° 7 7 °	+6.7 90.4 40.4 40.4	ቀብ - ۲ የ ዕ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ሰ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ - የ ስ ስ - የ ስ - የ ስ ስ - የ ስ - የ ስ ስ ስ ስ - የ ስ ስ	46.7 055 40.4 40.4	44.7 4.05 4.05 4.05	+6+7 +62 23+4	405 605 625 48*/	48.7 (トン)・ロンニ54 055。 (2742)51455 - 2711.M 50.9 (-)・UM・5612 - M 4.01 (2740)502432 - M
:'	2 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5	-0 -100 -100 -150 R-45 -2	•0 •100 •100 •100 •100 •100 •100 •100 •	•0 •200 •250 •570 •570	•0 •300 •300 •300	2 • 2 4 0 0 • 0 0 0 0 • 0 0 0 • 0 0 • 0 0 • 0 0 • 0	•0 •360 •370 8•40 •5•2	000 • 2000 • 200 • 20 • 20 • 20 • 20	2.54 000 000 000 000 000 000	• 5 • 5 • 5 • 5 • 5 • 5 • 5 • 5 • 5 • 5	4.54 0.120 0.120 0.120 0.00 0.00	באו הטטבנטבט) אפיש אבעיבטער כן אארין יוסט ליבר אטאנט (דארי) יוסט ליבר אטאנט (דארי) יוסט אבעיבטלין יוסט יוסט יוסט אבערטטיין יוסט יוסט יוסט יוסט יוסט יוסט יוסט יוסט
1											SN3	
1, 1 ¹ 4 2,	d 9 S	າລະ	חתר	NOC	Y AM	오덕호	Я A M	837	NAL	230	AON	120
3 5 1 1 2								== Sង	JT3MAPAQ .	נארא זאיאטו		
		•••••					•••••					

- --- --------

		YEAR- ====		ITHLY RESU		JT) ===						
44040F0G	IC CYCLE (	LUMPONENT										
	001	NUV	DEC	JAN	FLB	MAR	APR	MAY	10H	JUL	AUG	SEP
UIL MUISTURE( % RLCIPATIDI(CM) TTINFILTR.(CM) VAPUTRAISP.(C4) VRFACE RUNDFF(C4) R= RUNDFF(C4) ILLD (C4)	) 7 - 5 - 5 5 - 0 2 2 - 7 2 4) 2 - 5 - 1 5 - 14	25.8 7.86 9.02 2.72 2.34 2.30 7.14	25.d 7.56 5.32 2.72 2.u 4 2.30 5.14	25.8 7.80 5.02 2.72 2.34 2.30 5.14	25.8 7.80 5.02 2.72 2.09 2.50 2.50 2.14	25.0 7.36 5.32 2.72 2.04 2.30 5.14	25.8 7.86 5.02 2.72 2.34 2.30 5.14	25.8 7.86 5.02 2.72 2.34 2.30 5.14	25 - 8 7 - 85 5 - 02 2 - 72 2 - 84 2 - 30 5 - 14	25.8 7.86 5.02 2.72 2.94 2.30 5.19	25 • 8 7 • d5 5 • 02 2 • 72 2 • 34 2 • 30 5 • 14	25.8 7.02 2.02 2.02 2.03 2.03 2.03 2.03 2.03 2
ATIJ PAZAPA(JZ)	1.00	1.00	1.03	1.0)	1.60	1.00	1.00	1.00	1.00	1.00	1.00	1.00
POLLUTAN	T MASS 1N	PUT TU COI	LUMN(UG)									
	UCT	NUN	DEC	JAN	FEB	MAR	APR	MAY	JUN	JUL	AUG	SEP
RECIPATION THER(UPPLR) THER(LUMER)	• 0 • 0 • 0	- 0 - 0 - 0	• 0 • J • V	• 0 • 0 • 0	.0 .0 .0	• 0 • 0 • 0	• 0 • C • C	•0	• 0 • 0 • 0	- 0 - U - Ú	• 0 • 0 • 0	• 0 • 0
JTAL INPUT	.0	• 0	• 0	•0	۰.0	• 0	• 0	• 0	• 0	•0	•0	• 0

50

; •

Ļ													
İ	MIDULE	DIL 2014:											
	VJLATILIZEU UTHL? SIAKS AJJ. UH JJIL 14HJJILIZLU-CLC DLWRADJU HYJKJLYZLU-MOI HYJKJLYZLJ-JI HYURJLYZJ-CLC CUAPLEXLU UTHLR TAKIJ. IK JIL AJIST. 14 JJL AIR	-0 -0 -122 -0 -1725-01 -1725-01 -1725-04 -0 -0 -0 -0 -0 -0 -0 -0 -0 -0 -0 -0 -0	•0 •943E=01 •374E=01 •141C=10 •477E=04 •3 •0 •3 •362E=02 •525E=13	•0 •704E-01 •J •185E-01 •127L-05 •J55L-04 •0 •0 •24JE-02 •J47E-02	.0 .5 .5 .5 .0 .1 .2 .7 .0 .2 .2 .2 .0 .0 .1 .0 .2 .2 .0 .2 .2 .0 .2 .2 .0 .2 .2 .0 .2 .0 .2 .2 .0 .2 .2 .2 .2 .2 .2 .2 .2 .2 .2 .2 .2 .2	.0 .426L+01 .J .d99202 .c17E-J6 .212L-34 .J .U .U .119L-02 .1720J	• 0 • 350E - 01 • 6 • 651E - 02 • 453E - 06 • 170E - 04 • 0 • 0 • 0 • 404E - 03 • 131E - 03	.0 .271E-01 .0 .5u7E-02 .3+3E-06 .141E-04 .0 .0 .0 .0 .0 .0 .101E-03	.0 .235E-01 .0 .344E-02 .203E-00 .116E-04 .0 .0 .517E-03 .750E-04	.0 .200E-01 .0 .3112-02 .213E-06 .945E-05 .0 .0 .0 .0 .0 .0 .0 .000E-C4	• 0 • 0 • 16 3C - 01 • 0 • 230E - 02 • 15 3F - 05 • H0 2C - 05 • 0 • 0 • 0 • 0 • 10 E - 0 3 • 450E - 04	.0 .0 .143E-01 .0 .125E-62 .125E-62 .02E-05 .0 .0 .0 .0 .253E-03 .375E-04	.0 .143E-01 .0 .175E-02 .1172-06 .645E-05 .0 .5 .5 .5 .5 .5 .5 .5 .5 .5 .5
i	LU428 33	DIL ZONE:											
	INTO GRAATER VOLATILIZEO UIMER SINNS AJS. UN SIL IMM JILIZEO -CEC DEGRADED MYD (JEYZED - MUI MYD (JEYZED - MUI MYD (JEYZED - CEC CJ. MELKED UTMER JR (43) I N JIL (JIST. I N JUL AIX	.0 .0 .500 y/2. .0 21.2 .9612-03 .1 .1 .0 .1 .1 .0 .1 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .0 .1 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0	.0 .0 .3 .0 .3 .0 .5 .5 .5 .5 .0 .2 .7 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0	.0 .0 .0 .0 .0 .0 .0 .2 .1 43E-02 .2 .2 .0 .2 .2 .0  .2 .2 .1 9 .7 61	.U .U .0 .0 .0 .14.4 .136c-02 .2 .2 .0 .0 .20 .0 .0 .10 .741	.U .J .J .J .U .U .22.6 .129E-02 .221 .U .200 .J .200 .J .200 .J .200 .J .200 .J .200 .J .200 .300 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .000 .0000 .000 .000 .000 .000 .000 .0000 .000 .000 .000 .0000 .0000 .000 .0000 .000	• 0 • 0 • 0 • 0 • 30 • 9 • 122E - 02 • 213 • 0 • 213 • 0 • 213 • 667	• 0 • 6 • 7 7 6 • 0 • 2 • 1 • 1 • 6 • 0 • 0 • 2 • 1 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0	.0 .0 748. .0 27. y .1102-02 .154 .0 .154 .0 .154 .0 .0 .154 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0	-0 -0 722. -0 26.6 -105E-02 -191 -0 -209 -0 -209 -573	• 0 • 0 • 0 • 0 • 0 • 1 • 1 • 1 • 1 • 7 • 5 • 5 • 5	.C .0 .0 .73. .0 .4.1 .951E-03 .173 .0 .225 .0 .225 .5 .5 .5 .5	.0 .0 .0 .22.7 .906E-0J .172 .0 .136 .3.41 .435

1

;",

----

i

1

. ....

-- ----.......

(M)HT936 .J69 .XAM	5 ° G V	4.47	• 601	*6F I	• 04 1	•105	535.	.263.	.445	• *25	•536	• ୬ ୩ ୮
רברטאטזידר, אנרירטאבי גנירירטאבי גניאנטירירטאני גניאנטירטאני	• 2325-02 • 7375-01 • 715-01	•112 •9882-07 •1196-01 •5506-05	• 4712 • 477E - 27 • 587E - 21	• 1095 - 03 • 9095 - 03 • 1095 - 05	.1995-02 .6196-01 .6196-02	•115 •116-01 •118-05 •118-05	2/12 •0177-03 •0406-03 •1016-05	6//0 •0995-03 •1915-05	•161E-02 •161E-01 •161E-02	9715 • 5 4 65 - 6 J • 5 2 96 - 6 J • 1 4 65 - 62	.1392-02 .5572-03 .1392-02	5415-03 • 5*15-03 • * 936-01 • 1756-05
נוסאט-אט-אנסרב אוי-אנסרר אוי-אנסררק אטוסנאס-אנססרב	5.01 • • 55 • 0	0 • • 1 • 1 • • 0	   	•0 •500 •17 •020	۰ ۵۵۵ ۱۹۶۰ ۱۹۶۰	• 143 • 143 • 143	0. 110 5.20 0.210	•0 •11 AF -01 1•14 •500	0 • 99 • E = 0 1 • 1 • 5 • 1 • 5	•0 • • • • • • • • • • • • • • • • • • •	00-3607- 9607- 960-1 1000-	•0 • • 0 • • • • • • • • • • • • • • • •
K∃490≻67c1 [0M 1021082 k 140152 k 29440-01:401	- 10 - 11 - 11 - 10 - 10 - 10 - 10 - 10	14.1 74.4 473.	696. 695. 695.	•852 24•6 170•	• 852 60 69 • 5 69 • 5	•897 •125 •71 •225	• PG7 P11• C4•1 C97•	589* *0376-01 *58 *58	₹89* •14\E-01 1•34 •163	*667 10-3009 77*1 891*	538° ** 31E-01 1 * 07 * 1 51	• 692 • 711 • 710 • 700 • 700
			שרע מע נים									

.1

:!

YLAR - 1	ANNUAL SUMMARY	الا
111111		
TUTAL INPUTS		•
UPPLE SULL ZONE	205.	
MIDJLE SJIL ZUNË Lumer Suil Zjne	•0 •190E+J4	
ATDRULJGIC CYCLE CUMPONEN	115	
AVERAGE SULE MULSTURE( %)	25.8	11
TUTAL PRECIPATION(CM) TUTAL INFILTRATION (CM)	74 • 3 63 • 2	
TUTAL EVAPOLIANSP.(CM) TUTAL SUAFACE RUNIFF(CM)	32.0	
TJTAL YILD (CN)	61 • 7	
PULLUTANT MASS DISTRIBUTI	UN (UG)	1
UPPER ZUNE:		
TULAL SURFACE RUNDEF		1
	• 4 J J • 4 J J	i
FIAL IMAUSILIZED-CEC		
101 VE HY) (JEYZED-MUI 131 AL HY) (JEYZED-SJI	.121E-01	•
T IT IL HYDYULYZLO-C.C TUTAL CUMPLA:D	• J • + J7E - 02	1
TUTAL UTALA TRANS. Fland IN SULL NUIST.		1
FINAL IN SOLL AIR	• J) C = 01	
		1-
ALUDLE SUIL ZUNE:		
TUIAL VULATILIZED Tulal UTALR SINKS	• 0	
FINAL ADS. OF SULL FINAL LANJSILIZED-CEC	-143E-01	
IJIAL HESRAULD	• 112 • 7 • 7 5 - 25	
TUTIL HYURULYZED-UU	241E-03	
TUTAL CJAPLEXED TJTAL JTALA TXANS.	• 0 • 9	
FIGAL IN SULL AUST.	•23%=03 •37%=04	

## FINAL IN SULL AIR

.375E-04

1			,
	LOWER SOTE ZON :		j
Ţ		· · ·	'n
ų.	TUT 4L LITU UZAATE ?	• •	
1	TUTAL VULATALIZED	•0 •2-1-0	
	FLAL ADS. JA SULL	0.3.).	
1	TJTAL DEGRADED CEC	ل ه د د د د	1
,	TUTAL HYURULYZLD-401	.13/C-01	
,	TJTAL HYJKJLY/ED-501	2.44	
4	T JT AL CUAPLESLU	2	;
11	Fine it stit Milste	• 3 JU 3 • 4 1	
4	FINAL IN SULL ALM	• • > 5	;
-			
≥	- AN WALF POLLUTANT C	NCENTRATIONS-(11./4) ) 00 (11./1)	
Ĩ		UNCENTRATIONS-TOOMET OR TOSTOT	
ö			
•.			
5	MD. STURE-UPPER	• 5 2 3	
		2. 14	
78	FRE LIGAND-UPPER	• < 4 J 2 U 3 •	
77			1
11			;
2	MOISTURE-MIDDLE	• 502	
24	AIN-MIDULL	• 5 10	•
24	FREE LIJAHO-MIODLE	• U	'
μį.			
	M.11 . F.102 F = 1 (1=5 ⁻ -2	17702	,
FI.	SULLEUALA	. 0 U D E - 0 I	•
	AIR+LUAE? Far: 116AAD+13459	•72+E=0J	;
in;		• • • •	1
1	MAX. PJL.DLPTH(M)	J. 16	
5			
171			'
1			
isi.			
1			
11			
1			
-			

		· · · · · · · · · · · · · · · · · · ·											
													•
							· · ·						
	YL	AR- 2 MOI	THLY INPL	JT PARAME	TERS								
	001	NUV	DEC	JAN	FER	MAR	APG	MAY	JUN	JUI	AUG	SED	
	•••			•					••••			52.	
- CLIMATIC DA	DAMETEU												
CEIMATIC PA	IN ASIC LER.	<b>J</b>											
LATITUDE(D.G) TLMP.(JE. C)	42.5	42.5	42.5	42.5	42.5	42.5 8.40	42.5 8.40	42.5	42.5 H.40	42.5	42.5 8.40	42.5	
CLUUG CVR(FRAC.)	- 750	. 320	. 120	. 320	. 120	. 320	.320	. 320	.320	.320	. 120	.320	
ALBEDI(-)	• 100	• 700 • 360	. 100	. 300	-700 -Juu	.300	.700	•700	• 705	. 100	.300	•700 •300	
E /APUI. (C4/DAY)	.0	• Ú	• Ĵ	• 0	• 0	• 0	• 0	• 0	• Ŭ	• 0	• Ĵ	• 0	
PHECIP.(C4)	7.04	7.84	7.84	7.84	7.64	7.44	7.04	7.84	7.84	7.54	7.54	7.34	
M. [IML RA[4(DAYS) M. [2](+)	• J20 2• J3	- 320 7-66	. 120	.32J 9.09	- 320	• 320 9 • 08	・3こい い。いろ	- 750 60 - 6	• 323 V• 03	• 326 8 • 6	•320 8•39	-320	
M. JLASJA(JAYJ)	33.4	3.44	30.4	30.4	30.4	30.4	30.4	30.4	30.4	30.4	30.4	39.4	
-													
RUN DATA-SE	T 1												
MUIS. CUIC-UP. (UG/M	1	• 0	• 0	• 0	• 2	•0	•0	• 0	•0	• 0	• 0	•0	
NJIJ. CJ.(C-41.(UJ/	4L) •J	•0		.0	• •	•0		-0	•0	.0		.0	
	() •0	• 0	• 0	•0	•0	• 0	•0	•0	•0	• 0	•0	•0	
PJL . I :P-L (UJ/50.C"	:) .U	.ŏ	- Č	. ŏ	.0	•0	•0	.0	•0	.ŏ	.0	. 5	
50K. NU 1JFF(1=4+0=*	4) -0	• •	• 0	• 0	• 0	• 0	- 0	•0	• 0	. 0	• 0	• 0	
	.12												
CONC. IN RAIN(UG/ML	.) •0	• 0	• 0	• 0	• 0	• 0	•0	•0	•0	• 0	•0	•0	
1 4. 3r 3r 4. 0 - M(JG/37	ČM) U	, •Ŭ	.ŭ		÷ŏ	.č		.0	.ŭ	č	÷	10	
Tili ンドリ とオニンチビ(リン/っぷ。 ちてはたちゃい(リン/ンルットート	.см) .0	• •	• 6	.0		-0	• U • D	•0	•0	• 0	• • •	•0	
31455-4(Ju/3).(1)	.0	•0	.0	.0		-0	• •	•0	.0	.0	.0	-0	
51+53+6(36/53+64) 6162340017+3(3+/53+	•0 •1	•0	• • •	• • • • •	- U - D	•0 • u	- 0 - 0	•0	•0	•0 •ນ	0. . J	•9 •0	
LIU. I. PUT-1 (J.J. J. J.	- U		• •		- 0	- Ŏ		• •	•0	• 2	-0	-3	
L16.1.1.PUT-L(J3/33.6		• 0	• 3	• 0	• 3	- 0	• 0	.0	.0	• 0	• 0	• 0	
l													
---	-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	--------------------------------------------------------------------------------------------	-------------------------------------------------------------------------	--------------------------------------------------------------------------------------	-------------------------------------------------------------------------	------------------------------------------------------------------------------	----------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------	------------------------------------------------------------------------------------------	-----------------------------------------------------------------------------------------------------	-------------------------------------------------------------------------	------------------------------------------------------------------------	---------------------------------------------------------------------------------------
Ī		¥ć 4 F	- 2 NÚM	ITHL/ RESU	ULIS (OUTP	(TU)							
	HYDROLOGIC	CYCLE COP	PONENTS										
		oct	NON	DEC	MAL	FEB	MAR	APR	MAY	NUL	JUL	AUG	SEP
	SUIL MOISTURE( X) PRECIPATION(CM) NET INFILTR.(CM) LVARUTRANSP.(CM) SURFACE RUNDFF(CM) SHR RUNDFF(CM) YILLD (C4) RATIO PA/4PA(S2)	25.8 7.46 5.02 2.72 2.34 2.33 5.14 1.00	25.d 5.02 2.72 2.84 2.30 5.14 1.00	25.8 5.02 2.72 2.84 2.30 5.14 1.00	25.8 7.86 5.02 2.72 2.34 2.30 5.14 1.00	25.0 7.36 5.02 2.72 2.84 2.3J 5.14 1.00	25.8 7.86 5.02 2.72 2.84 2.30 5.14 1.00	25.8 7.80 5.02 2.72 2.84 2.30 5.14 1.00	25.8 7.80 5.02 2.72 2.84 2.30 5.14 1.00	25.3 7.35 5.02 2.72 2.84 2.30 5.14 1.00	25.8 7.30 5.02 2.72 2.30 5.14 1.00	25.8 7.80 5.02 2.72 2.84 2.30 5.14 1.00	25.8 7.95 5.02 2.72 2.34 2.30 5.14 1.00
	POLLUTANT	MASS INP	UT TU COLO	JMN (UG) ·									
	PRECIPATION Other(UPPER) Other(MIJ9LE) Uther(E042R)	• 0 • 0 • 0	• U • O • D • D	• 0 • 0 • 0 • 0	• 0 • 0 • 0	. U . U . U	• 0 • 0 • 0	• 0 • 0 • 0	• 0 • 0 • 0	• 0 • 0 • 0	••••	• 0 • 0 • 0	• 0 • 0 • 9 • 0
	TUTAL LOPUT	• 0		• 0	• •	•0`	• 0	•0	•0	•0	• 0	• 9	•0
	POLLUTANT	MASS DIST	NO ITUEI S	IN CULUMN	(UG)			-					
	102 N3440	L ZONE:											
	SURFACE RUNDFF VJLATILIZED UTHER SINKS ADS. UN SOIL IAMUBILIZED-CEC DIGADED MYDRUCYZLD-MJI HYDRUCYZLD-SII HYDRUCYZLD-CLC COMPLEAL) DIH.R THATS.	.0 .168E-J1 .0 15.8 .0 2.J9 .150E-J3 .750E-J3 .0 .397E-J3 .J	.0 .141E-01 .0 14.0 .0 1.70 .1322-03 .02E-02 .0	•0 •120E=01 •0 12.4 •0 1.4) •1.12E=03 •53E=02 •3 •307E=03	.0 .102E-01 .0 I.1.1 .9562-04 .5252-02 .0 .367E-03	.0 .076E-02 .0 10.0 .0 .922E-04 .471E-02 .0 .307E-03	.0 .758E-02 .0 .0 .7 .7 10E-04 .424E-02 .3 .3 .7 E-03	.U .055E - 02 .U .U .517 .616E - 04 .314E - 02 .U .J:(7E - 03)	•0 •573E-02 •0 7.41 •0 •537E-04 •537E-04 •340E-02 •0 •537E-03	• 0 • 50 3E - 02 • 0 • 76 • 0 • 0 25 • 47 1E - 04 • 31 7E - 03 • 31 7E - 03	•0 •443E-02 •0 •18 •550 •4152-04 •2892-02 •337E-03	•0 •J92E-02 •0 •480 •367L-04 •265E-02 •( •194f-03	• 0 • 340E- 02 • 0 5 • 21 • 432 • 320E- 94 • 2435- 02 • 337E- 03
	14 3816 4313T. 14 37 5 5 5 5	• 325 • • 7 * • * 1	•274 •37• -31	+233 +3330-71	•192 •192-01	•172 •••• • •01	•148 •15=-01	•129 •1371-01	•113 •104E•31	.992E-61 .144L-61	-876E-01 -1.7/E-01	.1121-01	. 070E - 01 . 107E - 31

P

MIDDLE	SOIL ZUNE:											
VJLATILIZED   UFHLH SI4KS   ADS. JH JJL   IMAUJULIZED-CEC   D_G4AUJJ   HYJKUYZJ-HUI   HYJKUYZJ-SI   HYJKULAIST.   HYJULAIH	-0 -994E-02 -0 -115E-02 -739E-07 -499E-03 -0 -0 -0 -155E-03 -225E-04	•0 •J •B72E-02 •J •J •J •J •J •J •J •J •J •J	•0 •0 •744E-02 •0 •582E-03 •552E-03 •537E-05 •0 •0 •0 •0 •103E-03 •153E-04	.0 .0 .006E-02 .0 .511E-03 .3512-07 .2742-05 .J .0 .0 .775E-04 .112E-04	.0 .0 .453E-02 .0 .426E-03 .292E-07 .240E-03 .0 .0 .0 .5 .15-04 .750E-05	•0 •5 •5 •3 •1 •2 •1 •2 •1 •2 •0 •0 •0 •0 •0 •5 •1 7 ±-0 4 •7 5 5 =0 •5	•0 •0 •276E-02 •0 •145E-03 •145E-03 •145E-05 •0 •0 •0 •0 •250E-04 •375E-05	•0 •0 •276E-02 •0 •17GE-03 •117E-07 •125L-05 •0 •0 •0 •259E-04 •375E-05	•0 •0 •276E-02 •170E-03 •117E-07 •125E-05 •0 •0 •0 •0 •0 •254E-04 •375E-05	- 0 - 0 - 0 - 1 2 6 5 - 0 3 - 0 7 7 5 - 0 8 - 9 3 8 5 - 0 6 - 5 - 5 - 5 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	······································
LU4ER S	JIL ZONE:											
INTJ GRAATER VULATILIZED UTHER STAKS AJS. UH SUIL 144USILIZED-CEC ULGRAUED MYURLYZED-MJI MYURUYZEU-CEC CUMPLEYED-CEC CUMPLEYED UTHER TRANS IN SULL AUTST IN SULL AUR	.0 .9 .0 21.8 .104E-03 .106 .0 .137 .0 J.25 .472	. 1) . 1 0 08. . 0 2 0 . 101 . 1 10 . 1 10 . 1 30 . 1 0 . 4 50	•0 •0 •0 •0 •1 •3 •4 •1 •3 •2 •2 •2 •4 30	• 0 • 0 • 5 • 5 • 7 • 7 • 7 • 7 • 7 • 7 • 7 • 7	.0 .0 .0 .5 .0 .14.1 .7175-03 .145 .0 .251 .0 2.70 .392	.3 .0 .5 .3 .0 .1 .0 .1 .1 .0 .1 .1 .0 .1 .0 .1 .0 .1 .0 .0 .1 .0 .0 .1 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0	-0 -0 -0 -17- -0 -0 -0 -0 -0 -0 -0 -0 -0 -0 -0 -0 -0	.0 .0 .5 .5 .5 .0 .0 .0 .0 .0 .1 .1 .1 .1 .1 .1 .1 .1 .1 .1 .1 .1 .1	.0 .0 .0 15.2 .001E-03 .12 .2 .2 .2 .2 .2 .2 .2 .2 .2 .2 .2 .2 .2	.0 .0 .0 471. .0 14.6 .570±-03 .1.'4 .0 .236 .0 .235 .0 .17 .315	.U .0 .457. .0 .14.0 .552E-0J .121 .146 .C .2 .00 .302	• C • G • G • G • J • J • J • J • G • J • C • J • C • C • C • C • C • C • C • C • C • C

-----

-----

____

_ __ __

i fr

;

.

-----

ATT JUAN -MAN -

-----

----

-

>0	LLUTANT C	DHCENTRA	TUNS-(UG)	<u>AC) DA (1</u>	JG/G)								
MUISTUPE-U SJIL-UPPLR Ain-UPPER LIGXU-UPP	PPEN En	• 840E-01 • 798 • 343E-01 208•	•703E-01 •700 •2896-01 230•	• 601E-01 • 628 • 2465-01 258•	•514c-01 •562 •210t-01 258•	•4435-01 •505 •1416-01 258•	.3+32-01 .455 .157E-01 254.	• 333E-01 • 412 • 130E-01 258•	•2916-01 •374 •1196-01 250•	•256E-01 •341 •105E-01 258•	.225L-01 .312 .724L-02 258.	•200£-01 •266 •819E-02 250•	.178F-01 .253 .723E-02 258.
MUISTUZE-M SJIL-4IUJLI 4I2-4IUJLZ LIS 40-4IJ	IDDLË t VLË	・030E-01 - 753 - 245ビー01 - 0	-500E-31 -001 -20-2-31 -3	• 400E-01 • 50 4 • 10 4E- 31 • 0	•J00E-01 •457 •123E-01 •0	.200E-01 .343 .3152-02 .0	- 200E-01 - 34 J - 91 dE-02 - 9	.100E-01 .209 .439E-02 .t	•100E-01 •209 •409E-02 •0	•100E-01 •203 •403L-02 •9	• 0	•0 •0 •0 •0	• 0 • č • 0
MUI 3TURE-L SJIL-LURER AIN-LURER LIGANJ-LJR	Ŭ#ÉR Ĺĸ	•126E-02 •477E-01 •515E-03 •775	•1202-02 •4018-J1 •4922-03 •775	•1155-02 •4465-01 •4695-03 •775	•1102-02 •432E-01 •445E-03 •775	•1055-02 •4185-01 •4285-03 •775	・1002-02 ・4352-01 ・410E-03 ・775	.458E-03 .372E-01 .372E-03 .775	•917L-03 •360E-01 •375L-03 •775	•373E-03 •363E-01 •3575-03 •775	•8418-03 •3578-01 •3446-63 •775	.807L-33 .337E-01 .33JC-03 .775	.774E-03 .330E-01 .310E-93 .775
MAR. PUL.	DEPTH(H)	417.	448.	411.	510.	541.	572.	60J.	634.	605.	640.	727.	759.

. .....

7	YEAR -	AN JUAL SUMMARY	·	
T				~
i.				
1				
1	IJIAL IRPUIS +-			
i.	1130. 11 COLL 2016			
!	MTEN 6 3311 7365	• • •		
÷.	- 12 E 3 E 3 I E 2 3 E	• •		
	CJWER JOIE ZUNG	•9		
!				
1	HYDRULOGIC CYCLE COMPON	14Tu +-		
÷.				
ų.	AVERAJE JOIL HOISTURE( %)	25.8		
1	TUTAL PRICIPATION(CM)	94.5		
9	TOTAL INFILTHATION (CM)	60.2		
1	TUTAL EVAPOTRANSP.(CA)	32.0		
'i	TUTAL SURFACE RUNJEF(CM)	34 • 1		
ų.	TUTAL GRA RUNDFF(CA)	27.6		;
ï	TUTAL VIELD (CM)	51.7		i
1				1
1	. OD . ITANT MACE DISTUIN	LINE COULD BE		
1	FOCEDIATI MASS DISTRICT	10N (007		•
1	UPPER /ONE:			
1				
1				
1	TOTAL SURFACE RUNDEF	• 0		i
	TUTAL VULATILIZED	•¥37E+01		1
	TUTAL UTHER SINKS	• J		1
	FINAL ADS. UN SUIL	5.21		1
	Fline IMAUBILIZED-CEC	• J		1
	TUTAL DEURADED	15•3		
	TUTAL HYDRULYZED-MOI	.7246-03		1
٩į.	IOLAL HADKOLASEN-POI	.3275-01		t
<u>:</u>	LILVE HATATES-CEC	• •		
i		• 4 4 JE * 02		1
וי	10146 01/167 1/(AMS). 61.31 1/1 6011 1/071 1	•U		'
	F 1 4 40 144 0010 TUIDIA P 1 6 41 191 - 191 49-	• 0 /JE = 01		
1	TINNE TH JOIE WIN	• 1 3 36. ~ 01		
1	NIDDLE SOIL ZONE:			
3				
4				
ΥÌ.	TUTAL VOLATILIZED	.0		i.
'i I	TUTAL UTIER STAKS	• 0		1
1	FINAL ADS. UN SOIL	• 0		
11	FINAL IFAUDILILLUTCLC	+ J 		
1	TJTAL ULGRAULU TJTAL MANLAN A'KNEMMT	◆サウメビデリビ えりょう		
1	TATAL HYDROLYLU-MUL TATAL HYDROLYLA+101	• J Z Z C = VO . J A A F • JA		
1				
	TUTAL CUMPLEXLU	. U		
	TUTAL UTALK TRANS.	. u		
	FIMAL IN SUIL MUIST.	.0		
	FINAL IN SOIL AIR	• 0		
	A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL			~

#### LOVEN SOIL ZON :

11140	I I U GREATER	• J
すって っに	VOLATALIZED	- U
「」「ーレ	JINER SINKS	• 0
Flink		443.
FLIAL	14463161266-666	• )
エンエッビ	JE JRAJEU	207.
T JI NL	HYJRJE/ZED-701	.317ċ-02
1-1-6	162-03:1364	1.35
TUTAL	HYJRULY_LD-CLC	• J
TJTML	CU14LE C.D	2.21
TUTIL	UTIL & TRANSA	• J
+1	IN JUL MULT.	2.30
FLINAL	TH SULL ALL	. 2 23

#### -- AVERAGE PULLUTANT CONCENTRATIONS-(UG/ML) OR (UG/G) --

1.53

MJ15TURE-UPPER	• 907E + 01
53iL-UPPER	• 716
AIR-UPPER	• J72E - 01
FREE LIGRAD-UPPER	28J•

NOL STURE-MIDDLE	•677E-01
Scil-41ddlê	•335
Alê-Middlê	•277E-01
Fri: Livand-41ddlê	•0
MULITURE-EUMER	• 1145- J2
Sule-Eumer	• 4 325-01
Alk-Eumer	• 4 J75-93
Fréi Elgand-Lower	• (34)

A14-1	LUMLA
FRC_	LIGANU-LOWER
MAX.	20L.DEPTH(4)

#### 

۰. 1,1 1'1

> • 1.4

> > ، . ا. .

:-

RE - references

## APPENDIX RE *

#### REFERENCES

Adams, R.T.; Jurisu, F.M. Simulation of pesticide movement on small agricultural watersheds. Report No. EPA-600/3-76-006. Athens, GA: U.S. Environmental Protection Agency Research Laboratory; 1976. Available from: NTIS, Springfield, VA; PB 259 933.

Benjamin, J.R.; Cornell, C.A. Probability, statistics, and decision for civil engineers. New York, NY: McGraw-Hill; 1970.

Biggar, J.W.; Nielsen, D.R. Miscible displacement: I. Behavior of tracers. Soil Sci. Soc. Amer. Proc. 26:125-128; 1962.

Bode, L.E.; Day, C.L.; Gebbart, M.R.; Goering, C.E. Prediction of trifluralin diffusion coefficients. Weed Sci. 21(5):485-489; 1973.

Crawford, N.H.; Donigian, A.S., Jr. Pesticide transport and runoff model for agricultural lands. Report No. EPA-660/2-74-013. Washington, DC: U.S. Environmental Protection Agency; 1973.

Davidson, J.M.; Brusewitz, G.H.; Baker, D.R.; Wood, A.L. Use of soil parameters for describing pesticide movement through soils. Report No. EPA R-800364. Washington, DC: U.S. Environmental Protection Agency; 1974.

Day, P.R.; Forsythe, W.M. Hydrodynamic dispersion of solute in the soil moisture stream. Soil Sci. Soc. Amer. Proc. 21:477-480; 1958.

Donigian, A.S., Jr.; <u>et al</u>. Agricultural Runoff Management (ARM) model version II. Report No. EPA-600/3-77-098. Athens, GA: U.S. Environmental Protection Agency Research Laboratory; 1977.

Donigian, A.S., Jr.; Crawford, N.H. Modeling pesticides and nutrients on agricultural lands. Report No. EPA-600/2-7-76-043. Athens, GA: U.S. Environmental Protection Agency Research Laboratory; 1976.

Eagleson, P.S. Dynamics of flood frequency. Water Resour. Res. 8(4):878-898; 1972.

Eagleson, P.S. Climate, soil, and the water balance, a framework for their analytical coupling. Cambridge, MA: Massachusetts Institute of Technology; 1977.

Eagleson, P.S. Climate, soil, and vegetation, 1, introduction to water balance dynamics. Water Resour. Res. 14(5):705-712, Paper 8W0184; 1978.

References mainly not contained in the various Appendices.

Eagleson, P.S. Climate, soil, and vegetation, 2, the distribution of annual precipitation derived from observed storm sequences. Water Resour. Res. 14(5):713-721, Paper 8W0185; 1978.

۰.

Eagleson, P.S. Climate, soil, and vegetation, 3, a simplified model of soil moisture movement in the liquid phase. Water Resour. Res. 14(5):722-730, Paper 8W0186; 1978.

Eagleson, P.S. Climate, soil, and vegetation, 4, the expected value of annual evapotranspiration. Water Resour. Res. 14(5):731-740, Paper 8W0188; 1978.

Eagleson, P.S. Climate, soil, and vegetation, 6, dynamics of the annual water balance. Water Resour. Res. 14(5):749-764, Paper 8W0207; 1978.

Eagleson, P.S. Climate, soil, and vegetation, 7, a derived distribution of annual water yield. Water Resour. Res. 14(5):765, Paper 8W0189; 1978.

Eagleson, P.S. The annual water balance. J. Boston Soc. Civil Engineers, Spring issue; 1979.

Eagleson, P.S. The annual water balance. J. ASCE. HY8(105):923-941; 1979.

Eliss, B.G.; Knwzek, B.D. Adsorption reactions of micronutrients in soils. Mortvedt, J.J.; Giordano, P.M.; Lindsay, W.L. eds. Micronutrients in Agriculture. Madison: Soil Science Society of America: 59-78; 1972.

Enfield, C.G. Rate of phosphorus sorption by fine Oklahoma soils. Soil Sci. Soc. Amer. Proc. 38:404; 1974.

Fairbridge, R.W.; Finkl, C.S. The encyclopedia of soil science. Part 1. Pennsylvania, PA: Dowden, Hutchinson and Ross, Inc.; 1979.

Farmer, W.J.; Igue, K.; Spencer, W.F. Effect of bulk density on the diffusion and volatilization of dieldrin from soil. J. Environ. Quality 2:107-109; 1973.

Farmer, W.J.; Igue, K.; Spencer, W.F.; Martin, J.P. Volatility of organochlorine insecticides from soil: I. Effect of concentration, temperature, air flow rate, and vapor pressure. Soil Sci. Soc. Amer. Proc. 36:443-447; 1972.

Foster, G.R. Soil erosion modeling: special considerations for nonpoint pollution evaluation of field-sized areas. Overcash, M.R.; Davidson, J.M. eds. Environmental impact of nonpoint source pollution. Ann Arbor, MI: Ann Arbor Science Publishers, Inc.; 1980.

Foster, G.R.; Huggins, L.; Meyer, L.D. Simulation of overland flow on short field plots. Water Resour. Res. 4(6):1179-1187; 1968.

Foster, G.R.; Lane, L.J.; Nowlin, J.D.; Laflen, J.M.; Young, R.A. A model to estimate sediment yield from field-sized areas: development of model. Report No. A-2361. Laxenburg, Austria: International Institute for Applied Systems Analysis; 1980.

Foster, G.R.; Meyer, L.D. Mathematical simulation of upland erosion using fundamental erosion mechanics. Presented at Sediment Yield Workshop, November 1972, Oxford, MS; 1972.

Foster, G.R.; Meyer, L.D. A closed-form soil erosion equation for upland areas. In: Shen, H.W., ed. Sedimentation symposium to honor Professor Hans Albert Einstein. Fort Collins, CO: Colorado State University. pp. 12-1 to 12-19; 1972.

Foster, G.R.; Meyer, L.D.; Onstand, C.A. A runoff erosinity factor and variable slope length exponents for soil loss estimates. Transactions .ASAE, 20(4):683; 1977.

Foster, G.R.; Meyer, L.D.; Onstand, C.A. An erosion equation derived from basic equation principles. Transactions of ASAE, 20(4):678; 1977.

Gardner, W.R. Some steady-state solutions of the unsaturated moisture flow equation with application to evaporation from a water table. Soil Sci. 85(4):228-232; 1958.

Hamaker, J.W. Decomposition: quantitative aspects. Goring, C.A.I; Hamaker, J.W., eds. Organic chemicals in the soil environment, Vol. I. New York, NY: Marcel Dekker; 1972.

Hamaker, J.W. Diffusion and volatilization. Goring, C.A.I.; Hamaker, J.W., eds. Organic chemicals in the soil environment, Vol. I. New York, NY: Marcel Dekker, 1972.

Hartley, G.S. Evaporation of pesticides. In: Pesticidal formulations research, physical and colloidal chemical aspects, advances in chemistry series, 86. Washington, DC: American Chemical Society; 1969.

Holton, H.N.; <u>et al</u>. Moisutre tension data for selected soils on experimental watersheds. Report No. ARS 41-144. U.S. Department of Agriculture. 1968.

Igue, K.; Farmer, W.J.; Spencer, W.F.; Martin, J.P. Volatility of organochlorine insecticides from soil: II. Effect of relative humidity and soil water content on dieldrin volatility. Soil Sci. Soc. Amer. Proc. 36:447-450; 1972.

Jurinak, J.J.; Grenney, W.J.; Woodbridge, G.L.; Riley, J.P.; Wagenet, R.J. A model on environmental transport of heavy metals originating from stack derived particulate emission in semi-arid regions. Logan, UT: Utah State University; 1977. Jury, W.A.; Grover, R.; Spencer, W.F.; Farmer, W.J. Predicting vapor losses of soil-incorporated triallate. Unpublished, undated manuscript received as personal communication from W.J. Farmer, December 4, 1979.

.Kay, B.O.; Elrick, D.E. Adsorption and movement of lindane in soils. Soil Sci. 104:314-322; 1967.

Knisel, W.G.; Foster, G.R. CREAMS--A system for evaluating bestmanagement practices. Personal correspondence with G.R. Foster; 1979.

Konrad, J.G.; Chester, G.; Bauer, K. Description and calibration of a pollutant loading model--LANDRUN. Report No. EPA-R005142. International Joint Commission, Menomonee River Pilot Watershed Study, Wisconsin Department of Natural Resources, University of Wisconsin; 1978.

Lindsay, W.L. Inorganic phase equilibria of micronutrients in soils. Mortvedt, J.J.; Geordano, P.M.; Lindsay, W.L., eds. Micronutrients in agriculture. Madison: Soil Science Society of America, pp. 41-57; 1972.

Lyman, W.; <u>et al</u>. Research and development methods for estimating physicochemical properties of organic compounds of environmental concern. Arthur D. Little, Inc., Phase II final report to U.S. Army. To be published in 1981 by McGraw-Hill, New York, NY.

Mayer, R.; Letey, J.; Farmer, W.J. Models for predicting volatilization of soil-incorporated pesticides. Soil Sci. Soc. Am. Proc., 38, 563-68; 1974.

Meyer, L.D.; Wischmeier, W.H. Mathematical simulation of a process of soil-erosion by water. Trans. ASAE 12(6):754-762; 1969.

Moe, P.G. Kinetics of the microbial decomposition of the herbicides IPC and CIPC. Environ. Sci. and Technol. 4(50):429-431; 1970.

Moldenhauer, W.C.; Long, D.C. Influence of rainfall energy on soil loss and infiltration particles: I. Effect over a range of texture. Soil Sci. Soc. Amer. Proc. 28(6):813-817.

Neibling, W.H.; Foster, G.R. Estimating deposition and sediment yield from overland flow processes. Proceedings of the international symposium on urban hydrology hydraulics and sediment control; Lexington, Kentucky; 1977.

Novotny, V. Hydrological and hydraulical conceptual models applicable to overland and water transport modeling. Madison, WI: University of Wisconsin, Water Resources Center; 1976.

Novotny, V.; Tran, H.; Simsiman, G.V.; Chester, G. Mathematical modeling of land runoff contaminated by phosphorus. J. Water Pollution Control pp. 101-112; 1978.

Onstand, C.A.; Foster, G.R. Erosion modeling on a watershed. Transactions of the ASAE, 18(2):288; 1975.

Penman, H.L. Natural evaporation from open water, bare soil, and grass. Proc. Roy. Soc. (London), Ser. A. 193:120-145; 1948.

Perry. J.H. Chemical Engineering Handbook, 4th ed. New York, NY: McGraw-Hill; 1963.

Philip, J.R. Theory of infiltration. Chow, V.T., ed. Advances in hydroscience 5:215-296. New York, NY: Academic Press; 1969.

Rifai. M.N.E.; Kaufman, W.J.; Todd, D.K. Dispersion phenomena in laminar flow porous media. Sanitary Eng. Res. Lab. and Div. C.E. Report 3. Berkeley, CA: University of California; 1956.

Ryden, J.C.; <u>et al</u>. Potential of an eroding urban soil for the phosphorus enrichment of streams. J. Environ. Quality 1(4):430; 1972.

Shearer, R.C.; Letey, J.; Farmer, W.J.; Klute, A. Lindane diffusion in soil. Soil Sci. Soc. Amer. Proc. 37(2):189-193; 1973.

Smith, J.M. Chemical engineering kinetics, 2nd ed. New York, NY: McGraw-Hill; 1970.

Spencer, W.F.; Claith, M.M.; Farmer, W.J. Vapor density of soil-applied dieldrin as related to soil-water content, temperature and dieldrin concentration. Soil Sci. Soc. Amer. Proc. 33:509-511; 1969.

Swann, R.L.; McCall, P.J.; Unger, S.M. Volatility of pesticides from soil surfaces. Unpublished and undated manuscript received as personal communication from Dow Chemical, USA, Midland, MI; November 16, 1979.

Yalin, Y.S. An expression for bed-load transportation. J. Hyd. Div., Proc. ASCE 89(HY3):221-250; 1963.

# MI - miscellaneous

### APPENDIX MI

### SYMBOLS & MISCELLANEOUS

The appendix will contain sections such as

- 1.0 INDEX
- 2.0 DEFINITIONS
  - 2.1 Soil Moisture Content (see next page)
- 3.0 NOTATIONS (also presented in each scientific appendix separately)
- 4.0 FORTRAN/NONFORTRAN VARIABLE CORRESPONDENCE
- 5.0 LIST OF FIGURES (also presented in each appendix)
- 6.0 LIST OF TABLES (also presented in each appendix)
- 7.0 TABLE FOR NUMERICAL CALCULATIONS
- 8.0 TYPICAL INPUT/OUTPUT (attached)

#### 2.1 SOIL MOISTURE CONTENT

The soil moisture content  $(\theta)$  is defined as:

$$\theta = V_w / V_t$$
 inmL/mL = cm³/cm³

where:

 $V_{t} = V_{s} + V_{w} + V_{a}$   $V_{t} = \text{total volume of soil}$  matrix (mL)  $V_{s} = \text{volume of solid portion (mL)}$   $V_{w} = \text{volume of water (mL)}$  $V_{a} = \text{volume of air (mL)}$ 

Like the porosity (n), volumetric moisture content is usually reported as a decimal fraction or a percent. For saturation flow,  $\theta=n$ ; for unsaturated flow,  $\Theta(n)$ .

Above definition is employed by most of hydrology engineers and SESOIL. Soil scientists, however, frequently use as a moisture content definition the effective "degree of medium saturation" (s), that is equivalent to the volume of active soil moisture (water) divided by the effective volume of voids . In this case

 $s = \theta/n$   $(0 \le s \le 1)$ or  $\theta = s \cdot n$   $(0 \le \theta \le n)$ 

where n is the effective medium porosity, or the effective volume of voids divided by the total volume.

۷a

v_w

V_s

#### MARCOS BONAZOUNTAS

Dr. Bonazountas, a civil and environmental engineer, is a senior member of the Bio/Enviro Systems Section of Arthur D. Little, Inc. who specializes in mathematical environmental modeling.

Since joining Arthur D. Little, Inc., in 1977, Dr. Bonazountas has coordinated modeling teams and has developed, applied and validated models for: soils, groundwater, biology, water resources and air; multi-media models for pollutant fate in the environment; estuarine and hydrodynamic models with a continuous emphasis on both the physics and the chemistry of the environmental systems; and water resources economic and optimization models. Recent models developed include: an unsaturated soil/biotic zone model, which can be interfaced with any numerical finite difference or finite element groundwater model; a time series water resource model; a civil engineering construction containment model for hazardous waste sites; an oceanic oil slick model; a river quality and sedimentation model; an economic model for waste underground injection; an environmental exposure/ risk model for toxic substances, and a river basin model for the Magarin Dam and the Jordan River in Jordan. He recently headed a multi-media (air, water, soil, biota) mathematical effort whose objective was to describe fate. pathways, and exposure of toxics in the environment. The SESOIL model encompasses his lengthy experience in soil, groundwater and general environmental modeling.

Dr. Bohazountas received his engineering degree in civil engineering (1969) from the National Technical University of Athens, Greece; his doctoral degree in river hydromechanics and sediment transportation (1973) from the Technical University of Munich, Germany; and his diploma in computer sciences (1969) from the Data Processing Institute in Athens, Greece. He undertook further studies at the Massachusetts Institute of Technology and at Harvard University. He is a member of the ASCE, AGU, AWRA, IAIIR, and has authored several publications concerning his research.

# JANET M. WAGNER

Ms. Wagner, a chemist and computer scientist, is a member of the Analytical and Environmental Chemistry Section. Her interests focus on the use of computers for solving environmental and chemical problems.

Since joining Arthur D. Little, Inc. in 1978, Ms. Wagner has been involved in the application of numerical and mathematical techniques to the modeling of environmental processes, in the development of software packages for data management, and in the analysis of several toxic substances in the environment: Recent projects, other than SESOIL, include design and implementation of a data management system for a project involving utility disposal wastes, which is designed to facilitate data entry and storage and to perform further statistical and numerical analyses on subsets of the data; analysis of chemical properties, laboratory studies, and environmental models to assess the environmental fate of some of the priority pollutants for EPA risk assessment reports; and a multi-media environmental computer modeling effort, encompassing air, soil and water-body modeling issues, such as air quality, river flow and quality, and exposure to humans and biota. Ms. Wagner was a member on the ADLPIPE/DIS project, where she contributed to the development, implementation, and quality control of large FORTRAN based software packages used for piping system design and pipe stress analysis. Prior to joining Arthur D. Little, Inc., Ms. Wagner' participated in the design and implementation of computer models in the areas of quantum chemistry and chemical kinetics:

In her computer applications, Ms. Wagner has employed the computer languages of FORTRAN, APL, BASIC, ALGOL and COBOL; has worked on IBM, VAX, CDC, UNIVAC, and Data General systems; and has developed and worked with computer graphics packages.

Ms. Wagner received her B.A. in Chemistry and Computer Sciences (1980) from Williams College, Massachusetts.